County Council Of Howard County, Maryland

2010 Legislative Session

Legislative Day No. 8

Resolution No. 103-2010

Introduced by: The Chairperson at the request of the County Executive

A RESOLUTION approving and authorizing the execution of a Development Rights and Responsibilities Agreement by and between Camilla Carroll and Philip D. Carroll and Howard County, Maryland in accordance Title 16, Subtitle 17 of the Howard County Code.

Introduced and read first time J - 4 6 2010. By order <u>Stephen DeGendre</u> Administrator Read for a second time at a public hearing on July 19_____, 2010. By order Stephen Mfredende This Resolution was read the third time and was Adopted___, Adopted with amendments ____, Failed___, Withdrawn___, by the County Council on J. 4 29 2010.

Certified By <u>Gtohen Megundre</u> Stephen LeGendre, Administrator

NOTE: ([text in brackets]) indicates deletions from existing law; TEXT IN SMALL CAPITALS indicates additions to existing law; Strike-out indicates material deleted by amendment; Underlining Indicates material added by amendment.

WHEREAS, Article 66B, § 13.01 of the Maryland Annotated Code grants
Howard County the authority to establish procedures and requirements for the
consideration and execution of Development Rights and Responsibilities Agreements;
and
WHEREAS, the Howard County Council adopted Council Bill No. 4-2010 on
March 1, 2010, effective May 4, 2010, that enacted Sections 16.1700 *et seq.* of the
Howard County Code authorizing the County to enter into Development Rights and

9 10 Responsibilities Agreements; and

WHEREAS, a Development Rights and Responsibilities Agreement (the "Agreement") between Camilla Carroll and Philip D. Carroll (collectively, the "Carrolls") and Howard County, substantially in the form attached to this Resolution as Exhibit A, is intended to constitute a Development Rights and Responsibilities Agreement as provided for in Article 66B and the Howard County Code; and

16

WHEREAS, the Carrolls own certain real property in Howard County, Maryland,
described on Howard County Tax Map No. 2523, Parcel 71 consisting of approximately
892.6 acres ("Doughoregan" or "Property"), as shown on Exhibit 1 to the Agreement;
and

21

WHEREAS, the home on the Property, Doughoregan Manor, was the country home of Charles Carroll of Carrollton; a planter, landowner, politician, United States Senator, and the only Roman Catholic signer of the Declaration of Independence; and

WHEREAS, Charles Carroll of Carrollton was the last surviving signer of the Declaration of Independence and is buried in the family chapel attached to the north end of the Manor; and

29

WHEREAS, Doughoregan Manor is the only home of a signer of the Declaration
of Independence that is still in family ownership; and

1

1	WHEREAS, as part of a comprehensive, multi-part strategy for the preservation
2	of Doughoregan, 75 acres are already protected in a permanent preservation easement;
3	and
A	
4	to the County is authorized to
5	WHEREAS, by passage of Council Bill No. 32-2010, the County is authorized to
6	enter into an Installment Purchase Agreement to acquire a permanent agricultural land
7	preservation easement on 500 acres at Doughoregan; and
8	
9	WHEREAS, the Carrolls are in the process of developing a strategy for
10	preserving the approximately 90-acre historic core that includes the Manor House and
11	numerous historic outbuildings; and
12	
13	WHEREAS, the Carrolls intend to develop approximately 221.1 acres of
14	Doughoregan with not more than 325 single family detached residential dwelling units,
15	substantially in the manner as set forth in Exhibit 2 to the Agreement; and
16	
17	WHEREAS, also as part of the Agreement, the Carrolls are offering to donate
18	approximately 36 acres as an addition to the Kiwanis-Wallis Park; and
19	
20	WHEREAS, on or about March 19, 2010, the Carrolls petitioned Howard County
21	to enter into the Agreement; and
22	
23	WHEREAS, pursuant to Section 16.1704 of the Howard County Code, on or
24	about April 1, 2010, Howard County accepted the Carroll's petition and initiated the
25	process of considering a Development Rights and Responsibilities Agreement; and
26	
27	WHEREAS, pursuant to Section 16.1704(b)(2) of the Howard County Code, the
28	Agreement was referred to the Howard County Planning Board for a determination of
29	whether the Agreement is consistent with the 2000 General Plan, as amended; and
30	

.

2

.

1	WHEREAS, at a public meeting held on June 17, 2010, the Howard County		
2	Planning Board determined that the Agreement was consistent with the General Plan an		
3	a copy of the determination by the Howard County Planning Board is attached to the		
4	Agreement as Exhibit 5; and		
5			
6	WHEREAS, a public hearing was held before the County Council on July 19,		
7	2010; and		
8			
9	WHEREAS, the criteria set forth in Title 16, Subtitle 17 of the Howard County		
10	Code have been met and the County Executive may execute the Agreement.		
11			
12	NOW, THEREFORE, BE IT RESOLVED by the County Council of Howard		
13	County, Maryland this $\frac{\partial q^{\prime}}{\partial t}$ day of $\frac{\partial u}{\partial t}$, 2010 that the Development Rights		
14	and Responsibilities Agreement, substantially in the form attached as Exhibit A, having		
15	met the criteria set forth in Title 16, Subtitle 17 of the Howard County Code is hereby		
16	approved.		
17			
18	AND BE IT FURTHER RESOLVED, that the County Executive is hereby		
19	authorized to execute the Development Rights and Responsibilities Agreement in the		
20	name of and on behalf of the County.		

•

.

.

EXHIBIT A

DEVELOPMENT RIGHTS AND RESPONSIBILITIES AGREEMENT

THIS DEVELOPMENT RIGHTS AND RESPONSIBILITIES AGREEMENT (this "Agreement"), made as of the _____ day of _____, 2010, by and between CAMILLA CARROLL and PHILIP D. CARROLL, individuals, (collectively referred to as the "Carrolls") and HOWARD COUNTY, MARYLAND, a body corporate and politic of the State of Maryland ("Howard County"). The Carrolls and Howard County are hereinafter referred to collectively as the "Parties".

RECITALS

1. Maryland law, Article 66B, § 13.01 of the Maryland Annotated Code ("Article 66B"), grants Howard County the authority to establish procedures and requirements for the consideration and execution of Development Rights and Responsibilities Agreements.

2. Howard County Council adopted Council Bill No. 4-2010 on March 1, 2010, effective May 4, 2010, creating Sections 16.1700 *et seq.* of the Howard County Code authorizing Howard County to enter into Development Rights and Responsibilities Agreements ("County Ordinance").

3. This Agreement is intended to constitute a Development Rights and Responsibilities Agreement as provided for in Article 66B and the County Ordinance.

4. The Carrolls own certain real property in Howard County, Maryland, described on Howard County Tax Map No. 2523, Parcel 71 consisting of 892.6 AC \pm as shown and described on the attached **EXHIBIT 1** hereto and made part hereof ("Doughoregan" or "Property").

5. The Carrolls intend to develop a 221.1 AC \pm portion of Doughoregan as depicted on the attached **EXHIBIT 2** (the "Site") with not more than 325 single family detached residential dwelling units substantially in the manner as set forth in **EXHIBIT 2** (the "Project"). A legal description of the Site is attached hereto as **EXHIBIT 3**.

6. The names of all parties having an equitable or legal interest in Doughoregan, including lien holders, are set forth in the title opinion letter, attached hereto as **EXHIBIT 4**, and made a part hereof.

7. On or about March 19, 2010, the Carrolls petitioned Howard County to enter into this Agreement.

8. On or about April 1, 2010, Howard County reviewed this petition and determined to accept this petition and initiate the process of considering a Development

Rights and Responsibilities Agreement.

9. This Agreement was negotiated between the Carrolls and the Howard County Executive.

10. This Agreement was referred to the Howard County Planning Board for determination of whether this Agreement is consistent with the 2000 General Plan as amended. At a public meeting held on <u>June 17</u>, 2010, the Howard County Planning Board determined that this Agreement was consistent with the General Plan. The determination by the Howard County Planning Board is attached hereto as **EXHIBIT 5**.

11. On <u>July 19</u>, 2010, the Howard County Council held a duly advertised public hearing on this Agreement in accordance with Howard County law, and approved this Agreement on <u>July 29, 2010</u> by Council Resolution <u>No. 103-2010</u>.

12. Any other agreements between the Carrolls and Howard County, including a signed commitment letter dated May 18, 2010 for the sale of the development rights to 500 acres of the Property under the Howard County Agricultural Land Preservation Program (the "Commitment Letter"), a copy of which is attached hereto as **EXHIBIT 6**, an Option Agreement for Agricultural Preservation Easement for the portion of the Property in the Commitment Letter (the "Option Agreement"), a copy of which is attached hereto as **EXHIBIT 7**, remains in full force and effect and is intended to be harmonious with this Agreement.

NOW, THEREFORE, in consideration of the foregoing recitals, which are not merely prefatory but are hereby incorporated into and made a part of this Agreement, and the mutual covenants and agreements as set forth below, and for other good and valuable consideration, the receipt and sufficiency of which the Parties hereby acknowledge, the Carrolls and Howard County hereby agree as follows:

ARTICLE I

DEFINITIONS

1.1 <u>Definitions</u>. Unless otherwise defined in this Agreement, the following words, when used in this Agreement, shall have the following meanings:

A. "Agreement" means this Development Rights and Responsibilities Agreement.

B. "APFO Approval" means approval by Howard County Department of Planning and Zoning of applicable Adequate Public Facilities Ordinance tests for the Project or any portion thereof in accordance with the provisions of Section 16.1100 et seq. of the Howard County Code.

C. "Howard County" means Howard County, Maryland, a body corporate and politic.

D. "Carrolls" means Camilla Carroll and/or Phillip D. Carroll and/or their respective agents, purchaser(s), heir(s), assignee(s) or successor(s) in interest.

E. "Commitment Letter" means the executed letter dated April 14 May 18, 2010 for the Carrolls' sale of the development rights on 500 acres of the Property in Howard County pursuant to the Agricultural Preservation Act. Upon the execution of this Agreement and the Option Agreement, notwithstanding paragraph 10 of the Commitment Letter, the obligations of the Parties shall be governed by this Agreement and the Option Agreement and binding subject to the provisions of these agreements.

F. "County Code" or "Howard County Code" means the Code of Howard County, Maryland and any other laws of Howard County including its regulations, resolutions, the Howard County Design Manual, master plan(s), design criteria or any other provision having the force and effect of law which are in effect on the Effective Date of this Agreement.

G. "County Ordinance" means Howard County Council Bill No. 4-2010 creating Sections 16.1700 *et seq.* of the Howard County Code and authorizing Development Rights and Responsibilities Agreements.

H. "Development Approval" means that final governmental approval of Subdivision Plat(s), Site Development Plans, Adequate Public Facilities Ordinance ("APFO") requirements, sewer and/or water approvals, as applicable, has been obtained for the Project or each respective portion and phase of the Project, and that all conditions of said approval have been satisfied with all applicable appeal period(s) having expired without the filing of any appeal, or, if an appeal(s) was filed, a final unappealable decision.

I. "DPZ" means the Howard County Department of Planning and Zoning.

J. "Effective Date of this Agreement" means the date the last party executes this Agreement.

K. "Improvements" means those improvements to be made to roads, sewer, water, storm water, and other facilities necessary to service the Project.

L. "Land Records" means the land records of Howard County, Maryland.

M. "Parkland" means the parcel which shall contain a total of thirtysix (36) acres of land, more or less, be adjacent to Kiwanis-Wallis Park and shall substantially conform to the area indicated on EXHIBIT 2, and shall be conveyed to the County by the Carrolls as set forth in Section 3.4 hereof.

N. "Phase" means any portion of the Project that has received Development Approval.

O. "Planning Board" means the Planning Board for Howard County Maryland.

P. "Property" means the Carroll's fee simple real property in Howard County, Maryland, shown on Howard County Tax Map No. 25 23 as Parcel 71 consisting of 892.6 AC ± as shown and described on the attached EXHIBIT 1 hereto and made part hereof.

Q. "Site" means the 221.1 AC \pm portion of the Property as depicted on the attached **EXHIBIT 2** (the "Site") which shall be developed with not more than 325 single family detached residential dwelling units substantially in the manner as set forth in **EXHIBIT 2** (the "Project"). A legal description of the Site is attached hereto as **EXHIBIT 3**.

R. "Subdivision Plat" means a final plat(s) of subdivision for the Project or any phase or portion thereof, prepared in accordance with the County Code and approved by DPZ and/or the Planning Board.

S. "Subdivision Regulations" means the Title 16, Subtitle 1 of the Howard County Code in effect on the Effective Date of this Agreement.

T. "Zoning Regulations" or "HCZR" means the Howard County Zoning Regulations in effect on the Effective Date of this Agreement.

ARTICLE II ZONING, LIMITATIONS, PLAN APPROVALS AND PLAN CONSISTENCY

2.1 <u>Zoning and Plan Designations</u>. The Site is zoned R-ED on the Howard County Zoning Map. The Site is a) located within the Planned Service Area for Water

and Sewer as contained in the 2000 General Plan, as amended; and (b) designated as "Residential Areas" on Map 7-1 of the General Plan Policies Map 2000-2020, as amended. In the event that the zoning for the Site is no longer R-ED (or the subsequent equivalent of such zoning classification) or the Site is no longer within the Planned Service Area for Water and Sewer as contained in the 2000 General Plan or any subsequent Howard County General Plan, the obligations and responsibilities of the Parties in this Agreement shall terminate without further action.

2.2 <u>Developable Area</u>. The physical area of the Site that is available for development consists of those areas that are generally shown on the concept plan attached hereto as EXHIBIT 2. With the exception of the area of land described in the provisions of Section 3.4 below, the developable areas are intended to encompass all of the land areas of the Site which are not precluded from development by the Subdivision Regulations. The Parties acknowledge that the exact size, shape and configuration of one or more of these areas may be adjusted from the area shown on EXHIBIT 2 based on final surveying, engineering, and design of the Project. Except as provided in Section 8.1C of this Agreement, this developable area may not be expanded, reduced, limited or otherwise altered by any legislative, executive or quasi-judicial action of Howard County including, but not limited to, a comprehensive rezoning, a piecemeal rezoning, or the enactment of ordinances, resolutions, rules or regulations, or the interpretation thereof (such as forest conservation or stream buffer ordinances), which would result in a reduction of the developable areas of the Site.

2.3 <u>Development Limitations</u>. The permissible uses on the Site shall be limited to single-family detached housing units and uses accessory thereto under Section 107 of the Zoning Regulations. The development requirements of the Project, such as setbacks, lot area, building height, open space, lighting, and parking, will be subject to the provisions and limitations as set forth in the Zoning Regulations and other applicable provisions of the County Code.

A. The total density of residential development on the Site shall not exceed three hundred twenty-five (325) single-family detached dwelling units as defined in the Zoning Regulations. The Carrolls shall execute and record in the land records of Howard County a perpetual covenant to prohibit the construction of single-family attached dwelling units or apartment units on the site, and to prohibit the construction of more than 325 single-family detached units on the site. This covenant shall be made between the Carrolls and either Howard County, a homeowner's association of the site, or other community association, at the discretion of Howard County, within 90 days after Howard County exercises the option under Section 1.2 of the Option Agreement, unless such period is extended by Howard County.

B. The design of the Site, including the lots, roads and open spaces indicated thereon, shall substantially conform to the concept plan attached hereto as

EXHIBIT 2.

2.4 <u>APFO Approval</u>. The Carrolls shall be required to obtain APFO Approval for the Project. <u>Project</u>, for all aspects of the development of the Site, in all of its phases, for all APFO tests.

2.5 <u>Other Development Approvals and Permits</u>. All sections of the Project shall be required to obtain all applicable Development Approvals and all other applicable requirements of the County Code for land development, including but not limited to site improvement plans (for water, sanitary sewer, storm water management and storm drainage and sediment and erosion control improvements), forest conservation requirements, Subdivision Plat(s) including any required dedications of land consistent with the County Code, building permits, and occupancy permits. The Project shall also be required to obtain all applicable Maryland or federal approval(s) and permits.

2.6 <u>Regulation and Master Plan Consistency</u>. Howard County has determined that the Project and this Agreement are consistent with the Howard County General Plan. The Planning Board has determined that this Agreement is consistent with the Howard County General Plan.

2.7 <u>Public Health, Safety and Welfare</u>. Howard County has determined that development of the Project in accordance with the Zoning Regulations and County Code and the terms and provisions of this Agreement will ensure that the public health, safety and welfare of the citizens of Howard County are protected.

ARTICLE III <u>ROAD IMPROVEMENTS, SEWER IMPROVEMENTS,</u> <u>WATER IMPROVEMENTS AND OTHER COMMUNITY AMENITIES</u>

3.1 <u>Road Improvements</u>.

A. <u>APFO Required Road Improvements</u>. The Carrolls shall be required, as a condition of APFO Approval, to make or contribute funds towards certain road improvements to serve the Project, and to convey any rights-of-way needed for the construction of the road improvements as required by APFO.

B. <u>Road Connection to Burnside Drive</u>. The Carrolls agree not to construct any roads that would connect the Site or any portion thereof to Burnside Drive. Furthermore, Howard County agrees to honor Howard County Council Resolution 43-1989's termination of Burnside Drive and, therefore, not require the connection of the Site to Burnside. Howard County shall, furthermore, neither exercise nor threaten to exercise its power of eminent domain so as to connect Burnside Drive to the Site

1. In the event that some other governmental or regulatory agency would require an easement or other accommodation in order to facilitate emergency vehicle access to the Site, the Carrolls may propose an alternative plan to provide such access provided, however, that in no event shall the Carrolls provide for the aforementioned road connection to Burnside.

2. In furtherance of the action necessary to fulfill the intent of this Section 3.1.B, the Carrolls shall subdivide and create an open space lot comprising the land fronting on Burnside Drive (the "Burnside Drive Open Space Lot"). The Carrolls shall work with the residents of Ridge Lake and Chateau Ridge, including the Chateau Ridge Lake Community Association ("CRLCA"), to effect a transfer by deed for the fee simple interests to the Burnside Drive Open Space Lot to the CRLCA and/or other to an entity or individuals identified by Howard County the CRLCA for sum of one dollar (\$1.00). The transfer of the Burnside Drive Open Space Lot shall be submitted to DPZ with the final deed for the Subdivision Plat, and shall be recorded in the Land Records with the Subdivision Plat.

C. To further effect the purposes of the intent of this Section 3.1.B, the Carrolls shall ensure that the Subdivision Plat will create at least two (2) residential single-family detached lots near and impeding the terminus of Burnside Drive such that Burnside Drive could not be extended westward without the demolition of homes on these two or more lots.

3.2 Sewer Improvements.

A. The Parties agree with the goal of minimizing increases in wastewater flow and the nutrient concentration in wastewater directed to the Little Patuxent Wastewater Reclamation Plant as a condition of Development Approval. In furtherance of this obligation, the Carrolls agree to abide by the terms of the proposed plan of action as set forth in Exhibit 8. The Carrolls shall be responsible for the payment of costs for the construction of sewer utilities for the Project to achieve the goal. Such costs may include the wastewater facility design, engineering, testing, bonding and, if necessary, acquisition of off-site easements for the improvements determined appropriate by the wastewater flow study ("Wastewater Mitigation Cost") provided, however, that in no event shall the Carrolls be required to expend greater than One Million Dollars (\$1,000,000) for the Wastewater Mitigation Cost.

B. In addition to the Wastewater Mitigation Costs, the Carrolls shall construct and/or be obligated to incur the following improvements and/or costs:

1. Replacement and increase in sizing of portions of the existing eight (8) inch sewer line currently on the Site, which portions are determined in the reasoned

opinion of Howard County to be undersized;

2. Application, incorporation and payment of all fees for Metropolitan District incorporation, which application due to its ministerial nature shall be approved by Howard County.

C. No building permit(s) for residential dwelling unit(s) on the Site shall be issued until (i) Capital Project S-6274 is funded for construction and (ii) the existing sewer line improvements in Section 3.2.B.1 are completed. A grading permit shall not be construed to be a building permit.

3.3 <u>Water Improvements</u>. The Carrolls shall be required to make certain water improvements for the Project as a condition of Development Approval. The Carrolls shall be responsible for the payment of all construction of water utilities for the Project, to include site design, engineering, testing and bonding. Additionally, the Carrolls shall construct and incur the following improvements and costs:

A. All necessary permit approval(s);

B. Application, incorporation and payment of all fees for Metropolitan District incorporation, which application due to its ministerial nature shall be approved by Howard County.

3.4 <u>Dedication of Land to Howard County</u>. The Carrolls shall dedicate to Howard County a parcel of land on the east side of the Site for use as a County park or other public, recreational use determined by the County to best serve the interest of the public ("Parkland"). This parcel shall contain a total of thirty-six (36) acres of land, more or less, be adjacent to Kiwanis-Wallis Park and shall substantially conform to the area indicated on EXHIBIT 2. The conveyance of this parcel to the County shall be dedicated as Open Space in accordance with the County Code.

The timeline for dedication of this parcel to the County shall be as follows:

A. The Carrolls shall submit a proposed plat of the Parkland to DPZ within one hundred eighty (180) days after execution of this Agreement.

B. The Carrolls shall convey the deed for the Parkland to Howard County with the submission of the first Subdivision Plat for County signature and recordation in accordance with the Subdivision Regulations following the fulfillment of the Conditions Precedent of section 1.2 of the Option Agreement. In the alternative, the Carrolls may in their sole discretion deed the Parkland to Howard County in any legal manner as directed by Howard County.

C. The Parkland to be conveyed to Howard County shall be eligible to be treated in accordance with the provisions of the HCZR and/or County Code as Open Space and is eligible to be included toward satisfaction of subdivision, site plan and/or other Development Approval requirements.

ARTICLE IV <u>PRESERVATION OF AGRICULTURAL LANDS AND</u> <u>LIMITATIONS ON DEVELOPMENT</u>

Agricultural Preservation. The Carrolls have agreed to place an agricultural land preservation easement on approximately 500 acres of the Property in accordance with Howard County's Agricultural Land Preservation Program pursuant to the terms of the Commitment Letter and the Option Agreement. A legal description of the 500 acres is attached hereto as **EXHIBIT 9**. Nothing in this Agreement shall be construed to supersede either the Carrolls' or Howard County's obligations under the Option Agreement.

- 4.1
- A. Notwithstanding any provision in this Agreement to the contrary, all of the obligations and responsibilities in this Agreement shall terminate in the event that Howard County does not exercise its option under the Option Agreement
- B. Prior to Howard County's exercise of its option under the Option Agreement, The Carrolls shall not submit and Howard County shall have no obligation to approve for recordation any Subdivision Plat for the Property creating any individual lots meeting the bulk requirements of R-ED zoning for single family detached housing.
- C. The Parties further agree that each shall take all actions required to promptly return the Site to a rural conservation zoning classification (now known as "RC") in the event the County has not acquired the agricultural land preservation easement under the Commitment Letter by January 1, 2013, which date may be extended under the terms of the Option Agreement. Upon the final, unappealable approval of the rural classification zoning approval for the Site, the County shall terminate its right to acquire the Agricultural Land Preservation Easement under the Commitment Letter.
- 4.2 <u>Covenant Against Residential Subdivision of the Core</u>. The Carrolls shall

execute and record In the Land Records a perpetual covenant to prevent further subdivision of the approximately ninety-four (94) acres of the property around the main Manor House of the Property (the "Core") with the Howard County Conservancy or other organization determined to be acceptable to Howard County which organization shall hold the enforcement rights to the restrictive covenant. <u>This covenant shall be made within 90 days after Howard County's exercise of the option under Section 1.2 of the Option Agreement, unless such period is extended by Howard County. A legal description of the Core is attached hereto as EXHIBIT 10.</u>

- 4.3 The Parties agree that the use of certain funds to be received by the Carrolls pursuant to the anticipated development of the Site shall be used to ensure funding for the restoration and ongoing maintenance of the Manor House and historic outbuildings on the Property. The Parties agree that the Carrolls will be obligated to:
 - 4.3.1 Establish a fund in the amount of Two Million Dollars (\$2,000,000) to be specifically allocated towards such purpose (the "Fund");
 - 4.3.2 To provide evidence to Howard County, at such times as Howard County may reasonably request (but not more often than once per year), of such Fund and invoices evidencing the use of such funds for the purposes as specified in paragraph 4.3 above.
 - 4.3.2.1 The funding of the Fund will be accomplished by the Carrolls placing in the Fund after-tax sums received by the Carrolls from the Project, in an amount equal to fifty percent (50%) of any such after-tax sums, until the Fund has reached a sum of Two Million Dollars (\$2,000,000), minus any sums the Carrolls have spent for such restoration and ongoing maintenance from the date of this Agreement to the date of such funding.
 - 4.3.2.2 The establishment of the Fund and the use of the proceeds thereof for such restoration and maintenance shall be the sole obligation of the Carrolls pursuant to this Agreement for such restoration and maintenance.

ARTICLE V DEVELOPMENT REVIEW

5.1 <u>Timely Development Review</u>. Howard County agrees to use its best

efforts in accordance with applicable rules and regulations to ensure that the processing and review of Development Approvals, including, but not limited to, APFO Approvals, subdivision plans, site development plans, issuance of building permits and occupancy permits, are performed in a succinct, timely manner, without undue delay, consistent with the County's current development review process, and that such processing and review will not be subjected to any delay or any moratorium except in accordance with the terms of this Agreement.

5.2 <u>Timely Submission of Documents</u>. The Carrolls agree to use their best efforts to submit and process plans and legal documents for the items set forth in Section 5.1 above in a timely manner.

ARTICLE VI SURVIVAL AND TRANSFER OF OBLIGATION

6.1 <u>Nature, Survival, and Transfer of Obligations</u>. The Carrolls agree that this Agreement shall run with the land and be binding upon and inure to the benefit of the Carrolls and their respective heirs, successors and assigns, and upon any and all successor owners of record of all or any portion of the <u>SiteProperty</u> (except owners of an individual lot improved as part of the Project and pursuant to a validly issued building permit). To assure that all such successors, assigns, and successor owners have notice of this Agreement and the obligations created by it, the Carrolls agrees <u>agree</u> that it shall:

A. Have this Agreement recorded among the Land Records of Howard County within twenty (20) days after the Effective Date of this Agreement; and

B. Incorporate, by reference, this Agreement into any and all real estate sales contracts entered into after the Effective Date of this Agreement for the sale of all or any portion of the <u>SiteProperty</u>.

C. Prior to the transfer of all or any portion of the Site<u>Property</u> (except the transfer of an individual lot solely for use as a private residence), or any equitable interest therein, require the transferee to execute an enforceable written agreement, in a form reasonably satisfactory to Howard County, binding transferee to this Agreement.

6.2 <u>Binding Upon Successors and Assigns of Howard County</u>. Howard County agrees that all obligations assumed by it under this Agreement shall be binding on it, its agencies, employees, governmental units, the Planning Board and its and their respective successors and assigns.

ARTICLE VII

BREACH AND REMEDIES

7.1 <u>Breach by Carrolls</u>. If the Carrolls shall fail or refuse to perform its obligations as required, then after sixty (60) days from receipt of written notice provided to the Carrolls by Howard County indicating the nature of the default and if the Carrolls have not cured the default, the County may seek equitable relief to enforce the terms and conditions of this Agreement either through a decree for specific performance or an injunction, or declare this Agreement null and void and cease the issuance of building permits and review of development plans. Should the remedies of specific performance or injunction not be available to Howard County because of actions of Carrolls, then Howard County shall be entitled to bring a legal action for damages.

7.2 <u>Breach by Howard County</u>. If Howard County shall fail or refuse to perform its obligations as required, then after sixty (60) days from receipt of written notice provided to Howard County by the Carrolls indicating the nature of the default and if Howard County has not cured the default, the Carrolls may seek equitable relief to enforce the terms and conditions of this Agreement either through a decree for specific performance or an injunction.

7.3 <u>Jurisdiction and Venue</u>. Jurisdiction and venue for any proceedings brought with respect to this Agreement shall be in the Circuit Court for Howard County, Maryland.

7.4 <u>Waiver of Trial by Jury</u>. Carrolls and Howard County do hereby waive trial by jury in connection with any proceedings brought to enforce the terms of this Agreement.

7.5 <u>Agreement to Petition for Rezoning</u>. In the event this Agreement is determined unenforceable by any Court, then the Parties agree that this paragraph shall survive such determination and the Carrolls agree to file a request to rezone the Site to a rural zoning classification equivalent to the RC zoning classification in effect on the date of this Agreement.

ARTICLE VIII EFFECT OF DEVELOPMENT REGULATIONS

8.1 Effect of Agreement.

A. Carrolls must comply with all federal, Maryland and local laws existing on the Effective Date of this Agreement.

B. Except as provided in Section 8.1.C herein, the laws, rules,

regulations and policies, which govern the use, density or intensity of the Site shall be the laws, rules, regulations and policies, if any, in force on the Effective Date of the Agreement.

C. If Howard County determines that compliance with a law(s) enacted or adopted after the Effective Date of this Agreement is essential to ensure the health, safety or welfare of residents of all or part of Howard County, this Agreement may not prevent Howard County from requiring the Carrolls to comply with those laws, rules, regulations and policies.

D. In the event Howard County takes any action to subject the Project to any new or modified law(s) after the Effective Date of this Agreement under Section 8.1.C above or modifies the Zoning or Comprehensive Plan land use designations of the Site, or the density of the Project, as set forth in this Agreement, which reduces the number of residential dwelling units by more than three and one-half percent (3.5%) ("Howard County Action"), Carrolls shall be relieved of any remaining obligation under this Agreement. Additionally, the Carrolls shall be relieved of any obligation to make any payments required under this Agreement that are not due and payable and are unpaid as of the date of the Howard County Action.

8.2 <u>Approvals Required</u>. Carrolls shall obtain all approvals necessary under any provision of local, Maryland or federal law before proceeding with development of the Project. This Agreement does not address any approvals required by Maryland or federal law, and Carrolls shall be responsible for obtaining any approvals required by local, Maryland, or federal law.

8.3 <u>Fees</u>. Carrolls shall pay all fees (specifically including but not limited to excise taxes, surcharges and water and sewer connection fees) required by the County at the time of the Effective Date of this Agreement at the rate in effect at the time the fee is due. Nothing in this Agreement shall be construed as a waiver or reduction of any such fee.

8.4 <u>Growth Control Delay</u>. In the event that a "Growth Control Delay" (as hereinafter defined) is imposed, then any deadline concerning (i) the Carrolls' obligation to construct, install, fund or post financial guarantees for (a) the infrastructure improvements required pursuant to any Development Approval for the Project in accordance with the phasing requirements set forth therein, and (b) the road improvements described in Article III of this Agreement; and, (ii) the Carrolls' obligation to dedicate the Parkland to the County or the obligations contained in the provisions of Article III of this Agreement, shall be extended for one (1) additional day for each day during which such Growth Control Delay exists, and the Project shall not be subjected to any additional regulation, legislation, limitation, phasing, contributions,

ł

penalties or delay in construction, or issuance of zoning certificates/building permits solely as a result of the Growth Control Delay. The term "Growth Control Delay" shall mean any and all delays caused during APFO Approval and/or the implementation or declaration by the United States Government, State of Maryland, Howard County, and/or any agency, department, division and/or branch thereof for purposes of a limitation, prohibition, restriction and/or phasing upon the review, recording, development and construction upon lots on the Site as intended by Carrolls, which is deemed to be essential to ensure the public health, safety, and welfare of County residents as determined in accordance with Section 8.1.C above.

ARTICLE IX <u>MISCELLANEOUS</u>

9.1 <u>Time of Essence</u>. Time is of the essence in the performance of all terms and provisions of this Agreement.

9.2 <u>Term</u>.

A. This Agreement shall constitute covenants running with the land and shall run with and bind the <u>SiteProperty</u>. This Agreement shall terminate and be void on the tenth (10^{th}) anniversary of the Effective Date of this Agreement, unless extended by an amendment to this Agreement complying with all procedures required in this Agreement, the County Ordinance and Maryland Law or in accordance with Section <u>8.5</u> <u>8.4</u> above or unless terminated by agreement of the Parties or as permitted by law.

B. Nothing in this Section shall be construed to supersede the term(s) as set forth in any other agreement(s) between the Carrolls and Howard County

9.3 <u>Notices</u>. All notices and other communications in connection with this Agreement shall be in writing and shall be deemed delivered to the addressee thereof (I) when delivered in person on a business day at the address set forth below; or (2) on the third business day after being deposited in any main or branch United States post office for delivery by properly addressed, postage prepaid, certified or registered mail, return receipt requested, at the address set forth below

Notices and communications to the Carrolls shall be addressed and delivered to the following address:

Camilla and Phillip D. Carroll 3500 Manor Lane Ellicott City, MD 21042 with a copy to:

Joseph Rutter 5300 Dorsey Hall Drive Ellicott City, MD 21042

Sang W. Oh, Esq. 5100 Dorsey Hall Drive Ellicott City, MD 21042

James L. Wright, Esq. 750 E. Pratt Street Suite 900 Baltimore, MD 21202

Notices and communications to the Howard County shall be addressed and delivered to the following address:

Howard County Executive 3430 Courthouse Drive Ellicott City, MD 21043

With a copy to:

Howard County Solicitor 3430 Courthouse Drive Ellicott City, MD 21043 Director, Department of Planning and Zoning 3430 Courthouse Drive Ellicott City, MD 21043

Chair, Howard County Council 3430 Courthouse Drive Ellicott City, MD 21043

By notice complying with the requirements of this Section, each party shall have the right to change the address or addressee or both for all future notices and communications to such party, but no notice of a change of address shall be effective until actually received.

9.4 <u>Amendments</u>. The Parties to this Agreement may terminate <u>amend</u> the Agreement by mutual consent after Howard County holds a public hearing and complies with all applicable laws of the County Ordinance concerning amendment of a Development Rights and Responsibilities Agreement. All amendments to this Agreement shall be in writing and shall be executed by Howard County and the Carrolls. Unless the Planning Board determines that the proposed amendment is consistent with the Howard County General Plan, the Parties may not amend or terminate this Agreement, and Howard County may not suspend or terminate the Agreement. Unless the Planning Board determines that the proposed amendment is consistent with the Howard County General Plan, the Parties may not amend this Agreement.

9.5 <u>Termination or Suspension</u>. The Parties to this Agreement may terminate or suspend the Agreement by mutual consent after Howard County holds a public hearing and complies with all applicable laws concerning termination or suspension of a Development Rights and Responsibilities Agreement as set forth in Howard County Ordinance. If Howard County determines that a suspension or termination is essential to ensure the public health, safety or welfare, as determined in accordance with Section 8.1.C above, Howard County may suspend or terminate this Agreement following a public hearing. Any such unilateral termination of this Agreement by Howard County shall not in any way affect the validity of any Development Approvals which have been obtained for the Project, including, but not limited to, APFO Approvals.

9.6 <u>Authority to Execute</u>. Howard County and the Carrolls hereby acknowledge and agree that all required notices, meetings, and hearings have been properly given and held by Howard County with respect to the approval of this Agreement and agree not to challenge this Agreement or any of the obligations created by it on the grounds of any procedural infirmity or any denial of any procedural right. Howard County hereby warrants and represents to the Carrolls that the persons executing this Agreement on its behalf have been properly authorized to do so. The Carrolls hereby warrant and represent to Howard County that they are the fee simple, record owners of Doughoregan and the Site, (2) that each has the right, power and authority to enter into this Agreement and to agree to the terms, provisions, and conditions set forth herein and to bind the <u>SiteProperty</u> as set forth herein, (3) that all legal actions needed to authorize the execution, delivery and performance of this Agreement have been taken, and (4) each has been duly represented by attorneys.

9.7 <u>Governing Law</u>. This Agreement shall be governed by and construed in accordance with the laws of the State of Maryland.

9.8 <u>Consent to Jurisdiction</u>. The Parties irrevocably consent to the jurisdiction of the Circuit Court for Howard County, Maryland or any federal court sitting in the District of Maryland.

9.9 <u>Remedies Cumulative</u>. Each right, power and remedy of a party provided for in this Agreement, or any other agreement between the Parties, now or hereafter existing, shall be cumulative and concurrent and in addition to every other right, power or remedy provided for in this Agreement or any other agreement between the Parties, now or hereafter existing.

9.10 <u>Severability</u>. In case any one or more of the provisions contained in this Agreement shall for any reason be held invalid, illegal or unenforceable in any respect, such invalidity, illegality, or unenforceability shall not affect any other provision of the Agreement, and this Agreement shall be construed as if such invalid, illegal, or unenforceable provision had never been contained in this Agreement.

9.11 <u>Recordation</u>. Any party may record this Agreement among the Land Records of Howard County, Maryland. In the event the Agreement is terminated following an appeal as provided for in Section 9.12 below, the Parties agree to execute and record a document in the aforesaid Land Records to terminate this Agreement.

9.12 <u>Appeals</u>. Both Maryland law and the County Ordinance may allow any person aggrieved by this Agreement to file an appeal. If the effect of the decision in such Appeal revises this Agreement in any way, any party to this Agreement may terminate the Agreement by providing notice to all Parties within thirty (30) days of the date that the decision in the appeal becomes final and all appeals thereof have been finally determined.

9.13 <u>No Obligation to Approve</u>. This Agreement shall not be interpreted or construed to impose any legal obligation on Howard County or any of its boards, agencies, commissions or employees to approve any development, use, density or intensity other than as provided specifically in this Agreement.

9.14 <u>No Third Party Beneficiary Status</u>. The Parties specifically agree that this Agreement is not intended to create in the public or any member thereof, third party beneficiary status in connection with the performance of the obligations under this Agreement.

9.15 <u>Lien holders</u>. All persons with a lien interest in the <u>Site Property</u> have executed this Agreement, and those lien holders with a power of sale have subordinated such liens to the position of Howard County under this Agreement.

IN WITNESS WHEREOF, the Parties have hereunto set their hands on the date first above written.

[Signatures on Following Page]

CAMILLA CARROLL	
	(SEAL)
PHILIP D. CARROLL	
(SEAL)
AGREED and APPROVED:	
HOWARD COUNTY, MARY	LAND
BY: Ken Ulman Howard County Execu	
AL SUFFICIENCY	
20	
	PHILIP D. CARROLL PHILIP D. CARROLL AGREED and APPROVED: HOWARD COUNTY, MARY BY: Ken Ulman Howard County Execut

Margaret Ann Nolan County Solicitor

[Notaries on Following Page]

STATE OF MARYLAND, _____ COUNTY, TO WIT:

I HEREBY CERTIFY that on this ______ day of ______, 2010, before me, the subscriber, a Notary Public of the State of Maryland, in and for the County aforesaid, personally appeared CAMILLA CARROLL, personally known to me or proven to be the individual named herein and executed this Agreement for the purposes stated therein.

AS WITNESS my Hand and Notarial Seal.

My Commission Expires: ______

STATE OF MARYLAND, _____ COUNTY, TO WIT:

I HEREBY CERTIFY that on this _____ day of _____, 2010, before me, the subscriber, a Notary Public of the State of Maryland, in and for the County aforesaid, personally appeared PHILIP D. CARROLL, personally known to me or proven to be the individual named herein and executed this Agreement for the purposes stated therein.

AS WITNESS my Hand and Notarial Seal.

Notary Public

My Commission Expires: _____

STATE OF MARYLAND, ______ COUNTY, TO WIT:

I HEREBY CERTIFY that on this _____ day of ______, 2010, before me, the subscriber, a Notary Public of the State of Maryland, in and for the County aforesaid, personally appeared KEN ULMAN, the County Executive for Howard County, Maryland, who acknowledged the same to be the act of the County and that he executed the foregoing Agreement for the purposes therein contained by signing in my presence the name of Howard County, Maryland as County Executive.

AS WITNESS my Hand and Notarial Seal.

Notary Public

My Commission Expires: _____

ATTORNEYS' CERTIFICATION

THIS IS TO CERTIFY that the undersigned are members, in good standing, of the Bar of the Court of Appeals of Maryland, and that the within instrument was prepared by the undersigned or under their supervision.

Upon Recordation Please Return To:

Sang W. Oh, Esq. Talkin & Oh, LLP 5100 Dorsey Hall Drive Ellicott City, MD 21042

•

LIENHOLDER CONSENTS AND SUBORDINATION OF INTERESTS

The undersigned lienholder does hereby consent to the aforesaid Agreement, and by the signature of its Trustee, does hereby agree that such instrument shall be subordinate to and shall survive any sale under its Deed of Trust dated______, and recorded among the Land Records of Howard County, Maryland in Liber ____, folio ____

WITNESS:

LIST OF EXHIBITS

- Exhibit 1 Property Description
- Exhibit 2 -- Concept Plan
- Exhibit 3 Description of Site
- Exhibit 4 Attorney title opinion certifying as to legal and equitable owners
- Exhibit 5 Planning Board Determination of Consistency
- Exhibit 6 Commitment Letter
- Exhibit 7 Option Agreement
- Exhibit 8 Wastewater Treatment Plan of Action
- Exhibit 9 Description of Agricultural Preservation Parcel
- Exhibit 10 Description of Core

•

EXHIBIT 1

•

,

Property Description

EXHIBIT 1

March 18, 2010

DESCRIPTION OF A 892.622 ACRE PARCEL PROPERTY OF PHILIP D. CARROL CAMILLA CARROLL SECOND AND THIRD ELECTION DISTRICTS HOWARD COUNTY, MARYLAND

BEING a parcel of land, situate and lying in the Second and Third Election Districts of Howard County, Maryland; said parcel of land being part of the land which, by Will of Nina R. Carroll dated June 4, 1979, as amended by its First Codicil dated November 28, 1986 and its Second Codicil dated April 10, 1987; Nina R Carroll having departed this life on February 11, 1989; said will having been probated in the Surrogates Court Of New York County, New York on April 11, 1989, and the Estate of Nina R. Carroll is filed in the Register of Wills of Howard County, Maryland as Estate No. 17-7868-276; said property also being described in a Mortgage dated October 31, 1962 and recorded among the Land Records of Howard County, Maryland in Liber 394 At Folio 64 and being all of Parcel 1 as described in the aforesaid Liber 394 at Folio. 64; said parcel of land being more particularly described, as now surveyed, in the Maryland State Coordinate System NAD '83 Datum, as projected by Howard County Geodetic Control, as follows:

SHER, OLLINS

ARTER, IC. A ENGINEERANG CONSISTANTS LOTO SURVEYING

72 8aliloore Hullonei Pike :eli City, Maryland 21042)) 481-2855 (410) 750-3784 BEGINNING FOR THE SAME at a point 33.0 feet, more or less, measured at right, angles northerly from the centerline of paving as now exists, twenty-four (24) feet wide more or less, and being on the Northern margin of the sixty-six foot wide Right of Way of Frederick Road (Maryland Route 144), formerly known as the Baltimore to Frederick Turnpike Road;

1

. • •

:

1 10 1

••••••••••••

* • · · ·

• •

• • • • •

a tall of a strategy to a strategy of

said point being opposite Manor Lane and at the beginning of the First or South 73°02'48" East 3140.09 foot Deed Line of Parcel 1, recorded in Liber 394 at Folio 64; thence running with Frederick Road and running with and binding on the said First Deed Line of Parcel 1, as now surveyed,

1)

South 73°02'02" East 3140.09 feet to a point at the beginning of the Second or South 25°58'45" East 2921.96 foot Deed Line of Parcel 1, recorded in Liber 394 at Folio 64; said Second Deed Line also being the following: the Third or North 18°43' West 323.01 foot line of the adjacent conveyance from Albert S. Singleton and Elsie B. Singleton, his wife, to Philip G. Wilkins and Corinne Elizabeth Wilkins, his wife, by deed dated June 29, 1963 and recorded among said Land Records in Liber 403 at Folio 669; and the North 26°02'23" West 1737.32 foot and the North 23°56'49" West 55.21 foot outlines, as shown on a plat entitled "Plat of Survey, Kiwanis Club of Ellicott City, Inc." recorded among said Land Records as Plat No. 10119; and the North 19°061/West 180.00 foot outline, as shown on a subdivision plat entitled "Map of Pine Orchard Meadow; Section Three" recorded among said Land Records in Plat Book 6 at Page 53; and also the North 18°39'00" West 532.45 foot outline, as shown on a subdivision plat entitled "Pine Orchard Meadows, Section Four" and recorded among said Land Records in Plat Book 17 at Page 75; thence intending to bind on and run with said Second Deed Line of Parcel 1, recorded in Liber 394 at Folio 64, as now surveyed, the following three (3) courses and distances:

SHER, DLLINS

ARTER, C.

LENSINEERONI CONSLETANTS LAND RURVEYORS

'2 Balilinary Nulloual Pixe alt City, Maryland 21042) 481-2355 (410) 760-3784

- South 26°02'24" East 2,155.13 feet, passing over a Concrete Monument found 89.57 feet on line, to a Rebar with "FCC-106" Cap set; thence continuing with the aforesaid Second Deed Line of Liber 394 at Folio 64,
- South 24°00'53" East 55.21 feet to a 1 inch Iron Pipe found; thence still continuing with said Second Deed Line of Liber 394 at Folio 64.
- 4) South 25°53'43" East, 711.66 feet to a 0.4 foot by 0.4 foot wide Concrete Monument found at the beginning of the Third or South 21°10'35" East 412.58 foot Deed Line of Parcel 1, recorded in Liber 394 at Folio 64; said Third Deed Line of Parcel 1 also being the North 14°00'00" West 136.25 foot outline, as shown on the aforesaid plat of "Pine Orchard Meadows, Section Four" recorded in Plat Book 17 at Folio 75, and also the South 14°00'00" East 276.25 foot outline, as shown on a subdivision plat entitled "Pine Orchard Meadows, Section 5" and recorded among said Land Records in Plat Book 22 at Folio 11; thence binding along said Third Deed Line of Parcel 1; recorded in Liber 394 at Folio 64, as now surveyed,
- 5) South 21913'25" East 412.50 feet to a Rebar with "FCC-106" Cap set at the beginning of the Fourth or South 87°17'37' West 523.99 foot Deed Line of Parcel 1, recorded in:Liber 394 at Folio 64; said Fourth Deed Line of Parcel 1 also being the North 87°16'56" East 523.27 foot outline, as shown on Sheet 4 of the subdivision plats entitled "Centennial Manor, Section One, Area Three, Lots 85-108", recorded among said Land Records as Plat C.M.P. No. 7675, thence

• • •

<u>1</u>4.

. .

· • • •

ARTER,

SHER.

C. ENGINEERING CONSULTANTS AND SURVEYORS

2 Ballinete Hallanai Pite 18 City, Karyland 21042 451-2855 410) 758-3784 binding along said Fourth Deed Line of Parcel 1, recorded in Liber 394 at Folio 64, as now surveyed,

South 87°11'19" West 523.27 feet to a Concrete Monument found at the beginning of the Fifth or South 07°19'03" East 2783.20 foot Deed Line of Parcel 1 recorded in Liber 394 at Folio 64; said Fifth Deed Line of Parcel 1 also being the following: the North 07°14'30" West 245.00 foot outline, as shown on the aforesaid plat of "Centennial Manor, Section One, Area Three, Lots 85-108" recorded as Plat C.M.P. No. 7675; and the North 07°14'30" West 930.98 foot outline, as shown on Sheets 3 and 2 of the subdivision plats entitled "Centennial Manor, Section One, Area One, Lots 1-22" recorded among said Land Records as Plat Nos. 6797 and 6796, respectively; and the North 06°45'30" West 600.00 foot outline, as shown on a subdivision plat entitled "Section Three, Chateau Ridge" recorded among said Land Records in Plat Book 21 at Folio 11; and the North 06°45'30? West 700.00 foot outline as shown on a subdivision plat entitled "Section Five, Chateau Ridge" recorded among said Land Records in Plat Book 22 at Folio 46; and also the North 06°45'30" West 300.00 foot outline, as shown on a subdivision plat entitled "Section Six, Chateau Ridge". recorded among said Land Records in Plat Book 22 at Folio 82; thence binding along said Fifth Deed Line of Parcel 1, recorded in Liber 394 at Folio 64, as now surveyed,

7)

, .

1. EHEINEERIND CONSULTANTS LIND SURVEYORS

sher, Ollins

ARTER,

1C.

٠.

6)

72 Balümere Hettonai Pike :ak City, Maryland 21042 1) 461-2855 (410) 750-3784 South 07°13'17" East 2779.90 feet to a Granite Stone found at the beginning of the Sixth or South 61°55'21" West 642.37 foot Deed Line of Parcel 1 recorded

1 S 1 1 1 1

. . .•]

2

in Liber 394 at Folio 64; said Sixth Deed Line of Parcel 1 also being the North 62°06'38" East 641.44 foot outline, as shown on a subdivision plat entitled "Plat of Revision, Kingsbridge @ Burleigh Manor, Lots 763 thru 778" recorded among said Land Records as Plat No. 12607; thence binding along said Sixth Deed Line of Parcel 1, recorded in Liber 394 at Folio 64, as now surveyed,

- 8) South 62°05'05" West 641.25 feet to a Granite Stone found at the beginning of the Seventh or North 73°14'48" West 4021.13 foot Deed Line of Parcel 1, recorded in Liber 394 at Folio 64; said Granite Stone also being at the end of the Seventh or South 73°14'48" East 4021.23 foot Deed Line of Parcel 2 described in the aforesaid Liber 394 at Folio 64; thence binding along the Seventh thru Thirteenth Deed Lines of Parcel 1, recorded in Liber 394 at Folio 64, and also binding reversely along the Seventh thru First Deed Lines of Parcel 2, recorded in Liber 394 at Folio 64, as now surveyed, for the following seven (7) courses and distances:
- 9) North 73°16'34" West 4018.45 feet to a Rebar with "FCC-106" Cap set at a point near the southeastern side of Manor Lane, being a twelve (12) foot wide gravel drive, as now exists; thence running with said Manor Lane,
- North 14°04'34" East 24.75 feet to a Rebar with "FCC-106" Cap set, thence crossing over said Manor Lane,
- North 65°25'26" West 2143.94 feet, to a 1.25 inch diameter Iron Pipe found;
 thence

sher, Jluns

ARTER, C.

i exeduerang consultants Luko surviyans

'2 Baltanara Hallodal Pika ott City, Maryland 21642) 451-2855 (410) 760-3784

- 12) North 70°10'26" West 1456.72 feet, to a 1.25 inch diameter Iron Pipe found;thence
- 13) North 78°34'25" West 791.22 feet to a point; thence
- 14) North 73°34'25" West 387.55 feet to a point; thence
- 15) South 69°10'35" West 31.37 feet to a point 23.8 feet, more or less, measured at right angles southeasterly from the centerline of paving, eighteen (18) feet wide, as now exists, of Folly Quarter Road, formerly known as the Vineyard Road; said point also being at the beginning of the Fourteenth or North 53°00'48" East 577.50 foot Deed Line of Parcel 1 recorded in Liber 394 at Folio 64; thence running with said Folly Quarter Road and running with and binding on the Fourteenth thru Sixteenth Deed Lines of Parcel 1, recorded in Liber 394 at Folio 64, as now surveyed, for the following three (3) courses and distances:
- 16) North 51°40'56" East 568.09 feet to a point; thence crossing into and running within said Folly Quarter Road, on or near the centerline of the existing paving thereof,
- 17) North 39°59'12" East 2458.50 feet to a point; thence
- 18) North 26°28'43" East 2376.00 feet to a point on or near the intersection of the centerlines of existing paving of said Folly Quarter Road with Frederick Road (Maryland Route 144), as now exists, twenty four (24) feet wide and formerly known as the Baltimore to Frederick Turnpike Road; said point also being the beginning of the Seventeenth or South 82°36'44" East 189.50 foot Deed Line of Parcel 1, recorded in Liber 394 at Folio 64; thence running within said Frederick

SHER, OLLINS

ARTER, IC.

1. ENGINEERING GONSLETANTS LAND SURVEYORS

72 Balilinare Hallonal Pike 2011 Dity, Naryland 21842 3) 461-2866 (410) 750-3784 Road, on or near the centerline of existing paving, and running with and binding on said Seventeenth Deed Line of Parcel 1 recorded in Liber 394 at Folio 64, as now surveyed,

- South 82°54'28" East 189.50 feet to a point; thence leaving said centerline of
 Frederick Road and running with and binding the Eighteenth or North
 14°06'44" West 35.47 foot Deed Line of Parcel 1 recorded in Liber 394 at Folio
 64, as now surveyed,
- 20) North 17°12'33" West 32.08 feet to a point at the beginning of the Nineteenth or South 82°36'44" East 1394.98 foot Deed Line of Parcel 1 recorded in Liber 394 at Folio 64; thence running with the aforesaid Frederick Road and running with and binding on said Nineteenth Deed Line of Parcel 1 recorded in Liber 394 at Folio 64, as now surveyed.
- South 82°35'58" East 1394.98 feet to the point of beginning; containing 892.622
 acres of land, more or less.

SUBJECT TO an existing 20' Right-Of-Way for a Sewer Main described in a "Deed Of Easement" conveyed from Philip D. Carroll and Camilla Carroll to Howard County, Maryland by Deed dated December 24, 1994 and recorded among the Land Records of Howard County, Maryland in Liber 3422, Folio 357.

ALSO SUBJECT TO AND TOGETHER WITH all matters shown on a plat entitled "Amended Plat Of Easement, Sending Parcel, Property Of Philip Carroll And Camilla Carroll"

ISHER, OLLINS

ARTER, IC. 11 FREINHERTING CONSIGNATION

:72 Ballmore Hallonel Pike coh City, Haryland 21042 0) 481-2865 (410) 750-3784

and recorded among the Land Records of Howard County, Maryland as Plat No. 19928, and a

"Deed Of Preservation Easement" from Philip D. Carroll and Camilla Carroll to The Howard County Conservancy, Inc. and Howard County, Maryland by Deed dated October 4, 2006 recorded among the aforesaid Land Records in Liber 10295, Folio 58.

ALSO SUBJECT TO AND TOGETHER WITH any and all terms and conditions that might exist to the rights, use and maintenance of Folly Quarter Road and Frederick Road, Maryland Route 144, to Howard County, Maryland and the Maryland State Highway Administration.

SUBJECT TO AND TOGETHER WITH all conditions, covenants, easements, restrictions and rights-of-way of record, which may apply.

This description was prepared without the benefit of a Title Report.



Prepared By Checked By

WO #05022-3002

SHER, DLLINS

ARTER, C.

1 ENBUTERORY CONSULTANTI Land Sontetions

72 Buillmore Hatlonal Pite eti Cliy, Haryland 21842 1) 401-2855 (410) 760-3784


.

.

Concept Plan

,



Description of Site

.

.

.

March 18, 2010

DESCRIPTION OF A 221.064 ACRE PARCEL PART OF THE PROPERTY OF PHILIP D. CARROLL AND CAMILLA CARROLL SECOND AND THIRD ELECTION DISTRICTS HOWARD COUNTY, MARYLAND

BEING a parcel of land, situate and lying in the Second and Third Election Districts of Howard County, Maryland, said parcel of land being part of the land which by Will of Nina R. Carroll dated June 4, 1979, as amended by its First Codicil dated November 28, 1986 and its Second Codicil dated April 10, 1987; Nina R. Carroll having departed this life on February 11, 1989; said Will having been probated in the Surrogates Court of New York County, New York on April 11, 1989, and the Estate of Nina R. Carroll is filed in the Register of Wills of Howard County, Maryland as Estate No. 17-7868-276, said property also being described in a Mortgage dated October 31, 1962 and recorded among the Land Records of Howard County, Maryland in Liber 394 at Folio 64, and being part of Parcel 1 described in the aforesaid Liber 394 at Folio 64; said parcel being more particularly described, as now surveyed in the Maryland State Coordinate System NAD' 83 Datum, as projected by Howard County Geodetic Control, as follows:

SHER, DLLINS

ARTER, C.

. EHAMAELRING CONSULTANTS 'AND SURVEYORS

2. Balikmore Hallonal Pike ni City, Manyland 21842 1 451-2655 410) 759-3784 BEGINNING FOR THE SAME at a point on the Northern margin of the sixty-six foot wide Right of Way of Frederick Road (Maryland Route 144), formerly known as the Baltimore to Frederick Tumpike Road, said point being at the beginning of the Second or South 25°58'45" East, 2921.96 foot Deed Line of the aforesaid Liber 394 at Folio 64 (Parcel 1); said Second Deed Line also being the following: the Third or North 18°43' West 323.01 foot line of the adjacent conveyance from Albert S. Singleton and Elsie B. Singleton, his wife, to Philip G. Wilkins and Corinne Elizabeth Wilkins, his wife, by deed dated June 29, 1963 and recorded among said Land Records in Liber 403 at Folio 669; and the North 26°02'23" West 1737.32 foot and the North 23°56'49" West 55.21 foot outlines, as shown on a plat entitled "Plat of Survey, Kiwanis Club of Ellicott City, Inc." recorded among said Land Records as Plat No. 10119; and the North 19°06' West 180,00 foot outline, as shown on a subdivision plat entitled "Map of Pine Orchard Meadow, Section Three" recorded among said Land Records in Plat Book 6 at Page 53; and also the North 18°39'00" West 532.45 foot outline, as shown on a subdivision plat entitled "Pine Orchard Meadows, Section Four" and recorded among said Land Records in Plat Book 17 at Page 75; thence intending to bind on and run with said Second Deed Line of Parcel 1, recorded in Liber 394 at Folio 64, as now surveyed, the following three (3) courses and distances:

- South 26°02'24" East 2,155.13 feet, passing over a Concrete Monument found 89,57 feet on line, to a Rebar with "FCC-106" Cap set; thence continuing with the aforesaid Second Deed Line of Liber 394 at Folio 64,
- South 24°00'53" East 55.21 feet to a I inch Iron Pipe found; thence still continuing with said Second Deed Line of Liber 394 at Folio 64,
- 3) South 25°53'43" East, 711.66 feet to a 0.4 foot by 0.4 foot wide Concrete Monument found at the beginning of the Third or South 21°10'35" East 412.58 foot Deed Line of Parcel 1, recorded in Liber 394 at Folio 64; said Third Deed Line of Parcel 1 also being the North 14°00'00" West 136.25 foot outline, as

SHER, DLLINS

NATER, C.

ENOINEERINA CONSULTANTS AND SURTYYYRS

2 Ballizova Rutionat Pita sti City, Maryianii 21042 461-2855 410) 730-3784 shown on the aforesaid plat of "Pine Orchard Meadows, Section Four" recorded in Plat Book 17 at Folio 75, and also the South 14°00'00" East 276.25 foot outline, as shown on a subdivision plat entitled "Pine Orchard Meadows, Section 5" and recorded among said Land Records in Plat Book 22 at Folio 11; thence binding along said Third Deed Line of Parcel 1, recorded in Liber 394 at Folio 64, as now surveyed,

- 4) South 21°13'25" East 412.50 feet to a Rebar with "FCC-106" Cap set at the beginning of the Fourth or South 87°17'37" West 523.99 foot Deed Line of Parcel 1, recorded in Liber 394 at Folio 64; said Fourth Deed Line of Parcel 1 also being the North 87°16'56" East 523.27 foot outline, as shown on Sheet 4 of the subdivision plats entitled "Centennial Manor, Section One, Area Three, Lots 85-108", recorded among said Land Records as Plat C.M.P. No. 7675, thence binding along said Fourth Deed Line of Parcel 1, recorded in Liber 394 at Folio 64, as now surveyed,
- 5) South 87°11'19" West 523.27 feet to a Concrete Monument found at the beginning of the Fifth or South 07°19'03" East 2783.20 foot Deed Line of Parcel 1 recorded in Liber 394 at Folio 64; said Fifth Deed Line of Parcel 1 also being the following: the North 07°14'30" West 245.00 foot outline, as shown on the aforesaid plat of "Centennial Manor, Section One, Area Three, Lots 85-108" recorded as Plat C.M.P. No. 7675; and the North 07°14'30" West 930.98 foot outline, as shown on Sheets 3 and 2 of the subdivision plats entitled "Centennial Manor, Section One, Area One, Lots 1-22" recorded among said Land Records

sher, Jllins

ARTER, C.

: ENGLALEBLING GUNSUAJANTS VAND SURVEYONS

2 Ballindra National Piae sti City, Maryland 21842) 481-2855 410) 758-3784 as Plat Nos. 6797 and 6796, respectively; and the North 06°45'30" West 600.00 foot outline, as shown on a subdivision plat entitled "Section Three, Chateau Ridge" recorded among said Land Records in Plat Book 21 at Folio 11; and the North 06°45'30" West 700.00 foot outline as shown on a subdivision plat entitled "Section Five, Chateau Ridge" recorded among said Land Records in Plat Book 22 at Folio 46; and also the North 06°45'30" West 300.00 foot outline, as shown on a subdivision plat entitled "Section Six, Chateau Ridge" recorded among said Land Records in Plat Book 22 at Folio 82; thence binding along said Fifth Deed Line of Parcel 1, recorded in Liber 394 at Folio 64, as now surveyed,

6) South 07°13'17" East 2779.90 feet to a Granite Stone found at the beginning of the Sixth or South 61°55'21" West 642.37 foot Deed Line of Parcel 1 recorded in Liber 394 at Folio 64; said Sixth Deed Line of Parcel 1 also being the North 62°06'38" East 641.44 foot outline, as shown on a subdivision plat entitled "Plat of Revision, Kingsbridge @ Burleigh Manor, Lots 763 thru 778" recorded among said Land Records as Plat No. 12607; thence binding along said Sixth Deed Line of Parcel 1, recorded in Liber 394 at Folio 64, as now surveyed,
7) South 62°05'05" West 641.25 feet to a Granite Stone found at the beginning of the Seventh or North 73°14'48" West 4021.13 foot Deed Line of Parcel 1, recorded in Liber 394 at Folio 64; said Granite Stone also being at the end of the Seventh or South 73°14'48" East 4021.23 foot Deed Line of Parcel 2 described in the aforesaid Liber 394 at Folio 64; thence running with and binding part of

SHER, OLLINS

ARTER, C.

: ENELDECERUID CONSULTAITS LAND SURVEYORS

2 Balilmore Hellonal Fike 38 GHy, Maryland 21042) 461-2855 A10) 758-3784 the aforesaid Seventh Deed Line of Liber 394, Folio 64, as now surveyed, the following course and distance:

- 8) North 73°16'34" West, 1,764.49 feet to a point; said point being on and a distance of 2,253.96 feet from the end of the Seventh or North 73°14'48" West, 4021.13 foot Deed Line of the aforesaid Liber 394 at Folio 64 (Parcel 1); thence leaving said Seventh line of Liber 394 at Folio 64 (Parcel 1) and running for new lines of division, the following twenty-five (25) courses and distances:
- 9) North 16°43'26" East, 858.48 feet to a point; thence
- 10) North 33°51'48" East, 651.81 feet to a point; thence
- 11) North 67°04'12" East, 462.42 feet to a point; thence
- 12) North 23°09'51" West, 507.53 feet to a point; thence
- 13) North 37°01'20" West, 903.13 feet to a point of curvature; thence
- 468.79 feet along the arc of a curve to the right, having a radius of 580.00 feet, a central angle of 46°18'34" and subtended by a chord bearing and distance of North 13°52'03" West, 456.13 feet to a point of tangency; thence
- 15) North 09°17'14" East, 113.42 feet to a point of curvature; thence
- 16) 136.14 feet along the arc of a curve to the right, having a radius of 325.00 feet, a central angle of 24°00'04" and subtended by a chord bearing and distance of North 21°17'16" East, 135.15 feet to a point of reverse curvature; thence
- 131.70 feet along the arc of a curve to the left, having a radius of 170.00 feet, a central angle of 44°23'10" and subtended by a chord bearing and distance of North 11°05'43" East, 128.43 feet to a point of reverse curvature; thence

SHER, JLLINS

ARTER, C. HIMBERINA CONSIGNATS LAND SURVEYORS

"2 Bailfraoro Haltonal Pike oli City, Maryland 21842) 461-2855 "410) 758-3784 18) 89.85 feet along the arc of a curve to the right, having a radius of 285.00 feet, a central angle of 18°03'45" and subtended by a chord bearing and distance of North 02°03'59" West, 89.47 feet to a point of tangency; thence

19) North 06°57'53" East, 62.64 feet to a point of curvature; thence

- 20) 219.63 feet along the arc of a curve to the right, having a radius of 285.00 feet, a central angle of 44°09'14" and subtended by a chord bearing and distance of North 29°02'30" East, 214.24 feet to a point of reverse curvature; thence
- 21) 283.12 feet along the arc of a curve to the left, having a radius of 275.00 feet, a central angle of 58°59°14" and subtended by a chord bearing and distance of North 21°37'30" East, 270.78 feet to a point of compound curvature; thence
- 22) 470.43 feet along the arc of a curve to the left, having a radius of 595.00 feet, a central angle of 45°18'00" and subtended by a chord bearing and distance of North 30°31'07" West, 458.27 feet to a point of tangency; thence

23) North 53°10'07" West, 259.20 feet to a point of curvature; thence

- 312.10 feet along the arc of a curve to the right, having a radius of 330.00 feet, a central angle of 54°11'16" and subtended by a chord bearing and distance of North 26°04'29" West, 300.60 feet to a point of tangency; thence
- 25) North 01°01'09" East, 232.32 feet to a point of curvature; thence
- 26) 121.72 feet along the arc of a curve to the left, having a radius of 150.00 feet, a central angle of 46°29'32" and subtended by a chord bearing and distance of North 22°13'37" West, 118.40 feet to a point of reverse curvature; thence

Sher,)llins

ATER, C,

ENGLIGELARGI CONSULTANTI AND SURVEYORS

2 Saillatore Hatloani Pito H City, Meryferd 21042 481-2855 110) 758-3784

- 27) 346.71 feet along the arc of a curve to the right, having a radius of 315.00 feet, a central angle of 63°03'47" and subtended by a chord bearing and distance of North 13°56'30" West, 329.47 feet to a point of tangency; thence
- 28) North 17°35'24" East, 225.85 feet to a point; thence
- 29) North 27°46'18" West, 35.58 feet to a point; thence
- 30) North 73°08'00" West, 180.91 feet to a point; thence
- 31) North 19°32'01" West, 18.41 feet to a point; thence
- 32) North 73°02'02" West, 328.95 feet to a point; thence
- 33) North 16°57'58" East, 65.50 feet to a point on the aforesaid Northern margin of Frederick Road (Maryland Route 144); said point being on and a distance of 1,432.93 feet from the beginning of the First or South 73°02'48" East, 3140.09 feet Deed Line of the aforesaid Liber 394 at Folio 64 (Parcel 1); thence running with and binding on said First line of Liber 394 at Folio 64 (Parcel 1) and the aforesaid Northern margin of Frederick Road (Maryland Route 144), the following course and distance:
- South 73°02'02" East, 1,707.16 feet to the point of beginning; containing
 221.064 acres of land, more or less.

SUBJECT TO an existing 20' Right-Of-Way for a Sewer Main described in a "Deed Of Easement" conveyed from Philip D. Carroll and Camilla Carroll to Howard County, Maryland by Deed dated December 24, 1994 and recorded among the Land Records of Howard County, Maryland in Liber 3422, Folio 357.

SHER, OLLINS

ARTER, C.

L ERBOITERING CONSULIANTS LAND SURVEYORS

12 Daltimers Heliosel Pite ofi City, Naryland 21042 † 401-2055 (410) 750-3764 ALSO SUBJECT TO AND TOGETHER WITH any and all terms and conditions that might exist to the rights, use and maintenance of Folly Quarter Road and Frederick Road, Maryland Route 144, to Howard County, Maryland and the Maryland State Highway Administration.

SUBJECT TO AND TOGETHER WITH all conditions, covenants, easements, restrictions and rights-of-way of record, or proposed, which may apply.

This description was prepared without the benefit of a Title Report.



Prepared By DBS Checked By 9

WO #05022-3002

SHER,)LLINS

NRTER, C.

EHODREEHTRO CONSULTANTS AND SURVEYORS

2 Bailimers Hellona) Pite n City, Maryland 21042 481-2855 410) 750-3784



I:\2005\05022\dwg\05022-3001 Parcel 3 Exhibit.dwg, 3/18/2010 10:01:06 AM, dstreett,

Attorney Title Opinion Certifying as to Legal and Equitable Owners

. LAW OFFICES OF TALKIN & OH, LLP COLUMBIA OFFICE 5100 DORSEY HALL: DRIVE ELLICOTT CITY, MARYLAND 21042-7870

(410) 964-0300 (301) 596-6500 Pax: (410) 964-2008

June 22, 2010

Ms. Charlotte R. Dryden Real Estate Services Howard County Department of Public Works 3430 Courthouse Drive Ellicott City, Maryland 21043

Re: "Doughoregan Manor", 3500 Manor Lane, Ellicott City

Dear Ms. Dryden:

The following is a title report for the above-referenced property (the "Property").

Fee simple title in the Property is good and marketable as vested of record in Philip D. Carroll (also known as Philip Carroll) and Camilla Carroll, as tenants in common, as follows:

- 1. Deed from Charles Carroll, Junior to Philip A. Carroll, dated February 24, 1923 and recorded among the Land Records of Howard County, Maryland (the "Land Records") in Liber H.B.N. No. 117, folio 113.
- 2. Last Will and Testament of Philip A. Carroll (deceased July 8, 1957, survived by his wife, Nina R. Carroll) dated May 9, 1946, as amended by a Codicil thereto dated October 16, 1953, admitted to probate by Howard County Register of Wills (Estate No. 2046); Item Second devised all real property and improvements to his wife, Nina R. Carroll, for life, with a power of appointment (to devise through her Will upon her death) over all property located in Howard County, Maryland (including the Property).
- 3. Last Will and Testament of Nina R. Carroll (deceased February 11, 1989) dated June 4, 1979, as amended by a Codicil thereto dated November 28, 1986 and a Second Codicil thereto dated April 10, 1987, admitted to probate by the Howard County Register of Wills (Estate No. 7868); Item Third, as

Ms. Charlotte R. Dryden June 22, 2010 Page 2

> amended, exercised the power of appointment granted by the Will of Philip A. Carroll and devised the Property to "such of the lineal descendants of my son, Philip Carroll, as shall survive me, in equal shares, <u>per stirpes</u>, as tenants in common", and the then-surviving lineal descendants of Philip Carroll were Philip D. Carroll (also known as Philip Carroll) and Camilla Carroll.

The Property is known as 3500 Manor Lane, Ellicott City, Maryland 21042, and is located in the Third Election District of Howard County, Maryland, containing 876.562 acres of land, more or less (per SDAT records). The Property is shown on Tax Map 23, Grid 10, Parcel 71, property tax account number 03-281779.

The Property is not encumbered by any open financing.

The Property is subject to the following exceptions and agreements:

- 1. Right of Way dated October 5, 1908, from Thomas P. O. Donnell and James O. Donnell to W. Raymond Cross, recorded among the Land Records in Liber W.W.L.C. 87, folio 662.
- 2. Deed of Easement dated May 23, 1977, from Nina R. Carroll to the Maryland Historical Trust, recorded among the Land Records in Liber 826, folio 542.
- 3. Right of Way Easement dated May 20, 1980, from Nina R. Carroll to Baltimore Gas and Electric Company, recorded among the Land Records in Liber 1047, folio 333.
- 4. Right of Way Easement dated March 29, 1981, from Nina R. Carroll to Baltimore Gas and Electric Company, recorded among the Land Records in Liber 1051, folio 633.
- 5. Deed of Preservation Easement dated October 4, 2006, by and among Philip Carroll and Camilla Carroll, The Howard County Conservancy, Inc., and Howard County, Maryland, recorded among the Land Records in Liber 10295, folio 58.
- 6. Plat entitled "Density Sending Plat, Property of Philip Carroll and Camilla Carroll", which Plat is recorded among the Land Records as Plat Number 18572.

Ms. Charlotte R. Dryden June 22, 2010 Page 3

7. Plat entitled "Amended Plat of Easement, Sending Parcel, Property of Philip Carroll and Camilla Carroll", which Plat is recorded among the Land Records as Plat Number 19928.

Title was run through May 28, 2010.

Copies of the title documents, encumbrances and exceptions noted above are enclosed. If you have any questions, please do not hesitate to get in touch with me.

Sincerely, Talkin & Oh, LLP

the Ett.

By:

Jonathan E. Greenstein Of Counsel

enclosures

Received for Record 28" Feb 1923 at 9 o'clock A. M. Same day recorded and

HBN 117/113

Hast B. Uslforerk.

118

	ŀ
CHARLES CARROLL,	\$78.00 Duly
. Junior,	Cancelled
DEED TO	Revenue
PHILIP A. OARROLL.	Bramps.
	þ

examined per.

THIS DEED Made this 24th day of February, ix the year mineteen hundred and twenty-three, by and between Charles Carroll, Junior, otherwise known as Charles Bancroft Carroll, ummarried, of Howard County, in the State of

Maryland, party of the first part, and Philip A. Carroll, of the City and State of ' New York, party of the second part.

WITNESSETH: that for and in consderation of the sum of seventy-eight thousand dollars (\$78,000.) this day paid, the receipt whereof is hereby soknowledged, the said party of the first part doth hereby grant and convey unto the said party of the second part, his heirs and assigns, in fee simple.

All those two treate or parcels of land situate and lying in Howard County, in the State of Maryland, which were conveyed to Anita Carroll by two deeds; the first from Charles Carroll and Caroline Carroll, his wife, dated June 7th, 1866, and recorded 'on the eighth day of June 1866 among the Land Records of Howard County in Liber W. -|W. W. No. 25, folio 316, etc., and the second from Matthias Hammond and wife, dated ; the 24th day of November 1867, and recorded among the said Land Records in Liber W.

114

said Charles Carroll, Jr., the grantor berein, after the death of the said Charles Carroll, to sell, give or in any other manner dispose of said two tracts or parcels of land or any part thereof, or any interest therein, free from any of the conditions or provisions in said deed set forth, provided, however, that should said Charles Carroll or Charles Carroll, Jr. desire to sell, give or otherwise dispose of said property, or any part thereof or any interest therein, pursuant to and in exercise of said power, the same shall first be offered to the said Royal Phelps Carroll, if living, and if not, to the said Philip A. Carroll, in fee and unencumbered with the buildings and improvements thereon, for the sum of sixty thousand (\$60,000) dollars, and in addition thereto such sum of money as shall represent the fair value of all permanent improvements which shall have been placed upon said property by Charles Carroll or Charles Carroll, Jr.; the said Charles Carroll and the said Royal Phelps Carroll having departed this life, the said Charles Carroll, Jr. being desirous of solling and the said hohilip A. Carroll being desirous of buying, and it having been agreed as to the fair value of all permanent improvements which have been placed on said property by said Charles Carroll and Charles Carroll; Jr.

TOGETHER WITH THE buildings and improvements thereon, and all the rights, roads, Alleyo, ways, waters, watercourses, privileges, easements, appurtenences and advantages to the same belonging or in anywise appertaining.

TO HAVE AND TO HOLD said described property unto and to the proper use of the said Philip A. Carroll, his heirs and assigns, in fee simple.

AND the said Charles Carroll, Junior, does hereby covenant that he will warrant sepecially the property hereby granted and conveyed and that he will execute such further assurances thereof as may be necessary.

WITNESS the hand and seal of the said grantor.

WITNESS: · Charles Ca H. L. Drake.

Charles Carroll. Jr. (SEAL)

STATE OF MARYLAND, CITY OF BALTINORE, SSI

I HEREBY CERTIFY that on this 24th day of February 1923, before me the subsoriber 'a Notary Public of the State of Maryland, in and for the City of Baltimore aforesaid, personally appeared Charles Carroll, Junior, the grantor hereinbefore named, and acknowledged the aforegoing deed to be his act.

AD WINNERS MIL Louis and unknowled and

10.2046 Colate of PHILIP A. C.	ARROLL			-	
Late of NEW YORK COUNTY. N. Y., died_	July 8th,	_19_57_,_	t	estate	
WERE CRANTED 1 HOND	RETIES	BOLICITORS			
a R. Carroll \$10,000. New Amsterdam	Oasualty Co.	Markel, Veaze 1904 First N Baltimore 2,	tiona	1 5 -	
Executors in Maryland East 70th, Street		J. B. R _g ndo	l Carr		
Y _o rk, N. Y.	\$ ⁸	· · · · · · · ·			
DATE PROCEEDINGS	NAME OF VOLUE	IE LIBER	No.	20Tio	
	Wills	R.L.P.	11	303·	
" Petition for Letters Testamentary in Maryla	md Proceedings	R.L.P.	·44 İ	215	
" Bond of Executors	Adm. Bonds	R.L.P.		455 .	
" Oath of Executors	n n	R.L.P.	8	455 l	
" Order approving Bond & Grant of Letters	Proceedings	R.L.P.	44	216	
" Order to give Notice to Greditors	n n	R.L.P.		217	
	, n	R.L.P.	1	217 .	
" Order to appraise Real Egtate " Order to appraise Personal Egtate	tr	R.L.P.		217 ·	
 Approximent of Resident Agent 	1 12	R.L.P.	: `` i	217	
26 Report of Notice to Creditors	1 11 '	R.L.P.		255.	
	Filed				
" Information Report	' Proceedings	R.L.P.	ւհև	276	
. 10. Order passing Inventory - Real Estate	Inventories	R.L.P.	1 11	259	
Inventory - Real Estate Note: Order passing Inventory - Personal Estate	Proceedings	R.L.P.		276	
	Inventories	R.L.P.	1	1	
3. " Inventory - Personal Estate		R.L.P.	1 .	j 373	
11 Petition & Order to transfer Personal Esta		R.L.P.	1	596	
t. 21. Order passing Administration Account	Adm. Accounts	1		173	
" Administration Account - First & Final .	Adm. Accounts	, <u>1</u> ,1,1,1,1,1,1,1,1,1,1,1,1,1,1,1,1,1,1,	ί το	1 112	
· ·	:		1	•	
	· · ·	i.	,	± , , ;	
			1	1 5	
	• .	l	1	,	
	•		ı	. 	
	•		I I	1 1	
· · ·	•	•	1 3	1	
	: .		ŧ '	1	
· · ·	•	ł	ł	1	
	•	, İ.	•	• .	
		:	1	; · 、	
			ı	•	
	:	l •	: `	i	
• • •	•	•	1	1	
	••	ł		1	
· ·	•	Į.	1	J.	

All those tracts of land which the said Philip A. Carroll, late of New York City, owned situate and lying in the Second and Third Election Districts of Howerd County binding on Carroll's Lane, Paul's Lane and the east side of Vineyard Road and known as Doughoregan Manor; 776 acres of which were devised to him under the Last Will and Testament of John Lee Carroll, late of Howard County, State of Maryland, of record in Liber RD of W No. 5, Folio 81, one of the Wills Records of Howard County aforesaid; 1204 acres of which was granted and conveyed unto Philip A. Carroll by deed dated the 24th day of February, 1923, and recorded among the Land Records of Howard County in Liber HBN No. 117, Folio 113 etc., from Charles Carroll, Jr.; 128 acres more or less of which was granted and conveyed unto Philip A. Carroll by deed dated the 15th. day of September, 1939, was recorded among the Land Records of Howard County in Liber BM Jr. No. 164, Folio 403, from Thomas P. O'Donnell and Ada D. O'Donnell, his wife, and five acres of which was granted and conveyed unto Philip A. Carroll by deed dated the 13th day of April, 1948, and recorded among the Land Records of Howard County in Liber MWB No. 203, Folio 483, from John Yarrow Eccles, executor. The said four tracts contained 2113 acres more or less, all of which sold lots are adjoining. ¥400,000.00

(2)

All that farm in the Fifth Election District which the said Philip A. Carroll owned, binding on Homewood Road and Clarksville Pike (old route 29) and known as the "Benedict Farm" containing 320 acres which were devised to him under the Last Will and Testament of John Lee Carroll, late of Howard County, State of Maryland, and recorded among the Wills Records of Howard County in Liber RD of W No. 5. Folio 81.

(1)

All that tract of wood land in the Fifth Election District of Howard County which the said Philip A. Carroll owned, having no frontage on any road, but having an easement or right of way one-half mile long over the lands of R. G. Harper Carroll II, to Clarksville Pike near tract No. (2) above, containing 60 acres which was devised to Philip A. Carroll under the Last Will and Testament of John Lee Carroll, late of Howard County, State of Maryland, and recorded among the Wills Records of Howard County in Liber RD of W No. 5, Folio 81. (2) and (3) \$50,000.00

÷

(3)



I, <u>PHILIP A. CARROLL</u>, a resident of the Borough of Manhattan, City, County and State of New York, hereby make, publish and declare this my <u>LAST WILL AND TESTAMENT</u> hereby revoking all former Wills by me made.

FIRST

I give and bequeath all my books, pictures, works of art, jewelry, wearing apparel, personal effects, household furniture, furnishings and utensils, silverware, plate linen, automobiles, farm and garden implements and equipmen and livestock, and all other tangible personal property to my wife, <u>NINA R. CANROLL</u>, to her own use absolutely-

8764

In case my said wife does not survive me, I give and bequeath such thereof as may at the time of my death be situated at my place, Doughoregan Manor, in Howard Count Maryland, to whichever of my children may take the portion of my said place described in Paragraph I of Article Second hereof, and I give and bequeath all the rest thereof to my Other children, in equal shares.

SECOND

I give and devise all my lands and real property



buildings and improvements thereon and all appurtehances, to my wife, <u>NINA R. OARHOLL</u>, for her life, with the powers hereinefter in Article Third hereof provided, and upon her death, I give and devise the same or any part thereof, to such of my children or lineal descendants as she may by Last Will and Testament appoint, in such shares and propor tions as she may designate.

In case my said wife does not survive me or in os having survived me, she fails in whole or in part effective to exercise the power of appointment hereinabove given to is with respect to my lands and real property in Howard County Maryland, then I thereupon dispose of my said lands and rea property, or such part thereof as may not be effectively disposed of by my said wife, as follows:

I. I give and devise Doughoregan Manor House, together with about seven hundred and fifty (750) acre more or less, of my surrounding lands, and all buildin and improvements thereon, and all appurtenances, constituting all my real property lying north of the center line of Paul's Lane extended to the easterly boundary of my property and being that portion of the

Sr(in)

property which was conveyed to me by Charles Carrol Jr. by Deed, dated February 24, 1923, and recorded the land records of Howard County on March 1, 1925; In Liber H. B. H. No. 117, Folio 113, or so much thereof as may not have been effectively appointed by my said wife, to the oldest of my children who may survive my said wife and myself, or if no child of mine shall so survive, to the oldest of my lineal descendants who may so survive.

II. I give and devise all my remaining lands and real property in Howard County, Waryland, togeth with all buildings and improvements thereon erected and all appurtenances, or so much thereof as may not have been effectively appointed by my said wife, in equal shares, to my son, <u>JOHN LEE CARROLL</u>, and my daughter, <u>MARY CARTER CARROLL</u>, or to the survivor or them if only one of them shall survive my wife and myself, or if neither shall so survive, to my son, <u>PHILIP OARROLL</u>, or if he shall not so survive, to the oldest of my lineal descendants then survivine. In case all my lineal descendants shall die durin the lifetime of my said wife, then, upon the death of the

8766

last survivor of such lineal descendants, I direct that property in which my said wife shall have taken a life of tate or interest hereunder shall vest in her, her heirs assigns absolutely, and I give, devise and bequeath the

same accordingly.

THIRD

I authorize and empower my said wife to sell, lease (for any term or terms, although the same may exten after her death), mortgage, exchange, or grant options wi respect to all or any part of the real property in which may hold a life estate pursuant to Article Second hereof, for such price and upon such terms and conditions as to credit or otherwise as in her absolute discretion she may deem advisable, and I direct that no purchaser or mortgag of any such property shall be required to look to the app cation of the purchase money or of any money advanced to wife upon security of the said property. Any money or pr perty representing the proceeds of any sale, mortgage or exchange of any of the property in which my wife may hold a life estate pursuant to Article Second hereof may be expended by my wife to the extent she deems desirable in

8767

the maintenance, repair and improvement of the remaining real property held by her, or may be invested by her in such stocks, common or preferred, bonds, notes or other property, real or personal, as in her absolute disarction she may deem advisable, and I give, devise and bequeath to my wife a life estate or interest in any of the said prooeeds so held or invested by her, and grant to her the power to dispose of the same upon her death to or among whichever of my children or lineal descendants are may by her Last Will and Testament appoint, in such shares and proportions as she may designate.

8768

I direct that no bond shall be required of my said wife in connaction with the life estates devised and bequeathed to her hereunder, and that she shall be under no inability for waste or for depreciation of any of the property, real or personal, in which she may at any time hold a life interest; or for any losses indurred by her in respect thereto, it being my intention, in view of changing conditions, that her powers and discretion with respect to the retantion, menagement or investment thereof shall be absolute.

FOURTH



taxes, or succession duties, upon any bequest or devise under this my Will shall be paid, without proration with respect thereto, out of my residuary estate.

PIFTH

I give, devise and bequeath all the rest, resident and remainder of my property, real and personal, of every kind whatsoever and wheresoever situated to my Trustees hereinafter named, <u>IN TRUST</u> however, for the uses and purposes following:

I. To apply the income thereof to the use o my wife, NINA R. CARDOLL, during her life.

8769

II. Upon the death of my said wife to convey transfer and pay over the principal to or among whichever of my children or lineal descendants she may by her Last will and Testament appoint, in such shares, estates and proportions, in trust or otherwise as she may designate.

III. In case my said wife does not survive me in case, having survived me, she fails in whole or in par affectively to exercise the power of appointment given to her in the preceding paragraph, then I give, devise and bequeeth my said residuary estate, or such part of the



principal thereof as my wife, having survived me, fails effectively to dispose of by will, to my lineal descendant in being at the death of the survivor of my wife and myse in equal shares per stirpes.

SIXTH

My Executors and Trustees shall have full power and authority, in their discretion, to hold and retain any the property coming into their hands hereunder in the sam form of investment as that in which it is received by the and to invest and reinvest the same and any trust funds he by them hereunder, in such amounts as they see fit, in su stocks, bonds, or other securities or property, real, or personal as in their discretion they may deem advisable, although the same may not be of the character permitted for trustees investments by the ordinary rules of law. They also, have full power and authority, in their discretion, sell, lease, improve, mortgage or exchange the whole or a part of the property belonging to me at the time of my de or at any time held by them hereunder, whether real or pe upon such terms and conditions as may to them seem advisa

8770

I also authorize and empower my Executors and

claims against or on behalf of my estate as they may deem expedient and to extend the time of payment; to consent to and participate in any reorganization, merger or other ohange of corporate structure and to deposit any property with any protective, reorganization or similar committee, and to delegate discretionary power thereto and to pay pe of its expenses; to exercise all conversion, subscription voting and other rights and to grant proxies discretionar or otherwise; to appoint agents to act in their behalf an to delegate discretionary powers to such agents and to register any property in the name of their nomines or hol it unregistered or in such other form that, title shall pe by delivery.

SEVENTH

I direct my Trustees to apply the entire income of all securities at any time held by them hereunder to t use of the beneficiery for whom they are held, irrespect of the price paid for them or of their market value at ar time; it being intended hereby that no part of such incom shall be applied as a sinking fund to offset any loss of



- 8



premium upon, or market value of, such securities; all stock dividends shall be treated as principal and added to the trust fund, so far as may be permitted by law, and all cash dividends, except liquidating dividends, shall be distributed to the income beneficiary.

Upon the termination of the trust estate for the benefit of my wife hereunder, interest, dividends and othe income accrued or declared but not yet due and payable, subject to any charges against it, shall not be paid to th estate of my wife but shall be distributed to those entitie to the principal of my estate, in the same shares, estates and proportions.

In the division or distribution of my estate or any trust under this Will, or any part thereof, including any distribution or payment to themselves as Trustees of the trusts under this Will, my Executors and Trustees may make division, distribution or payment in kind, or partly in kind and partly in money, and for any such purpose may determine the value of any property divided or distributed which determination shall, to the extent permitted by law be binding on all persons interested

EIGHTH

I direct that the provisions of this my will in favor of my said wife, shall be in lieu and par of all her right of dower in my estate.

NINTH

In the event that my said wife and I shall peris in or as a result of a common disaster so that it shall be difficult or impossible to ascertain which of us survived the other, I direct that for the purposes of this Will it shall be deemed that I survived my said wife.

TENTH

L'appoint my said wife, <u>ENA A. CAMOLE</u>, and my son, <u>WILLEP CARROLE</u>, Executors of and Trustees under this my will. In case either of the aforesaid Trustees, or an person appointed as an associate trustee, as hereinafter provided, should at any time fail to qualify or cease to as trustee hereunder for any reason. I authorize the othe trustee then acting by an instrument in writing, signed a acknowledged by him or her, to appoint an associate trust and any associate so appointed shall have the same rights powers and obligations as if originally appointed in this Will, but bending such appointment of an associate trustne any remaining trustee shall have full authority to act as sole tructee hereunder for as long as he or she may deem

desirable. I direct that no security for the faithful performance of his or her auties be required of either Executorior Trustee hereby appointed, nor of any associate trustee, whether a child of mine mentioned in this Will or other person appointed as above provided; should by law security be required of any associate trustee, I request th such security be in a nominal amount.

Flexing Q Carrier

The FORTGOING INSTRUMENT was, on the day of the date thereof, signed, sealed, published and declared by PHILIP A. CARROLL, the Testator therein named, as and for his EAST WILL AND TESTAMENT, in the presence of us, who, at his request, in his presence and in the presence of each other, have hereunto subscribed our names as witnesses.

M. Bandenet 1802 Hear

I. PHILIP A. CARROLL, nereby make, publish and ceclare this to be a Codicil to my Last Will and Testament eaving date the 9th day of May, 1946.

<u>EIRST</u>

I liereby grant to my wlfe, NINAIR, CARROLL a powe to appoint to such person or persons, including the estate of my said wife, and in such shares and proportions as my sald wife may designate at any time and from time to time during her life by Deed or by her Last Will and Testament any or all of my lands and real property situated in Howa County, Maryland, Logether with any or all buildings and improvements thereon and appuntenances thereto, and any or all money or property representing the proceeds of any sale, mortgage on exchange of any of such real property, or any investment or reinvestment thereof, in which my wife may hold a 11 Perestate pursuant to Article SECOND or Article THIRD of my said will.

SECOND

glve and bequeath any and all commissions to

as Thustee of any trust find administered by me during my

1120, in squal shares, to such of my wife, NHWA R. CARROLE,

and my children, as shall survive me; provided, howev that if any child of mine should predecesse m Jeavin lineal descendant or lineal descendants me surviving, such lineal descendant or lineal descendants shall take equal parts, per stinges, the share or the whole or such

commissions to which such deceased child of mine would have been entitled, if he or she had survived me.

THIRD:

have been entitled, if he or she

give and begieath any and all sums to which Leam entibled, on to which my estate may become entitled reason of my membership in the firm of Shearman & Sterlin Wright, in equal shares, to such of my wife. MINA A. LOARROL nd my children, as shall survive me; providel, however, c 10 any child of mine should predecease me leaving a lineal descendant of lineal descendants me surviving, such lineal descendant of lineal descendants shall take, if equal part ber stippes, the share or the whole of such partnership istributions to which such deceased child of mine world

- POURTH

high declare what said will and this codicil together constitute my Last will and Testamont.-

TH WITHERS MERCOF, I have hereintd set my hand

and soal this

As thus amonded, I hereby ratify, republish and

confirm my aadd will bearing date the 9th day of May ... 19

day of Cotober

The reardoing instraiment was, on the day of date thereof, signed, sealed, rablished and declared by PHILLP A. CARROLL, the Testator therein named, as and for a CODICIL to his Last Will and Testament, bearing date Nay 9, 1940 - In the presence of us, who, at his request, in this presence and in the presence of each other, have hereunto subscribed our

names as witherses

posiding at 99/201

pessiding, at

residing at

<u>7868</u>	late of MINA B	GAROLL !		No. 1117 and	
ale of MEN YORK,	NEW YORK	died	FEBRUARY 11	-19-19) testa	te
Sons to whom leiters were granted	амоцит Сноя то	SURETIES	8	SOLICITORS	
P A, CARROLL Carter Carroll Ziegi Lee Carroll Dudert Brothers Ark Avenue Drk, New York 10136	•	:		JOHN C. NURPHY 516 N. Charles Street Baltimore, Naryland (301) 625-4828	2120
		•			, .
,	PROCEEDINGS .		NAME OF VOLUME	LIBER No. FC	L10
INITIAL A RESIDENT AGENT F AND HARY	COPY) RTIFJED COPIES) OF PROCEEDINGS FROM PPDINTNENT ORMS FOR JOHN LEE CA CARTER CARROLL ZIEGI TS OF MARYLAND PROPI	H ARROLL "	PROCEEDINGS	·	
NOTICE TO CREDIT	•	• KIT .*		•	• _
APPRAISAL OF MAR	YLAND ASSETS	*	'n		
HARYLAND ESTATE	TAX RETURN	, R		Marcon	
Amended Maryland 1	Batate Tax Return ·			MICROFICE	ТЕ,
Application by Fo Interitance Tax	reign Per. Rep to F	'ix		- METS	
Notice of Inchoat FROM DEPT OF TREA CERTIFICATE DISCHARGE	e Lien SURY-IRS- UNITED STA NG PROPERTY SUBJECT TO 1	TES ESTATE TAX LI	Cen		
N	l. N	Ĵ.	-		
					· .
	Samt me tillari i	F. Parce	N		•
·					,
			-		
I, NINA RYAN CARROLL, a resident of the City, County and the State of New York, do make, publish and declare this my LAST WILL AND TESTAMENT, hereby revoking all former Wills and Codicils made by me.

FIRST

By Article TENTH of an Indenture of Trust dated February 18, 1942, between DOROTHY H. RYAN as Grantor, and me, NINA M. RYAN, as Trustee, I am vested with the power to appoint a Successor Trustee to succeed me in the event of my death prior to the termination of the Trust, with all of the rights, duties, powers and immunities conferred upon the Trustees therein by the said Trust Indenture. Pursuant to such power, I hereby appoint my son, PHILIP CARROLL, as Successor Trustee to me with all of the powers, rights and duties possessed by me as Trustee.

SECOND

4642

A. Since my son PHILIP CARROLL now owns the tangible contents of Doughoregan Manor in Howard County, Maryland and is now farming the surrounding acres, I give and bequeath to him, but not by way of limitation, all farm and garden implements, equipment, livestock, and produce, together with all other tangible personal property there located used in connection with the farming operation. But if he is not and the state of the second second second second second second second second second second second second second

living then I give and bequeath the same to his son, PHILIP D. CARROLL; but if he is not living then to the Trustees of the Carroll Trust which is hereinafter provided for in subparagraph (c) of paragraph II of Article Third.

B. I give and bequeath the contents of my apartment at 33 East 70th Street, New York, N.Y. as follows:

- 1. To my son, PHILIP CARROLL, the two Shakespeare pictures, now in the living room, constituting part of a set at The Manor and all blue, white, gold and red china compotes, dishes, plates, urns, etc., constituting part of a set now at The Manor.
- 2. To my son, JOHN LEE CARROLL, the painting of The Manor by Felix Kelly now in the library of my apartment.
- 3. To my children, PHILIP CARROLL, JOHN LEE CARROLL and MARY CARTER CARROLL ZIEGLER, all my furs, jewelry, clothing and other personal effects, to be allocated among them as they should agree among themselves.
- 4. All the rest and remainder of the contents of my apartment, including, but not by way of limitation, the silverware, glass, china, furniture, pictures, rugs, books, ornaments and all other household goods and furnishings to my children, JOHN LEE CARROLL and MARY CARTER CARROLL ZIEGLER, to be allocated between them as they should agree.

In the event that any child of mine should predecease me, leaving children him or her surviving, then anything in this paragraph B to the contrary notwithstanding, such children shall receive and share equally the articles which my child would have received had he or she survived me.

4643

-2-

THIRD

In and by Article SECOND of the Last Will and Testament of my late husband, PHILIP A. CARROLL, dated May 9, 1946 together with Article FIRST of the Codicil thereto dated October 16, 1953, both of which were duly admitted to probate in the Surrogate's Court of New York County, New York, I was given a life estate in all of his real property situated in Howard County, Maryland, with all improvements thereon together with a general power to appoint the said property by Deed or Will.

Pursuant to the aforesaid power vested in me, I hereby appoint any and all of the aforesaid CARROLL lands and real property situated in Howard County, Maryland, together with any and all buildings and improvements thereon and appurtenances thereto, and any and all money or property representing the proceeds of any sale, mortgage or exchange of any such real property, and any investment or reinvestment thereof as follows:

4644

. .ť.

3

I. I appoint Doughoregan Manor House together with about seven hundred and fifty (750) acres, more or less, of the surrounding lands (said house and surrounding lands being hereinafter referred to as "The Manor"), and all buildings and improvements thereon and all appurtenances, constituting that part of the real property lying north of the center line of Paul's Lane extending to the easterly boundary of said property, being that portion of said property which was conveyed to PHILIP A. CARROLL, by CHARLES CARROLL, JR., by Deed, dated February 24, 1923, and recorded in the land records of Howard County on March 1, 1923, in Liber H. H. No. 117, Folio 113, to my son, PHILIP CARROLL, if living, or if he does not survive me, to his son, PHILIP D. CARROLL, if living, or, if PHILIP D. CARROLL does not survive me, to my son, JOHN LEE CARROLL, and my daughter, MARY CARTER CARROLL ZIEGLER, as trustees in trust however to hold and administer as provided in sub-paragraph (c) of paragraph II of this Article Third known as The Carroll Trust.

II. All of the remaining lands and real property in Howard County, Maryland over which I have the aforesaid power of appointment together with all buildings and improvements thereupon erected and all appurtenances and any and all money or property representing the proceeds of any sale, mortgage or exchange of any such property, or any investment or reinvestment thereof and I appoint as follows:

(a) I appoint an undivided one-fourth interest therein to my daughter, MARY CARTER CARROLL ZIEGLER, for her own use absolutely, if living, but if she is not living, I appoint the said interest to the trustees of The Ziegler Trust provided for in the following sub-paragraph (b) as an addition to that trust.

-4-

I appoint an undivided one-fourth interest (b) to my son, JOHN LEE CARROLL, and my daughter, MARY CARTER CARROLL ZIEGLER, as trustees of a trust (The Ziegler Trust) in trust nevertheless to hold, manage, invest and reinvest the same and to collect the income therefrom and to pay or apply the net income at least quarterly to, or to the use of, such members of a class composed of the children of my daughter, MARY CARTER CARROLL ZIEGLER, in such amounts and proportions as my trustees, in their discretion, shall deter-In disbursing the income mine from time to time. the trustees, in their uncontrolled discretion, may pay to, or apply the same to the use of, any one member of the class or apportion it for the . benefit of various members of the class to the exclusion of other members in such manner as they shall, from time to time, deem advisable without equality of treatment. Any net income not so distributed shall be added to the principal.

. . .

朝鮮情報投资**的新教育的**自然的教育和自然的特征和社会。1997年1997年19

人名住 把原制 打击

Upon the death of the last surviving descendant of my husband, PHILIP A. CARROLL, living at the time of his death in 1957, or upon the twentieth anniversary of my death, whichever event first occurs, or upon the death of the last survivor of the children of MARY CARTER CARROLL ZIEGLER prior

4646

thereto; this trust shall terminate and the trustees shall deliver and pay the principal then remaining in equal shares to each of the children of MARY CARTER CARROLL ZIEGLER then living, if any, and to the estate of each deceased child of MARY CARTER CARROLL ZIEGLER. 21.026:16

In the event neither my son, PHILIP CARROLL, (c)nor his son, PHILIP D. CARROLL, shall survive me I appoint an undivided one-half interest to my granddaughter, CAMILLA CARROLL, for her own use absolutely or to her estate if she should predecease me, but if either my aforesaid son or grandson should survive me, then in either such event the above appointment to my granddaughter, CAMILLA CARROLL shall be cancelled and shall be of no effect and in lieu thereof I appoint the aforesaid undivided one-half interest to my son, JOHN LEE CARROLL, and my daughter, MARY CARTER CARROLL ZIEGLER, as trustees of a trust (The Carroll Trust) in trust nevertheless to hold, manage, invest and reinvest the same to collect the income therefrom, and to pay or apply the net income therefrom at least quarterly to, or to the use of, such members of a class composed of the children of my son, JOHN LEE CARROLL, in such amounts and proportions as my trustees, in their discretion shall determine from In disbursing the income the Trustees, time to time.

in their discretion may pay to, or apply the same to the use of, any one member of the class or apportion it for the benefit of various members of the class to the exclusion of other members in such manner as they shall, from time to time, deem advisable without equality of treatment. Any net income not so distributed shall be added to principal. 法自己中国地。

Upon the death of the last surviving member of the descendants of my husband, PHILIP A. CARROLL, living at the time of his death in 1957 or upon the twentieth anniversary of my death, whichever event first occurs, or upon the death of the last survivor of the children of JOHN LEE CARROLL prior thereto, this trust shall terminate and the Trustees shall deliver and pay the principal then remaining in equal shares to each of the children of JOHN LEE CARROLL then living if any and to the estate of any deceased child of JOHN LEE CARROLL, provided however that if The Manor be included in said principal then and in that event The Manor shall be conveyed to such male child or children of JOHN LEE CARROLL as the trustees shall designate and to be charged against the distribution share or shares of said male or males so designated.

¢.

1648

FOURTH

A large portion of the Maryland real property subject to

-- 7 --

%防盗理或通货等到把按指用按照相关提供rr-1 ·····

my aforesaid power of appointment is now subject to an easement granted by me to Maryland Historical Trust. Since all of the various parcels of real property subject to my power of appointment are contiguous or nearly so, and have been and now are being maintained and operated as a unit primarily as a farm, it is my wish that the beneficiaries under paragraph II of Article Third herein coordinate their efforts with those of the owner of Doughoregan Manor to the extent that the law allows so that the operation of the various parcels will be devoted to the same enterprise if possible and as much as possible.

FIFTH

I appoint any and all property over which I may have a power of appointment by Will under Article Fifth of the last Will and Testament of my husband, the late PHILIP A. CARROLL, into three equal shares as follows:

I. I appoint one such share to my son, PHILIP CARROLL, absolutely, if he survives me, but if he does not, then to his son, PHILIP D. CARROLL and his daughter, CAMILLA CARROLL, in equal shares but if either predeceases me leaving descendants, such descendants shall take the share which my grandchild would have received had he or she survived me, but if there be none, to his or her estate.

4649

-8-

II. I appoint one such share to my son, JOHN LEE CARROLL, absolutely, if he should survive me, or if he should predecease me, to his descendants me surviving, in equal shares per stirpes.

1.20

的现在分词的影响的制作的。

III. I appoint one such share to my daughter, MARY CARTER CARROLL ZIEGLER, absolutely, if she should survive me, or if she should predecease me, to her descendants me surviving, in equal shares per stirpes.

IV. If the appointment of any share of the property over which I have a power of appointment by Will under Article Fifth of said Will of said PHILIP A. CARROLL should lapse or fail then I appoint such lapsed or failed share to the persons entitled to receive the balance of such property under the foregoing provisions of this Article in the proportions which their respective shares bear to each other.

4650

SIXTH

ALL THE REST, RESIDUE AND REMAINDER of my estate of whatsoever kind and wheresoever situated, I give, devise and bequeath as follows:

A. One third (1/3) thereof to my son, PHILIP CARROLL, if living, but if he is not living, then to his son, PHILIP D. 睜

ne herendense 2

CARROLL and his daughter, CAMILLA CARROLL, in equal shares, but if either of them be not living, then his or her share shall go to his or her descendants if any, me surviving and if none, then to his or her estate.

B. One-third (1/3) thereof to my son, JOHN LEE CARROLL if living, but if he is not living, to his descendants me surviving in equal shares per stirpes.

C. One-third (1/3) thereof to my daughter, MARY CARTER CARROLL ZIEGLER, if living, but if she is not living, then to her descendants me surviving in equal shares per stirpes.

D. Should any child of mine predecease me, leaving no descendants me surviving, then the share which the child would have received had he or she survived me, I give to my descendants who survive me in equal shares per stirpes.

۰,

SEVENTH

I direct that all estate, inheritance and succession taxes including interest assessed against my estate by federal or state governments with respect to property required to be included in my gross tax estate passing by this Will or otherwise shall be charged to and paid out of the principal of my residuary estate without contribution thereto from the persons benefited, but if such taxes exceed my residuary estate after the payment of debts and administration expenses, then and in that event I

-10-

51

direct that the excess tax be charged to the beneficiaries under paragraph II of Article Third herein in proportion to the value of their respective benefits received, such value to be based upon that value assessed in the federal estate tax proceeding in my estate. 二世 化乙基甲酰胺

EIGHTH

I hereby appoint my sons, PHILIP CARROLL and JOHN LEE CARROLL, and my daughter, MARY CARTER CARROLL ZIEGLER, to be executors of this my Will and I direct that no bond or other security be required of any of them as executors or as trustees of any trust that may come into existence under Article Tenth of this Will.

In and by Article Third I have appointed my son, JOHN LEE CARROLL, and my daughter, MARY CARTER CARROLL ZIEGLER, as trustees of the trusts created therein. I direct that neither of them shall be required to file a bond or other security as trustee for the performance of his and her duties as such.

In the event of the death, resignation or failure to qualify of any of my aforesaid children as executor or trustee I authorize and empower that child to appoint his or her own mulmillute trustee or nucceanor trustee. In the absence of much appointment I authorize and empower the other executors or trustees to appoint a substitute or successor executor or trustee as the case may be. Any appointments provided for

-11-

12

herein shall be made in writing and shall be filed in the court where this Will is probated.

NINTH

In addition to the powers vested in them by law, I grant to my executors and to my trustees (as to the latter in addition to the powers set forth in Article FOURTH herein), with respect to any and all property, whether real or personal belonging to me at the time of my death or which may at any time be held by them hereunder, the following powers which may be exercised at any time during the administration of my estate, the continuance of any trust hereunder until actual distribution of all property in their hands:

(1) To retain such property as an investment without regard to the proportion scuh property or property of a similar character, so held, may bear to the entire amount of my estate, or of the trust in which such property is held, and whether or not the same be within the class of securities in which Toustees are authorized by law or any rule of court to invest trust funds.

(2) To sell such property at either public or private sale for cash or on credit, to exchange such property, and to grant options for the purpose thereof.

4653

...

(3) To invest and reinvest in any property, in which a reasonably prudent person would invest, including but not by way of limitation, bonds, notes, debentures, mortgages, certificates of deposit, common and preferred stocks, and shares of interests in investment trusts, without being limited to the class of securities in which trustees are authorized by law or any rule of court to invest trust funds.

(4) To consent to and participate in any reorganization, consolidation, merger, combination, dissolution, change of capitalization or other changes of corporate structure, or any plan for any such change, and to consent to any contract, lease, mortgage, purchase, sale or other action by any corporation pursuant thereto.

1654

(5) To deposit any such property with any protective reorganization or similar committee, to delegate discretionary power thereto, and to pay part of its expenses and compensation and any assessments levied with respect to such property.

-13-

UNINE STORE STREET, STATE STREET, ST

(6) To exercise all conversion, subscription, voting and other rights of whatsoever nature pertaining to any such property, and to grant proxies, discretionary or otherwise, in respect thereto. 4.9<u>65-9</u>29-1-

(7) To manage any such real property as if the absolute owner thereof, including but not by way of limitation the power to lease, or grant options to lease the same for any term or terms, although in excess of five years, and although any such terms may extend beyond the period of the trust under this my Will, without application to any court, and to enter into any covenants or agreements relating to the property so leased or any improvements which may then or thereafter be created thereon.

(8) To borrow money, and to pledge or mortgage any such property, for any purpose.

(9) To satisfy any general legacy or any part thereof, including any legacy to themselves as Trustees of the trusts under this Will, in kind; and in the division or distribution of my estate or the trusts under this Will, or any part thereof, including any distribution or payment

4655

Ľ

to themselves as Trustees of any trust under this Will, to make partition, division or distribution of property, or such payment, in kind, and for any such purpose to determine the value of any such property, which determination shall, to the extent permitted by law, be conclusive and not subject to review.

(10) To apportion between income and principal charges or expenses incurred by, or taxes assessed against my estate or the trusts hereunder, or to allocate the same wholly to principal or wholly to income, and any such apportionment or allocation shall, to the extent permitted by law, constitute full protection to my said Executors and Trustees with respect to any action taken or payments made in reliance therein.

(11) To appoint agents to set in his or their behalf, and to delegate discretionary powers to such agents.

(12) To extend the time of payment of any obligation at any time owing by or to my estate, or the trusts under this Will and to compromise, settle or submit to arbitration upon such terms as they may deem advisable, or to release with or without consideration, any claim in favor of

1656

nenerita shi Uradikashi in

or against my estate or the trusts under this Will.

(13) To do all such acts, take all such proceedings and exercise all such rights and privileges, although not hereinbefore specifically mentioned, with relation to any such property as if the absolute owner thereof, and in connection therewith to make, execute and deliver any instrument and to enter into any covenants or agreements binding my estate or the trusts under this Will.

In distinguishing between income and principal hereunder, my Trustees shall not be required to amortize premiums paid for trust securities, and all stock and extraordinary dividends shall be treated as principal and added to the trust fund so far as permitted by law.

In the administation of my estate and of the trust hereunder, my Executors and my Trustees are hereby authorized to deposit any securities held by him or them in a custodian account in a bank or trust company, and to cause any such securities so deposited by them to be registered and held in the name of the nominee of such depositary without words indicating fiduciary capacity, and to pay the custodian fees . and other charges of such bank or trust company for the custody of such securities, the collection of income and other

-16-

custodian services, and to charge the same against income of the estate, or of the trust as the case may be, and I direct that no part of such charge shall be deducted from or applied against the compensation of my individual Executors or my individual Trustees hereunder.

No one dealing with my Executors or my Trustees shall be obliged to look to the application of any moneys or other properties paid or delivered to my Executors or my Trustees.

TENTH

If at any time any or all the principal of my residuary estate or any property, real or personal, appointed by any other Article of this my Will would, but for the provisions of this paragraph, vest in a minor, then anything hereinabove contained to the contrary notwithstanding, I give, devise, bequeath and appoint such property to my sons, PHILIP CARROLL and JOHN LEE CARROLL, and my daughter, MARY CARTER CARROLL ZIEGLER, as Trustees, IN TRUST, to hold, manage, invest and reinvest the same, to collect and receive the income therefrom and of the principal thereof as they in their absolute discretion shall deem advisable for the

-17-

support, maintenance and education of such minor, and to accumulate any income not so applied until such minor shall attain the age of twenty-one years; and upon such minor attaining such age, my Trustees shall transfer and pay over to such minor the principal and accumulated income then in their hands hereunder; or in the event of the death of such minor prior to attaining the age of twenty-one years, my Trustees shall thereupon transfer and pay over the principal and accumulated income then in their hands hereunder to the personal representative of such minor.

In any case, where my Trustees are authorized to make payment to a minor, my Trustees may make application of income or principal to or for the use of such minor either by payment directly to such minor or by payment for the benefit of such minor; to the parent of such minor or to the guardian of such minor or to the person with whom such minor may reside, or by paying the expenses of such minor, to such minor and the receipt of such minor, parent or guardian or such other person for all payments so made or other evidence of such expenditures shall be sufficient to discharge my Trustees from further liability for the amount so paid or expended.

ELEVENTH

Except as expressly provided herein by Article THIRD and FIFTH it is my intention not to exercise by this Will any power of appointment by Will which I may have at the time of my death.

IN WITNESS WHEREOF, I have hereunto set my hand and seal this 4th day of June 1979.

MAARIN Paperi Cand all (L.S.).

THE FOREGOING INSTRUMENT, was, on the day of the date thereof, signed, sealed, published and duclared by NINA RYAN CARROLL, the Testatrix therein named, as and for her LAST WILL AND TESTA-MENT, in the presence of us, who at her request, in her presence, and in the presence of each other, have hereunto subscribed our names as witnesses.

residing at <u>3)</u> residing at 41 /1/and Gunary Timble residing at 119 a word R. M. M. 105553 Campord Com 06965

I, Nina Ryan Carroll, a resident of the City, County and State of New York do make, publish and declare this as a Codicil to my Last Will and Testament dated June 4, 1979.

FIRST: I hereby revoke paragraph I of Article Third of my will and declare in its place and stead the following:

(a) I appoint Doughoregan Manor House together with about seven hundred and fifty (750) acres, more or less, of the surrounding lands (said house and surrounding lands being hereafter referred to as "The Manor"), and all buildings and improvements thereon and all appurtenances, constituting that part of the real property lying north of the center line of Paul's Lane extending to the easterly boundary of said property, being that portion of said property which was conveyed to PHILIP A. CARROLL, by CHARLES CARROLL, JR., by Deed, dated February 24, 1923, and recorded in the land records of Howard County on March 1, 1923, in Liber H. H. No. 117, Folio, to my grandson Philip D. Carroll and my granddaughter Camilla Carroll in equal shares as tenants in It is my wish that in the event that my granddaughter at common. anytime be vested with sole ownership of The Manor she will observe the Carroll family tradition that she exchange with some other male Carroll'descendant her interest in The Manor for other Howard County realty of then-equal value owned by such other male descendant.

4661 (b) If neither grandchild survives me, then in that event I appoint The Manor as hereinabove describe to my son Philip Carroll and to his estate.

SECOND

angalan di mananan a dina di kara a dina di kara anan anan sata a dina anan ana

As thus amended I hereby ratify and confirm my said Will of June 4, 1979 and declare that the Will and this Codicil together constitute my Last Will and Testament.

In Witness Whereof I have hereunto set my hand and seal this 25 day of Netry, 1986.

Man R. Lamolle.s.)

THE FOREGOING INSTRUMENT, was, on the day of the date thereof, signed, sealed, published and leclared by NINA RYAN CARROLL, the Testatrix therein named, as and for a codicil to her LAST VILL AND TESTAMENT, in the presence of us, who at her equest, in her presence, and .n the presence of each other, ave hereunto subscribed our ames as witnesses.

1464 residing at 220 E916 St 14/10/28

_____residing at 1763. 2 4 4. N 1C N / 10/28 Elane M' Smath,

4662

Generic T Shester residing at 710 Mart Lane

Stampers: (1 06905

SECOND CODICIL

TO

LAST WILL AND TESTAMENT

OF

NINA RYAN CARROLL

I, NINA RYAN CARROLL, domiciled in the City, County and State of New York, do hereby make, publish and declare this to be the Second Codicil to my Last Will and Testament dated June 4, 1979, as amended by my First Codicil dated November 28, 1986.

17

3,0,

I delete the last sentence of Paragraph A of Article SECOND of my said Will and substitute therefor the following:

ĭ.

"But if my son, PHILIP CARROLL, shall not survive me, then I give and bequeath the same to such of his lineal descendants as shall survive me, in equal shares, per stirpes."

I hereby amend Article THIRD of my said Will, as amended by my said First Codicil, in its entirety; Article THIRD shall now read as follows:

II.

"In and by Article SECOND of the Last Will and Testament of my late husband, PHILIP A. CARROLL, dated May 9, 1946, together with Article FIRST of the Codicil thereto dated October 16, 1953, both of which were duly admitted to probate in the Surrogate's Court of New York County, New York, I was given a life estate in all of his real property situated in Howard County, Maryland, with all improvements thereon together with a general power to appoint the said property by Deed or Will. "Pursuant to the aforesaid power vested in me, I hereby appoint any and all of the aforesaid CARROLL lands and real property situated in Howard County, Maryland, together with any and all buildings and improvements thereon and appurtenances thereto, and any and all money or property representing the proceeds of any sale, mortgage or exchange of any such real property, and any investment or reinvestment thereof as follows:

nie ander minnen inden und balle influende burben mes bit

"I. I appoint Doughoregan Manor House together with about seven hundred and fifty (750) acres, more or less, of the surrounding lands (said house and surrounding lands being hereinafter referred to as 'The Manor'), and all buildings and improvements thereon and all appurtenances, constituting that part of the real property lying north of the center line of Paul's Lane extending to the easterly boundary of said property, being that portion of said property which was conveyed to Philip A. Carroll, by Charles Carroll, Jr., by Deed, dated February 24, 1923, and recorded in the land records of Howard County on March 1, 1923, in Liber H. H. No. 117, Folio 113, as follows:

"A. If I shall die before January 1, 1990, to such of the lineal descendants of my son, PHILIP CARROLL, as shall survive me, in equal shares, per stirpes, as tenants in common, but if there be none, to my son, PHILIP CARROLL, if he shall survive me.

"B. If I shall not die before January 1, 1990, to my son, PHILIP CARROLL, if he shall survive me, but if he does not, to such of his lineal descendants as shall survive me, in equal shares, per stirpes, as tenants in common.

"II. I appoint all of the remaining lands and real property in Howard County, Maryland, over which I have the aforesaid power of appointment together with all buildings and improvements thereupon erected and all appurtenances and any and all money or property representing the proceeds of any sale, mortgage or exchange of any such property, or any investment or reinvestment thereof, as follows:

"A. An undivided one-fourth interest therein to my daughter, MARY CARTER CARROLL ZIEGLER if she shall survive me, but if she does not, to such of her lineal descendants as shall survive me, in equal shares, per stirpes, as tenants in common.

"B. An undivided one-fourth interest therein (i) if I shall die before January 1, 1990, to such of the lineal

- 2 -

4664

12.51

descendants of my daughter, MARY CARTER CARROLL ZIEGLER, as shall survive me, in equal shares, per stirpes, as tenants in common; or (ii) if I shall not die before January 1, 1990, to my daughter, MARY CARTER CARROLL ZIEGLER, if she shall survive me, but if she does not, to such of her lineal descendants as shall survive me, in equal shares, per stirpes, as tenants in common.

//经济中/风中与

r,

1665

te training of

አንሮቹ የሆኑን ምንድ የአንፍ የብዙ የምግር ላይ

"C. An undivided one-half interest therein (i) if I shall die before January 1, 1990, to such of the lineal descendants of my son, JOHN LEE CARROLL, as shall survive me, in equal shares, per stirpes, as tenants in common; or (ii) if I shall not die before January 1, 1990, to my son, JOHN LEE CARROLL, if he shall survive me, but if he does not, to such of his lineal descendants as shall survive me, in equal shares, per stirpes, as tenants in common.

"III. Should any child of mine predecease me, leaving no descendant surviving me, then the share which the child would have received had he or she survived me, I give to my descendants who survive me, in equal shares, per stirpes.

In the event that distribution of the property "IV. described in Paragraph I of this Article would be deemed a generation skipping transfer for federal tax purposes, I hereby direct my executors to allocate to said property the One Million Dollars (\$1,000,000) exemption applicable to generation skipping transfers pursuant to Section 2631. of the Internal Revenue Code, as amended. If said distribution would not be deemed a generation skipping transfer for federal tax purposes or to the extent the exemption is not needed to prevent imposition of the generation skipping tax, I hereby direct my executors to allocate said One Million Dollars (\$1,000,000) exemption, or the remaining amount of the exemption, among any property described in Paragraph II of this Article deemed to be generation skipping transfers for federal estate tax purposes in proportion to the value of said property."

III.

I amend Article EIGHTH by revoking in its entirety , C, the second paragraph of said Article. I amend the third paragraph of said Article by deleting from the first sentence the phrase, "substitute trustee or successor trustee", and

- 3 -

substituting therefor the phrase, "substitute or successor executor or trustee".

IV. I amend the first paragraph of Article NINTH by deleting the phrase "(as to the latter in addition to the powers set forth in Article FOURTH herein).".

I amend the first paragraph of Article TENTH of my said Will by deleting the phrase, "appointed by any other Article of this Will", and substituting therefor the phrase, "appointed by Article FIFTH or SIXTH of this Will".

v.

VI.

I hereby revoke my First Codicil dated November 28, 1986, to my said Last Will and Testament in its entirety being an amendment to Article THIRD which has been wholly restated in this Second Codicil.

4666

, _B, <.

VII.

As thus amended, I hereby ratify and confirm my said Will of June 4, 1979 and declare that the Will and this Second Codicil together constitute my Last Will and Testament.

In Witness Whereof I have hereunto set my hand and seal this $\int (2) day$ of April, 1987.

MANG (IN AN CANDOL (L.S.)

THE FOREGOING INSTRUMENT, was, on the day of the date thereof, signed, sealed, published and declared by NINA RYAN CARROLL, the Testatrix therein named, as and for a codicil to her LAST WILL AND TESTAMENT, in the presence of us, who at her request, in her presence, and in the presence of each other, have hereunto subscribed our names as witnesses.

1 to 1		500 I D
10002 100000	residing	at 67-66 108 49 At.
$\Delta $ - 1	•••	Forest Hill, N.Y.
Marry Z. Vied	residing	at 311 Cast Tandst.
1. NOO		Aber glock 114
KUJA 12. Mamer.	residing	at RY West CCT ST,
		New York, N.Y.
· · ·		¢

4667

的考虑到在这

STATE OF NEW YORK

95.;

COUNTY OF NEW YORK

Each of the undersigned, individually and severally, being duly sworn, deposes and says:

The within Codicil was subscribed in our presence and sight at the end thereof by NINA RYAN CARROLL, the within named testatrix, on the 10^{14} day of April, 1987, at 33 East 70th Street, Apartment 6F, New York, New York 10021.

Said testatrix at the time of making such subscription declared the instrument so subscribed to be her Codicil.

Each of the undersigned thereupon signed his and her name as a witness at the end of said Codicil, at the request of said testatrix and in their presence and sight and in the presence and sight of each other.

Said testatrix was, at the time of so executing said Codicil, over the age of eighteen years, and, in the respective opinions of the undersigned, of sound mind, memory and understanding and not under any restraint or in any respect incompetent to make a Codicil.

Said testatrix, in the respective opinions of the undersigned, could read, write and converse in the English language and was suffering from no defect of sight, hearing or speech, or from any other physical or mental impairment which would affect her capacity to make a valid Codicil. The Codicil was executed as a single, original instrument and was not executed in counterparts.

Each of the undersigned was acquainted with said testatrix at such time, and makes this affidavit at her request.

The within Codicil was shown to the undersigned at the time this affidavit was made, and was examined by each of them as to the signature of said testatrix and of the undersigned.

The foregoing instrument was executed by said testatrix and witnessed by each of the undersigned affiants

· • • •

under the supervision of <u>Shirun</u> Laurer

さ.

an

,it

:

N. 19.

Severally sworn to before me this $\underline{/\nu t \lambda}$ day of April, 1987

ë.l.l Notary Public

ELFINE C. REELT Notary Public, State of New York No. 41-471 3174 Qualified in Onlocos Caunty Commission Expires May 31, 1928

- 2 -

ESTATE OF NINA R. CARROLL

The Maryland assets consist of tangible and intangible personal property. The list of recipients under the will and codicils and their relationship is as follows:

Son: Philip Carroll P.R. * 3500 Menor Lane Ellicott City, Md. 21043

Son: John Lee Carroll P.K 215 East 72nd Street New York, New York 10021

Daughter: Mary Carter Carroll Zeigler P.R. P.O. Box 146 Free Union, Va. 22940

Grandchildren:

Philip D. Carroll * Apt. 252 1670 El Camino Real Menlo Park, CA 94025

Camilla Carroll 4 USARU APO New York 09053

John Lee Carroll, Jr. Box 199 Queenstown, Md. 21658[.]

Thomas T. Carroll 215 E. 72nd Street New York, New York 10021

Genevieve Anne carroll

MSA CE53-78

1999 - C.		
	PHONAN W. Johon D. Downwith Fon AND IN DOMITINGIANION ON One mit no 7100(21,00). Doitons	
	Wight or my may to us in hand paid shifts the reading or which is hardby he	
2 32 g . 3	"It navious Grach. Anon buyen, and a durther consideration of 50 conte por red.	
	for onen rollor pipe inid on herein elevation pression of the battors and pipe	
	alind 10 with we drives the Doundell where P. O. Donnell or "Alligate dity do interby wrant.	が開
24 14. - 14. - 14.	fo R. Revisiond Oronia nin noirs or analtan, the blint of way to any maintain operated	
	and repays a pine line. for the transportation of bill or gap and graot, built in and	なぞう
	The of the state of the selection of the selection and the round motoranty on a wood	何前
1777-1505. 1777-1705	and the oun our loude, " althore in Third bigsplot Howard Donney near or hanvindersonfa	\$1.77 (K SP
	- and address who an assistant on the North by Lana or raylor similar the work by the	
	and is if 100 Darrolly on the South by ladidat of Urilloo Darrolly, on the weat by links	
	at toy tor most with increase and orders be and tron bus sense who hald granter notice	
	a and the second of the second of the third invariance . Scoops for the hirdwoor nervolve	
	1 Oto is infinited to the half it. Investing Orosal, who insidely across to very ally unknows sings it	•••
	1917 Classific propriation for the indication in the indication of the provided on the state of the indication in the indication of the in	,
	AND NIDE CINE, Seid datember it not suturilly arroad upon, do be abourtailed and de-	•
	in all of py shree distinction for normany, one thousand to be appellined by the sale years	
	and the on oralization one by its respond theory, him house on and the problem of the second of the	
	and a second second second second second second second second second second second second second second second	•
	and a second second second second second second second second second second second second second second second Second second	
	Train Dilaras honoain provined unon the prometric allo price per una aparts said and all or said a set and a	
	and a second second second second second second second second second second second second second second second	
	and a second stand of the second of the second of the second second second second second second second second s	
	The second state of the se	
	timen der de und her vorge in monante or	
	and a second second second second second second second second second second second second second second second	
	Approximation and the second second second second second second second second second second second second second	
	Marine Manual Toward County . Townin	
	reicoly foorf it which is a bind its cars of Augules in the your hime, purpre us, the Sanar	
	the thousand the thousand the thousand the thousand the notion	
1. C.S.		
	den vo 101 Feanra 3 Heatr. 1909 ht 11:0 0100K All hand day resound and acailing	
	In the form for the second second	
and the second		

MARYLAND HISTORICAL TRUST

LIBERID 820 FOLID 542

THIS DEED OF EASEMENT, made this 23c/ day of May 1977, by and between NINA R. CARROLL, Grantor, and the MARYLAND HISTORICAL TRUST, Grantee,

826 - 542

15, 810

*** WITNESSETH ***

WHEREAS, Grantee is a body corporate and instrumentality of the State of Maryland created for the purpose generally of preserving and maintaining historical, aesthetic and cultural properties, all as is more particularly provided by law; and

WHEREAS, Grantor is the owner, pursuant to the Will of Philip A. Carroll, dated May 9, 1946, and a Codicil thereto dated October 16, 1953, of a life estate and a general power of appointment in improved real property known as Doughoregan Manor, located in the second and Third Election Districts of Howard County, State of Maryland, which property is hereinafter referred to as the Property and is more particularly described below; and

WHEREAS, the Property, including all the land covered by this easement, has substantial historic, aesthetic and cultural character and importance, and this easement will promote the preservation and maintenance of the Property and its historic aesthetic and cultural character; and

WHEREAS, Grantee has determined that this easement is exclusively for conservation purposes; and

WHEREAS, Grantee is possessed with the power and duty to accept, hold and administer this easement;

NOW, THEREFORE, in consideration of the premises Grantor, as life tenant and in partial exercise of her power of appointment referred to above, gives and conveys to Grantee an easement (hereinafter the "Easement") in all of that certain lot or parcel of land known as Doughoregan Manor, together with all of the improvements thereon and appurtenances, rights, and interests thereunto belonging, the Property being more particularly described as follows: Begining for the same at a cut granite stone heretofore set on the Northern margin of the sixty-six foot wide right-of-way of the Baltimore to Frederick Turnpike Road, the said point being directly opposite Manor Lane, and at the end of the fifteenth or N 70 degrees W 181 perches line of that land, which by deed dated June 7, 1866 and recorded among the Land Records of Howard County in Liber NWW 25, folio 316, etc. was granted and conveyed by Charles Carroll and Caroline Carroll, his wife, to Anita Carroll, wife of John Lee Carroll and running with the said road margin and reversely with the said fifteenth line plus a straight line prolongation thereof, with bearings corrected to True Meridian, as now surveyed;

AND IN THE REPORT AND A STREET AND A

IBERD826 FOLID543

(1) S 73 degrees 02 minutes 48 seconds E 3140.09 feet to intersect the thirteenth outline at 2921.96 feet on said line, thence running reversely with said line and passing over a concrete monument now set on the Southern margin of the said road right-ofway.

(2) S 25 degrees 58 minutes 45 seconds E 2921.96 feet to a concrete monument now set, thence running reversely with the twelfth and a part of the eleventh outlines respectively.

(3) S 21 degrees 10 minutes 35 seconds E 412.58 feat to a granite stone heretofore set, thence

(4) S 87 degrees 17 minutes 37 seconds W 523.99 feet to a granite stone heretofore set at the end of the third line of that land, the said land being a part of the land herein described, which by deed dated November 26, 1867 and recorded among the said Land Records in Liber WWW 27, folio 411 etc., was granted and conveyed by Matthais Hammond and Clara Hammond, his wife, to Anita Carroll, wife of John Lee Carroll and running reversely with the said third line

(5) S 07 degrees 19 minutes 03 seconds E 2783.20 feet to a granite stone heretofore set at the end of the minth or N 62 degrees E 34-3/4 perches line of the aforementioned conveyance from Charles Carroll and wife, thence running reversely with the said inith line

JBERD826 FOLIO544

(6) S 61 degrees 55 minutes 21 seconds W 642.37 feet to a granite stone heretofore set at the end of the eighth or N 62 degrees E 157.5 perches line of that tract designated as Lot J, Division 5 and recorded among the Partition Records of Howard County in Liber EPH No. 1, folio 35 etc., and running reversely with the eighth to second line inclusively

(7) S 61 degrees 52 minutes 51 seconds W 2549.50 feet
to a large granite stone with the loccers CC cut therein, thence
(8) S 27 degrees 49 minutes 20 seconds E 3024.41 feet

to a granite stone heretofore set.

(9) S 61 degrees 59 minutes 09 seconds W 180.43 feet to a concrete monument now set,

(10) S 27 degrees 01 minute 31 seconds # 513.33 feet to a granite stone heretofore set,

(11) S 49 degrees 13 minutes 56 seconds W 956.26 feet to a tall granite stone heretofore set,

(12) N 85 degrees 56 minutes 08 seconds W 1573.23 feet to a concrete monument now set,

(13) S 76 degrees 03 minutes 52 seconds W 1021.30 feat to a cut sandstone bearing the letter A cut therein, thence running reversely with the firstly described outline of the said Lot J in combination reversely with the seventeenth or N 76 degrees W 147 perches line plus the sixteenth line of that tract designated as Lot K in the aforementioned Partition Records

(14) S 76 degrees 13 minutes 08 seconds W 3570.74 feet to a stone marked five which is now reset, thence running reversely with the fiftmenth and fourteenth outlines of Lot K.

ELIBERD826 FOLIOS45

(15) S 78 degrees 20 minutes W 561.00 feet passing over a stone heretofore set 35.61 feet from the end thereof, to a concrete monument now set on the west edge of an abandoned roadway, thence

(16) N 17 degrees 55 minutes 00 seconds W 2314.65 feet to a concrete monument now set at the point the persimmon tree once stood, thence running reversely with the combined thirteenth and twelfth outlines

(17) N 03 degrees 44 minutes 03 seconds N 354.05 feet to a point in the center of the stream between the abuttments of an abandoned bridge site, thence reversely with the eleventh and tenth outlines respectively

(18) N 15 degrees 17 minutes 33 seconds W 165.00 feet to a concrete monument now set,

(19) N 19 degrees 47 minutes 33 seconds W 592.00 feet to an iron pipe heretofore set in the bed of Carroll's Mill Road, the said point also being at the end of the third or 5 24-1/2 degrees W 23 perches line of the thirdly described or 20 acres 3 roods and 4 square perches parcel which by deed dated September 15, 1939 and recorded among the said Land Records in Liber EM Jr. No. 164, folio 403 etc. was granted and conveyed by Thomas P. O'Donnell and Ada D. O'Donnell, his wife, to Philip A. Carroll, and running three courses with the said Carroll's Mill Road and with the fourth, fifth and a part of the sixth lines of said third parcel

(20) S 72 degrees 11 minutes 00 seconds W 57.75 feet to an iron pipe now set,

(21) N 87 degrees 49 minutes 00 seconds W 396.00 feet to an iron pipe now set,

(22) N 69 degrees 36 minutes 00 seconds W 206.70 feet to an iron pipe now set at the end of the third or 5 26 degrees 50 minutes W 209.10 foot line of that land which by deed dated January 12, 1939 and recorded among the said Land Records in Liber BM Jr. No. 163 folio 258 was granted and conveyed by Thomas P. O'Donnell and Ada D. O'Donnell, his wife, to Isaac Johnson, the said parcel being an exception to the aforementioned thirdly described parcel and not the firstly described parcel as set forth in the said conveyance from O'Donnell to Carroll, thence running reversely with the third and second lines of the Johnson lot

LIDERD826.FOLID546

(23) N 21 degrees 54 minutes 00 seconds E 209.10 feet to a concrete monument now set,

(24) N 69 degrees 36 minutes # 209.13 feet to a concrete monument now set at 161.23 feet on the eighth line of the aforementioned thirdly described parcel thence with a part of the eighth and all of the ninth outlines

(25)N 21 degrees 54 minutes 00 seconds E 610.97 feet to a concrete monument now set on the West edge of a flintstone quarry, thence running with the combined tenth and eleventh outlines, and excluding the said quarry

(26) S 09 degrees 51 minutes 00 seconds E. 176.55 feet to an iron pipe now set, thence with the twelfth and thirteenth lines respectively

(27) S 80 degrees 96 minutes 00 seconds E. 13.20 fect to an iron pipe now set.

(28) S 89 degrees 36 minutes E. 151.80 feet to a concrete monument now set, thence running with the fourteenth line of the said third parcel plus the seventh or N 2 degrees W. 9.8 perches outline of the tract firstly described in the aforementioned conveyance from O'Donnell to Carroll

(29) N 00 degrees 50 minutes 33 seconds E. 828.36 feet to an iron pipe now set in a white oak stump, thence with the eighth, ninth, tenth and eleventh lines of the said firstly described parcel

(30) N 18 degrees 33 minutes 07 seconds W. 112.20 feet to an iron pipe now set in a black oak stomp (31) N 24 degrees 56 minutes.53 seconds E. 1468.58 feet to a concrete monument now set

LIBEROB26 FULINO + 1

TATA A CALLER AND A CALLER AND A CALLER AND A CALLER AND A CALLER AND A CALLER AND A CALLER AND A CALLER AND A

(32) S 65 degrees 03 minutes 07 seconds E. 394.22 feet to a stone heretofore set beside a gully,

(33) N 27 degrees 16 minutes 05 seconds E. 653.39 feet to a concrete monument now set at the end of the third or S 27-1/2 degrees E 25-3/4 perches lines of that land, which by deed dated March 31, 1927 and recorded among the said Land Records in Liber H.B.N. No. 129, folio 532, etc. and running reversely with the third; second, and a part of the first outlines of said land

(34) N 32 degrees 28 minutes 55 seconds W. 417.44 feet to a poplar tree

(35) N 42 degrees 41 minutes 06 seconds W. 364.81 feet to a stone heretofore set

(36) N 13 degrees 31 minutes 04 seconds E. 1056.53 feet to a concrete monument heretofore set at the end of the third or S 37-1/2 degrees E. 625 foot line of that land, which by deed dated April 13, 1948 and recorded among the said Land Records in Liber M.W.B. No. 203, folio 483 etc., was granted and conveyed by John Yarrow Eccles, Executor of Will and Codicil of Helen Beatrice Cavendish Moyle Sherer, Late of Knightsbridge to Philip A. Carroll and running reversely with the said third line

(37) N 47 degrees 48 minutes 39 seconds W. 634.82 feet to the centerline of Vineyard Road as it is now located and passing over a concrete monument heretofore set on the Eastern boundary of said road, thence running four courses and distances with the said Vineyard Road and reversely with the second line of the said conveyance from Eccles to Carroll

(38) N S7 degrees 55 minutes 48 seconds E. 500.36 feet to the end of the twenty-first or S 56 degrees W 35 perches line of

, ·
the herein firstly mentioned conveyance to Anita Carroll, thence running reversely with the twenty-first to sixteenth lines of said land inclusively

LIBER D826 FOLIO548

patrice and the second

(39) N 53 degrees 00 minutes 48 seconds E. 577.50 feet,
(40) N 39 degrees 45 minutes 48 seconds E. 2458.50 feet,
(41) N 26 degrees 15 minutes 48 seconds E. 2376.00 feet
to the center-line intersection of Vineyard Road and the aforementioned Baltimore to Frederick Turnpike Road, thence one course
with the centerline of the said Turnpike.

(42) S 82 degrees 36 minutes 44 seconds E. 189.50 feet, thence

(43) N 14 degrees 06 minutes 44 seconds W. 35.47 feet to the Northern right-of-way margin,

(44) S 82 degrees 36 minutes 44 seconds E. 1394.98 feet to the point of the beginning, containing 2042.28 acres of land, more or less.

Exhibit A hereto consists of 40 pages, and includes as page 1 a schedule (which is recorded with this deed of easement) describing the documents, photographs and other things that are a part of the exhibit and that are filed at the offices of Grantee, that are not recorded herewith but are nonetheless as fully and completely incorporated into this deed of easement as though recorded herewith.

The Easement is Subject to any and all presently existing Valid encumbrances, easements and rights of way upon the Property.

The Easement shall be of a duration of thirty (30) years from the date hereof and shall then fully terminate. It is an easement in gross and as such is inheritable and assignable and runs with the land as a binding servitude and as an incorporeal property interest in the Property enforceable by Grantee, its successors and assigns with respect to the Property and against Grantor and Grantor's heirs, successors and assigns; and to that end Grantor covenants on behalf of herself, her heirs, successors and assigns, with Grantee, its successors and assigns, such covenants being deemed to run as a binding servitude until the date of termination, to do and to refrain from doing upon the Property each of the following stipulations, which contribute to the public purpose in that they aid significantly in the preservation and conservation of the Property:

_IBER0826 FOLIO549

TAL TO CONTRACTOR DESCRIPTION OF THE OWNER OF THE OWNER OF THE OWNER OF THE OWNER OF THE OWNER OF THE OWNER OWNE

1.78 F. 1.98

1. No industrial or commercial activities, with the exception of farming (including the sale of farm products to the public), shall be carried on on the Property, except such as can be carried on from a residential or farm building without alterations to the external appearance of the building.

2. Timberlands shall be managed in accordance with sound forestry practices, and trees may be selectively or clear cut from time to time in such manner as will not permanently alter the character of such lands as forest lands. Notwithstanding the provisions of the preceding sentence, timberland may to cleared for agricultural use or for any construction permitted by paragraph 4 hereof.

3. No dump of ashes, sawiust, bark, trash, rubbish or any other unsightly or offensive material, except that which is produced by or maintained for agricultural use, may be permitted on the Property visible from the public roads and highways.

4. No building or other structure shall be built or maintained on the Property other than those buildings or structures which are on the date hereof located on the Property as described and depicted in Exhibit A, except as follows:

 (a) fences and unpaved roads, when constructed or maintained for agricultural purposes, are not included within the meaning of "building" or "structure" as used herein; (b) any building or structure may be constructed and maintained for agricultural use, including residential houses for workers (including guards) on the Property and garages or other buildings or structures incident to such houses.

LIBERO826 FOLIOSSO

HICK STORES

WHEN THE PROPERTY OF THE PROPE

(c) no more than eight (B) additional houses may be constructed and maintained on the Property, together with outbuildings and structures reasonably related thereto, including (but not limited to) access rocds, wells, barns or stables, garages, swimming pools, tennis courts, garden structures and the like, provided that no such house shall be constructed except for the bona fide rersonal residential use of one or more of Grantor's lineal descendants, including lineal descendant. by adoption.

5. *Without the express written permission of the Director of the Maryland Historical Trust (hereinafter the "Officer"), no construction, alteration or remodeling or any other thing shall be undertaken or permitted to be undertaken c. the existing structures numbered 1, 3, 4, 5, 7 and 22 on page 4 of Exhibit A which would affect their exterior, as described and depicted in Exhibit A; provided, however, that the maintenance, reconstruction, repair, repainting or refinishing of any of said exteriors, damage to which has resulted from casualty loss, deterioration or wear and tear, shall be permitted without such written permission of the Officer provided that such maintenance, reconstruction, repair, repainting or refinishing is performed in a manner that will not materially alter the appearance thereof as they are as of this dato. The term exterior shall include general style and arrangement of such exterior, including the kind and texture of building materials and the type and style of all exterior windows, doors, light fixtures, signs and other similar features, but not the color of paint or other finish.

6. Grantor agrees, to the extent reasonably financia. feasible, to maintain the Manor House (Exhibit A, p. 4, No. 1) in good, clean and safe condition and shall maintain, repair and administer it to preserve its historic, aesthetic and cultural character and appearance as is described and depicted in Exhibit A; <u>provided</u>, that nothing herein shall require reconstruction if the Manor House is destroyed in whole or in part by casualty loss. This covenant is unpressly limited to the Kanor House and does not apply to any other imporvement on the Property. The obligations of this affirmative covenant are expressly declared not to apply to the existing structures numbered 3, 4, 5, 7 a.u 22, which are subjected to a negative covenant only in paragraph 5 above.

LIBEROB26 FOLIO551

· 10 -

C.

7. If at any time during the term of this Easement a public highway'is proposed to run across the Property between the existing Manor Lane and the eastern boundary of the Property in a generally north-south direction, such public highway may be constructed and maintained if (a) the Officer finds that such highway would be less deleterious to the historic, aesthetic and cultural character of the Property than is the existing Manor Lane, and (b) the public right of way over the existing Manor Lane as it crosses all or substantially all of the Property is surrendered.

1. 8. Grantee shall have the right to enter the Property for the purpose of inspecting the Property to determine whether there is compliance by Grantor with the terms of this Easement, provided that (a) much right may be excluded no more frequently than once every two years, (b) no more this, one representative of Grantee shall be entitled to participate in each inspection, and (a) the interiors of all buildings and structures shall be exempt from inspection.

9. Upon any breach of the terms of this Easement by Grantor, Grantee shall have the following rights, which shall be cumulative and shall be in addition to any other rights and remedies . available to grantee at law or in equity: (a) to require restoration of the Property to the condition required by this Easement;

ŝ.,

LIBERD826 FOLIO552

(b) to enjoin any breach or enforce any covenant hereof by ex parte, interlocutory, and final injunction; and

(c) to recover compensatory damages for any breach, which damages shall be applied to restoration of the Property to the condition required by this Easement.

No failure on the part of the Grantee to enforce any covenant or provision herein nor the waiver of any right hereunder by Grantee shall discharge or invalidate such covenant or provision or any other covenant, condition, or provision hereof, or affect the right of Grantee to enforce the same in event of a subsequent brutch or default.

10. In any event where the terms of this Easement require the consent of the Officer, such consent shall be requested by notice to the Officer and consent shall be deemed to have been given within forty-five (45) days after receipt of notice by the Officer unless the Officer gives notice to Grantor of specific reason for disapproval. In any event where the Officer gives such notice of disapproval. Grantor may appeal the disapproval to the Board of Trustees of the Maryland Historical Trust for review by it or by such person or agency as may be designated by it to make such review. Appeal shall be made by notice to the Officer given within forty-five days of receipt of notice of disapproval from the Officer.

11. Any notice required to be given by this Easement shall be in writing and may be given by certified or registered mail, with postage prepaid and return receipt requested, addressed to the Officer, as follows:

> Diroctor Maryland Historical Trust Shaw House 21 State Circle Annapolis, Md. 21401

or to the Grantee or the Officer at such other address as the . Officer may from time to time designate by notice to Grantor. Any notice given in the foregoing manner shall be deemed to have been given when deposited with the United States Post Office.

G

LIBEROB26 FULIUS53

12. This Easement is for the purpose of promoting and shall be construed so as to promote the pruposes of the statutes creating and governing Grantee and of Section 2-118 of the Real Property Article of the Annotated Code of Maryland and to preserve the historic, cultural, scenic and aesthetic character of the Property.

13. Grantee agrees that it will hold the Easement exclusively for conservation purposes, <u>i.e.</u>, that it will not transfer the Easement in exchange for money, other property, or services.

TO HAVE AND TO HOLD unto the Maryland Historical Trust, its successors and assigns, for thirty years.

Witness the following signatures and seals. WITNESS: GRANTOR:

WITNESS:

. 1

Maryland Historical Trust --

(SEAL)

Min R. Cassale

GRANTEE: `

(SEAL) ITS DIRECTOR

STATE OF MENTAND, NY COUNTY, to wit;

I HEREBY CERTIFY that on this, 23 day of MAY, in the year 1, before the subscriber, personally appeared Nina K. Carroll and acknowledged the foregoing deed to be her act.

Notary Public

My commission expires: 3.30.78 by powership Labry My Addr. Sale of her York My Addr. Sale of her York My Addr. Sale of her York Commy Commission Commy Commission Commy Commission Commy Commission Commy Commission Commy Commission Co

CHEROB26 FOLIOS54. - 13 -STATE OF MARYLAND (Inu Chunge COUNTY, to wit: I HEREBY CERTIFY that on this / day of func in the year 22, before the subscriber, personally appeared Orwin C., falboft, Director of the Maryland Historical Trust, and acknowledged the foregoing deed to be the act of said Trust. My commission expires: 7/1/7: to legal form and sufficiency which are 1977. ENERAL ě. × • Ň. ì

SCHEOD	
SCHEDD	
	IL OF EXHIBIT A
	•
Page 1 of 40 ' Schedule	
Page 2 of 40 Site Plan Page 3 of 40 Aerial Photograph	
Page 4 of 40 List of existing st	
Pige 6 of 40 Hanor House Southeau	
	st Facade main entrance detail . trance molding detail
Page 9 of 40 Entrance drive look	ing Southeast from Manor House toward Manor lane
Page 11 of 40 Hanor House Chapel S	st facade Northeast end Southeast facade
Page 12 of 40 Hanor Kouse Chapel 1 Page 13 of 40 Manor House Northwest	
Page 14 of 40 Hanor House Chapel N	forthwest facade
Page 16 of 40 Hanor Bouse Contral	letail of Northwest Entrance Section Northwest Facada
Page 17 of 40 Kanor House Central Page 18 of 40 Hortbyest Manor House	section Northwest facade, tower detail Re Garden looking West from Manor House
Page 19 of 40 Hanor House Northwys	at Farade looking Northeast
	t and of Southeast facade
Page 22 of 40 Manor House Southeas Page 23 of 40 Stons Spring House N	t face southwest wing detail
Page 24 of 40 Stone Spring House S	outhwest corner
Page 25 of 40 Stone Spring House S Page 26 of 40 Stone Spring House E	
Page 27 of 40 Stone and Frame tena Page 28 of 40 Stone and Frame tena	nt house Southwest corner
Page 29 of 40 Stone and Frame tena	nt house North facade
Page 30 of 40 Hanagers House West Page 31 of 40 Hanagers House South	
Page 32 of 40 Managers House South Page 33 of 40 Managers House North	
Page 34 of 40 Old Office Southweat Page 35 of 40 Old Office Southeast	facada
Page 36 of 40 Old Office Northwest	end
Page 37 of 40 Horse barn complex fi Page 38 of 40 Borse barn complex fi	rom East
Page 39 of 40 'Horse barn complex for Page 40 of 40 Horse barn complex for	rom Southeast
	,
•	
,	
	·
• •	
DCXGHOREGAN HANDE ELLICOY CIT:	SCHEDULE
HOWARD COUNTY	SCALE: PREPARED: 5//7 JAC
EASEMENT EXHIBIT NO. A, page 1 of 40	MARYLAND HISTORICAL TRUST NORTH
SIGNED ORIGINAL ON FILE WITH THE M.F	LT. O. 2041 GRANTOR GRANTEE

-ti





1 pr.a. 1200		25 FOLIOSSB	· · · · · · · · · · · · · · · · · · ·
			,
			•
	LIST OF EXIST	THE STRUCTURES	
BUILDINGS	Manor Bonse	•	
2.	Accompanying Building, Pool, Tenn	is Court arrack (Borse barn complex)	
, 14 .		-	
6.	Old Shop (apartments)' Stone Spring House	· .	.
	Old Machine Shed (frame	ffice	
9. 10.		building	
4 12	Brick Shop Building Three Siles (Two with roof, one w	(thout)	
1 716.	Large Pole Barn - used now as Mac	hine Shed	
18.	Block and metal Dairy Building, n Large Stone Building - in poor re	pair	
	Gate House (stone) Old Bath House		
. <u>TENANT HOU</u>			
*22.	No. 3421 - Frame two-story No. 3425 - Stone and frame two-st	οτγ	
24	No. 3429 - Frame two-story No. 3433 - Frame two-story	•	
1 26.	No. 3445 - Brick two-story No. 3457 - Rancher - frame constr	uction	
	No. 3461 - Stone two-story No. 3526 - Old Frame two-story	•	1
	No. 3536 - Frame two-story No. 4288 - Frame - two-story		
31,32,33.	On Mill Road, west side of Manor	Lane - House and two (7) Far	m buildings
34,35,36.	On Hill Road, east side of Hanor two (7) small buildings	Lane - 2-story stone and tra	
例 猜 "	. CNO (7) SMALL DULLALINGS		me house, and
	. two (r) heart purrounds	•	lard
			we house, and
	. two (r) heart purroungs	· · · ·	lard
			lard
			lard
			lard
	. two (r) Suarr mortungs		lard
			lard
DOUGHOREGA	AN MANOR	LIST OF EXISTING STRUCTURE	Malt: Um Curlitted Malt: Um Curlitted Ana Rectored Truck Anar Muneral attach Curches, nd 2 1401
DOUGHCREGA ELLICOTT C HOWARD COL		LIST OF EXISTING STRUCTURE SCALE :	Malt: Um Curlitted Malt: Um Curlitted Ana Rectored Truck Anar Muneral attach Curches, nd 2 1401
ELLICOTT C HOWARD COT	AN MAINOR TITY THY THY THY THY THY THY THY THY THY T	LIST OF EXISTING STRUCTURE SCALE: PREPARED: 5/77 MARYLAND HISTORICAL TI	104 2 Day 100 June 100 June Mart 8. Jun Confirted Anniel State Internet attack NOVE 1072 June 2 140
ELLICOTT C HOWARD COT	N MANOR ZITY JUTY	LIST OF EXISTING STRUCTURE SCALE: PREPARED: 5/77 MARYLAND HISTORICAL TI	Matt: Jon Carpitol



,

•

,

				I AND AND AND AND AND AND AND AND AND AND
	\$\$4(QIQ)	3/26/81)	R/V 31500	34
	Right of Way	AGREEMENT	r	
)51 EEEO 633	The underrigued hereby grass to BALTIM successors, licenses, and ardyn, for value reached, if and telephone here, including the measury poles, cros gays, conduits, eshes, suret lights and equipment undersigned situated on the vest side of Ha- being known as 4288 Manor Lane In 3rd Dist Phillip A. Carroll will by deed dated May 9th, 1946 Howard County in Liber R.L.P.	annu, electri in, over, und nor Lema s rict, Nova	r, idephone and other wine, mechons, or and through the property of the casth of Carroll Mill Road	r alter alter
0 2001 	Together with the right to: have access at properties; string wires between any poles and from t cut down trees adjacent to the wires to provide any errected under or over the lines. The lines are or art to be located baginain south of Carroll Mill Road at Heltimon numbered 170015 and extending in a wee feet. "Grantor also grants and cou of all anexestrances, for val on grantor's property."	ne neneri por ple clearance. ng at Nanon re Gau and pterly dire	No buildings or structures are to be r Lans approximately 1705 Ricetric Company pale now sotion approximately 1520	CATROIL, NURA R
		. 296	day of MAYER 4 19 51.	, ,
	WITNESS MY band and seal this WITNESS:	1 ⁻		
	1 for In Canel "	ina R. Car	roll RECD FE	LOLLE H
	STATE OF MARYLAND (DUNLY OF NEW YOLK) TO WIT: I HEREBY CHETIFY, that on this 2.1 before not, the subscriber, a Notary Public of the St aforemit personally another Sina B. CATOO	ate of therein	March 1001	0 RO2 TIQ:39 HO5/5/81
	aforeasid, personally appeared fina R. Carrol and acknowledged the foregoing agreement to be way made without monetary consideration.		I and deed, and sold act and deed	MARCE
	WITNESS my hand and Notarial scal.	Ellyu BLUYN G	C THAT THE ADD	1 29, 1981
Sioo	Mail 20: Balls. Ous & Clitte Room 401 Front St. P. S. 1306 1475 Bultining, ma 2120	.,	Child G. TRUTTLAND Child Patter State of How York L. Glauble to Micros Conference W. Configure to Micros Conference Openstates Explore Marca 30, 13462	•
	ی میساند میدود با این و این و در این و در این و در باین و میدود میدود این و میدود میدود این و میدود این و میدود ا	منجوع منضرة العالم فالموا		
	•	1		- 19 54
	, 		· · · ·	
6.1.X.V.Y	5 ⁻¹ 4			· · · · ·
	•			
				- Alina and a start of the
(24) 3 (m. 27)	مىيىنىيە بىرىيەيچۈ ئەت بەتە ئەتەتەتە يەلەرد بايىرۇر يارىي بارى. 			

٠

.

Subdivision No. RE-06-02

DEED OF PRESERVATION EASEMENT

THIS DEED OF PRESERVATION EASEMENT ("Preservation Easement") made this day of <u>Octobel</u>, 2006, by and between PHILIP CARROLL also known as Philip D. Carroll and CAMILLA CARROLL, individuals residing in Howard County, Maryland, having an address at 3500 Manor Lane, Ellicott City, Md, (the "Grantor") and THE HOWARD COUNTY CONSERVANCY, INC., a Maryland corporation and HOWARD COUNTY, MARYLAND, a body corporate and politic (hereinafter collectively referred to as the "Grantees").

WITNESSETH:

NDR HDDE Oct 12 2006 01:14 Fm

WHEREAS, The Howard County Conservancy, Inc. is a not-for-profit tax exempt organization within the meaning of Section 501(c)(3) of the Internal Revenue Service Code and is established to promote the preservation and protection of natural resources within Howard County, Maryland; and

WHEREAS, Howard County, Maryland is a body corporate and politic which is authorized pursuant to the Howard County Zoning Regulations to be the grantee of a preservation easement in the RC District (Rural Conservation); and

WHEREAS, the Grantees have been designated as two of the entities which may be parties to the Preservation Easement in accordance with the Howard County Zoning Regulations; and

WHEREAS, Grantor owns in fee simple 75 acres, more or less, of certain real property situate, lying and being in the Second Election District of Howard County, Maryland known as Parcel 71 on Tax Map 23 as shown on a Final Record Plat entitled, "Density Sending Plat Property of Philip Carroll and Camilla Carroll" and recorded as Plat No. $1e^{-72}$ among the Plat Book Records of Howard County, Maryland (the "Property"); and

WHEREAS, in consideration of the privilege of subdivision, the Grantor is willing to grant a perpetual preservation easement on a portion of the Property pursuant to the Howard County Zoning Regulations and Subdivision and Land Development Regulations applicable to cluster subdivisions within the RC District (Rural Conservation), whereby the Grantor restricts and limits the use of that portion of the Property shown and described as Preservation Easement, consisting of 75 acres of land, more or less, as shown on the aforesaid Final Record Plat ("the Preservation Parcel"), pursuant to the terms and conditions, and for the purposes hereinafter set forth, and Grantees are willing to accept such preservation easement as to the Preservation Parcel; and

WHEREAS, Grantor and Grantees have a common purpose in conserving the dominant scenic, cultural, rural, agricultural, woodland and wetland character of the Preservation Parcel, and, except as hereinafter provided, preventing the use or development of the Preservation Parcel for any purpose or in any manner that would conflict with the maintenance of the Preservation Parcel in its open-space conditions; and

Office of Law HOWARD COLNIGETHEER 208URT (Land Records) IMSA CE 53-10286) Book MDR 10295, p. 0058. Printed 09/24/2008. Online 19/10/2006.

The second second second second second second second second second second second second second second second s

WHEREAS, the Grantor and the Grantees acknowledge and agree that the primary purpose and use of the Preservation Parcel is Farming.

NOW, THEREFORE, in consideration of the mutual covenants, terms, conditions and restrictions hereinafter set forth, and other good and valuable consideration, the adequacy and receipt of which are hereby acknowledged, Grantor unconditionally and irrevocably hereby grants and conveys unto Grantees forever and in perpetuity a Preservation Easement of the nature and character and to the extent hereinafter set forth, with respect to the Preservation Parcel.

The purpose of this Preservation Easement is to preserve and protect the environment of the Preservation Parcel and to maintain permanently the open-space values of the Preservation Parcel and the dominant scenic, historic, cultural, rural, agricultural, woodland and wetland character of the Preservation Parcel.

To achieve these objectives, the following conditions and restrictions are set forth:

<u>ARTICLE I.</u>

1. As used in this Preservation Easement, the term "Grantor" means the Grantor, its personal representatives, heirs, successors and assigns and any other person(s) or entity(ies) now or hereafter having a legal interest in the fee simple title to the Preservation Parcel.

2. As used in this Preservation Easement, the term "Grantees" means the Grantees and their respective personal representatives, heirs, successors and assigns and any other person(s) or entity(les) now or hereafter having a legal ownership interest in and benefit of the easement granted herein.

ARTICLE II. DURATION OF EASEMENT

This Preservation Easement shall be perpetual. It is an easement in gross and as such is inheritable and assignable in accordance with Article IX and runs with the land as an incorporeal interest in the Preservation Parcel, enforceable with respect to the Preservation Parcel by Grantees against Grantor.

ARTICLE III. PERMITTED USES AND ACTIVITIES

Those uses permitted on the Rural Conservation (RC) Preservation Parcels, principal uses permitted as a matter of right, accessory uses and conditional uses, pursuant to the Howard County Zoning Regulations, Section 104 as of July 12, 2001, are permitted, except to the extent such use is prohibited in Article IV hereof. Any other uses, which become permitted uses in RC Preservation Parcels pursuant to any amendments to the Howard County Zoning Regulations subsequent to July 12, 2001 shall require the prior written approval of both Grantees. Grantor and Grantees understand, stipulate, and agree that all principal uses permitted as a matter of right and all accessory uses are compatible uses and, further, that the compatibility of any conditional uses with the aforesaid uses will be determined on a case-by-case basis by the Howard County Board of Appeals upon petition for approval of a particular conditional use.

HOWARD COUNTY CIRCUIT COURT (Land Records) [MSA CE 53-10285] Book MDR 10295, p. 0059. Printed 09/24/2008. Online 10/18/2006.

September 22, 2006.tdb.cs

Z. TO LOW THE COMPANY OF THE OWNER.

1,12

11

2

In addition, the Grantor is permitted on the Preservation Parcel:

(i) To construct, improve, repair, restore, alter, remodel, and maintain all structures, including accessory structures, designed for the purpose of serving the permitted uses of the Preservation Parcel identified in the Howard County Zoning Regulations; and

(ii) To construct and maintain reasonable means of access to all permitted uses and structures both within and outside of the Preservation Parcel; and

(iii) If the Property is served by a shared sewage disposal facility, to install and use a shared subsurface wastewater disposal field, reserve fields and collector pipes in accordance with the terms of this Preservation Easement and the developer agreement and declaration of covenants executed by the Grantor.

ARTICLE IV. PROHIBITED AND RESTRICTED ACTIVITIES

The following uses and activities are prohibited and/or restricted on the Preservation Parcel:

1. Industrial uses, and commercial or residential uses other than those enumerated in Section 104 of the Zoning Regulations are prohibited on the Preservation Parcel unless any such use becomes permitted by amendments to the Zoning Regulations subsequent to July 12, 2001 and prior written approval of both Grantees is given.

2. Display of billboards, signs or advertisements is prohibited on or over the Preservation Parcel, except (a) to state solely the name and/or address of the Preservation Parcel and/or the owners; (b) to advertise the sale or lease of the Preservation Parcel; (c) to advertise the sale of goods or services produced by permitted uses on the Preservation Parcel; or (d) to commemorate the history of the Property, its recognition under state or federal laws; provided that no sign or billboard on the Preservation Parcel shall exceed four feet by four feet. Multiple signs shall be limited to a reasonable number, shall be placed at least 50 feet apart, shall not damage living trees, and shall be placed in accordance with applicable local regulations.

3. Dumping of soil, trash, ashes, garbage, waste, abandoned vehicles, appliances or machinery, or other materials on the Preservation Parcel is prohibited, except that soil, rock, other earth materials, vegetative matter or compost may be placed (a) as reasonably necessary for agriculture and silviculture on the Preservation Parcel or (b) as may be reasonably necessary for the construction and/or maintenance of structures permitted under this Preservation Easement and means of access.

4. Excavating, dredging, mining, or removing loam, gravel, soil, rock, sand, coal, petroleum and other materials are prohibited, except (a) for the purpose of combating erosion or flooding, (b) for agriculture and silviculture on the Preservation Parcel, or (c) for the construction and/or maintenance of permitted structures, homesites, means of access and wildlife habitat.

5. Diking, draining, filling or removing wetlands is prohibited without the advance written approval of both Grantees.

HOWARD COUNTY CIRCUIT COURT (Land Records) [MSA CE 53-10285] Book MDR 10295, p. 0060, Printed 09/24/2008, Online 10/18/2006.

September 22, 2006.tdh.cs

UNIA 10295 MINO61

6. Managing and harvesting of all forested areas on the Preservation Parcel shall be in accordance with the Maryland Forest Practices Guidelines or comparable provisions of any guidelines or regulations which may replace the Forest Practices Guidelines in the future or as they may be amended from time to time.

7. No building, facility, or other structure shall be constructed on the Preservation Parcel after the date of execution of this Preservation Easement, except for that which is permitted in Article III hereof.

8. The total number of residential structures (including but not limited to principal dwelling units, farm tenant houses, mobile homes, caretakers' dwellings and accessory apartments) on the Preservation Parcel shall not exceed None.

9. If a shared subsurface wastewater disposal field, reserve fields and collector pipes are located on the Preservation Parcel, the following restrictions apply to that area of the Preservation Parcel where such fields and pipes are located:

(a) structures are prohibited, except that the Howard County Health Department may approve the installation of playground or athletic equipment of types which do not affect the operations or performance of the shared sewage disposal facility.

(b) earth moving, grading and other land disturbing activities are prohibited unless prior written approval by Howard County is given.

(c) coverings or toppings such as gravel, asphalt or concrete, which impede the growth of vegetation, are prohibited.

(d) the cultivation of crops (other than hay) or other farming practices which may disturb the soil are prohibited unless prior to written approval by the Maryland Department of the Environment ("MDE"), the Howard County Health Department and Howard County Soil-Conservation Service is given.

(e) planting of trees is prohibited. Existing trees need only be removed prior to the installation of the original and reserve subsurface wastewater disposal fields if required by the Howard County Health Department.

10. The further subdivision of the Preservation Parcel is prohibited.

ARTICLE V. PRESERVATION PARCEL DESCRIPTION

HOWARD COUNTY CIRCUIT COURT (Land Records) [MSA CE 53-10285] Book MDR 10295, p. 0061, Printed 09/24/2008. Online 10/13/2006.

September 22, 2006.tdh.cs

and the second second second second second second second second second second second second second second second

4

2. The existing improvements on the Preservation Parcel consist of None. In the event that either The Howard County Conservancy, Inc. ("Conservancy") or the Audubon Society of Central Maryland, Inc. ("Audubon Society") is a Grantee of this Preservation Easement, color slides and aerial photographs of the Preservation Parcel obtained at the time of execution of this Preservation Easement shall be kept on file with the Conservancy and/or the Audubon Society, as appropriate, or their respective successors-in-interest in perpetuity. The color slides and aerial photographs shall be utilized to confirm the conditions existing at the time that this Preservation Easement is executed and shall be operative between the parties for such purposes.

ARTICLE VI. MAINTENANCE

1. Grantor shall be responsible for the maintenance of the Preservation Parcel consistent with the terms and conditions of this Preservation Easement.

2. In the event that the Audubon Society is a Grantee of this Preservation Easement, Grantor shall develop a maintenance plan for the Preservation Parcel in consultation with and agreed to by the Audubon Society in order to ensure that the Preservation Parcel is maintained to benefit wildlife and natural habitats.

3. Grantor shall establish and maintain a vegetative buffer strip along the N/A River (Creek, etc.). The minimum width of the buffer strip shall be seventy-five (75) feet along the N/A River (Creek, etc.), except as may be reasonably necessary for (a) forest or wildlife management; (b) recreational water uses and associated structures; (c) hunting, fishing, or trapping; or (d) access to water. Manure and compost shall not be stored within seventy-five (75) feet of streams. Pesticides, insecticides, herbicides or fertilizers shall not be used or deposited within seventy-five (75) feet of streams.

A. All rights reserved by Grantor or activities not prohibited by this Preservation Easement shall be exercised so as to prevent or to minimize damage to water quality, air quality, land/soil stability and productivity, wildlife, scenic and cultural values, and the natural topographic and open space character of the Preservation Parcel.

ARTICLE VII. ENFORCEMENT

1. Upon any breach of the terms of this Preservation Easement by Grantor, Grantees may, after reasonable notice to Grantor, require that the Preservation Parcel be restored promptly to the condition required by this Preservation Easement. In addition, each of the Grantees shall have the independent right to enforce, by any proceeding at law and/or in equity, all restrictions, covenants, conditions and provisions of this Preservation Easement. Grantees' aforesaid remedies shall be cumulative. There shall be and there is hereby created and declared to be a conclusive presumption that any violation or breach or attempted violation or breach of any of the aforesaid restrictions, covenants, conditions or and provisions cannot be adequately remedied by action at law or exclusively by recovery of damages.

2. No failure on the part of Grantees to enforce any covenant or provision hereof shall be deemed a waiver of the right to do so thereafter, nor shall a failure to enforce discharge or invalidate such covenant or any other covenant, condition, or provision hereof or affect the right of Grantees to HOWARD COURTING to an enint therewash of same in the event) of same interest of the reach coverage of the right of Grantees to HOWARD COURT of the same in the event) of same interest of the reach coverage of the right of Grantees to HOWARD COURT of the same in the event) of same interest of the reach coverage of the right of Grantees to HOWARD COURT of the same in the event) of same interest of the reach coverage of the rea

5

September 22, 2006.tdb.cs

UNER | 0295 FALOO63

Grantees; their directors, officers, employees, contractors and agents, have the right, 3. after reasonable notice to the record owner of the Preservation Parcel, to enter the Preservation Parcel at reasonable times for the purposes of: (i) inspecting the Preservation Parcel to determine whether the Grantor is complying with the terms, conditions and restrictions of the Preservation Easement; and (ii) remediating any damage to the Preservation Parcel resulting from Grantor's breach of any of the aforesaid terms, conditions and restrictions. Further, if a shared subsurface wastewater disposal field is located on the Preservation Parcel, Howard County and representative of MDE have the right to enter the Preservation Parcel for the purpose of inspecting, maintaining, repairing and replacing the shared sewage disposal facility serving the Property; provided, that Howard County and/or MDE shall be required to give notice to Grantees prior to undertaking any construction. The Grantees and MDE shall coordinate inspection visits to the Preservation Parcel and shall coordinate written and oral communications to the Grantor in response to all of the foregoing inspections. This right of inspection does not include access to the interior of buildings and structures. The Grantees shall not be deemed guilty of any manner of trespass for any such entry, remediation, inspection, maintenance, repair or replacement.

4. If Grantor is found to have breached any of Grantor's obligations under this Preservation Easement, Grantor shall reimburse Grantees for any costs and expenses incurred by Grantees in enforcing the terms of this Preservation Easement, including but not limited to court costs and reasonable attorney's fees, and in remediating any damage to the Preservation Parcel pursuant to Paragraph 3, above.

5. The Grantor agrees to indemnify, hold harmless and defend each of the Grantees, its directors, officers, employees, contractors and agents from and against any and all claims, actions, demands, damages, liability and expenses in connection with loss of life, personal injury, bodily injury and/or damage to or loss of property that arises from the exercise by any Grantee of the right-of-entry granted herein and any activity undertaken on the Preservation Parcel in connection therewith; provided, that no Grantee shall be indemnified, held harmless or provided the cost of a defense by the Grantor for claims, actions, demands, damages, liability and expenses arising from that Grantee's own negligent act or omission, or that of its directors, officers, employees, contractors and agents.

6. Each Grantee has independent authority to enforce the provisions of this Preservation Easement and may do so in its sole and absolute discretion; provided, that such authority shall not be deemed to create an obligation to enforce, above and beyond the enforcement obligations conferred on Howard County by law. In the event that any Grantee does not agree as to whether the Preservation Easement terms are being met, any Grantee may proceed, with reasonable advance notice to the other Grantee and the Grantor, with enforcement actions without the consent of the other Grantee.

ARTICLE VIII. PUBLIC ACCESS

The granting of this Preservation Easement does not convey to the public the right to enter the Preservation Parcel for any purpose other than using any trails or other public recreational facilities which now or hereafter exist on the Preservation Parcel as a permitted Public use pursuant to Article III of this Preservation Easement.

HOWARD COUNTY CIRCUIT COURT (Land Records) (MSA CE 53-10285) Book MDR 10295, p. 0063, Printed 09/24/2008, Online 10/10/2006,

September 22, 2006.tdh.cs

UNA 10295 MINO64

ARTICLE IX. MISCELLANEOUS

Each of the Grantees may assign, upon prior written notice to Grantor, its rights under 1. the Preservation Easement to Howard County, Maryland Environmental Trust ("MET"), Maryland Historical Trust ("MHT"), any land conservation organization that has been approved by resolution of the Howard County Council, or an incorporated homeowners' association and only with assurances that the purposes of this Preservation Easement will be maintained; provided, that there shall always be two Grantees or assignees. If any Grantee or assignee shall abandon this Preservation Easement or the rights and duties of enforcement herein set forth, or be dissolved and the terms of the dissolution fail to provide a successor, then such Grantce or assignee shall assign its rights to Howard County, MET, MHT, any land conservation organization that has been approved by resolution of the Howard County Council, or an incorporated homeowners' association; provided, that there shall always be two Grantees or assignees. In the event that such Grantee or assignee fails to make the aforesaid assignment, then, Grantor, its successor and/or assigns, shall institute in a court of competent jurisdiction a proceeding to appoint an appropriate successor as Grantee; provided, that there shall always be two Grantees or assignees. Any such successor shall be Howard County, MET, MHT, a land conservation organization that has been approved by resolution of the Howard County Council, or an incorporated homeowners' association. No assignment may be made by the Grantees or assignces of their rights under this Preservation Easement unless such Grantees or assignces, as a condition of such assignment, require the assignee to carry out the preservation purposes of this Preservation Easement.

2. Grantor shall notify Grantees in writing of the names and addresses of any party to whom the Preservation Parcel is to be transferred at or prior to the time said transfer is consummated. Grantor further agrees to make specified reference to this Preservation Easement in a separate paragraph of any subsequent deed or other legal instrument by which any interest in the Property or Preservation Parcel is conveyed.

3. The provisions of this Preservation Easement do not replace, abrogate or otherwise set aside any local, state or federal laws, requirements or restrictions applicable to the Preservation Parcel.

4. This instrument sets forth the entire agreement of the parties with respect to the Preservation Easement and supersedes all prior discussions, negotiations, understandings or agreements relating to the Preservation Easement. Any amendments to this Preservation Easement shall be in writing, signed by each of the parties. If any provision is found to be invalid, the remainder of the provisions of this Preservation Easement, and the application of such provision to persons or circumstances other than those as to which it is found to be invalid, shall not be affected thereby.

5. Grantees shall record this instrument in timely fashion in the land records of Howard County, Maryland and may re-record it at any time as may be required to preserve their rights under this Preservation Easement.

HOWARD COUNTY CIRCUIT COURT (Land Records) [MSA CE 53-10285] Book MDR 10295, p. 0064. Printed 09/24/2008. Online >0/18/2006.

September 22, 2006, (di., cs

and the second state of th

1

7

·UNIR | 0295 MM065

Any notices by Grantor to Grantees pursuant to any provision hereof shall be sent by 6. registered or certified mail, return receipt requested, addressed to: The Howard County Conservancy, Inc., P.O. Box 175, Woodstock, Maryland 21163 and Howard County, Maryland, c/o Director, Department of Planning and Zoning, 3430 Court House Drive, Ellicott City, Maryland 21043 or to such other addresses as Grantees may establish in writing on notification to Grantor.

In any case where the terms of this Preservation Easement require the consent of 7. Grantees, such consent shall be requested by notice to Grantees. Such consent shall be deemed to have been given unless within forty-five (45) days after receipt of notice Grantees mail notice to Grantor of disapproval and the reason therefore.

This Preservation Easement does not in any manner prohibit or otherwise restrict **8**. MDE from enforcing the requirements of COMAR 26.04.05, or any amendments thereto, and any other laws and regulations governing on site water and sewage disposal systems.

Neither the Conservancy nor the Audubon Society shall be subject to any claims for 9. damages as a result of or in connection with this Preservation Easement or the exercise of or failure to exercise its rights hereunder. Each shall only be subject to claims for equitable relief.

TO HAVE AND TO HOLD unto Philip Carroll, Camilla Carroll and The Howard County Conservancy, Inc. their successors and assigns, forever. The covenants agreed to and the terms, conditions, restrictions and purposes imposed as aforesaid shall be binding upon Grantor and Grantees, their personal representatives, heirs, successors and assigns and all other successors to them in interest, and any person(s) or entity(ies) having a legal interest in the fee simple title to the Preservation Parcel or a legal ownership interest in and benefit of the Preservation Easement and shall be binding upon and continue as a servitude running in perpetuity with the Preservation Parcel.

All references herein to "Grantor" shall be deemed plural if more than one person has an interest in the property herein conveyed to the Grantee. Any pronoun reference herein shall be deemed to apply to the appropriate gender or person, as the case may be. The term Grantor shall mean its respective successors or assigns.

IN WITNESS WHEREOF, Grantor and Grantees have hereunder set their hands and seals the day and year hereinabove written.

GRANTOR:

WITNESS:

(SEAL) PHILIP CARROLL also known as Philip D. Carroll (SEAL) ARROL

HOWARD COUNTY CIRCUIT COURT (Land Records) [MSA CE 53-10285] Book MDR 10295, p. 0065. Printed 09/24/2008. Online 10/18/2006. 8

September 22, 2006.1dh.cs

CONTRACTOR OF STREET

LWAR | 0295 MM066

ACCEPTED by the Grantees on this $\frac{44h}{1}$ day of October . 2006.

ATTEST:

5

Calcb Gould Carol Filipczak Assistant Corporate Secretary

ATTEST:

Raquel Sanudo Chief Administrative Officer

APPROVED:

3/04

James M. Ivin, Director Department of Public Works

APPROVED:

J. h Marsha S. McLaughlin, Director

Department of Planning and Zoning

APPROVED FOR SUFFICIENCY OF FUNDS:

Sharon F. Greisz, Director Department of Finance

APPROVED FOR FORM AND LEGAL SUFFICIENCY this Styl day of (Qol ADLA . 2006

Barbara M. Cook

County Solicitor

POWARD COUNTY CIRCUIT COURT (Land Records) [MSA CE 53-10285] Book MDR 10295, p. 0066, Printed 09/24/2008, Online 10/18/2006.

9

September 22, 2006.tdh.cs

/

TAR\LD/06/Phillip Carroll Property-125

THE HOWARD COUNTY CONSERVANCY, INC. a Maryland corporation

By: (SEAL)

Ann Holmes Jones President

HOWARD COUNTY, MARYLAND

By: > (SEAL) ames N. Robey County Executive

LINER | 0295 mLin067

GRANTOR: STATE OF MARYLAND, HOWARD, COUNTY TO WIT:

I HEREBY CERTIFY that on this <u>27</u>th day of <u>September</u>, 2006, before me, the subscriber, a Notary Public of the State of Maryland, in and for the County aforesaid, personally appeared Philip Carroll also known as Philip D. Carroll and Camilla Carroll the Grantor named in the within Deed of Preservation Easement, and each acknowledged the same to be his or her act.



Kinda K. Kehman

My Commission Expires: 2-2-09

GRANTEE: STATE OF MARYLAND, HOWARD, COUNTY TO WIT:

I HEREBY CERTIFY that on this 2th day of OCTOBER, 2006, before me, the subscriber, a Notary Public of the State of Maryland, in and for the County aforesaid, personally appeared Ann Homes Jones the President of The Howard County Conservancy, Inc., known to me (or satisfactorily proven) to be the person whose name is subscribed to the within instrument and (s)he acknowledged that (s)he executed the within and aforegoing instrument to be his/her act on behalf of said body corporate for the uses and purposes contained therein and in my presence signed and sealed the same.

AS WITNESS my Hand and Notarial Seal.



Kinda K. Lehman Notary Public

H.3WARD COUNTY CIRCUIT COURT (Land Records) [MSA CE 53-10285] Book MDR 10295, p. 0067. Printed 09/24/2008. Online 10/10/2006

September 22, 2006.tdh.cs

A CONTRACTOR OF A CONTRACTOR OF A CONTRACTOR OF A CONTRACTOR OF A CONTRACTOR OF A CONTRACTOR OF A CONTRACTOR OF

10

UNR 10295 MINO68

STATE OF MARYLAND, Baltimals COUNTY, TO WIT:

I HEREBY CERTIFY that on this <u>4</u>(t) day of <u>(Yetchu)</u>, 2006, before me, the subscriber, a Notary Public of the State of Maryland, in and for the County aforesaid, personally appeared James N. Robey, the County Executive for Howard County, Maryland, a party to the within Deed of Preservation Easement, who acknowledged the same to be the act of the county and that he executed the foregoing Deed of Preservation Easement for the purposes therein contained by signing in my presence, the name of Howard County, Maryland, by himself as County Executive.

AS WITNESS my Hand and Notarial Seal.

My Commission Expires:

ithinland ₽Ÿ C, S COUN

THIS IS TO CERTIFY that this instrument was prepared by Howard County, Maryland, a party named in the within Deed of Preservation Easement.

Tina D. Hackett, Chief Real Estate Services Division

After Recording, Return To: Howard County, Maryland Real Estate Services Division 3430 Court House Drive Ellicott City, Md 21043

HOWARD COUNTY CIRCUIT COURT (Land Records) [MSA CE 53-10285] Book MDR 10295, p. 0068. Printed 09/24/2008. Online 10/18/2006.

September 22, 2006. idh.cs

And the second states in the second states in the

11





•

EXHIBIT 5

Planning Board Determination of Consistency

.

.

1 **DEPARTMENT OF PLANNING & ZONING** BEFORE THE Petitioner: Camilla Carroll and Philip D. Carroll* 2 PLANNING BOARD OF 3 × HOWARD COUNTY, MARYLAND 4 **Development Rights and Responsibilities** 5 Agreement 6 7 **MOTION:** To recommend to the County Council that the Development Rights and 8 Responsibilities Agreement for the Doughoregan Manor is consistent with the 9 General Plan as amended by Council Bill 9-2010 10 VOTE: 5-0 * 11 12 13 On June 17, 2010, the Planning Board of Howard County, Maryland, considered the petition of 14 Camilla Carroll and Phillip D. Carroll for a proposed Development Rights and Responsibilities 15 Agreement ("DRRA") for the Doughoregan Manor. 16 The Petitioners were represented by Sang Oh, Esquire. Also present and representing the 17 Petitioner was Joseph Rutter. 18 19 Public Testimony to the Board 20 The DRRA with attached Exhibits 1-10 and Department of Planning and Zoning Technical 21 Staff Report were presented to the Board for its consideration. The Department of Planning and Zoning ("DPZ") recommended a finding of General Plan consistency between the DRRA and 22 23 General Plan 2000 as amended by CB 9-2010. 24 Mr. Paul Johnson of the Office of Law explained to the Board their role, which is to 25 determine whether or not the petition is consistent with the General Plan. 26 Mr. Oh stated that the Planning Board was only charged with finding whether or not the 27 DRRA is consistent with the General Plan and not to make recommendations on the details of the 28 agreement not related to the consistency with the General Plan. He noted specific parts of the agreement that address General Plan issues: compliance with water and sewer regulations; enhancing 29 30 the County's park system; agricultural preservation; and historic preservation. 31 A number of persons testified in opposition to the petition. Mr. Amit Pramanik testified that the plan was being rushed through the system with a false 32 33 sense of emergency which was not allowing adequate public scrutiny and benefiting only the

1

petitioner. Testimony noted several areas believed to be inconsistent with the General Plan including
 waste treatment, environmental degradation and fiscal impact.

3

4

5

15

16

17

18

19

20

21

22

23

24

Mr. Harry Carnes expressed concerns about sewage and waste water treatment and specific language included in the agreement pertaining to Burnside Drive. Comments were also critical of the technical staff report presented.

Mr. Victor Ilenda, speaking on behalf of the Chateau Ridge Lake Community Association 6 (CRLCA), reiterated that this is the first time Howard County will be entering into a DRRA and that 7 it should be done with the utmost caution. He referenced the section of the DRRA that states 8 Burnside Drive would remain closed, but noted that concept plans have been created for the proposal 9 and should also indicate where an alternate entrance would be made if needed to ensure a connection 10 to Burnside Drive wouldn't be opened. He stated that since the CRLCA was not an incorporated 11 entity, transfer of the proposed open space to block access to Burnside Drive should be to County 12 ownership not CRLCA's. He also noted concerns about and the potential impacts on the community 13 of on-site wastewater treatment. 14

Several persons spoke with concerns of Burnside Drive, which abuts a portion of the Doughoregan Manor property to be developed.

Mr. Ken Aldrich provided the Board with several photos of current road conditions and indicated how added traffic would only worsen those conditions.

Mr. Ted Baruch reiterated that the State Highway Administration's comments requested two points of connection to the proposed development and suggested an another entrance besides Burnside Drive be identified now to address this issue.

Mr. Larry Jeeter spoke in opposition of the proposed DRRA as currently written with concerns including the opening of Burnside Drive, increasing traffic and adequate roads and retention of existing environmental features of the area where development is proposed.

Ms. Christina Delmont-Small stated that the DRRA is not consistent with the General Plan based on a fallacy of composition and specifically noted concerns regarding sewage and the protection of environmental resources. Concerns were also raised regarding the availability of public records and a request was made to provide verification of the negotiation process leading to the proposed DRRA.

Mr. Kal Bhatti stated that proposal was inconsistent with the General Plan's smart growth policies noting that the addition of 325 dwelling units represented low density growth and that expanding public services perpetuates suburban sprawl and depletes needed resources.

2

Ms. Cathy Hudson stated that while the agreement has many good parts, there should be added emphasis on protecting and preserving the historic core of the property given its significance. Suggestions included stronger language that calls for no further development and further clarification on the intent of the preservation strategy rather than just saying there would be no subdivision.

Mr. Johnson noted that even with this agreement, the Carroll family is still subject to all other County laws and also that requirements set forth in the DRRA are above beyond and also run with the land and not the owners. Although the agreement proposes a ten year term, the County still retains all rights to change laws affecting the proposal based on the health, safety, and welfare of the County.

Mr. Oh's response to the public's testimony included further explanation of the intent to ensure Burnside Drive stays closed. Exhibit 2 of the DRRA was highlighted to show that in order for the County to connect through Burnside Drive, they would have to condemn two homes as well as the open space lot abutting Burnside Drive. Mr. Oh also stated that he didn't hear anything presented to the Board that specifically showed how the DRRA was inconsistent with the General Plan. He also addressed questions of the Board regarding Ms. Hudson's testimony considering landominiums, as well as ten years being sufficient to complete execution of the DRRA.

Board Discussion and Recommendation

Transportation

The Board agreed that language in the DRRA adequately addressed community concerns regarding the closing of Burnside Drive although this issue is not related to the General Plan. The Board noted that the County's Adequate Public Facility Ordinance (APFO) would address concerns related to the adequacy of transportation infrastructure at the time new development occurs. Concerns raised about access to the proposed development would also be reviewed during the County's subdivision review process.

7 <u>Sewer</u>

With regard to concerns raised by the additional sewage generated by the proposed development, the Board was in agreement that the General Plan Amendment (CB9-2010) to include additional acreage into the Planned Service Area would not have been passed if the issue was not adequately evaluated by the County. It was noted that the County's request to include an on-site treatment facility was initially considered by the petitioners, yet after receiving much community objection, has been removed as an option in favor of studying additional alternatives that would

÷

.

handle wastewater nutrient reduction off-site as outlined in the agreement. While fiscal impacts associated with the proposal were seen as valid concerns, the Board agreed that it was a budget issue and outside the purview of evaluating the agreement on the basis of consistency with the General Plan and furthermore would not substitute its judgment for that of those with the knowledge and expertise to make a recommendation in the County's best interests.

Technical Report and Information

6

7

8

9

10

11

12

13

14

15

The Board agreed that the inclusion of information provided in previous Technical Staff Reports and Board Recommendations on the General Plan amendment and the Zoning amendment for Doughoregan Manor would enhance public understanding of the process as a whole and specific parts. The law enabling the County to enter into a DRRA has been in effect for a short period of time. This being the first DRRA the County has executed, the Board felt the public could benefit from additional background, even if redundant. Staff also explained that citizens seeking to review the DRRA files at the DPZ had to be redirected to the Office of Law, which has the files to respond to a Freedom of Information Act request. The Board recommended that a better format for hearing a DRRA, and all points included, be made if there were another request.

16 Historic Preservation

The Board had diverging opinions related to the preservation strategy for the historic core 17 included in the agreement. Although Doughoregan Manor is a National Historic Landmark, the 18 property is a place of private residence and deserves the benefit of allowing the petitioners to work 19 on their preservation efforts in good faith. The General Plan speaks broadly on the importance of 20 preserving historic resources, but does not specifically say how that should occur for individual 21 properties, nor does it say anything about the development of preservation plans for individual 22 resources other than through a comprehensive preservation plan developed by the County. To 23 suggest in this instance that a property plan needs to be part of the agreement, and the petitioner and 24 the County work together to preserve the site, could be construed as far reaching. The petitioner has 25 also invested considerable resources in the upkeep of the property. The County architectural 26 historian has been on the grounds numerous times to document the preservation activities now taking 27 place and noted the quality of work being employed. In addition, the petitioner has been negotiating 28 with the Howard County Conservancy on the final conservation easement that will prohibit any 29 future subdivision of the historic core. By these accounts, the intent to preserve the core is 30 memorialized in the various strategies outlined in the agreement. 31

One Board member, however, expressed disappointment that of all of the different parts of the DRRA proposal, the one containing the least amount of detail is the preservation strategy for the

historic core itself. In fact, the petitioners presented the proposal to expand the planned service area 1 as a strategy to preserve the historic home. However, within the DRRA there is no mention of 2 minimum preservation standards beyond limiting further subdivision of the core. 3 4 Consistency with the General Plan 5 The Board was in agreement that the DRRA is consistent with the following sections of the 6 General Plan: 3.1 Ensure the critical mass of high quality, strategically located farmland is protected from 7 development: addressed in Article IV, Agricultural Preservation. Out of 900 total acres, placing 500 8 acres in agricultural preservation (in addition to 75 acres previously preserved) is a significant 9 10 critical mass of agricultural resources. 3.4 Protect Water Resources: addressed in Article III (3.3). The agreement is also consistent with 11 12 the General Plan Water Resources Element. Concentrating the area to be developed within the 13 Planned Water and Sewer Service Area prevents the proliferation of septic fields; there is also a strategy to reduce the wastewater nutrient load from this development. 14 4.18 Enhance the County park system and recreational facilities: addressed in Article III (3.4.). 15 Donation of land at no cost to be added to Kiwanis-Wallis Park will allow expansion and enrichment 16 17 of current programs offered at this popular recreational facility. 5.14 Maintain or enhance the landscape character of roads: addressed as part of Exhibit 2 through 18 the orientation of housing along Route 144. Dedication of additional right-of way and the orientation 19 of homes along Route 144 are both appropriate and protect the character of Route 144. 20 5.18 Establish a comprehensive County-wide historic preservation program: addressed in Article 21 IV (4.2). The historic core easement will prevent future subdivision. Also the Concept Plan for 22 23 development in Exhibit 2 locates all new development behind the topographical ridgeline, which will 24 preserve existing vistas. 6.8 Secure better protection of environmental and landscape resources within new developments: 25 addressed as part of Article IV, Agricultural Preservation. The proposed R-ED zoning is the most 26 27 environmentally sensitive zone available within the PSA. 28 Additional Points Addressed by DRRA During its meeting on the Doughoregan zoning petition ZB1087M, the Board noted several 29 issues expected to be addressed in the DRRA. In reviewing the DRRA, the Planning Board found 30 31 that the following points are adequately addressed: • Places formally approximately 500 acres of the Property into the Agricultural Land 32 Preservation: addressed as part of Article IV, Agricultural Preservation. 33

1	Ó	Recordation of covenants to prevent further subdivision of the core 94 acres of property
2		around the
3		Manor, with the Howard County Conservancy holding the easement: addressed in Article IV
4		(4.2).
5	٥	Compliance with fire safety requirement via sprinklers within the dwellings to eliminate any
6		need for
7		fire safety access through a secondary access at Burnside Road: addressed as part of plan to
8		install sprinkler system throughout the proposed development.
9	•	Prohibits any development access via Burnside Road. Addressed as part of Article III (3.1).
10	Ø	Confirmation of the County's commitment to honor the intent of Resolution 43's termination
11		of
12		Burnside Road: addressed as part of Article III (3.1)(B).
13	o	Confirmation of the donation of approximately of 34 to 36 acres of land to Howard County
14		for the
15		Department of Recreation and Parks: addressed in Article III (3.4.)
16	0	Obligates the provision of a nitrogen pre-treatment plant and the expansion of sewer line
17		capacity by entering into a Major Facilities Agreement with the Department of Public
18		Works: addressed in Article III (3.2) (A) reassessment of nutrient concentration strategy
19		confirmed by a wastewater flow study.
20	a	Caps the number of dwelling units at 325 units as shown in the plan presented to the
21		Planning Board: addressed as part of Article II (2.3) (A).
22	a	Requires the design for the development to substantially conform to the conceptual plan
23		presented to the Planning Board: addressed as part of Article II (2.3) (B).
24	•	Affirms that the development will: (1) Comply with all stormwater management
25	1	requirements; (2) Will not proceed until all initial necessary public utility improvements are
26		made; and (3) Comply with the Adequate Public Facilities Ordinance: addressed as part of
27		Article II (2.4) (2.5) and Article III (3.1) (A).
28	0	DRRA is to be valid for a minimum 5 year period with a recommendation to extend the
29		validity until
30		the completion of the development, although it was expressed that perhaps a having a longer
31		10 year period might be prudent: addressed as part of Article IX (9.2) (A),
32		

.

Finally, the Board noted that from a legal perspective, the Option Agreement included in the DRRA is a strong tool that binds both parties to the terms of the DRRA.

Motion:

1 2

3

4

5

6 7

8

9

10 11

12 13

14

15

16

17

18

19 20

21

22

23

24 25 26

27

28

29

30

Mr. Grabowski made a motion that the Board make an advisory determination that the Developer's Rights and Responsibilities Agreement for the Doughoregan Manor is consistent with the General Plan as amended by CB 9-2010. Ms. CitaraManis seconded the motion. The motion passed by a vote of 5 to 0.

For the foregoing reasons, the Planning Board of Howard County, Maryland, on this 17th day of June, 2010, determines that the Development Rights and Responsibilities Agreement for the Doughoregan Manor proposed by Camilla Carroll and Philip D. Carroll, is consistent with the Howard County General Plan, as amended by Council Bill 9-2010 as noted above.

HOWARD COUNTY PLANNING BOARD TPG Dombrowski. Chair David Grabowski, Vice-Chair Manie ſamm∀ taral Paul Yelder ATTEST: Marsha S. McLaughlin, Executive Secretary

EXHIBIT 6

•

Commitment Letter

.



HOWARD COUNTY OFFICE OF COUNTY EXECUTIVE 3430 Courthouse Drive Ellicott City, Maryland 21043 # 410-313-2013

Ken Ulman Howard County Executive

www.howardcountymd.us FAX 410-313-3051 TDD 410-313-2323

May 18, 2010

VIA OVERNIGHT MAIL

Camilla Carroll and Philip Carroll c/o Camilla Carroll 3500 Manor Lane Ellicott City, Maryland 21042

Dear Ms, Carroll and Mr, Carroll:

Howard County, Maryland is pleased to offer to purchase the development rights as defined in Section 15.502 of the Howard County Code ("Development Rights") on the parcel of land consisting of approximately 500 acres, more or less, which you own in Howard County, Maryland and which is located at 3500 Manor Lane, Ellicott City, Howard County, Maryland (the "Land") pursuant to (i) an installment purchase agreement (the "Installment Purchase Agreement") between Howard County, Maryland, as purchaser (the "County") and Camilla Carroll and Philip Carroll, as sellers (the "Seller") and (ii) a Deed of Agricultural Land Preservation Easement between the County and the Seller (the "Deed of Easement"), upon and subject to the terms and conditions hereinafter set forth:

1. <u>Purchase Price</u>. The County offers to purchase the Development Rights from the Seller for a total purchase price of no more than \$19,100,000.00, (the "Purchase Price") which is based on a price of \$38,200.00 per acre, rounded to the next highest \$1,000.00. The Seller will receive 10% of the Purchase Price at settlement. The balance of the Purchase Price shall be paid to the Seller or Seller's assignee in 20 equal annual installments commencing on August 15, 2011 and on each August 15 thereafter. This Purchase Price is based on the following agreements and conditions with respect to the Land which shall be set forth in the Deed of Easement:

- (a) The Land is currently a portion of a parcel of record. The Land may be divided into no more than five (5) 50+ acre parcels;
- (b) Ten (10) one-acre lots may be subdivided from the Land upon the approval of the Agricultural Land Preservation Board and the Department of Planning and Zoning; and
- (c) A maximum of fifteen (15) tenant houses may be permitted on the Land upon the approval of the Agricultural Land Preservation Board and the Department of Planning and Zoning.

2. Interest on Unpaid Balance of Purchase Price. Interest on the unpaid balance of the Purchase Price shall accrue from the date of settlement ("Closing Date") and shall be paid to the Seller or Seller's assignee in 40 semiannual payments on February 15 and August 15 in each year after the Closing Date. Interest shall accrue and be payable at (a) the interest rate, as determined by the County's Director of Finance at the time of settlement of the first Batch 14 property to settle, which is equal to the greater of the average of the United States Treasury Yields or AAA Tax-Exempt General Obligation Yields, in each case for the closest available date to each principal payment installment date for the first Batch 14 property to settle, or (b) 4% per annum, whichever is lower. Interest shall be calculated on the basis of a 360-day year of twelve 30-day months.

3. <u>Documentation</u>. All instruments and documents required hereby or affecting the Land, or relating to the Seller's capacity and authority to sell the Development Rights and to execute the Documents and such other documents, instruments, opinions, assurances, consents and approvals as the County may request and all procedures connected herewith shall be subject to the approval, as to form and substance, of the County, the
Camilla Carroll and Philip Carroll property May 18, 2010 Page 2 of 4

County's counsel, and Miles & Stockbridge P.C. ("Bond Counsel"). All of the documents evidencing the installment purchase of the Development Rights (the "Documents"), including without limitation, the Installment Purchase Agreement and the Deed of Easement, shall be prepared by counsel for the County or by Bond Counsel. In addition to those items specifically set forth in this commitment, the Seller shall furnish to the County, prior to the Closing Date, such additional instruments, documents, opinions and materials as the County may require, all of which shall be satisfactory to the County in all respects.

4. <u>Conditions Precedent to Closing</u>. As a condition precedent to the County's obligation to close, not less than seven (7) days prior to the Closing Date:

(a) The Seller shall furnish to the County, a Subordination Agreement in recordable form from each mortgagee and other lienholder having a lien on all or any portion of the Land subordinating such person's interest in the Land to the rights of the County under the Deed of Easement, which Subordination Agreement shall be satisfactory in all respects to the County.

(b) If the Seller is not a natural person acting only in the Seller's individual capacity, Seller shall furnish to the County an opinion of Seller's Counsel, which is satisfactory in form and substance to the County, stating that Seller has the authority to convey the development rights to the County and has taken all actions necessary to validly exercise that authority.

(c) The County shall have received a title insurance binder with a commitment to issue a title insurance policy in the amount of the Purchase Price insuring the acquisition of the easement on the Land and ownership of the Development Rights by the County subject only to those exceptions to title as are approved by the County and its counsel, and with affirmative insurance on such matters as the County may require. The County will secure the title insurance.

(d) The County shall have verified with the Soil Conservation District that the Land is subject to an approved Soil Conservation and Water Quality Plan that reflects current conditions and activities on the Land.

5. <u>Expenses</u>. All costs relating to the recording of the Deed of Easement and any Subordination Agreement, all title examination charges, the premium for the title insurance policy, and the fees of Bond Counsel for a reasonable number of hours of time expended on consultation with legal or financial advisors of the Seller and the out-of-pocket expenses of Bond Counsel shall be paid by the County.

The Seller shall pay the fees and expenses of the Seller's own counsel and, if necessary, licensed engineer or surveyor, and all costs of preparation and recording of other documents, if any, required to perfect the title to the Land and provide a metes and bounds description for the Deed of Easement.

6. <u>Termination by County</u>. This commitment is being made in reliance upon the information supplied by the Seller to the County in connection with the sale of the Development Rights. If the County, acting in good faith, should determine that any such information or supporting representation of a material nature is false, inaccurate, incomplete or misleading, the County may rescind and cancel this commitment.

7. <u>Brokerage</u>. The County shall pay no fee or commission to any broker or agent in connection with the purchase of the Development Rights, and the Seller hereby agrees to indemnify and hold harmless the County against all claims for brokerage fees and commissions.

8. <u>Receipt of Opinion of Bond Counsel</u>. It is a condition precedent to the closing of the transaction contemplated hereby that the County and the Seller receive an opinion from Bond Counsel, dated the Closing Date, to the effect that under existing laws, regulations, rulings and decisions, interest paid under the Installment Purchase Agreement is not includable in the gross income of the Seller (or any holder of the Installment Purchase Agreement) for federal income tax purposes, which opinion may assume continuous compliance with certain

Camilla Carroll and Philip Carroll property May 18, 2010 Page 3 of 4

covenants in the Tax Certificate and Compliance Agreement to be executed and delivered by the County on the date of delivery of the Certificate and may be otherwise limited in accordance with its terms.

9. <u>Acknowledgement of Seller with Regard to Tax Consequences of Transaction</u>. The Seller acknowledges that the Seller has made an independent investigation and has consulted with attorneys, accountants and others selected by the Seller with respect to all tax considerations related to the transaction contemplated hereby (other than the matter described in Section 8 hereof), and the Seller certifies that the Seller has not looked to or relied upon the County or any of its officials, agents or employees, or to Bond Counsel, with respect to any of such matters.

10. <u>Commitment Subject to Approval by Seller of Final Documents</u>. Final Documents for execution, consistent with this commitment, shall be satisfactory to the Seller in form and substance. In the event that the Documents are not acceptable to Seller for any reason (including the failure to execute a Development Rights and Responsibility Agreement) and cannot be made so, or this transaction is not acceptable to the Seller for any reason, Seller, at Seller's option, shall be released from this commitment.

11. <u>Acceptance and Counterparts</u>. To accept this commitment, a copy of this commitment must be signed (and the signature witnessed) by each person with an ownership interest in the property and the copy with original signature of the Seller, delivered to the County no later than May 20, 2010. This commitment may be executed in counterparts, each of which shall be considered an original and all of which shall, together, constitute a single instrument.

12. <u>Commitment Subject to Enaotment of Approval Ordinance</u>. Because the Installment Purchase Agreement is a multi-year contract subject to the provisions of Section 612 of the County's charter, the Installment Purchase Agreement must be approved by ordinance ("Approval Ordinance") and the County is not authorized to execute and deliver the Installment Purchase Agreement until the Approval Ordinance is enacted. The Approval Ordinance will be prefiled for introduction to the County Council of Howard County (the "Council") on May 27, 2010, for the Approval Ordinance to be considered by the 'Council in June. If for any reason the Approval Ordinance is not enacted by October 1, 2010, this commitment shall be null and void and the County and Seller shall have no further obligation hereunder.

13. Assignment Prohibited. This commitment may not be assigned or in any way transferred by the Seller.

14. <u>Entire Agreement</u>. No statements, agreement or representations, oral or written, which may have been made to the Seller or to any employee or agent of the Seller, either by the County or by any employee, agent or broker acting on the Seller's behalf, with respect to the purchase of the Development Rights on the Land, including the Purchase Price, the Deed of Easement, or Installment Purchase Agreement, shall be of any force or effect, except to the extent stated in this commitment, and all prior agreements and representations with respect to the matters in this commitment are merged herein. This commitment may not be changed except by written agreement signed by the Seller and the County.

15. Governing Law. The Seller agrees that this commitment and the Documents shall be governed by and construed under the laws of the State of Maryland.

16. <u>Closing Date: Survival</u>. This transaction may be closed after all conditions precedent to closing have been met. Unless the Seller and the County enter into a Development Rights and Responsibilities Agreement concerning the Land and the Seller's property adjacent to the Land, this transaction must be fully closed by February 1, 2011 or this commitment shall be deemed null and void. Unless otherwise agreed to by the County, the closing shall take place in the main office of the County or at the office of Bond Counsel in Baltimore, Maryland, as the County might designate. The terms and conditions of this commitment shall survive the closing; provided, however, that if any of the terms and conditions of this. Camilla Carroll and Philip Carroll property May 10, 2010 Page 4 of 4

commitment shall conflict with any of the terms and conditions of the Documents, the terms and conditions of the Documents shall prevail. The terms of this commitment shall supersede in full any prior commitment issued by the County in connection with the transaction contemplated hereby.

We are pleased to make this offer to you. Enclosed for your review is an updated score sheet and amortization schedule. Also attached is a sample deed of easement and a sample installment purchase agreement. Please indicate your acceptance of this commitment by signing and returning to us one of the executed originals of this letter no later than May 20, 2010. If not accepted prior to May 21, 2010, the offer set forth in this commitment letter terminates automatically and shall have no further force and effect.

ATTEST:

Lonnie Robbins Chief Administrative Officer

HOWARD COUNTY, MARYLAND

Bv: Ken Ulman

County Executive

Approved for Form and Legal Sufficiency this $\underline{ \mathcal{A}}_{\mathcal{A}}$ day of $\underline{ \mathcal{A}}_{\mathcal{A}}$, 2010:

MargareyAnn Nolan County Bolicitor

THE FOREGOING TERMS AND CONDITIONS WITH RESPECT TO THE LAND ARE HEREBY AGREED TO AND ACCEPTED THIS ______ DAY OF _____, 2010.

Witness:

amilla Carroll

Witness:

Philip Carroll

EXHIBIT 7

Option Agreement

.

OPTION AGREEMENT

THIS OPTION AGREEMENT (this "Option"), is made this _____ day of _____, 2010 ("Effective Date") by and between by and between CAMILLA CARROLL and PHILIP D. CARROLL, individuals, (collectively referred to as the "Carrolls") and Howard County, MARYLAND, a body corporate and politic of the State of Maryland ("Howard County"). The Carrolls and Howard County are hereinafter referred to collectively as the "Parties".

RECITALS

2. The Carrolls intend to develop a 221.1 AC \pm portion of Doughoregan as depicted on **EXHIBITS 2** and **3** (the "Site") of the DRRA with not more than 325 single family detached residential dwelling units substantially in the manner as set forth in EXHIBIT 2 (the "Project").

3. The Carrolls, furthermore, intend to sell and the County intends to purchase a perpetual "Agricultural Land Preservation Easement" on 500 acres of the Property more particularly described in the Exhibit A attached hereto ("Agricultural Preservation Parcel") subject to the terms and conditions contained in the Commitment Letter dated May 18, 2010 setting forth the terms and conditions for the County's purchase of the Development Rights on the Agricultural Preservation Parcel for \$19,100,000.00 (the "Purchase Price") between the County and the Carrolls, a copy of which is attached as Exhibit B (as amended in the DRRA to delete paragraph 10, the "Commitment Letter"), which is incorporated by reference.

4. Notwithstanding anything in the Commitment Letter to the contrary, either expressed or implied, the Parties intend that the obligations contained in the Commitment Letter shall be binding upon the Parties upon occurrence of the conditions precedent set forth in Section 1.2 below, whereupon the County shall exercise its rights under this Agreement and upon the failure of the conditions precedent set forth herein the Carrolls shall perform their obligations under this Agreement.

5. The Parties are entering into this Option to effectuate their mutual intent as specified in Recital No. 4.

NOW, THEREFORE, in consideration of the DRRA, the Commitment Letter, the recitals set forth herein, and for other good and valuable consideration, the receipt and adequacy of which they each acknowledge, Carrolls grant to Howard County and Howard

County accepts, the exclusive option to obtain the perpetual Agricultural Land Preservation Easement in, over, on, and through the Agricultural Preservation Parcel as set forth in the Commitment Letter as amended by the DRRA as set forth herein. The Carrolls and the County agree that the sale of the Agricultural Land Preservation Easement to the County is a material term of the DRRA and without the consideration of the sale of the Agricultural Land Preservation Easement to the County on the terms and conditions set forth in the Commitment Letter, the County would not have agreed to the DRRA.

Section 1. Option Term and Exercise of Option.

1.1 <u>Term</u>. The term of the Option ("Option Period") shall begin on the Effective Date of this Option, and shall expire (unless sooner exercised by written notice from the County) on January 1, 2013 ("Outside Date"). In the event any one of the conditions precedent to exercise set forth in Section 1.2.1-1.2.3 have not yet been determined as of the Outside Date, then the Option Period shall be extended automatically until (i) all conditions precedent have been determined or (ii) the termination of the DRRA. If any of the conditions precedent have not yet been determined as of the Outside Date, the Carrolls may take action to return the Property to a rural conservation zoning classification (now known as "RC"). Upon the final, unappealable approval of the rural classification zoning approval for the Site, the County shall terminate its right to acquire the Agricultural Land Preservation Easement under this Option Agreement and the Commitment Letter.

1.2 <u>Exercise</u>. Howard County shall exercise the Option upon the occurrence of all of the following conditions precedent by providing written notice to the Carrolls:

1.2.1 The final, unappealable approval of amendment(s) to the Howard County General Plan for extension of the Planned Service Area for water and sewerage for the Site including any amendment(s) to the Master Plan for Water and Sewerage and any application(s) for incorporation into the Howard County metropolitan district.

1.2.2 The final, unappealable approval of the DRRA; and

1.2.3 The final, unappealable approval of R-ED zoning approval for the Site.

1.3 Effect of Exercise. Upon the exercise of the Option, Howard County shall become entitled and obligated to purchase the Agricultural Land Preservation Easement from the Carrolls, and the Carrolls shall become obligated to sell the Agricultural Land Preservation Easement to Howard County on the terms and conditions set forth in the Commitment Letter. Furthermore, by exercising the Option, Howard County shall become obligated to the Carrolls, their heirs, successors and/or assigns to full performance under this Option.

1.4 <u>Effect of Conditions Precedent</u>. The Parties agree no Subdivision Plat for the Property creating any individual lots meeting the bulk requirements of R-ED zoning for single family detached housing shall be submitted by the Carrolls or reviewed by the County until all of the conditions precedent set forth in Sections 1.2.1 through 1.2.3 are satisfied.

1.5 <u>Failure of Conditions Precedent</u>. The Parties further agree that each shall take all actions required to promptly return the Property to a rural conservation zoning classification (now known as "RC") in the event any one of the conditions precedent in Sections 1.2.1 through 1.2.3 hereof is not satisfied. Upon the final, unappealable approval of the rural classification zoning approval for the Site, the County shall terminate its right to acquire the Agricultural Land Preservation Easement under the Commitment Letter.

Section 2. <u>Settlement</u>.

2.1 <u>Settlement Date</u>. Upon exercise of the Option, the Carrolls and Howard County shall set a mutually agreeable date for settlement to occur within ten (10) business days of the exercise of the Option. As set forth in the Commitment Letter, the County shall pay the Carrolls 10% of the Purchase Price of \$19,100,000.00 in cash and issue an installment purchase agreement for the balance of the Purchase Price and the Carrolls shall execute and deliver to County the Deed of Agricultural Land Preservation Easement as described in the Commitment Letter.

2.2 <u>Title</u>. The title to the Agricultural Land Preservation Easement shall be subject only to the effect of the matters recorded among the Land Records of Howard County and deemed acceptable to the County, as set forth in the Commitment Letter. While this Option is in effect, Carrolls shall not transfer all or any portion of their interest in the Agricultural Preservation Parcel or otherwise encumber the title to the Agricultural Land Preservation Easement by any lien, or other interest. The Carrolls agree to cause any lender or lien holder or other person having an interest in the Agricultural Preservation Parcel to subordinate their interest in the Agricultural Preservation Parcel to the County's Agricultural Land Preservation Easement and such subordination agreement shall be in a form acceptable to the County and recorded among the Land Records of Howard County.

Section 3. <u>Specific Performance</u>. If Howard County exercises its Option and the Carrolls fail to agree to a settlement date and consummate settlement, the Carrolls consent to Howard County obtaining a decree against them for specific performance as set forth herein. The Carrolls agree that the County entered into the Commitment Letter, this Option Agreement, and the DRRA because of the Carrolls' agreement to conserve and protect the 500 \pm acres pursuant to the Howard County Agricultural Land Preservation Act, and that there are no liquidated damages or other types of monetary damages that would be adequate to compensate the County for the Carrolls' breach of that promise. Therefore, Howard County may file, in lieu and as a substitute for monetary damages, a complaint against the Carrolls in the Circuit Court for Howard County for specific

performance of their promise to execute and deliver to Howard County a Deed of Agricultural Land Preservation Easement, pursuant to the Commitment Letter, the DRRA, and this Option Agreement. A proposed form of complaint is attached hereto as Exhibit C. The Carrolls appoint and authorize [NAME, ADDRESS] as their agent to receive service of process and filings in the action. Except for such service, as an expedient compromise of the complaint and without admitting liability, the Carrolls knowingly, intelligently, and voluntarily waive all rights, defenses, and claims, from whatever source derived, both procedural and substantive, that they may have to Howard County's action to enforce its rights for specific performance of the acquisition of the Agricultural Land Preservation Easement, and the Carrolls consent to Howard County moving for entry of a consent decree against them for specific performance and to the entry of such a decree in the form attached hereto as Exhibit D.

Section 4. <u>General</u>.

4.1. <u>Notices</u>. Any communication to be given to a party shall be in writing, shall be deemed to have been given on the 3rd business day after being sent as certified or registered mail in the United States mails, postage prepaid, return receipt requested, or on the next business day after being deposited with a reputable overnight courier service as follows:

In the case of the Carrolls notice shall be sent to: Camilla and Phillip D. Carroll 3500 Manor Lane Ellicott City, MD 21042

with a copy to:

Joseph Rutter 5300 Dorsey Hall Drive Ellicott City, MD 21042

Sang W. Oh, Esq. 5100 Dorsey Hall Drive Ellicott City, MD 21042

James L. Wright, Esq. 750 E. Pratt Street Suite 900 Baltimore, MD 21202

In the case of the County, notice shall be sent to: Director of Department of Planning and Zoning 3930 Courthouse Drive Ellicott City, Maryland 21043 With a copy to:

County Solicitor of Howard County 3930 Courthouse Drive Ellicott City, Maryland 21043

4.2. <u>Effect; Binding; Amendment; Counsel</u>. This Option shall become effective on its execution and delivery by each party. This Option may be amended only by a document signed by each party. The Parties agree that this Option shall run with the land and be binding upon and inure to the benefit of each party and their respective heirs, successors and assigns, and upon any and all successor owners of record of all or any portion of the Site. Each party has entered into this Option and the Commitment Letter after having the opportunity to receive advice of legal counsel and represents to the other that each understands their respective legal obligations under this Option.

4.3. <u>Governing Law; Jurisdiction</u>. This Option shall be governed by and construed and enforced in accordance with the laws of the State of Maryland. The parties hereby irrevocably accept and submit to the jurisdiction of the Circuit Court for Howard County, Maryland, in any action brought to enforce this Option, suit, action or proceeding and further waive any objection and any right of immunity on the ground of venue, the inconvenience of any forum or the jurisdiction of such courts or from the execution of judgments resulting therefrom.

4.4. <u>Rules of Construction</u>. The enumeration and headings of the sections of this Option are merely for convenience of reference and do not constitute representations or warranties, do not impose any obligations whatever and have no substantive significance. Unless the context otherwise requires, whenever used in this Option the singular will include the plural, the plural will include the singular, and the masculine gender will include the neuter or feminine gender and vice versa. As to the Carrolls, each of the obligations in this Option and the Commitment letter shall be joint and several. The Parties agree that neither shall be considered the primary drafter of this Option.

4.5 <u>Conflict with DRRA; Survival</u>. The Parties hereby agree that any conflict between the terms of the DRRA and this Option shall be interpreted to give this Option its full force and effect to allow the County to acquire the Agricultural Land Preservation Easement. The Parties further agree that this Option shall survive the invalidation of all or any portion of the DRRA.

WITNESS/ATTEST:

CAMILLA CARROLL

_____(SEAL)

PHILIP D. CARROLL

_____(SEAL)

AGREED and APPROVED:

HOWARD COUNTY, MARYLAND

ATTEST:

BY: _____(SEAL)

Lonnie R. Robbins Chief Administrative Officer Ken Ulman Howard County Executive

APPROVED AS TO FORM AND LEGAL SUFFICIENCY

this _____ day of _____ 20_.

Margaret Ann Nolan County Solicitor

[Notaries on Following Page]

STATE OF MARYLAND, _____ COUNTY, TO WIT:

I HEREBY CERTIFY that on this _____ day of _____, 2010, before me, the subscriber, a Notary Public of the State of Maryland, in and for the County aforesaid, personally appeared CAMILLA CARROLL, personally known to me or proven to be the individual named herein and executed this Option for the purposes stated therein.

AS WITNESS my Hand and Notarial Seal.

My Commission Expires: ______ Notary Public

STATE OF MARYLAND, _____ COUNTY, TO WIT:

I HEREBY CERTIFY that on this _____ day of _____, 2010, before me, the subscriber, a Notary Public of the State of Maryland, in and for the County aforesaid, personally appeared PHILIP D. CARROLL, personally known to me or proven to be the individual named herein and executed this Option for the purposes stated therein.

AS WITNESS my Hand and Notarial Seal.

Notary Public

My Commission Expires: _____

STATE OF MARYLAND, _____ COUNTY, TO WIT:

I HEREBY CERTIFY that on this _____ day of ______, 2010, before me, the subscriber, a Notary Public of the State of Maryland, in and for the County aforesaid, personally appeared KEN ULMAN, the County Executive for Howard County, Maryland, who acknowledged the same to be the act of the County and that he executed the foregoing Option for the purposes therein contained by signing in my presence the name of Howard County, Maryland as County Executive.

AS WITNESS my Hand and Notarial Seal.

Notary Public

My Commission Expires:

ATTORNEYS' CERTIFICATION

THIS IS TO CERTIFY that the undersigned are members, in good standing, of the Bar of the Court of Appeals of Maryland, and that the within instrument was prepared by the undersigned or under their supervision.

Upon Recordation Please Return To:

Sang W. Oh, Esq. Talkin & Oh, LLP 5100 Dorsey Hall Drive Ellicott City, MD 21042 Exhibit A Description of Agricultural Land Preservation Easement

.

.

.

.

EXHIBIT 2

March 18, 2010

DESCRIPTION OF A 500.000 ACRE AGRICULTURAL PRESERVATION EASEMENT CONSISTING OF 241.638 ACRE PART ONE AND 258.362 ACRE PART TWO PART OF THE PROPERTY OF PHILIP D. CARROLL AND CAMILLA CARROLL SECOND AND THIRD ELECTION DISTRICTS HOWARD COUNTY, MARYLAND

BEING a parcel of land, situate and lying in the Second and Third Election Districts of Howard County, Maryland, said parcel of land being part of the land which by Will of Nina R. Carroll dated June 4, 1979, as amended by its First Codicil dated November 28, 1986 and its Second Codicil dated April 10, 1987; Nina R. Carroll having departed this life on February 11, 1989; said Will having been probated in the Surrogates Court of New York County, New York on April 11, 1989, and the Estate of Nina R. Carroll is filed in the Register of Wills of Howard County, Maryland as Estate No. 17-7868-276, said property also being described in a Mortgage dated October 31, 1962 and recorded among the Land Records of Howard County, Maryland in Liber 394 at Folio 64, and being part of Parcel 1 described therein; said parcel being more particularly described, as now surveyed in the Maryland State Coordinate System NAD' 83 Datum, as projected by Howard County Geodetic Control, as follows:

FISHER, COLLINS & CARTER, INC.

CAVIL EMILICIERNEL CONSCLUENTS And LINED SURVEYORS

10272 Baltimots Nakoasi Pita Filicoli Giy, Maryiats 21042 1410) 481-2835 Fax (410) 780-3784 (PART ONE) BEGINNING FOR THE SAME at a point on or near the intersection of the centerlines of existing paving of Folly Quarter Road with Frederick Road (Maryland Route 144); said point being at the beginning of the Seventeenth or South 82°36'44" Bast, 189.50 foot Deed Line of the aforesaid Liber 394 at Folio 64 (Parcel 1); thence running within said Frederick Road, on or near the centerline of paving, and running with and binding on all of the said Seventeenth line of said Liber 394 at Folio 64 (Parcel 1), and with all of the Highteenth, Nineteenth and a portion of the First line of said Liber 394 at Folio 64 (Parcel 1), as now surveyed the following four (4) courses and distances:

- 1) South 82°54'28" East, 189.50 feet to a point; thence
- 2) North 17°12'33" West, 32.08 feet to a point; thence
- 3) South 82°35'58" East, 1,394.98 feet to a point; thence
- 4) South 73°02'02" East, 2.22 feet to a point on and being a distance of 3,137.87 feet from the end of the aforesaid First or South 73°02'48" East, 3140.09 foot Deed Line of the aforesaid Liber 394 at Folio 64 (Parcel 1) and also being the Northern margin of the sixty-six foot wide Right of Way of Frederick Road (Maryland Route 144); thence leaving said First line of Liber 394 at Folio 64 (Parcel 1) and the Northern margin of Frederick Road (Maryland Route 144); to run over, across and through the aforesaid Liber 394, Folio 64 the following thirteen (13) courses and distances;
- 5) South 06°25'26" West, 152.29 feet to a point; thence
- 6) South 04°05'30" West, 282.20 feet to a point of curvature; thence
- 253.27 feet along the arc of a non-tangential curve to the left, having a radius of
 1,301.42 feet, a central angle of 11°09'01" and subtended by a chord bearing and
 distance of South 02°54'01" East, 252.87 feet to a point; thence
- 8) South 07°27'35" East, 191.10 feet to a point; thence
- 9) South 09°24'23" East, 392.33 feet to a point; thence
- 10) South 09°00'09" East, 293,76 feet to a point; thence

FISHER, COLLINS & CARTER, INC.

ayıl Breditindin Coksultatışı 1941 lako surveyori

(0272 Baltinista Haliazai Piza Ellicoti Gily, Maryizod 23042 (410) 461-2855 722 (410) 760-8784

		· · ·
	11)	South 09°47'54" East, 393.83 feet to a point; thence
ŀ	12)	South 09°03'17" East; 615.16 feet to a point; thence
	13)	South 09°03'55" East, 291.12 feet to a point; thence
	14)	North 79°49'03" West, 1,832.40 feet to a point; thence
	15)	South 19°08'23" West, 1,237,51 feet to a point; thence
	16)	North 77°07'13" West, 68.06 feet to a point; thence
	17)	South 19°08'11" West, 1,248.08 feet to a point on and at a distance of 279.07 feet
		from a 1.25 inch diameter Iron Pipe found at the beginning of the Tenth or North
		70°15'08" West, 1456.95 foot Deed Line of the aforesaid Liber 394 at Folio 64
		(Parcel 1); thence running with and binding on the remaining portion of said
		Tenth line of Liber 394 at Folio 64 (Parcel I), as now surveyed, the following
		course and distance:
1	8)	North 70°10'26" West, 1,177.65 feet to a 1.25 inch diameter Iron Pipe found at
		the end thereof; thence running with and binding on the Bleventh, Twelfth,
		Thirteenth, Fourteenth, Fifteenth and Sixteenth lines of said Liber 394 at Folio 64
		(Parcel 1), the following six (6) courses and distances;
. 1	9)	North 78°34'25" West, 791,22 feet to a point; thence
2	0)	North 73°34'25" West, 387.55 feet to a point; thence
2	1)	South 69°10'35" West, 31,37 feet to a point, thence
2:	2)	North 51°40'56" East, 568.09 feet to a point; thence
2:	3)	North 39°59'12" East, 2,458.50 feet to a point; thence
24	4)	North 26°28'43" East, 2,376.00 feet to the point of beginning, containing 241.638
		acres, more or less.

•

.

FISHER, COLLINS & CARTER, INC.

•

.

CIVIL ENGINEERING CONSULTANTS 2001 LANII SURIVEYONS

10772 Baltimere Halleos) Plur Elitent Ciy, Maryland 21042 (410) 401-2155 For (410) 750-3184

.

(PART TWO) BEGINNING FOR THE SAME at a point on the Northern margin of the sixty-six foot Right of Way of Frederick Road (Maryland Route 144), said point being on and a distance of 53.08 feet from the beginning of the First or South 73°02'48" East, 3'140.09 foot Deed Line of the aforesaid Liber 394 at Folio 64 (Parcel 1); thence running with and binding on a portion of said First line of Liber 394 at Folio 64 (Parcel 1), and the Northern margin of Frederick Road (Maryland Route 144), the following course and distance:

- South 73°02'02" East, 1,379.85 feet to a point thereon; thence leaving the aforesaid First line of Liber 394 at Folio 64 and the Northern margin of Frederick Road (Maryland Route 144) and running for new lines of division, the following twenty-five (25) courses and distances;
- 2) South 16°57'58" West, 65.50 feet to a point; thence
- 3) South 73°02'02" East, 328.95 feet to a point; thence
- 4) South 19°32'01" East, 18.41 feet to a point; thence
- 5) South 73°08'00" East, 180.91 feet to a point; thence
- 6) South 27°46'18" East, 35.58 feet to a point; thence
- 10) South 17°35'24" West, 225.85 feet to a point of curvature; thence
- 11) 346,71 feet along the arc of a curve to the left, having a radius of 315.00 feet, a central angle of 63°03'47" and subtended by a chord bearing and distance of South 13°56'30^u East, 329.47 feet to a point of reverse curvature; thence
- 121.72 feet along the arc of a curve to the right, having a radius of 150.00 feet, a central angle of 46°29'32" and subtended by a chord bearing and distance of South 22°13'37" East, 118.40 feet to a point of tangency; thence

FISHER, COLLINS & CARTER, INC.

CIVIL EXHIDEEHDEN GUUSLEDINTS And LAND SHEVSYDEN

19272 Baltimura Hellovaf Pika Ellivut Citz, Morgiana 21042 1410) 481-2855 Fax (910) 755-9784

- 10) South 01°01'09" West, 232.32 feet to a point of curvature; thence
- 312.10 feet along the arc of a curve to the left, having a radius of 330.00 feet, a central angle of 54°11'16" and subtended by a chord bearing and distance of South 25°04'29" East, 300.60 feet to a point of tangency; thence
- 12) South 53°10'07" East, 259.20 fast to a point of curvature; thence
- 470,43 feet along the arc of a non-tangential curve to the right, having a radius of
 595.00 feet, a central angle of 45°18'00" and subtended by a chord bearing and
 distance of South 30°31'07" East, 458.27 feet to a point of compound curvature;
 thence
- 14) 283.12 feet along the arc of a curve to the right, having a radius of 275.00 feet, a central angle of 58°59'14" and subtended by a chord bearing and distance of South 21°37'30" West, 270.78 feet to a point of reverse curvature; thence
- 15) 219.63 feet along the arc of curve to the left, having a radius of 285.00 feet, a central angle of 44°09'14" and subtended by a chord bearing and distance of South 29°02'30* West, 214.24 feet to a point of tangency; thence
- 16) South 06°57'53" West, 62.64 feet to a point of curvature; thence
- 17) 89.85 feet along the arc of a curve to the left, having a radius of 285.00 feet, a
 central angle of 18°03°45" and subtended by a chord bearing and distance of
 South 02°03'59" East, 89.47 feet to a point of reverse curvature; thence
- 18) 131.70 feet along the arc of a curve to the right, having a radius of 170.00 feet, a central angle of 44°23'10" and subtended by a chord bearing and distance of South 11°05'43" West, 128.43 feet to a point of reverse curvature; thence

FISHER, COLLINS & CARTER, INC.

CIVIL ENRICEERDA CONSULTANTS 428 LAND GUDYEYURT

10272 Hallimore Hallonai Pixe Ellicoit Gig, Margiard 21042 (410) 451-2835 Fax (410) 750-3784

19)	136.14 feet along the arc of a curve to the left, having a radius of 325.00 feet, a
	central angle of 24°00'04" and subtended by a chord bearing and distance of
	South 21°17'16" West, 135.15 feet to a point of tangency; thence
20)	South 09°17'14" West, 113.42 feet to a point of curvature; thence
21)	468.79 feet along the arc of a curve to the left, having a radius of 580.00 feet, a
	central angle of 46°18'34" and subtended by a chord bearing and distance of
	South 13°52'03" East, 456.13 feet to a point of tangency; thence
22)	South 37°01'20" Bast, 903.13 feet to a point; thence
23)	South 23°09'51" East, 507.53 feet to a point; thence
24)	South 67°04'12" West, 462.42 feet to a point; thence
25)	South 33°51'48" West, 651,81 feet to a point; thence
26)	South 16°43'26" West, 858.48 feet to a point on and a distance of 1,764.49 feet
	from a Granite Stone found and held at the beginning of the Seventh or North
	73°14'48" West, 4021.13 foot Deed Line of the aforesaid Liber 394 at Folio 64
	(Parcel 1); thence running with and binding on the remaining portion of said
	Seventh line of Liber 394 at Folio 64 (Parcel 1) and with all of the Eighth and a
	portion of the Ninth or North 65°30'08" West, 2143,94 foot Deed Line of said
	Liber 394 at Follo 64 (Parcel 1), the following three (3) courses and distances:
27)	North 73°16'34" West, 2,253.96 feet to a point; thence
28)	North 14°04'34" East, 24.75 feet to a point; thence
29)	North 65°25'29" West, 39.81 feet to a point; said point being on and a distance of
	2,104,13 feet from a 1.25 inch Iron Pipe found at the end of the aforesaid Ninth
	line of Liber 394 at Folio 64 (Parcel 1); said point also being on the easterly

.

•

.

FISHER, COLLINS & CARTER, INC.

•

.

.

k

CIVIL, IMALILIANAS CONSTITUTIS Rea Land Surveyora

10232 Habileters Habbest Pika Shirok GBJ, Maryland 21042 "410) 487-2855 Fat (410) 350-3784 outline of a plat entitled "Amended Plat Of Easement, Sending Parcel, Property Of Philip Carroll and Camilla Carroll" and recorded as Plat No. 19928; thence I leaving the aforesaid Ninth line of Liber 394 at Folio 64 (Parcel 1) and running with and binding on part of said easterly outline of Plat No. 19928, the following

course and distance:

- 30) North 14°13'15" East, 1,478.90 feet to a point thereon; thence leaving the aforesaid easterly outline of Plat No. 19928 and running over, across and through the aforesaid Liber 394, Folio 64 (Parcel 1) the following twenty-two (22) courses and distances:
- 31) South 85°27'17" East, 421.22 feet to a point; thence

32) South 64°29'11" East, 192.86 feet to a point; thence

33) South 47°22'36" East, 333.45 feet to a point; thence

34) South 27°46'03" East, 102.62 feet to a point; thence

35) South 73°56'56" East, 224.55 feet to a point; thence

36) South 82°25'01" East, 149.17 feet to a point; thence

37) North 12°14'25" East, 389.04 feet to a point; thence

38) North 23°29'52" West, 478.60 feet to a point; thence

39) North 16°15'45" West, 750.96 feet to a point; thence

40) North 81°30'55" West, 730.33 feet to a point; thence

41) North 00°27'40" East, 207.41 feet to a point; thence

42) North 00°40'30" West, 76.57 feet to a point; thence

43) North 06°06'01" West, 117.35 feet to a point; thence

44) North 09°05'56" West, 310.97 feet to a point; thence

FISHER, COLLINS & CARTER, INC.

CAVIL FHODEERCHA COISLAIANTA Ana LAND SUBVEYDDI

10272 Ballins an Hallowel Plat Ethioth Efit, Maryland 21042 (410) 451-(855 Fax (410) 750-3784

- 45) North 09°03'18" West, 615.62 feet to a point; thence
- 46) North 09°47'54" West, 393.78 feet to a point; thence
- 47) North 09°00'07" West, 293.32 feet to a point; thence
- 48) North 09°24'23" West, 391.82 feet to a point; thence
- 49) North 07°27'35" West, 190.70 feet to a point of curvature; thence
- 243.37 feet along the arc of a curve to the right, having a radius of 1,251,42 feet, a central angle of 11°08'33" and subtended by a chord bearing and distance of
 North 02°55'27" West, 242.98 feet to a point of tangency; thence
- 51) North 04°05'30" Hast, 280,56 feet to a point; thence
- 52) North 06°25'26" Bast, 141.96 feet to the point of beginning; containing 258.362 acres, more or less.

SUBJECT TO AND TOGETHER WITH any and all terms and conditions that might exist to the rights, use and maintenance of Folly Quarter Road and Frederick Road, Maryland Route 144, to Howard County, Maryland and the Maryland State Highway Administration.

SUBJECT TO AND TOGETHER WITH all conditions, covenants, easements, restrictions and rights-of-way of record, which may apply.

This description was prepared without the benefit of a Title Report,



Prepared By Checked By

WO #05022-3002

FISHER, COLLINS & CARTER, INC.

GIYIL ENGLASSERDIƏ CONSULTANTS aud LAND SURVITTORS

10272 Batsmort Kalloca) Pika Eniroti City, Nartiacd 21042 (410) 481-285 Fax (410) 750-3784



1:1/2005/050222/dvg/05022-3001 Parcel 1 Exhibit.dvg, 3/18/2010 9:59:18 AM, ds0reat, 1:1



i:\2005\05022\dvvg\05022-3001 Parcei 2 Extibit.dvvg, 3/18/2010 9-49:20 AM, dstreet. 1:1

Exhibit B

Commitment Letter to Acquire Agricultural Land Preservation Easement

.

.

.

.

.

.

.

,

.

.

1. e . e



HOWARD COUNTY OFFICE OF COUNTY EXECUTIVE 3430 Courthouse Drive Ellicott City, Maryland 21043 # 410-313-2013

Ken Ulman Howard County Executive www.howardcountymd.us FAX 410-313-3051 TDD 410-313-2323

May 18, 2010

VIA OVERNIGHT MAIL Camilla Carroll and Philip Carroll c/o Camilla Carroll 3500 Manor Lane Ellicott City, Maryland 21042

Dear Ms. Carroll and Mr. Carroll:

Howard County, Maryland is pleased to offer to purchase the development rights as defined in Section 15.502 of the Howard County Code ("Development Rights") on the parcel of land consisting of approximately 500 acres, more or less, which you own in Howard County, Maryland and which is located at 3500 Manor Lane, Bilicott City, Howard County, Maryland (the "Land") pursuant to (I) an installment purchase agreement (the "Installment Purchase Agreement") between Howard County, Maryland, as purchaser (the "County") and Camilla Carroll and Philip Carroll, as sellers (the "Seller") and (ii) a Deed of Agricultural Land Preservation Easement between the County and the Seller (the "Deed of Basement"), upon and subject to the terms and conditions hereinafter set forth:

1. <u>Purchase Price</u>. The County offers to purchase the Development Rights from the Seller for a total purchase price of no more than \$19,100,000.00, (the "Purchase Price") which is based on a price of \$38,200,00 per acre, rounded to the next highest \$1,000.00. The Seller will receive 10% of the Purchase Price at settlement. The balance of the Purchase Price shall be paid to the Seller or Seller's assignee in 20 equal annual installments commencing on August 15, 2011 and on each August 15 thereafter. This Purchase Price is based on the following agreements and conditions with respect to the Land which shall be set forth in the Deed of Easement:

- (a) The Land is currently a portion of a parcel of record. The Land may be divided into no more than five (5) 50+ sore parcels;
- (b) Ten (10) one-acre lots may be subdivided from the Land upon the approval of the Agricultural Land Preservation Board and the Department of Planning and Zoning; and
- (c) A maximum of fifteen (15) tenant houses may be permitted on the Land upon the approval of the Agricultural Land Preservation Board and the Department of Planning and Zoning.

2. <u>Interest on Unpaid Balance of Purchase Price</u>. Interest on the unpaid balance of the Purchase Price shall accrue from the date of settlement ("Closing Date") and shall be paid to the Seller or Seller's assignee in 40 semiannual payments on February 15 and August 15 in each year after the Closing Date. Interest shall accrue and be payable at (a) the interest rate, as determined by the County's Director of Finance at the time of sottlement of the first Batch 14 property to settle, which is equal to the greater of the average of the United States Treasury Yields or AAA Tax-Exempt General Obligation Yields, in each case for the closest available date to each principal payment installment date for the first Batch 14 property to settle, or (b) 4% per annum, whichever is lower. Interest shall be calculated on the basis of a 360-day year of twelve 30-day months.

3. <u>Documentation</u>. All instruments and documents required hereby or affecting the Land, or relating to the Seller's capacity and authority to sell the Development Rights and to execute the Documents and such other documents, instruments, opinions, assurances, consents and approvals as the County may request and all procedures connected herewith shall be subject to the approval, as to form and substance, of the County, the

Camila Carroll and Philip Carroll property May 18, 2010 Page 2 of 4

County's counsel, and Miles & Stookbridge P.C. ("Bond Counsel"). All of the documents evidencing the installment purchase of the Development Rights (the "Documents"), including without limitation, the Installment Purchase Agreement and the Deed of Basement, shall be prepared by counsel for the County or by Bond Counsel. In addition to those items specifically set forth in this commitment, the Seller shall furnish to the County, prior to the Closing Date, such additional instruments, documents, opinions and materials as the County may require, all of which shall be satisfactory to the County in all respects.

4. <u>Conditions Precedent to Closing</u>. As a condition precedent to the County's obligation to close, not less than seven (7) days prior to the Closing Date:

(a) The Seller shall furnish to the County, a Subordination Agreement in recordable form from each mortgagee and other lienholder having a lien on all or any portion of the Land subordinating such person's interest in the Land to the rights of the County under the Deed of Easement, which Subordination Agreement shall be satisfactory in all respects to the County.

(b) If the Seller is not a natural person aoting only in the Seller's individual capacity, Seller shall furnish to the County an opinion of Seller's Counsel, which is satisfactory in form and substance to the County, stating that Seller has the authority to convey the development rights to the County and has taken all actions necessary to validly exercise that authority.

(c) The County shall have received a title insurance blnder with a commitment to issue a title insurance policy in the amount of the Purchase Price insuring the acquisition of the casement on the Land and ownership of the Development Rights by the County subject only to those exceptions to title as are approved by the County and its connsel, and with affirmative insurance on such matters as the County may require. The County will secure the title insurance.

(d) The County shall have verified with the Soil Conservation District that the Land is subject to an approved Soil Conservation and Water Quality Plan that reflects current conditions and activities on the Land,

5. <u>Expenses</u>. All costs relating to the recording of the Deed of Easement and any Subordination Agreement, all title examination charges, the premium for the title insurance policy, and the fees of Bond Counsel for a reasonable number of hours of time expended on consultation with legal or financial advisors of the Seller and the out-of-pocket expenses of Bond Counsel shall be paid by the County.

The Seller shall pay the fees and expenses of the Seller's own counsel and, if necessary, licensed engineer or surveyor, and all costs of preparation and recording of other documents, if any, required to perfect the title to the Land and provide a metes and bounds description for the Deed of Easement.

6. <u>Termination by County</u>. This commitment is being made in reliance upon the information supplied by the Seller to the County in connection with the sale of the Development Rights. If the County, acting in good faith, should determine that any such information or supporting representation of a material nature is false, inaccurate, incomplete or misleading, the County may rescind and cancel this commitment.

7. <u>Brokerage</u>. The County shall pay no fee or commission to any broker or agent in connection with the purchase of the Development Rights, and the Seller hereby agrees to indemnify and hold harmless the County against all claims for brokerage fees and commissions.

8. <u>Receipt of Opinion of Bond Counsel</u>. It is a condition precedent to the closing of the transaction contemplated hereby that the County and the Soller receive an opinion from Bond Counsel, dated the Closing Date, to the effect that under existing laws, regulations, rulings and decisions, interest paid under the Installment Purchase Agreement is not includable in the gross income of the Seller (or any holder of the Installment Purchase Agreement) for federal income tax purposes, which opinion may assume continuous compliance with certain

 $\sim C$

Carnilla Carroll and Philip Carroll property May 18, 2010 Page 3 of 4

covenants in the Tax Certificate and Compliance Agreement to be executed and delivered by the County on the date of delivery of the Certificate and may be otherwise limited in accordance with its terms.

9. <u>Approvledgment of Seller with Regard to Tax Consequences of Transaction</u>. The Seller acknowledges that the Seller has made an independent investigation and has consulted with attorneys, accountants and others selected by the Seller with respect to all tax considerations related to the transaction contemplated hereby (other than the matter described in Section 8 hereof), and the Seller certifies that the Seller has not looked to or relied upon the County or any of its officials, agents or employees, or to Bond Counsel, with respect to any of such matters.

10. <u>Commitment Subject to Approval by Seller of Final Documenta</u>. Final Documents for execution, consistent with this commitment, shall be satisfactory to the Seller in form and substance. In the event that the Documents are not acceptable to Seller for any reason (including the failure to execute a Development Rights and Responsibility Agreement) and cannot be made so, or this transaction is not acceptable to the Seller for any reason, Seller, at Seller's option, shall be released from this commitment.

11. <u>Acceptance and Counterparts</u>. To accept this commitment, a copy of this commitment must be signed (and the signature witnessed) by each person with an ownership interest in the property and the copy with original signature of the Seller, delivered to the County no later than May 20, 2010. This commitment may be executed in counterparts, each of which shall be considered an original and all of which shall, together, constitute a single instrument.

12. <u>Commitment Subject to Enactment of Approval Ordinance</u>. Because the Installment Purchase Agreement is a multi-year contract subject to the provisions of Section 612 of the County's charter, the Installment Purchase Agreement must be approved by ordinance ("Approval Ordinance") and the County is not authorized to execute and deliver the Installment Purchase Agreement until the Approval Ordinance is enacted. The Approval Ordinance will be prefiled for introduction to the County Council of Howard County (the "Council") on May 27, 2010, for the Approval Ordinance to be considered by the Council in June. If for any reason the Approval Ordinance is not enacted by October 1, 2010, this commitment shall be null and void and the County and Seller shall have no further obligation hereunder.

13. <u>Assignment Prohibited</u>. This commitment may not be assigned or in any way transferred by the Seller.

• 14. <u>Entire Agreement</u>. No statements, agreement or representations, oral or written, which may have been made to the Seller or to any employee or agent of the Seller, either by the County or by any employee, agent or broker acting on the Seller's behalf, with respect to the purchase of the Development Rights on the Land, including the Purchase Price, the Deed of Easement, or Installment Purchase Agreement, shall be of any force or effect, except to the extent stated in this commitment, and all prior agreements and representations with respect to the matters in this commitment are merged herein. This commitment may not be changed except by written agreement signed by the Seller and the County.

15. <u>Governing Law</u>. The Seller agrees that this commitment and the Documents shall be governed by and construed under the laws of the State of Maryland.

16. <u>Closing Date: Survival</u>. This transaction may be closed after all conditions precedent to closing have been met. Unless the Seller and the County enter into a Development Rights and Responsibilities Agreement concerning the Land and the Seller's property adjacent to the Land, this transaction must be fully closed by Pebruary 1, 2011 or this commitment shall be deemed null and void. Unless otherwise agreed to by the County, the closing shall take place in the main office of the County or at the office of Bond Counsel in Baltimore, Maryland, as the County might designate. The terms and conditions of this commitment shall survive the closing; provided, however, that if any of the terms and conditions of this

CC

Camilla Carroll and Philip Carroll property May 10, 2010 Page 4 of 4

commitment shall conflict with any of the terms and conditions of the Documents, the terms and conditions of the Documents shall prevail. The terms of this commitment shall supersede in full any prior commitment issued by the County in connection with the transaction contemplated hereby.

We are pleased to make this offer to you. Enclosed for your review is an updated score sheet and amortization schedule. Also attached is a sample deed of easement and a sample installment purchase agreement. Please indicate your acceptance of this commitment by signing and returning to us one of the executed originals of this letter no later than May 20, 2010. If not accepted prior to May 21, 2010, the offer set forth in this commitment letter terminates automatically and shall have no further force and effect.

ATTEST:

ophie Nobbins

Chief Administrative Officer

HOWARD COUNTY, MARYLAND

Bv: Ken Ulman

County Executive

Approved for Form and Legal Sufficiency this 20 day of , 2010: Margaret/Ann Nolan

County Bolicitor

THE FOREGOING TERMS AND CONDITIONS WITH RESPECT TO THE LAND ARE HEREBY AGREED TO AND ACCEPTED THIS ______ DAY OF _____, 2010.

Witness:

Camilla Carroll

Witness:

Philip Carroll

Complaint for Speemer encounter									
HOWARD COUNTY, MARYLAND	*	IN THE							
3430 Courthouse Drive Ellicott City, Maryland 21043	*	CIRCUIT COURT							
Plaintiff	*	FOR							
v.	*	HOWARD COUNTY							
CAMILLA CARROLL	*								
3500 Manor Lane Ellicott City, Maryland 21042	*	Case No.:							
SERVE ON:	*								
[name] [address]	*								
[city, state, zip]	*								
and	*								
PHILIP D. CARROLL [address]	*								
[city], New York [zip]	*								
SERVE ON: [name]	*								
[address] [city, state, zip]	*								
Defendants	*								
* * * *	*	* * * * *							

Exhibit C Complaint for Specific Performance

VERIFIED COMPLAINT FOR SPECIFIC PERFORMANCE

*

Howard County, Maryland (the "County"), by its attorneys, Margaret Ann Nolan, County Solicitor, and Gary W. Kuc, Senior Assistant County Solicitor, sues Camilla Carroll and Philip D. Carroll (the "Carrolls"), Defendants, for specific performance of their promise to execute and deliver to the County a Deed of Agricultural Land Preservation Easement for 500 acres.

PARTIES

1. Plaintiff Howard County, Maryland is a body corporate and politic of the State of Maryland.

2. Defendant Priscilla Carroll is a resident of Howard County, Maryland.

3. Defendant Philip D. Carroll is a resident of [city / county], New York.

4. The Carrolls own certain real property in Howard County, Maryland, described on Howard County Tax Map No. 2523, Parcel 71 consisting of 892.6 acres \pm ("Doughoregan" or the "Property").

JURISDICTION

5. This Court has subject matter jurisdiction over this equitable action for specific performance pursuant to § 1-501 of the Courts and Judicial Proceedings Article of the Annotated Code of Maryland.

6. This Court has personal jurisdiction over the Carrolls pursuant to § 6-102 and § 6-103 of the Courts and Judicial Proceedings Article of the Annotated Code of Maryland.

7. Furthermore, the Carrolls have waived any defense based on lack of subject matter jurisdiction or lack of personal jurisdiction, pursuant to § 4 of an Option Agreement between the parties. A true and correct copy of the fully executed Option Agreement is attached hereto and incorporated as Exhibit A.

VENUE

8. Venue is proper in this Court pursuant to § 6-201 and § 6-202 of the Courts and Judicial Proceedings Article of the Annotated Code of Maryland.

9. Furthermore, the Carrolls have waived any defense based on venue pursuant to § 4 of the Option Agreement between the parties.

FACTS

10. The Howard County Agricultural Land Preservation Act is codified in § 15.500 through § 15.521 of the Howard County Code (the "Act"). A true and correct copy of the Act is attached hereto and incorporated as Exhibit B.

11. The purpose of the Act is to protect the health, safety, and well-being of present and future residents of Howard County by conserving and protecting agricultural land as an environmental resource of major importance and as the basis of a viable sector of the County's economy. § 15.500(c).

12. To effectuate the Act's purpose, the County acquires the development rights on eligible land by purchase of the land, by purchase of the development rights on the land, and by acceptance of dedicated remainders from cluster subdivisions as provided by law. \S 15.500(c).

13. The County may acquire an agricultural land preservation easement on land in the County by, among other things, purchasing the development rights on eligible land.§ 15.503(a).

14. The parties have a valid and enforceable contract for the purchase by the County of the Carrolls' development rights in 500 acres of Doughoregan. The contract consists of a number of written agreements between the parties.

15. The first agreement is a Commitment Letter dated May 18, 2010. A true and correct copy of the fully executed Commitment Letter is attached hereto and

incorporated as Exhibit C. The Commitment Letter contains the terms and conditions of the County's offer to purchase the Carrolls' development rights in their Agricultural Preservation Parcel, including the purchase price of \$19,100,000.00; an Installment Purchase Agreement concerning payment of the purchase price; and certain agreements and conditions for the land that is to be set forth in the Deed of Agricultural Land Preservation Easement (the "Deed of Easement"). A true and correct copy of the Deed of Easement is attached hereto and incorporated as Exhibit D.

16. On [month date], 2010, the parties entered into an Option Agreement. A true and correct copy of the fully executed Option Agreement is attached hereto and incorporated as Exhibit A. The Option Agreement grants the County the exclusive option to obtain the perpetual Deed of Easement in, over, and through the Agricultural Preservation Parcel described in the Commitment Letter, as amended by the parties' Development Rights and Responsibilities Agreement.

17. Contemporaneous with execution of the Option Agreement, the parties entered into a Development Rights and Responsibilities Agreement (the "DRRA"). A true and correct copy of the fully executed DRRA is attached hereto and incorporated as Exhibit E. The DRRA modifies the Commitment Letter by deleting paragraph 10. That paragraph had granted the Carrolls to right to be released from the Commitment Letter if the final documents for execution were not satisfactory to them in form and substance.

18. On [month day], 2010, the parties entered into the Installment Purchase Agreement. A true and correct copy of the fully executed Installment and Purchase Agreement is attached hereto and incorporated as Exhibit F.

19. All preconditions for the County's right to exercise its Option have occurred. Option Agreement § 1.2.

¹ 20. On [month day], 2010, the County exercised its Option by giving written notice to the Carrolls. Option Agreement § 1.2. A true and correct copy of the notice is attached hereto and incorporated as Exhibit G.

21. The exercise of the Option entitled and obligated the County to purchase the Deed of Easement from the Carrolls, and obligated the Carrolls to sell the Deed of Easement to the County. Option Agreement § 1.3.

22. Upon exercise of the Option, the parties were required to set a mutually agreeable date for settlement to occur within ten (10) business days of the exercise of the Option. Option Agreement § 2.1.

23. The Carrolls failed to agree to a settlement date and consummate settlement, despite the County's demand to do so on [month day], 2010. A true and correct copy of the demand is attached hereto and incorporated as Exhibit H.

24. At all relevant times, the County was ready, willing, and able to perform under the terms of the parties' contract and all preconditions for settlement were satisfied.

25. The Carrolls failed to schedule and attend settlement and deliver the Deed of Easement to the County and take the necessary action to satisfactorily perform under the parties' contract.

26. The Carrolls agree that there are no liquidated damages or other types of monetary damages that would be adequate to compensate the County for their failure to

perpetually conserve and protect the 500 acres \pm pursuant to the Act as promised. Option Agreement § 4.

27. The Carrolls agree that the County entered into the Commitment Letter, the Option Agreement, and the DRRA because of the Carrolls' promise to conserve and protect the $500 \pm acres$ pursuant to the Act. Option Agreement § 4.

COUNT I – SPECIFIC PERFORMANCE

28. The Plaintiff adopts by reference the allegations contained in paragraphs 1 through 27 of this Complaint with the same effect as if fully set forth.

29. The County and the Carrolls have a valid and enforceable contract for the purchase by the County of the Carrolls' development rights in $500 \pm \text{acres}$ of Doughoregan as set forth in the Deed of Easement.

30. The Carrolls breached the contract by failing to agree to settle and by not consummated settlement.

31. The County was ready, willing, and able to perform under the terms of the contract and all preconditions for settlement were satisfied.

32. The County has no adequate remedy at law.

WHEREFORE, Plaintiff Howard County, Maryland demand judgment that the contract between the parties be specifically enforced, and that the Defendants be ordered to execute and deliver the Deed of Easement to the County, and for such other relief as the Court deems proper.

VERIFICATION

I solemnly affirm under the penalties of perjury that the contents of the foregoing Verified Complaint for Specific Performance are true to the best of my knowledge, information, and belief.

HOWARD COUNTY, MARYLAND

Date: _____

By:

[name] Administrator Howard County Agricultural Land Preservation Program

Respectfully submitted,

HOWARD COUNTY OFFICE OF LAW

Margaret Ann Nolan County Solicitor

Gary W. Kuc Senior Assistant County Solicitor George Howard Building 3430 Courthouse Drive Ellicott City, Maryland 21043 Counsel for Plaintiff 410-313-2103 410-212-3292

Date: _____

Exhibit D Consent Decree for Specific Performance

HOWARD COUNTY, MARYLAND	*	IN THE
Plaintiff	*	CIRCUIT COURT
ν.	*	FOR
CAMILLA CARROLL, ET AL.	*	HOWARD COUNTY
Defendants	*	Case No.:
* * * * *	*	* * * * *

CONSENT DECREE FOR SPECIFIC PERFORMANCE

Plaintiff, Howard County, Maryland (the "County") and Defendants, Camilla Carroll and Philip D. Carroll (the "Carrolls"), hereby represent and acknowledge that they agree to enter into this Consent Decree (collectively, the "Parties").

1. The Carrolls own certain real property in Howard County, Maryland, described on Howard County Tax Map. No. 2523, Parcel 71 consisting of $892.6 \pm acres$ (the "Property"). The Carrolls intend to develop a 221.1 acre \pm portion of the Property with single family detached residential dwellings. The Howard County Agricultural Land Preservation Act is intended to protect the health, safety, and well-being of present and future residents of Howard County by conserving and protecting agricultural land as an environmental resource of major importance and as the basis of a viable sector of the County's economy. Howard County Code §§ 15.500 – 15.521 (the "Act").

2. The Parties entered into a contract for the purchase by the County of the Carrolls' development rights in 500 \pm acres of the Property pursuant to the Act. In exchange, the Carrolls agreed to execute and deliver to the County a Deed of Agricultural Land Preservation Easement regarding these rights (the "Deed of Easement").
3. Howard County filed a Verified Complaint for Specific Performance against the Carrolls (the "Complaint"). The County alleged that all preconditions for settlement had occurred and the County was ready, willing, and able to perform the contract, and that the Carrolls breached the contract by failing to agree to a settlement date and consummate settlement regarding the Deed of Easement. The County demands judgment that the contract between the Parties be specifically enforced, and that the Carrolls be ordered to execute and deliver to the County the Deed of Easement. Contemporaneously with the Complaint, the County filed a Consent Motion for Entry of Consent Decree for Specific Performance, with a copy of this order attached (the "Consent Motion").

4. Prior to the commencement of this action, the Carrolls consented to the County suing them for specific performance in the event of their breach of the contract, and they knowingly, intelligently, and voluntarily waived all rights, defenses, and claims, from whatever source derived, both procedural and substantive, that they may have to the action, except for service of process and filings in the action. *See* Option Agreement § 4, attached as Exhibit A to the Complaint. The record in this action indicates that the County served the Complaint and Writs of Summonses, as well as the Consent Motion, on the appointed and authorized agent of the Carrolls.

5. The Parties have reached agreement concerning resolution of this action and consent to entry of this Consent Decree for Specific Performance.

NOW, THEREFORE, IT IS HEREBY ORDERED, ADJUDGED AND DECREED AS FOLLOWS:

2

6. That the Consent Motion be and the same is hereby GRANTED;

7. That within ten (10) business days of the entry of this Consent Decree the Carrolls execute and deliver to the County the Deed of Easement;

8. That contemporaneously with the Carrolls' delivery of the Deed of Easement, the County pay the Carrolls in accordance with the terms and conditions of the Parties' contract.

9. That this Consent Decree for Specific Performance shall automatically terminate and this action shall be closed upon the County's recordation of the Deed of Easement in this Land Records of this Court, subject to payment of any open court costs.

IT IS SO DECREED this _____ day of _____, 20__:

Judge, Circuit Court for Howard County

IT IS SO AGREED AND CONSENTED TO:

Date

ł

CAMILLA CARROLL

IT IS SO AGREED AND CONSENTED TO:

Date

PHILIP D. CARROLL

IT IS SO AGREED AND CONSENTED TO:

Date

[NAME] Attorneys for the Carrolls

IT IS SO AGREED AND CONSENTED TO:

Date

[NAME] Administrator, Howard County Agricultural Land Preservation Program

IT IS SO AGREED AND CONSENTED TO:

Date

Margaret Ann Nolan, County Solictor Gary W. Kuc, Senior Assistant County Solicitor

Wastewater Treatment - Plan of Action

WASTEWATER TREATMENT -- PLAN OF ACTION

In order to implement the provisions of Section 3.2.D of this Agreement, the Carrolls will undertake a feasibility study which will analyze the alternatives for treating the wastewater flow from the development proposed for the Doughoregan Manor property.

The Carrolls will retain an environmental engineering consultant who is acceptable to the County and who has expertise and experience in analyzing sewer system flows and preparing engineering reports in conformance with the Howard County Volume II Design Manual, and evaluating wastewater flow and nutrient loading adjustment at the Little Patuxent Water Reclamation Plant in accordance with the laws and regulations of Howard County and Maryland Department of the Environment.

The consultant will draft a scope of work which will be reviewed and approved by the County. The study will include the following minimum components:

- Develop an estimate of wastewater flow for the project, per Volume II design standards, including an estimated nitrogen loading to the public sewer system. The estimate shall also include the projected development timeline.
- Evaluate options for minimizing wastewater flows and/or nutrient production for the project.
- Evaluate the establishment of a fund to be used by the Counties Bureau of Utilities to fund projects that will provide additional nutrient reduction to the Little Patuxent Water Reclamation Plant.
- Assist the County in identifying and obtaining offset nutrient credits to balance the addition of the 325 homes on the Site to the Metropolitan District.
- Prepare and submit both draft and final reports for review and approval by the County.

The goal of the consultant's evaluation is to minimize wastewater flow and nutrient concentration in the wastewater from this project. The alternatives shall address and consider the impact on County residents and the County with respect to odor, transportation of sludge, capital costs, operating costs, public health and environmental impacts.

Upon completion of the study and acceptance by the County, the Carrolls and the County will review the results with the Maryland Department of the Environment. The County shall determine the final wastewater system improvements necessary for the wastewater created by the site to best meet the goal of minimizing wastewater flow and nutrient concentration in wastewater directed to the Little Patuxent Wastewater Reclamation Plant.

Description of Agricultural Preservation Parcel

.

EXHIBIT 🕊

March 18, 2010

DESCRIPTION OF A 500.000 ACRE AGRICULTURAL PRESERVATION EASEMENT CONSISTING OF 241.638 ACRE PART ONE AND 258,362 ACRE PART TWO PART OF THE PROPERTY OF PHILIP D. CARROLL AND CAMILLA CARROLL SECOND AND THIRD ELECTION DISTRICTS HOWARD COUNTY, MARYLAND

BEING a parcel of land, situate and lying in the Second and Third Election Districts of Howard County, Maryland, said parcel of land being part of the land which by Will of Nina R. Carroll dated June 4, 1979, as amended by its First Codicil dated November 28, 1986 and its Second Codicil dated April 10, 1987; Nina R. Carroll having departed this life on February 11, 1989; said Will having been probated in the Surrogates Court of New York County, New York on April 11, 1989, and the Estate of Nina R. Carroll is filed in the Register of Wills of Howard County, Maryland as Estate No. 17-7868-276, said property also being described in a Mortgage dated October 31, 1962 and recorded among the Land Records of Howard County, Maryland in Liber 394 at Folio 64, and being part of Parcel 1 described therein; said parcel being more particularly described, as now surveyed in the Maryland State Coordinate System NAD' 83 Datum, as projected by Howard County Geodetic Control, as follows:

SHER, JLLINS

ARTER, C.

LAND SUBVEYORS

'2 Ballineers Hollonel Pite oli City, Maryland 21042) 461-2855 [410] 760-3784 (PART ONE) BEGINNING FOR THE SAME at a point on or near the intersection of the centerlines of existing paving of Folly Quarter Road with Frederick Road (Maryland Route 144); said point being at the beginning of the Seventeenth or South 82°36'44" East, 189,50 foot Deed Line of the aforesaid Liber 394 at Folio 64 (Parcel 1); thence running within said Frederick Road, on or near the centerline of paving, and running with and binding on all of the said Seventeenth line of said Liber 394 at Folio 64 (Parcel 1), and with all of the Eighteenth, Nineteenth and a portion of the First line of said Liber 394 at Folio 64 (Parcel 1), as now surveyed the following four (4) courses and distances:

- 1) South 82°54'28" East, 189.50 feet to a point; thence
- 2) North 17°12'33" West, 32.08 feet to a point; thence
- 3) South 82°35'58" East, 1,394.98 feet to a point; thence
- 4) South 73°02'02" East, 2.22 feet to a point on and being a distance of 3,137,87 feet from the end of the aforesaid First or South 73°02'48" East, 3140.09 foot Deed Line of the aforesaid Liber 394 at Folio 64 (Parcel 1) and also being the Northern margin of the sixty-six foot wide Right of Way of Frederick Road (Maryland Route 144); thence leaving said First line of Liber 394 at Folio 64 (Parcel 1) and the Northern margin of Frederick Road (Maryland Route 144) to run over, across and through the aforesaid Liber 394, Folio 64 the following thirteen (13) courses and distances:
- 5) South 06°25'26" West, 152.29 feet to a point; thence
- 6) South 04°05'30" West, 282.20 feet to a point of curvature; thence
- 253.27 feet along the arc of a non-tangential curve to the left, having a radius of 1,301.42 feet, a central angle of 11°09°01" and subtended by a chord bearing and distance of South 02°54'01" East, 252.87 feet to a point; thence
- 8) South 07°27'35" East, 191.10 feet to a point; thence
- 9) South 09°24'23" East, 392.33 feet to a point; thence
- 10) South 09°00'09" East, 293,76 feet to a point; thence

3HER,)LLINS

NRTER, C.

ENGUREERONE CONSULTANTS AND SURVEYORS

2 Balilicore Hallonal Pike itt City, Haryland 21042 481-2855 410) 750-3784

- 11) South 09°47'54" East, 393.83 feet to a point; thence
- 12) South 09°03'17" East; 615.16 feet to a point; thence
- 13) South 09°03'55" East, 291.12 feet to a point; thence
- 14) North 79°49'03" West, 1,832.40 feet to a point; thence
- 15) South 19°08'23" West, 1,237.51 feet to a point; thence
- 16) North 77°07'13" West, 68.06 feet to a point; thence
- 17) South 19°08'11" West, 1,248.08 feet to a point on and at a distance of 279.07 feet from a 1.25 inch diameter Iron Pipe found at the beginning of the Tenth or North 70°15'08" West, 1456.95 foot Deed Line of the aforesaid Liber 394 at Folio 64 (Parcel 1); thence running with and binding on the remaining portion of said Tenth line of Liber 394 at Folio 64 (Parcel 1), as now surveyed, the following course and distance:
- 18) North 70°10'26" West, 1,177.65 feet to a 1.25 inch diameter Iron Pipe found at the end thereof; thence running with and binding on the Eleventh, Twelfth, Thirteenth, Fourteenth, Fiftcenth and Sixteenth lines of said Liber 394 at Folio 64 (Parcel 1), the following six (6) courses and distances:
- 19) North 78°34'25" West, 791.22 feet to a point; thence
- 20) North 73°34'25" West, 387.55 feet to a point; thence
- 21) South 69°10'35" West, 31,37 feet to a point; thence
- 22) North 51°40'56" East, 568.09 feet to a point; thence
- 23) North 39°59'12" East, 2,458.50 feet to a point; thence
- North 26°28'43" East, 2,376.00 feet to the point of beginning, containing 241.638
 acres, more or less.

3HER,)llins

ARTER, C.

, ERBOLERIMI CONSULTANTS 'ARU SURVEYORS'

2 Ballingta Haliocal Pike Iti Bit, Maryland 21042 | 461-2855 | 410] 750-3784 (PART TWO) BEGINNING FOR THE SAME at a point on the Northern margin of the sixty-six foot Right of Way of Frederick Road (Maryland Route 144), said point being on and a distance of 53.08 feet from the beginning of the First or South 73°02'48" East, 3140.09 foot Deed Line of the aforesaid Liber 394 at Folio 64 (Parcel 1); thence running with and binding on a portion of said First line of Liber 394 at Folio 64 (Parcel 1), and the Northern margin of Fréderick Road (Maryland Route 144), the following course and distance:

- South 73°02'02" East, 1,379.85 feet to a point thereon; thence leaving the aforesaid First line of Liber 394 at Folio 64 and the Northern margin of Frederick Road (Maryland Route 144) and running for new lines of division, the following twenty-five (25) courses and distances;
- 2) South 16°57'58" West, 65.50 feet to a point; thence
- 3) South 73°02'02" East, 328.95 feet to a point; thence
- 4) South 19°32'01" East, 18.41 feet to a point; thence
- 5) South 73°08'00" East, 180.91 feet to a point; thence
- 6) South 27°46'18" East, 35.58 feet to a point; thence
- 10) South 17°35'24" West, 225.85 feet to a point of curvature; thence
- 346.71 feet along the arc of a curve to the left, having a radius of 315.00 feet, a central angle of 63°03'47" and subtended by a chord bearing and distance of South 13°56'30" East, 329.47 feet to a point of reverse curvature; thence
- 121.72 feet along the arc of a curve to the right, having a radius of 150.00 feet, a central angle of 46°29'32" and subtended by a chord bearing and distance of South 22°13'37" East, 118.40 feet to a point of tangency; thence

Sher, Jllins

ARTER, C.

. ENERGEBRICH GURSULDUNTS LAND SURVEYORS

2 Ballinore Netional Pike 38 City, Marytané 21062 3401-2855 410) 760-3764

- 10) South 01°01'09" West, 232.32 feet to a point of curvature; thence
- 312.10 feet along the arc of a curve to the left, having a radius of 330.00 feet, a central angle of 54°11'16" and subtended by a chord bearing and distance of South 26°04'29" East, 300,60 feet to a point of tangency; thence

12) South 53°10'07" East, 259.20 feet to a point of curvature; thence

- 470.43 feet along the arc of a non-tangential curve to the right, having a radius of
 595.00 feet, a central angle of 45°18'00" and subtended by a chord bearing and
 distance of South 30°31'07" East, 458.27 feet to a point of compound curvature;
 thence
- 14) 283.12 feet along the arc of a curve to the right, having a radius of 275.00 feet, a central angle of 58°59'14" and subtended by a chord bearing and distance of South 21°37'30" West, 270.78 feet to a point of reverse curvature; thence
- 219.63 feet along the arc of curve to the left, having a radius of 285.00 feet, a central angle of 44°09'14" and subtended by a chord bearing and distance of South 29°02'30" West, 214.24 feet to a point of tangency; thence
- 16) South 06°57'53" West, 62.64 feet to a point of curvature; thence
- 17) 89.85 feet along the arc of a curve to the left, having a radius of 285.00 feet, a central angle of 18°03'45" and subtended by a chord bearing and distance of South 02°03'59" East, 89.47 feet to a point of reverse curvature; thence
- 131.70 feet along the arc of a curve to the right, having a radius of 170.00 feet, a
 central angle of 44°23'10" and subtended by a chord bearing and distance of
 South 11°05'43" West, 128.43 feet to a point of reverse curvature; thence

Sher,)llins

NRTER, C.

EHECKEERCERI CORSOLYNITS NITI SURVEYORS

t Battlmore National Pixe It Gity, Maryland 21042 481-2855 210) 759-9784

- 19) 136.14 feet along the arc of a curve to the left, having a radius of 325.00 feet, a central angle of 24°00'04" and subtended by a chord bearing and distance of South 21°17'16" West, 135.15 feet to a point of tangency; thence
- 20) South 09°17'14" West, 113.42 feet to a point of curvature; thence
- 468.79 feet along the arc of a curve to the left, having a radius of 580.00 feet, a central angle of 46°18'34" and subtended by a chord bearing and distance of South 13°52'03" East, 456.13 feet to a point of tangency; thence
- 22) South 37°01'20" East, 903.13 feet to a point; thence
- 23) South 23°09'51" East, 507.53 feet to a point; thence
- 24) South 67°04'12" West, 462.42 feet to a point; thence
- 25) South 33°51'48" West, 651.81 feet to a point; thence
- 26) South 16°43'26" West, 858.48 feet to a point on and a distance of 1,764.49 feet from a Granite Stone found and held at the beginning of the Seventh or North 73°14'48" West, 4021.13 foot Deed Line of the aforesaid Liber 394 at Folio 64 (Parcel 1); thence running with and binding on the remaining portion of said Seventh line of Liber 394 at Folio 64 (Parcel 1) and with all of the Eighth and a portion of the Ninth or North 65°30'08" West, 2143.94 foot Deed Line of said Liber 394 at Folio 64 (Parcel 1), the following three (3) courses and distances:
- 27) North 73°16'34" West, 2,253.96 feet to a point; thence
- 28) North 14°04'34" East, 24.75 feet to a point; thence
- 29) North 65°25'29" West, 39.81 feet to a point; said point being on and a distance of 2,104.13 feet from a 1.25 inch Iron Pipe found at the end of the aforesaid Ninth line of Liber 394 at Folio 64 (Parcel 1); said point also being on the easterly

SHER, JLLINS

NRTER, C.

ENGOLEERING CONSULTANTS AKD SURVEYORS

2 Bahlunors Halloost Pixe in City, Neryland 21642 451-2855 410) 750-3784 outline of a plat entitled "Amended Plat Of Easement, Sending Parcel, Property Of Philip Carroll and Camilla Carroll" and recorded as Plat No. 19928; thence

- leaving the aforesaid Ninth line of Liber 394 at Folio 64 (Parcel 1) and running with and binding on part of said easterly outline of Plat No. 19928, the following course and distance:
- North 14°13'15" East, 1,478.90 feet to a point thereon; thence leaving the aforesaid easterly outline of Plat No. 19928 and running over, across and through the aforesaid Liber 394, Folio 64 (Parcel 1) the following twenty-two (22) courses and distances;
- 31) South 85°27'17" East, 421.22 feet to a point; thence
- 32) South 64°29'11" East, 192.86 feet to a point; thence
- 33) South 47°22'36" East, 333.45 feet to a point; thence
- 34) South 27°46'03" East, 102.62 feet to a point; thence
- 35) South 73°56'56" East, 224.55 feet to a point; thence
- 36) South 82°25'01" East, 149.17 feet to a point; thence
- 37) North 12°14'25" East, 389.04 feet to a point; thence
- 38) North 23°29'52" West, 478.60 feet to a point; thence
- 39) North 16°15'45" West, 750.96 feet to a point; thence
- 40) North 81°30'55" West, 730.33 feet to a point; thence
- 41) North 00°27'40" East, 207.41 feet to a point; thence
- 42) North 00°40'30" West, 76,57 feet to a point; thence
- 43) North 06°06'01" West, 117.35 feet to a point; thence
- 44) North 09°05'56" West, 310.97 feet to a point; thence

Sher, Dllins

ARTER, C.

ENGOLEERCHU GONSLATANTS And Surveydhs

2 Battimpto National Pixe in City, Maryland 21042 061-2855 010) 750-3784

- 45) North 09°03'18" West, 615.62 feet to a point; thence
- 46) North 09°47'54" West, 393.78 feet to a point; thence
- 47) North 09°00'07" West, 293.32 feet to a point; thence
- 48) North 09°24'23" West, 391.82 feet to a point; thence
- 49) North 07°27'35" West, 190.70 feet to a point of curvature; thence
- 50) 243.37 feet along the arc of a curve to the right, having a radius of 1,251.42 feet, a central angle of 11°08'33" and subtended by a chord bearing and distance of North 02°55'27" West, 242.98 feet to a point of tangency; thence
- 51) North 04°05'30" East, 280.56 feet to a point; thence
- 52) North 06°25'26" East, 141.96 feet to the point of beginning; containing 258.362 acres, more or less.

SUBJECT TO AND TOGETHER WITH any and all terms and conditions that might exist to the rights, use and maintenance of Folly Quarter Road and Frederick Road, Maryland Route 144, to Howard County, Maryland and the Maryland State Highway Administration.

SUBJECT TO AND TOGETHER WITH all conditions, covenants, easements, restrictions and rights-of-way of record, which may apply.

This description was prepared without the benefit of a Title Report.



Prepared By Checked By

WO #05022-3002

3HER,)LLINS

NRTER, C.

EURINEERING CONSULTAITS AND SURVEYINS

2 Battimere Hallona) Pika ili City, Maryland 21042 481-2855 410) 750-3784



1:\2005\05022\dwg\05022-3001 Parcel 1 Exhibit dwg, 3/18/2010 9:59:18 AM, dstreeft, 1:1



Exhibit 10

Description of Core

March 18, 2010

DESCRIPTION OF A 94.558 ACRE PARCEL PART OF THE PROPERTY OF PHILIP D. CARROLL AND CAMILLA CARROLL SECOND AND THIRD ELECTION DISTRICTS HOWARD COUNTY, MARYLAND

BEING a parcel of land, situate and lying in the Second and Third Election Districts of Howard County, Maryland, said parcel of land being part of the land which by Will of Nina R. Carroll dated June 4, 1979, as amended by its First Codicil dated November 28, 1986 and its Second Codicil dated April 10, 1987; Nina R. Carroll having departed this life on February 11, 1989; said Will having been probated in the Surrogates Court of New York County, New York on April 11, 1989, and the Estate of Nina R. Carroll is filed in the Register of Wills of Howard County, Maryland as Estate No. 17-7868-276, said property also being described in a Mortgage dated October 31, 1962 and recorded among the Land Records of Howard County, Maryland in Liber 394 at Folio 64, and being part of Parcel 1 described in the aforesaid Liber 394 at Folio 64; said parcel being more particularly described, as now surveyed in the Maryland State Coordinate System NAD' 83 Datum, as projected by Howard County Geodetic Control, as follows;

Sher, Jllins

ARTER, C. ENGLIGEERING CONSULTANTS AND SURVEYORS

2 Ballimore Kellonal Pike ti City, Maryland 21042 461-2865 \$10) 760-3784 BEGINNING FOR THE SAME at a point on the Northern margin of the sixty-six foot wide Right of Way of Frederick Road (Maryland Route 144), formerly known as the Baltimore to Frederick Turnpike Road, said point being on and a distance of 2.22 feet from the beginning of the First or South 73°02'48" East, 3140.09 foot Deed Line of the aforesaid Liber 394 at Folio 64 (Parcel 1); thence running with and binding on a portion of said First line of Liber 394 at Folio 64 (Parcel 1) and the aforesaid Northern margin of Frederick Road (Maryland Route 144) the following course and distance:

- South 73°02'02" East, 50.86 feet to a point thereon, said point being a distance of 3,087.01 feet from the end thereof; thence leaving the aforesaid First line of Liber 394 at Folio 64 (Parcel 1) and the aforesaid Northern margin of Frederick Road (Maryland Route 144) and running over, across and through the aforesaid Liber 394, Folio 64 the following twenty-two (22) courses and distances:
- 2) South 06°25'26" West, 141.96 feet to a point; thence
- 3) South 04°05'30" West, 280.56 feet to a point of curvature; thence
- 243.37 feet along the arc of a curve to the left, having a radius of 1,251.42 feet, a central angle of 11°08'33" and subtended by a chord bearing and distance of South 02°55'27" East, 242.98 feet to a point of tangency; thence
- 5) South 07°27'35" East, 190.70 feet to a point; thence
- 6) South 09°24'23" East, 391.82 feet to a point; thence
- 7) South 09°00'07" East, 293.32 feet to a point; thence
- 8) South 09°47'54" East, 393.78 feet to a point; thence
- 9) South 09°03'18" East, 615.62 feet to a point; thence
- 10) South 09°05'56" East, 310.97 feet to a point; thence
- 11) South 06°06'01" East, 117.35 feet to a point; thence
- 12) South 00°40'30" East, 76.57 feet to a point; thence
- 13) South 00°27'40" West, 207.41 feet to a point; thence
- 14) South 81°30'55" East, 730.33 feet to a point; thence
- 15) South 16°15'45" East, 750.96 feet to a point; thence

HER, LLINS

RTER,

VEINTERDITI CONSULTANTS VID SURVEYORS

Jahlimora Hallonal Piko City, Maryland 21017 81-2255 0) 750-3784

- 16) South 23°29'52" East, 478.60 feet to a point; thence
- 17) South 12°14'25" West, 389.04 feet to a point; thence
- 18) North 82°25'01" West, 149.17 feet to a point; thence
- 19) North 73°56'56" West, 224.55 feet to a point; thence
- 20) North 27°46'03" West, 102.62 feet to a point; thence
- 21) North 47°22'36" West, 333.45 feet to a point; thence
- 22) North 64°29'11" West, 192.86 feet to a point; thence
- 23) North 85°27'17" West, 421.22 feet to a point on the easterly outline of a plat entitled "Amended Plat Of Basement, Sending Parcel, Property Of Philip Carroll and Camilla Carroll" and recorded among the Land Records of Howard County, Maryland as Plat No. 19928; thence running with and binding on the easterly and northerly outlines of said plat, the following three (3) courses and distances;
- 24) North 14°13'15" East, 205.25 feet to a point; thence
- 25) North 76°10'22" West, 1,044.43 feet to a point; thence
- 26) North 77°07'13" West, 1,128.53 feet to a point at the northwesterly corner of the aforesaid Plat No. 19928; thence leaving the outline of said Plat No. 19928 and continuing to run over, across and through the aforesaid Liber 394, Folio 64, the following eleven (11) courses and distances;
- 27) North 19°08'23" East, 1,237.51 feet to a point; thence
- 28) South 79°49'03" East, 1,832.40 feet to a point; thence
- 29) North 09°03'55" West, 291.12 feet to a point; thence
- 30) North 09°03'18" West, 615.16 feet to a point; thence
- 31) North 09°47'53" West, 393.83 feet to a point; thence

SHER, SLLINS

RTER, D. EXELICTERINA CONSULTANTS VICE SUMYEYORS

Bahimoro Hallonaj Piro † 5Hy, Niaryland 21042 481-2855 10] 750-3784

- 32) North 09°00'09" West, 293.76 feet to a point; thence
- 33) North 09°24'23" West, 392.33 feet to a point; thence
- 34) North 07°27'35" West, 191.10 feet to a point of curvature; thence
- 35) 253.27 feet along the arc of a curve to the right, having a radius of 1,301.42 feet, a central angle of 11°09'01" and subtended by a chord bearing and distance of North 02°54'01" West, 252.87 feet to a point of tangency; thence
- 36) North 04°05'30" East, 282.20 feet to a point; thence
- North 06°25'26" East, 152.29 feet to the point of beginning; containing 94.558
 acres of land, more or less.

SUBJECT TO AND TOGETHER WITH any and all terms and conditions that might exist to the rights, use and maintenance of Folly Quarter Road and Frederick Road, Maryland Route 144, to Howard County, Maryland and the Maryland State Highway Administration.

SUBJECT TO AND TOGETHER WITH all conditions, covenants, easements, restrictions and rights-of-way of record, or proposed, which may apply.

This description was prepared without the benefit of a Title Report.



Prepared By Checked B

WO #05022-3002

SHER, SLLINS

NRTER, C. ENRICLERIONE CONSTRATTS

2 Bakimose Hallons) Pike Iti City, Maryland 21042 461-2855 410) 750-3784

AND SURVEYORS



Amendment 1 to Council Resolution No. 103-2010

BY: Chairperson at the request of the County Executive and co-sponsored by Courtney Watson, Calvin Ball, Greg Fox, Mary Kay Sigaty and Jennifer Terrasa Legislative Day No. 9 Date: July 29, 2010

Amendment No.

(The amendment:

- 1. Corrects a tax map reference;
- 2. Inserts the date of the public hearing before the County Council;
- 3. Inserts the date of Council approval;
- 4. Inserts the Resolution number;
- 5. Corrects a reference to the date of the Commitment Letter;
- 6. Substitutes the word "Property" for the word "Site" in certain instances;
- 7. Corrects certain language related to amendments; and
- 8. Makes other similar technical corrections.)
- 1 Strike "25" and substitute "23" in the following instances:
- 2 1. In the Resolution, on page 1 in line 18;
- 3 2. In the Development Rights and Responsibilities Agreement (the "DRRA"),
- 4 attached to the Resolution as Exhibit A:
- 5 a. On page 1 in Recital number 4; and
- 6 b. On page 4 in the definition of "Property";
- 7 3. In the Option Agreement, attached to the DRRA as Exhibit 7, on page 1 in the
 8 first sentence of Recital number 1;
- 9 4. In the Complaint for Specific Performance, attached to the Option Agreement as
- 10 Exhibit C, on page 2 in number 4 under "Parties"; and
- 11 5. In the Consent Decree for Specific Performance, attached to the Option

12 Agreement as Exhibit D, on page 1 in item number 1.

13

15

1.

14 In the DRRA, attached to the Resolution as Exhibit A:

On page 2, in item number 11:

ABOPTED J	ly 29, 2010
FAILED	
SIGNATURE S	Jestim MJ. Yenke

1

16		a. In the first blank, insert "July 19,";	
17		b. In the second blank, insert "July 29, 2010"; and	
18		c. In the third blank, insert "No. 103-2010";	
19	2.	On page 3, in the definition of "Commitment Letter", strike "April 14" and	
20		substitute " <u>May 18</u> ";	
21	3.	In page 11, in Section 6.1, in the second sentence, strike "agrees" and substitute	
22		"agree";	
23	4.	On page 14, in Section 9.2, strike "8.5" and substitute "8.4";	
24	5.	On page 15, in Section 9.4, in the first sentence, strike "terminate" and substitute	
25		"amend";	
26	6.	On page 15, in Section 9.4, strike the last sentence and substitute: "Unless the	
27		Planning Board determines that the proposed amendment is consistent with the	
28		Howard County General Plan, the Parties may not amend this Agreement."; and	
29	7.	Strike "Site" and substitute "Property" in the following instances:	
30		a. On page 11:	
31		i. In the first sentence of Section 6.1;	
32		ii. In Section 6.1B; and	
33		iii. In Section 6.1C; and	
34		b. On page 14, in the first sentence of Section 9.2A;	
35		c. On page 16, in Section 9.6(2); and	
36		d. On page 17, in Section 9.15.	
37			

.

.

•

•

2

.

Amendment 2 to Council Resolution No. 103-2010

By: Courtney Watson and Greg Fox

1

Legislative Day 9 Date: July 29, 2010

Amendment 2

(This amendment would add a provision to the DRRA requiring the Carrolls to record a perpetual covenant at a specified time to permit only the construction of 325 single-family detached dwelling units on the Site, and would require that the covenant to prevent further subdivision of the "Core" be recorded within a specified period of time.)

1	In the Developer Rights and Responsibilities Agreement attached to the Resolution as Exhibit A,	
2	make the following changes:	
3		
4	On page 5, in Section 2.3A, after "Regulations." insert the following:	
5	"THE CARROLLS SHALL EXECUTE AND RECORD IN THE LAND RECORDS OF HOWARD COUNTY	
6	A PERPETUAL COVENANT TO PROHIBIT THE CONSTRUCTION OF SINGLE-FAMILY ATTACHED	
7	DWELLING UNITS OR APARTMENT UNITS ON THE SITE, AND TO PROHIBIT THE CONSTRUCTION	
8	of more than 325 single-family detached units on the site. This covenant shall	
9	BE MADE BETWEEN THE CARROLLS AND EITHER HOWARD COUNTY, A HOMEOWNER'S	
10	ASSOCIATION OF THE SITE, OR OTHER COMMUNITY ASSOCIATION, AT THE DISCRETION OF	
11	Howard County, within 90 days after Howard County exercises the option	
12	UNDER SECTION 1.2 OF THE OPTION AGREEMENT, UNLESS SUCH PERIOD IS EXTENDED BY	
13	HOWARD COUNTY."	
14		
15	On page 9, in Section 4.2, in the next to last line, after "covenant." insert the following:	
16	"This covenant shall be made within 90 days after Howard County's exercise of	
17	THE OPTION UNDER SECTION 1.2 OF THE OPTION AGREEMENT, UNLESS SUCH PERIOD IS	
18	EXTENDED BY HOWARD COUNTY."	
1 9	EXTENDED BY HOWARD COUNTY."	
	SIGNATURE STaghen M. Ledende	

Amendment <u>3</u> to Council Resolution No. 103-2010

BY: Greg Fox

Legislative Day No: _____

Co-sponsored by Courtney Watson

Date: July 29, 2010

Amendment No. 3

(This amendment would clarify the intent of Section 2.4 of the DRRA that the Carrolls are required to completely comply with APFO requirements in the development of the Site.)

1 In the Developer Rights and Responsibilities Agreement attached to the Resolution as Exhibit

2 A, make the following changes:

3

- 4 On page 5, in Section 2.4, strike "Project." and substitute "PROJECT, FOR ALL ASPECTS OF THE
- 5 DEVELOPMENT OF THE SITE, IN ALL OF ITS PHASES, FOR ALL APFO TESTS."

ADOPTED 2	429 2010
FAILED _	100
SIGNATURE	Itenten interendre

Amendment to __4_ Council Resolution No. 103-2010

BY: Courtney Watson

··· • ··

Legislative Day No: 9 Date: July 29, 2010

Amendment _4__

(This amendment makes technical corrections to section 3.1B2. of the DRRA.)

1 On page 7 of Exhibit A, delete after the word "Lot" through "other", insert "to an entity 2 <u>or</u>"; after "identified by" insert "<u>Howard County</u>" and strike "the CRLCA".

PHILIP AND CAMILLA CARROLL – PRESUBMISSION COMMUNITY MEETING (Howard County Code Sec. 16.128(b)) 3500 Manor Lane, Ellicott City, Maryland 21042 Tax Map 23, Grid 10, Parcel 71 Approx. 655.498 AC±

REPORT OF COMMUNITY INPUT MEETING HELD AT 6 P.M. ON TUESDAY, MAY 23, 2019 AT MILLER BRANCH LIBRARY, 9421 FREDERICK ROAD, ELLICOTT CITY, MARYLAND 21042

The following people were in attendance at that meeting:

Sang W. Oh (attorney), Joseph Rutter (consultant), Jeremy Rutter (consultant).

After a brief presentation by Petitioner of its request for a 5 year extension of the DRRA pertaining to Doughoregan/Westmount, the following questions and answers were provided:

- 1) I called Justin Tyler, planner of the day, at DPZ and he told me that there will be 800 new units? This is false. 325 total units. Nothing is changing in the DRRA other than the term.
- 2) What about the land that is to be donated to Kiwanis Wallis Park? Done
- 3) Why doesn't HCYP know about the donation? The donation was to recreation and parks, but HCYP should know.
- 4) Will the representatives for the property owners please identify themselves?
- 5) So the letter that some neighbors received referenced 655.49 acres. That doesn't match up to any of the acreages referenced in the DRRA. Can you explain? The acreages under the DRRA was explained.
- 6) Does your map show Manor Lane? No.
- 7) Will Manor Lane be used to divert traffic to Rt. 144? No.
- 8) The DRRA provides that all access including emergency must be approved. Has that been done? Yes.
- 9) Don't there need to be two access points? No. The ingress/egress to Westmount was approved with a dual lane
- 10) Have you done a traffic study? What are the numbers? This is not the forum to talk about that.
- 11) Do the Carrolls have any intention to build more houses after these remaining homes will be constructed? No except the fact that the Carrolls have 10 residual development rights.
- 12) Can you explain the status of the construction of the phases? The plans that have been submitted and approved for the project were explained.
- 13) Once you have the approval for the final phases, can you still modify the plan? *Unlikely*.
- 14) I just want to be clear that this is just a reaffirmation of the original plan with no changes other than the term? *Correct*

- 15) Are there going to be any traffic changes made to the intersection of Kiwanis Park? No.
- 16) So you need another 5 years? Yes
- 17) When will the last house be built? In about 5 years
- 18) All single family homes? Yes
- 19) How many kids per home? Difficult to predict.
- 20) Are all the kids going to be redistricted to the same school? That is a Board of *Education issue*.
- 21) What are the schools that districted to attend? Discussed
- 22) Will you strictly adhere to the terms of the DRRA in so far as your accessing Burnside Drive for construction? Yes, we will.
- 23) I'm disappointed that we don't have a powerpoint. The purpose of this meeting is about a DRRA. If you leave your email, we will provide you with a pdf of the exhibit that we have been showing.
- 24) What are the Carrolls doing for us? The terms of the DRRA contain all of the terms.
- 25) Would the Carrolls allow public access to the Doughoregan? No.
- 26) Is this going to be a rubber stamp for the County Council? A discussion ensued about the DRRA process.
- 27) What are the implications if the DRRA is not extended? Difficult to say. We would have to really examine the consequences.
- 28) If you don't get the extension, then does the whole thing get renegotiated? Unknown
- 29) The challenge is that during the past 10 years, our schools have gotten increasingly crowded. We ask you to put your thinking caps on to come up with solutions or we'll keep showing up. You have the ability to solve this. You have the deep pockets to solve this. *Do not agree*,
- 30) Would this be a taking to reduce the number form 325? Difficult to say.
- 31) All of the school tests have been passed? Yes
- 32) Are you going to inform your buyers that schools are over-crowded and that there is no room for the new kids? No. We passed the schools test and no phase of this development has had to wait 5 years for APFO. We're talking about 200 units over the next 6 years.
- 33) I personally don't have a problem with the Carroll family or the extension, but I am opposed to the amount of development that the County allows.
- 34) We don't have the money to buy the Turf Valley ES and have no money to build the school.
- 35) Can the 500 acres that the County purchased an easement on be rezoned? It can, but it can't be developed.
- 36) I'd like to ask about some legal aspects of the DRRA. What is the purpose of the 176 pages in the exhibits? *Title Report*. What is Exhibit 4? *Title Opinion* What is the purpose and why was it necessary to include all the State and probate records to show who owned what? *Just to be inclusive with information*. What is the purpose and why was it necessary to include ROW and easement? What is the purpose and why was it necessary to include language in Article 6, subsection C for the exception?

- 37) When are you going to submit the application? We have already filed the application. It is likely to go before the County Council not before September 2019. CORRECTION: THE APPLICATION HAS NOT YET BEEN FILED. LIKLEY TO BE FILED IN THE NEAR FUTURE. WE STILL ANTICIPATE THAT THE DRRA WILL BE CONSIDERED BY THE COUNTY COUNCIL THIS FALL OR WINTER.
- 38) Is there anyone from the County here? How will the County receive a summary of the comments/questions from this meeting? We will do our best to summarize, but if you have any additions/corrections, please forward to me.
- 39) You said you will not connect with Burnside Drive. Did you say that you could connect to Chateau Ridge? With the DRRA, we cannot.
- 40) What rights and responsibilities remain to be completed? Believe most of the responsibilities have been performed.
- 41) Does phase 4 have to be completed by 2020?
- 42) And no other terms are being changed? That's right.
- 43) How do we get a copy of the traffic report? It should be in the County's file.
- 44) Fire truck access plans have been approved by the County and State? Yes
- 45) Will you send a copy of the plan when you send the minutes? Sure
- 46) This meeting is done in accordance with the presubmission meeting rules. A reference is in error. What rules are you using? *Trying to fulfill the intent of those rules*.
- 47) So what happens to our comments? It will go to the administration and the County Council.
- 48) The studies are just archaic. 10 years ago. Traffic tests compound traffic over a number of years.
- 49) To summarize: extend 5 years; no access to Burnside; nothing purchased on Chateau Ridge to create a new access? *Correct*
- 50) Since nothing is being changed, why do we have to have this meeting? We're following the law.
- 51) When did you realize you would need an extension? Last year.
- 52) What is the financial impact to the Carrolls if the next phases don't happen? No answer
- 53) Will you re-do the traffic because there have been a couple of serious accidents.
- 54) People will want to cut through Chateau Ridge via Centennial to avoid Rt. 29.
- 55) This process is strange because although you're not seeking to change any of the real terms of the DRRA, why do we still have to have this meeting?
- 56) So we can open a whole can of worms if we want to?
- 57) The development rights are held by the County? No. County cannot sell these units.

The meeting concluded at approximately 7:10 p.m.



HOWARD COUNTY DEPARTMENT OF PLANNING AND ZONING 3430 Court House Drive Ellicott City, Maryland 21043 410-313-2350

Voice/Relay

FAX 410-313-3467

Amy Gowan, Director

TECHNICAL STAFF REPORT June 18, 2020

Planning Board Meeting of July 2, 2020 County Council Hearing to be scheduled

Case No./Petitioner:	At the request of Camilla Carroll and Philip D. Carroll
Request:	Petition to enter into a First Amendment to the Doughoregan Development Rights and Responsibilities Agreement (DRRA) to extend the term by a period of five (5) years. Advisory determination by the Howard County Planning Board regarding General Plan, <i>PlanHoward2030</i> , consistency for the amending previously executed agreements pursuant to Section 16.1701(b)(2) of the Howard County Code and Md. Land Use Code Ann. § 7-305.
Location:	Portion of Historic Doughoregan Manor at 3500 Manor Lane Ellicott City, Howard County, Maryland Tax Map 23, Grid 10, Originally Part of Parcel 71
Area of Site:	892 +/- ac
Existing Zoning:	R-ED: Residential Environmental Development
Proposed Zoning:	N/A
I. BACKGROUND	

The Petitioners desire to extend the terms of the Development Rights and Responsibilities Agreement between the County and Camilla and Philip Carroll for a portion of the Historic Doughoregan Manor property at 3500 Manor Lane (See Doughoregan DRRA as Attachment A) by a period of five (5) years. The Petitioners and have submitted a First Amendment (Attachment B) requesting an extension to the original DRRA, which was signed and executed on September 23, 2010. This purpose of the amendment is to extend the term of the current DRRA by a period of five years to September 23, 2025. No other changes are proposed to be made to the current DRRA.

Doughoregan Manor was the country home of Charles Carroll III of Carrollton, a signer of the Declaration of Independence for Maryland. Charles Carroll, planter, landowner, politician and US Senator, was the only Roman Catholic signer of the Declaration and its last surviving signer. Charles Carroll III is buried in the family chapel attached to the north end of the Manor. Doughoregan Manor is the only home of a signer of the Declaration of Independence that is still in family ownership.

DRRA Legislative Process

- CB 9-2010: Adopted April 5, 2010 General Plan 2000 amended to adjust the Planned Service Area boundary for water and sewer service to include 221.1 acres of the Doughoregan property. (see Technical Staff Report – Attachment C)
- Land Preservation Commitment Letter: May 18, 2010
 Signed a commitment letter for the sale and purchase of development rights under the Agricultural Land Preservation Program (ALPP) for 500 acres of the Doughoregan property.
- Planning Board DRRA Recommendation: June 17, 2010
 Determined the DRRA was consistent with the General Plan. (see Technical Staff Report- Attachment D)
- CB 32-2010: Adopted July 6, 2010 Authorized an Installment Purchase Agreement (IPA) to acquire a permanent Agricultural Land Preservation Easement on 500 acres of the Doughoregan property.
- CR 103-2010: Adopted July 29, 2010 Authorization to execute a DRRA.
- ZB 1087M: D&O Sept. 13, 2010 Authorized an amendment to the zoning map reclassifying 221.1 acres of the Doughoregan property from RC-DEO to R-ED.
- Signed DRRA: Executed Sept. 23, 2010

II. PROPOSAL

The Petitioner proposes to amend the Doughoregan DRRA, which expires on September 23, 2020 for an additional 5 years. This extension allows the current agreement to remain in place while the Petitioner completes the final phase of development. The extension does not grant any development rights, as the development can proceed regardless of an amended DRRA. The extension allows the terms of the agreement to apply during final development stages, one such term is a prohibition on connecting to Burnside Drive.

Pursuant to Section 16.1701 Procedures (b)(2) of the Howard County Code " An amendment to an agreement may be executed by the County Executive only after: ... (1) A pre-submission community meeting conducted using the same procedures established in subsections 16.128(b)—(f) of the Subdivision and Land Development Regulations... and (2) a public meeting before the Planning Board and an advisory determination by the Planning Board that the proposed amendment to the agreement is consistent with the plan."

A Presubmission Community Meeting on this Petition was held on May 23, 2019 according to Section 16.128 of the Howard County Code.

III. EVALUATION

The Technical Staff Report for the Planning Board advisory recommendation on the original DRRA (Attachment D) dated June 3, 2010 provided the following evaluation,

"The proposed Development Rights and Responsibilities Agreement commits the parties to the multi-part strategy for the subject property. GPA 2010-1 and ZB 1087M, which are the only two items subject to the Planning Board's review were both recommended for approval by the Planning Board on February 18, 2010. At that time, the Planning Board found GPA 2010-1 to

DocuSign Envelope ID: F36EB5C8-4F5D-4ED0-8A45-C7606B975CE4

be consistent with General Plan 2000. The County Council concurred in adopting CB 9-2010 to amend the PSA boundary as shown in Exhibit B. Accordingly, the DRRA which includes this PSA boundary amendment is consistent with General Plan 2000."

The DRRA was determined to be consistent with the 2000 General Plan, and those determinations carried over to *PlanHoward 2030*, as the portion of the property to be developed (known as Westmount Subdivision) was included in the PSA and the land use and place type designations did not change. Extending this DRRA remains consistent with the County's General Plan Place Type Map shown below.



5 (4089)

Additionally, it also comports with the following PlanHoward 2030 policies:

Chapter 4: Resource Conservation

Policy 4.1 – Promote additional agricultural preservation opportunities.

b. Other Preservation Options. Promote other preservation options such as the dedication of easements to the County through the subdivision process, the purchase of easements by the MALPF program, and the donation of easements to nonprofit land trusts.

Policy 4.10 – Expand on existing programs to enhance historic preservation and create an historic preservation plan.

Chapter 6: Growth

Policy 6.1- Maintain adequate facilities and services to accommodate growth.

a. Limit Planned Service Area Expansion. Zoning requirements for approved PSA expansion should include a development proposal that is consistent with the General Plan and established a transition that is compatible with and enhances surrounding communities and provides an environmental benefit.

DocuSign Envelope ID: F36EB5C8-4F5D-4ED0-8A45-C7606B975CE4

Policy 6.2 – Ensure that the County's needs for land for government facilities and land preservation are met in light of competing needs for housing and economic development.

Chapter 8: Public Facilities and Services

Policy 8.12 – Expand the County park system and recreational facilities.

Chapter 10: Community Design

Policy 10.1 - Protect and enhance established communities through compatible infill, sustainability improvements, and strategic public investments.

III. RECOMMENDATION

The Department of Planning and Zoning recommends a finding of General Plan consistency between the proposed amended Development Rights and Responsibilities Agreement and *PlanHoward 2030*.

— DocuSigned by: Amy Glonan

6/18/2020

Amy Gowan, Director

Date

Attachment A - Existing DRRA for Doughoregan

000281

DEVELOPMENT RIGHTS AND RESPONSIBILITIES AGREEMENT

THIS DEVELOPMENT RIGHTS AND RESPONSIBILITIES AGREEMENT (this "Agreement"), made as of the <u>23rd</u> day of <u>September</u>2010, by and between CAMILLA CARROLL and PHILIP D. CARROLL, individuals, (collectively referred to as the "Carrolls") and HOWARD COUNTY, MARYLAND, a body corporate and politic of the State of Maryland ("Howard County"). The Carrolls and Howard County are hereinafter referred to collectively as the "Parties".

RECITALS

INF FO SINE & CO.60 NECORDING FEE 73.00 IOTAL 53.00 Rest HJUDE ROFT & ACLOS NOR CON BIX & JOCO

Mcc

1. Maryland law, Article 66B, § 13.01 of the Maryland Amndiated Code 11 10 16 ("Article 66B"), grants Howard County the authority to establish procedures and requirements for the consideration and execution of Development Rights and Responsibilities Agreements.

2. Howard County Council adopted Council Bill No. 4-2010 on March 1, 2010, effective May 4, 2010, creating Sections 16.1700 *et seq.* of the Howard County Code authorizing Howard County to enter into Development Rights and Responsibilities Agreements ("County Ordinance").

3. This Agreement is intended to constitute a Development Rights and Responsibilities Agreement as provided for in Article 66B and the County Ordinance.

4. The Carrolls own certain real property in Howard County, Maryland, described on Howard County Tax Map No. 23, Parcel 71 consisting of 892.6 AC \pm as shown and described on the attached **EXHIBIT 1** hereto and made part hereof ("Doughoregan" or "Property").

5. The Carrolls intend to develop a 221.1 AC \pm portion of Doughoregan as depicted on the attached **EXHIBIT 2** (the "Site") with not more than 325 single family detached residential dwelling units substantially in the manner as set forth in **EXHIBIT 2** (the "Project"). A legal description of the Site is attached hereto as **EXHIBIT 3**.

6. The names of all parties having an equitable or legal interest in Doughoregan, including lien holders, are set forth in the title opinion letter, attached hereto as **EXHIBIT 4**, and made a part hereof.

7. On or about March 19, 2010, the Carrolls petitioned Howard County to enter into this Agreement.

8. On or about April 1, 2010, Howard County reviewed this petition and determined to accept this petition and initiate the process of considering a Development

Rights and Responsibilities Agreement.

9. This Agreement was negotiated between the Carrolls and the Howard County Executive.

10. This Agreement was referred to the Howard County Planning Board for determination of whether this Agreement is consistent with the 2000 General Plan as amended. At a public meeting held on <u>June 17</u> 2010, the Howard County Planning Board determined that this Agreement was consistent with the General Plan. The determination by the Howard County Planning Board is attached hereto as **EXHIBIT 5**.

11. On July 19, 2010, the Howard County Council held a duly advertised public hearing on this Agreement in accordance with Howard County law, and approved this Agreement on July 29, 2010 by Council Resolution No. 103-2010.

12. Any other agreements between the Carrolls and Howard County, including a signed commitment letter dated May 18, 2010 for the sale of the development rights to 500 acres of the Property under the Howard County Agricultural Land Preservation Program (the "Commitment Letter"), a copy of which is attached hereto as **EXHIBIT 6**, an Option Agreement for Agricultural Preservation Easement for the portion of the Property in the Commitment Letter (the "Option Agreement"), a copy of which is attached hereto as **EXHIBIT 7**, remains in full force and effect and is intended to be harmonious with this Agreement.

NOW, THEREFORE, in consideration of the foregoing recitals, which are not merely prefatory but are hereby incorporated into and made a part of this Agreement, and the mutual covenants and agreements as set forth below, and for other good and valuable consideration, the receipt and sufficiency of which the Parties hereby acknowledge, the Carrolls and Howard County hereby agree as follows:

ARTICLE I

DEFINITIONS

1.1 <u>Definitions</u>. Unless otherwise defined in this Agreement, the following words, when used in this Agreement, shall have the following meanings:

A. "Agreement" means this Development Rights and Responsibilities Agreement.

B. "APFO Approval" means approval by Howard County Department of Planning and Zoning of applicable Adequate Public Facilities Ordinance tests for the Project or any portion thereof in accordance with the provisions of Section

MCC CC
C. "Howard County" means Howard County, Maryland, a body corporate and politic.

D. "Carrolls" means Camilla Carroll and/or Phillip D. Carroll and/or their respective agents, purchaser(s), heir(s), assignee(s) or successor(s) in interest.

E. "Commitment Letter" means the executed letter dated May 18, 2010 for the Carrolls' sale of the development rights on 500 acres of the Property in Howard County pursuant to the Agricultural Preservation Act. Upon the execution of this Agreement and the Option Agreement, notwithstanding paragraph 10 of the Commitment Letter, the obligations of the Parties shall be governed by this Agreement and the Option Agreement and binding subject to the provisions of these agreements.

F. "County Code" or "Howard County Code" means the Code of Howard County, Maryland and any other laws of Howard County including its regulations, resolutions, the Howard County Design Manual, master plan(s), design criteria or any other provision having the force and effect of law which are in effect on the Effective Date of this Agreement.

G. "County Ordinance" means Howard County Council Bill No. 4-2010 creating Sections 16.1700 *et seq.* of the Howard County Code and authorizing Development Rights and Responsibilities Agreements.

H. "Development Approval" means that final governmental approval of Subdivision Plat(s), Site Development Plans, Adequate Public Facilities Ordinance ("APFO") requirements, sewer and/or water approvals, as applicable, has been obtained for the Project or each respective portion and phase of the Project, and that all conditions of said approval have been satisfied with all applicable appeal period(s) having expired without the filing of any appeal, or, if an appeal(s) was filed, a final unappealable decision.

I. "DPZ" means the Howard County Department of Planning and Zoning.

J. "Effective Date of this Agreement" means the date the last party executes this Agreement.

K. "Improvements" means those improvements to be made to roads, sewer, water, storm water, and other facilities necessary to service the Project.

L. "Land Records" means the land records of Howard County,

Maryland.

M. "Parkland" means the parcel which shall contain a total of thirtysix (36) acres of land, more or less, be adjacent to Kiwanis-Wallis Park and shall substantially conform to the area indicated on EXHIBIT 2, and shall be conveyed to the County by the Carrolls as set forth in Section 3.4 hereof.

N. "Phase" means any portion of the Project that has received Development Approval.

O. "Planning Board" means the Planning Board for Howard County Maryland.

P. "Property" means the Carroll's fee simple real property in Howard County, Maryland, shown on Howard County Tax Map No. 23 as Parcel 71 consisting of 892.6 AC \pm as shown and described on the attached EXHIBIT 1 hereto and made part hereof.

Q. "Site" means the 221.1 AC \pm portion of the Property as depicted on the attached EXHIBIT 2 (the "Site") which shall be developed with not more than 325 single family detached residential dwelling units substantially in the manner as set forth in EXHIBIT 2 (the "Project"). A legal description of the Site is attached hereto as EXHIBIT 3.

R. "Subdivision Plat" means a final plat(s) of subdivision for the Project or any phase or portion thereof, prepared in accordance with the County Code and approved by DPZ and/or the Planning Board.

S. "Subdivision Regulations" means the Title 16, Subtitle 1 of the Howard County Code in effect on the Effective Date of this Agreement.

T. "Zoning Regulations" or "HCZR" means the Howard County Zoning Regulations in effect on the Effective Date of this Agreement.

ARTICLE II ZONING, LIMITATIONS, PLAN APPROVALS AND PLAN CONSISTENCY

2.1 <u>Zoning and Plan Designations</u>. The Site is zoned R-ED on the Howard County Zoning Map. The Site is a) located within the Planned Service Area for Water and Sewer as contained in the 2000 General Plan, as amended; and (b) designated as

LIDER 12722 FOLIO 252

"Residential Areas" on Map 7-1 of the General Plan Policies Map 2000-2020, as amended. In the event that the zoning for the Site is no longer R-ED (or the subsequent equivalent of such zoning classification) or the Site is no longer within the Planned Service Area for Water and Sewer as contained in the 2000 General Plan or any subsequent Howard County General Plan, the obligations and responsibilities of the Parties in this Agreement shall terminate without further action.

2.2 <u>Developable Area</u>. The physical area of the Site that is available for development consists of those areas that are generally shown on the concept plan attached hereto as EXHIBIT 2. With the exception of the area of land described in the provisions of Section 3.4 below, the developable areas are intended to encompass all of the land areas of the Site which are not precluded from development by the Subdivision Regulations. The Parties acknowledge that the exact size, shape and configuration of one or more of these areas may be adjusted from the area shown on EXHIBIT 2 based on final surveying, engineering, and design of the Project. Except as provided in Section 8.1C of this Agreement, this developable area may not be expanded, reduced, limited or otherwise altered by any legislative, executive or quasi-judicial action of Howard County including, but not limited to, a comprehensive rezoning, a piecemeal rezoning, or the enactment of ordinances, resolutions, rules or regulations, or the interpretation thereof (such as forest conservation or stream buffer ordinances), which would result in a reduction of the developable areas of the Site.

2.3 <u>Development Limitations</u>. The permissible uses on the Site shall be limited to single-family detached housing units and uses accessory thereto under Section 107 of the Zoning Regulations. The development requirements of the Project, such as setbacks, lot area, building height, open space, lighting, and parking, will be subject to the provisions and limitations as set forth in the Zoning Regulations and other applicable provisions of the County Code.

A. The total density of residential development on the Site shall not exceed three hundred twenty-five (325) single-family detached dwelling units as defined in the Zoning Regulations. The Carrolls shall execute and record in the land records of Howard County a perpetual covenant to prohibit the construction of singlefamily attached dwelling units or apartment units on the site, and to prohibit the construction of more than 325 single-family detached units on the site. This covenant shall be made between the Carrolls and either Howard County, a homeowner's association of the site, or other community association, at the discretion of Howard County, within 90 days after Howard County exercises the option under Section 1.2 of the Option Agreement, unless such period is extended by Howard County.

B. The design of the Site, including the lots, roads and open spaces indicated thereon, shall substantially conform to the concept plan attached hereto as EXHIBIT 2.

M

2.4 <u>APFO Approval</u>. The Carrolls shall be required to obtain APFO Approval for the Project, for all aspects of the development of the Site, in all of its phases, for all APFO tests.

2.5 <u>Other Development Approvals and Permits</u>. All sections of the Project shall be required to obtain all applicable Development Approvals and all other applicable requirements of the County Code for land development, including but not limited to site improvement plans (for water, sanitary sewer, storm water management and storm drainage and sediment and erosion control improvements), forest conservation requirements, Subdivision Plat(s) including any required dedications of land consistent with the County Code, building permits, and occupancy permits. The Project shall also be required to obtain all applicable Maryland or federal approval(s) and permits.

2.6 <u>Regulation and Master Plan Consistency</u>. Howard County has determined that the Project and this Agreement are consistent with the Howard County General Plan. The Planning Board has determined that this Agreement is consistent with the Howard County General Plan.

2.7 <u>Public Health, Safety and Welfare</u>. Howard County has determined that development of the Project in accordance with the Zoning Regulations and County Code and the terms and provisions of this Agreement will ensure that the public health, safety and welfare of the citizens of Howard County are protected.

ARTICLE III

ROAD IMPROVEMENTS, SEWER IMPROVEMENTS, WATER IMPROVEMENTS AND OTHER COMMUNITY AMENITIES

3.1 <u>Road Improvements.</u>

A. <u>APFO Required Road Improvements</u>. The Carrolls shall be required, as a condition of APFO Approval, to make or contribute funds towards certain road improvements to serve the Project, and to convey any rights-of-way needed for the construction of the road improvements as required by APFO.

B. <u>Road Connection to Burnside Drive</u>. The Carrolls agree not to construct any roads that would connect the Site or any portion thereof to Burnside Drive. Furthermore, Howard County agrees to honor Howard County Council Resolution 43-1989's termination of Burnside Drive and, therefore, not require the connection of the Site to Burnside. Howard County shall, furthermore, neither exercise nor threaten to exercise its power of eminent domain so as to connect Burnside Drive to the Site

HOWARD COUNTY CIRCUIT COURT (Land Records) MDR 12722, p. 0253, MSA_CE53_12713. Date available 10/01/2010. Printed 06/04/2020.

1. In the event that some other governmental or regulatory agency would require an easement or other accommodation in order to facilitate emergency vehicle access to the Site, the Carrolls may propose an alternative plan to provide such access provided, however, that in no event shall the Carrolls provide for the aforementioned road connection to Burnside.

2. In furtherance of the action necessary to fulfill the intent of this Section 3.1.B, the Carrolls shall subdivide and create an open space lot comprising the land fronting on Burnside Drive (the "Burnside Drive Open Space Lot"). The Carrolls shall work with the residents of Ridge Lake and Chateau Ridge, including the Chateau Ridge Lake Community Association ("CRLCA"), to effect a transfer by deed for the fee simple interests to the Burnside Drive Open Space Lot to an entity or individuals identified by Howard County for sum of one dollar (\$1.00). The transfer of the Burnside Drive Open Space Lot shall be submitted to DPZ with the final deed for the Subdivision Plat, and shall be recorded in the Land Records with the Subdivision Plat.

C. To further effect the purposes of the intent of this Section 3.1.8, the Carrolls shall ensure that the Subdivision Plat will create at least two (2) residential single-family detached lots near and impeding the terminus of Burnside Drive such that Burnside Drive could not be extended westward without the demolition of homes on these two or more lots.

3.2 Sewer Improvements.

A. The Parties agree with the goal of minimizing increases in wastewater flow and the nutrient concentration in wastewater directed to the Little Patuxent Wastewater Reclamation Plant as a condition of Development Approval. In furtherance of this obligation, the Carrolls agree to abide by the terms of the proposed plan of action as set forth in Exhibit 8. The Carrolls shall be responsible for the payment of costs for the construction of sewer utilities for the Project to achieve the goal. Such costs may include the wastewater facility design, engineering, testing, bonding and, if necessary, acquisition of off-site easements for the improvements determined appropriate by the wastewater flow study ("Wastewater Mitigation Cost") provided, however, that in no event shall the Carrolls be required to expend greater than One Million Dollars (\$1,000,000) for the Wastewater Mitigation Cost.

B. In addition to the Wastewater Mitigation Costs, the Carrolls shall construct and/or be obligated to incur the following improvements and/or costs:

1. Replacement and increase in sizing of portions of the existing eight (8) inch sewer line currently on the Site, which portions are determined in the reasoned opinion of Howard County to be undersized;

Mcc

2. Application, incorporation and payment of all fees for Metropolitan District incorporation, which application due to its ministerial nature shall be approved by Howard County.

C. No building permit(s) for residential dwelling unit(s) on the Site shall be issued until (i) Capital Project S-6274 is funded for construction and (ii) the existing sewer line improvements in Section 3.2.B.1 are completed. A grading permit shall not be construed to be a building permit.

3.3 <u>Water Improvements</u>. The Carrolls shall be required to make certain water improvements for the Project as a condition of Development Approval. The Carrolls shall be responsible for the payment of all construction of water utilities for the Project, to include site design, engineering, testing and bonding. Additionally, the Carrolls shall construct and incur the following improvements and costs:

All necessary permit approval(s);

B. Application, incorporation and payment of all fees for Metropolitan District incorporation, which application due to its ministerial nature shall be approved by Howard County.

3.4 <u>Dedication of Land to Howard County</u>. The Carrolls shall dedicate to Howard County a parcel of land on the east side of the Site for use as a County park or other public, recreational use determined by the County to best serve the interest of the public ("Parkland"). This parcel shall contain a total of thirty-six (36) acres of land, more or less, be adjacent to Kiwanis-Wallis Park and shall substantially conform to the area indicated on EXHIBIT 2. The conveyance of this parcel to the County shall be dedicated as Open Space in accordance with the County Code.

The timeline for dedication of this parcel to the County shall be as follows:

A. The Carrolls shall submit a proposed plat of the Parkland to DPZ within one hundred eighty (180) days after execution of this Agreement.

B. The Carrolls shall convey the deed for the Parkland to Howard County with the submission of the first Subdivision Plat for County signature and recordation in accordance with the Subdivision Regulations following the fulfillment of the Conditions Precedent of section 1.2 of the Option Agreement. In the alternative, the Carrolls may in their sole discretion deed the Parkland to Howard County in any legal

manner as directed by Howard County.

C. The Parkland to be conveyed to Howard County shall be eligible to be treated in accordance with the provisions of the HCZR and/or County Code as Open Space and is eligible to be included toward satisfaction of subdivision, site plan and/or other Development Approval requirements.

ARTICLE IV <u>PRESERVATION OF AGRICULTURAL LANDS AND</u> <u>LIMITATIONS ON DEVELOPMENT</u>

<u>Agricultural Preservation</u>. The Carrolls have agreed to place an agricultural land preservation easement on approximately 500 acres of the Property in accordance with Howard County's Agricultural Land Preservation Program pursuant to the terms of the Commitment Letter and the Option Agreement. A legal description of the 500 acres is attached hereto as **EXHIBIT 9**. Nothing in this Agreement shall be construed to supersede either the Carrolls' or Howard County's obligations under the Option Agreement.

4.1

.

- A. Notwithstanding any provision in this Agreement to the contrary, all of the obligations and responsibilities in this Agreement shall terminate in the event that Howard County does not exercise its option under the Option Agreement
- B. Prior to Howard County's exercise of its option under the Option Agreement, The Carrolls shall not submit and Howard County shall have no obligation to approve for recordation any Subdivision Plat for the Property creating any individual lots meeting the bulk requirements of R-ED zoning for single family detached housing.
- C. The Parties further agree that each shall take all actions required to promptly return the Site to a rural conservation zoning classification (now known as "RC") in the event the County has not acquired the agricultural land preservation easement under the Commitment Letter by January 1, 2013, which date may be extended under the terms of the Option Agreement. Upon the final, unappealable approval of the rural classification zoning approval for the Site, the County shall terminate its right to acquire the Agricultural Land Preservation Easement under the Commitment Letter.
- 4.2 <u>Covenant Against Residential Subdivision of the Core</u>. The Carrolls shall execute and record In the Land Records a perpetual covenant to prevent

further subdivision of the approximately ninety-four (94) acres of the property around the main Manor House of the Property (the "Core") with the Howard County Conservancy or other organization determined to be acceptable to Howard County which organization shall hold the enforcement rights to the restrictive covenant. This covenant shall be made within 90 days after Howard County's exercise of the option under Section 1.2 of the Option Agreement, unless such period is extended by Howard County. A legal description of the Core is attached hereto as **EXHIBIT 10**.

- 4.3 The Parties agree that the use of certain funds to be received by the Carrolls pursuant to the anticipated development of the Site shall be used to ensure funding for the restoration and ongoing maintenance of the Manor House and historic outbuildings on the Property. The Parties agree that the Carrolls will be obligated to:
 - 4.3.1 Establish a fund in the amount of Two Million Dollars (\$2,000,000) to be specifically allocated towards such purpose (the "Fund");
 - 4.3:2 To provide evidence to Howard County, at such times as Howard County may reasonably request (but not more often than once per year), of such Fund and invoices evidencing the use of such funds for the purposes as specified in paragraph 4.3 above.
 - 4.3.2.1 The funding of the Fund will be accomplished by the Carrolls placing in the Fund after-tax sums received by the Carrolls from the Project, in an amount equal to fifty percent (50%) of any such after-tax sums, until the Fund has reached a sum of Two Million Dollars (\$2,000,000), minus any sums the Carrolls have spent for such restoration and ongoing maintenance from the date of this Agreement to the date of such funding.
 - 4.3.2.2 The establishment of the Fund and the use of the proceeds thereof for such restoration and maintenance shall be the sole obligation of the Carrolls pursuant to this Agreement for such restoration and maintenance.

ARTICLE V DEVELOPMENT REVIEW

5.1 <u>Timely Development Review</u>. Howard County agrees to use its best efforts in accordance with applicable rules and regulations to ensure that the processing

HOWARD COUNTY CIRCUIT COURT (Land Records) MDR 12722, p. 0257, MSA_CE53_12713. Date available 10/01/2010. Printed 06/04/2020.

MCC

and review of Development Approvals, including, but not limited to, APFO Approvals, subdivision plans, site development plans, issuance of building permits and occupancy permits, are performed in a succinct, timely manner, without undue delay, consistent with the County's current development review process, and that such processing and review will not be subjected to any delay or any moratorium except in accordance with the terms of this Agreement.

5.2 <u>Timely Submission of Documents</u>. The Carrolls agree to use their best efforts to submit and process plans and legal documents for the items set forth in Section 5.1 above in a timely manner.

ARTICLE VI SURVIVAL AND TRANSFER OF OBLIGATION

6.1 <u>Nature, Survival, and Transfer of Obligations</u>. The Carrolls agree that this Agreement shall run with the land and be binding upon and inure to the benefit of the Carrolls and their respective heirs, successors and assigns, and upon any and all successor owners of record of all or any portion of the Property (except owners of an individual lot improved as part of the Project and pursuant to a validly issued building permit). To assure that all such successors, assigns, and successor owners have notice of this Agreement and the obligations created by it, the Carrolls agree that it shall:

A. Have this Agreement recorded among the Land Records of Howard County within twenty (20) days after the Effective Date of this Agreement; and

B. Incorporate, by reference, this Agreement into any and all real estate sales contracts entered into after the Effective Date of this Agreement for the sale of all or any portion of the Property.

C. Prior to the transfer of all or any portion of the Property (except the transfer of an individual lot solely for use as a private residence), or any equitable interest therein, require the transferee to execute an enforceable written agreement, in a form reasonably satisfactory to Howard County, binding transferee to this Agreement.

6.2 <u>Binding Upon Successors and Assigns of Howard County</u>. Howard County agrees that all obligations assumed by it under this Agreement shall be binding on it, its agencies, employees, governmental units, the Planning Board and its and their respective successors and assigns.

ARTICLE VII BREACH AND REMEDIES

1A

7.1 <u>Breach by Carrolls</u>. If the Carrolls shall fail or refuse to perform its obligations as required, then after sixty (60) days from receipt of written notice provided to the Carrolls by Howard County indicating the nature of the default and if the Carrolls have not cured the default, the County may seek equitable relief to enforce the terms and conditions of this Agreement either through a decree for specific performance or an injunction, or declare this Agreement null and void and cease the issuance of building permits and review of development plans. Should the remedies of specific performance or injunction not be available to Howard County because of actions of Carrolls, then Howard County shall be entitled to bring a legal action for damages.

7.2 <u>Breach by Howard County</u>. If Howard County shall fail or refuse to perform its obligations as required, then after sixty (60) days from receipt of written notice provided to Howard County by the Carrolls indicating the nature of the default and if Howard County has not cured the default, the Carrolls may seek equitable relief to enforce the terms and conditions of this Agreement either through a decree for specific performance or an injunction.

7.3 <u>Jurisdiction and Venue</u>. Jurisdiction and venue for any proceedings brought with respect to this Agreement shall be in the Circuit Court for Howard County, Maryland.

7.4 <u>Waiver of Trial by Jury</u>. Carrolls and Howard County do hereby waive trial by jury in connection with any proceedings brought to enforce the terms of this Agreement.

7.5 <u>Agreement to Petition for Rezoning</u>. In the event this Agreement is determined unenforceable by any Court, then the Parties agree that this paragraph shall survive such determination and the Carrolls agree to file a request to rezone the Site to a rural zoning classification equivalent to the RC zoning classification in effect on the date of this Agreement.

ARTICLE VIII EFFECT OF DEVELOPMENT REGULATIONS

8.1 Effect of Agreement.

A. Carrolls must comply with all federal, Maryland and local laws existing on the Effective Date of this Agreement.

B. Except as provided in Section 8.1.C herein, the laws, rules, regulations and policies, which govern the use, density or intensity of the Site shall be the laws, rules, regulations and policies, if any, in force on the Effective Date of the

 \mathcal{M}_{cc}

· Agreement.

C. If Howard County determines that compliance with a law(s) enacted or adopted after the Effective Date of this Agreement is essential to ensure the health, safety or welfare of residents of all or part of Howard County, this Agreement may not prevent Howard County from requiring the Carrolls to comply with those laws, rules, regulations and policies.

D. In the event Howard County takes any action to subject the Project to any new or modified law(s) after the Effective Date of this Agreement under Section 8.1.C above or modifies the Zoning or Comprehensive Plan land use designations of the Site, or the density of the Project, as set forth in this Agreement, which reduces the number of residential dwelling units by more than three and one-half percent (3.5%) ("Howard County Action"), Carrolls shall be relieved of any remaining obligation under this Agreement. Additionally, the Carrolls shall be relieved of any obligation subsequent to the date of the Howard County Action, including any obligation to make any payments required under this Agreement that are not due and payable and are unpaid as of the date of the Howard County Action.

8.2 <u>Approvals Required</u>. Carrolls shall obtain all approvals necessary under any provision of local, Maryland or federal law before proceeding with development of the Project. This Agreement does not address any approvals required by Maryland or federal law, and Carrolls shall be responsible for obtaining any approvals required by local, Maryland, or federal law.

8.3 <u>Fees</u>. Carrolls shall pay all fees (specifically including but not limited to excise taxes, surcharges and water and sewer connection fees) required by the County at the time of the Effective Date of this Agreement at the rate in effect at the time the fee is due. Nothing in this Agreement shall be construed as a waiver or reduction of any such fee.

8.4 <u>Growth Control Delay</u>. In the event that a "Growth Control Delay" (as hereinafter defined) is imposed, then any deadline concerning (i) the Carrolls' obligation to construct, install, fund or post financial guarantees for (a) the infrastructure improvements required pursuant to any Development Approval for the Project in accordance with the phasing requirements set forth therein, and (b) the road improvements described in Article III of this Agreement; and, (ii) the Carrolls' obligation to dedicate the Parkland to the County or the obligations contained in the provisions of Article III of this Agreement, shall be extended for one (1) additional day for each day during which such Growth Control Delay exists, and the Project shall not be subjected to any additional regulation, legislation, limitation, phasing, contributions, penalties or delay in construction, or issuance of zoning certificates/building permits solely as a result of the Growth Control Delay. The term "Growth Control Delay" shall

Mcc

mean any and all delays caused during APFO Approval and/or the implementation or declaration by the United States Government, State of Maryland, Howard County, and/or any agency, department, division and/or branch thereof for purposes of a limitation, prohibition, restriction and/or phasing upon the review, recording, development and construction upon lots on the Site as intended by Carrolls, which is deemed to be essential to ensure the public health, safety, and welfare of County residents as determined in accordance with Section 8.1.C above.

ARTICLE IX

MISCELLANEOUS

9.1 <u>Time of Essence</u>. Time is of the essence in the performance of all terms and provisions of this Agreement.

9.2 <u>Term</u>.

A. This Agreement shall constitute covenants running with the land and shall run with and bind the Property. This Agreement shall terminate and be void on the tenth (10th) anniversary of the Effective Date of this Agreement, unless extended by an amendment to this Agreement complying with all procedures required in this Agreement, the County Ordinance and Maryland Law or in accordance with Section 8.4 above or unless terminated by agreement of the Parties or as permitted by law.

B. Nothing in this Section shall be construed to supersede the term(s) as set forth in any other agreement(s) between the Carrolls and Howard County

9.3 <u>Notices</u>. All notices and other communications in connection with this Agreement shall be in writing and shall be deemed delivered to the addressee thereof (I) when delivered in person on a business day at the address set forth below; or (2) on the third business day after being deposited in any main or branch United States post office for delivery by properly addressed, postage prepaid, certified or registered mail, return receipt requested, at the address set forth below

Notices and communications to the Carrolls shall be addressed and delivered to the following address:

Camilla and Phillip D. Carroll 3500 Manor Lane Ellicott City, MD 21042

with a copy to:

Mac

Joseph Rutter 5300 Dorsey Hall Drive Ellicott City, MD 21042

Sang W. Oh, Esq. 5100 Dorsey Hall Drive Ellicott City, MD 21042

James L. Wright, Esq. 750 E. Pratt Street Suite 900 Baltimore, MD 21202

Notices and communications to the Howard County shall be addressed and delivered to the following address:

Howard County Executive 3430 Courthouse Drive Ellicott City, MD 21043

With a copy to:

Howard County Solicitor 3430 Courthouse Drive Ellicott City, MD 21043 Director, Department of Planning and Zoning 3430 Courthouse Drive Ellicott City, MD 21043

Chair, Howard County Council 3430 Courthouse Drive Ellicott City, MD 21043

By notice complying with the requirements of this Section, each party shall have the right to change the address or addressee or both for all future notices and communications to such party, but no notice of a change of address shall be effective until actually received.

9.4 <u>Amendments</u>. The Parties to this Agreement may amend the Agreement by mutual consent after Howard County holds a public hearing and complies with all applicable laws of the County Ordinance concerning amendment of a Development Rights and Responsibilities Agreement. All amendments to this Agreement shall be in writing and shall be executed by Howard County and the Carrolls. Unless the Planning Board determines that the proposed amendment is consistent with the Howard County General Plan, the Parties may not amend this Agreement.

HOWARD COUNTY CIRCUIT COURT (Land Records) MDR 12722, p. 0262, MSA_CE53_12713. Date available 10/01/2010. Printed 06/04/2020.

MCC

10

9.5 <u>Termination or Suspension</u>. The Parties to this Agreement may terminate or suspend the Agreement by mutual consent after Howard County holds a public hearing and complies with all applicable laws concerning termination or suspension of a Development Rights and Responsibilities Agreement as set forth in Howard County Ordinance. If Howard County determines that a suspension or termination is essential to ensure the public health, safety or welfare, as determined in accordance with Section 8.1.C above, Howard County may suspend or terminate this Agreement following a public hearing. Any such unilateral termination of this Agreement by Howard County shall not in any way affect the validity of any Development Approvals which have been obtained for the Project, including, but not limited to, APFO Approvals.

9.6 <u>Authority to Execute</u>. Howard County and the Carrolls hereby acknowledge and agree that all required notices, meetings, and hearings have been properly given and held by Howard County with respect to the approval of this Agreement and agree not to challenge this Agreement or any of the obligations created by it on the grounds of any procedural infirmity or any denial of any procedural right. Howard County hereby warrants and represents to the Carrolls that the persons executing this Agreement on its behalf have been properly authorized to do so. The Carrolls hereby warrant and represent to Howard County that they are the fee simple, record owners of Doughoregan and the Site, (2) that each has the right, power and authority to enter into this Agreement and to agree to the terms, provisions, and conditions set forth herein and to bind the Property as set forth herein, (3) that all legal actions needed to authorize the execution, delivery and performance of this Agreement have been taken, and (4) each has been duly represented by attorneys.

9.7 <u>Governing Law</u>. This Agreement shall be governed by and construed in accordance with the laws of the State of Maryland.

9.8 <u>Consent to Jurisdiction</u>. The Parties irrevocably consent to the jurisdiction of the Circuit Court for Howard County, Maryland or any federal court sitting in the District of Maryland.

9.9 <u>Remedies Cumulative</u>. Each right, power and remedy of a party provided for in this Agreement, or any other agreement between the Parties, now or hereafter existing, shall be cumulative and concurrent and in addition to every other right, power or remedy provided for in this Agreement or any other agreement between the Parties, now or hereafter existing.

9.10 <u>Severability</u>. In case any one or more of the provisions contained in this Agreement shall for any reason be held invalid, illegal or unenforceable in any respect, such invalidity, illegality, or unenforceability shall not affect any other provision of the Agreement, and this Agreement shall be construed as if such invalid, illegal, or

HOWARD COUNTY CIRCUIT COURT (Land Records) MDR 12722, p. 0263, MSA_CE53_12713. Date available 10/01/2010. Printed 06/04/2020.

unenforceable provision had never been contained in this Agreement.

9.11 <u>Recordation</u>. Any party may record this Agreement among the Land Records of Howard County, Maryland. In the event the Agreement is terminated following an appeal as provided for in Section 9.12 below, the Parties agree to execute and record a document in the aforesaid Land Records to terminate this Agreement.

9.12 <u>Appeals</u>. Both Maryland law and the County Ordinance may allow any person aggrieved by this Agreement to file an appeal. If the effect of the decision in such Appeal revises this Agreement in any way, any party to this Agreement may terminate the Agreement by providing notice to all Parties within thirty (30) days of the date that the decision in the appeal becomes final and all appeals thereof have been finally determined.

9.13 <u>No Obligation to Approve</u>. This Agreement shall not be interpreted or construed to impose any legal obligation on Howard County or any of its boards, agencies, commissions or employees to approve any development, use, density or intensity other than as provided specifically in this Agreement.

9.14 <u>No Third Party Beneficiary Status</u>. The Parties specifically agree that this Agreement is not intended to create in the public or any member thereof, third party beneficiary status in connection with the performance of the obligations under this Agreement.

9.15 <u>Lien holders</u>. All persons with a lien interest in the Property have executed this Agreement, and those lien holders with a power of sale have subordinated such liens to the position of Howard County under this Agreement.

IN WITNESS WHEREOF, the Parties have hereunto set their hands on the date first above written.

[Signatures on Following Page]

Mac



Margaret Ann Nolan ρ

County Solicitor

[Notaries on Following Page]

.

STATE OF MARYLAND,
I HEREBY CERTIFY that on this <u>944</u> day of <u>Secter by</u> , 2010, before me, the subscriber, a Notary Public of the State of Maryland, in and for the County aforesaid, personally appeared CAMILLA CARROLL, personally known to me or proven to be the individual named herein and executed this Agreement for the purposes stated therein. AS WITNESS my Hand approximation Seal of
My Commission Expires:
STATE OF MARYLAND,
I HEREBY CERTIFY that on this <u>944</u> day of <u>Sptewey</u> , 2010, before me, the subscriber, a Notary Public of the State of Maryland, in and for the County aforesaid, personally appeared PHILIP D. CARROLL, personally known to me or proven to be the individual named herein and executed this Agreement for the purposes stated therein AS WITNES my Handrant, Avaarial Seal. My Commission Expire
STATE OF MARYLAND, Howard COUNTY, TO WIT:
I HEREBY CERTIFY that on this <u>3</u> day of <u>September</u> , 2010, before me, the subscriber, a Notary Public of the State of Maryland, in and for the County aforesaid, personally appeared KEN ULMAN, the County Executive for Howard County, Maryland, who acknowledged the same to be the act of the County and that he executed the foregoing Agreement for the purposes therein contained by signing in my presence the name of Howard County, Maryland as County Executive.
AS WITNESS my Hand and Notarial Seal.
My Commission Expires:
10

LISA STELLO O'BRIEN NOTARY PUBLIC STATE OF MARYLAND My Commission Expires February 20, 2011

ATTORNEYS' CERTIFICATION

1

THIS IS TO CERTIFY that the undersigned are members, in good standing, of the Bar of the Court of Appeals of Maryland, and that the within instrument was prepared by the undersigned or under their supervision.

 \odot

Upon Recordation Please Return To:

Sang W. Oh, Esq. Talkin & Oh, LLP 5100 Dorsey Hall Drive Ellicott City, MD 21042

LIENHOLDER CONSENTS AND SUBORDINATION OF INTERESTS

The undersigned lienholder does hereby consent to the aforesaid Agreement, and by the signature of its Trustee, does hereby agree that such instrument shall be subordinate to and shall survive any sale under its Deed of Trust dated______, and recorded among the Land Records of Howard

County, Maryland in Liber _____, folio _____

WITNESS:

2

LIST OF EXHIBITS

- Exhibit 1 Property Description
- Exhibit 2 -- Concept Plan
- Exhibit 3 Description of Site
- Exhibit 4 Attorney title opinion certifying as to legal and equitable owners
- Exhibit 5 Planning Board Determination of Consistency
- Exhibit 6 Commitment Letter
- Exhibit 7 Option Agreement
- Exhibit 8 Wastewater Treatment Plan of Action
- Exhibit 9 Description of Agricultural Preservation Parcel
- Exhibit 10 Description of Core



LIBER 12722 FOLD 271

EXHIBIT 1

March 18, 2010

DESCRIPTION OF A 892.622 ACRE PARCEL PROPERTY OF PHILIP D. CARROL CAMILLA CARROLL SECOND AND THIRD ELECTION DISTRICTS HOWARD COUNTY, MARYLAND

BEING a parcel of land, situate and lying in the Second and Third Election Districts of Howard County, Maryland; said parcel of land being part of the land which, by Will of Nina R. Carroll dated June 4, 1979, as amended by its First Codicil dated November 28, 1986 and its Second Codicil dated April 10, 1987; Nina R Carroll having departed this life on February 11, 1989; said will having been probated in the Surrogates Court Of New York County, New York on April 11, 1989, and the Estate of Nina R. Carroll is filed in the Register of Wills of Howard County, Maryland as Estate No. 17-7868-276; said property also being described in a Mortgage dated October 31, 1962 and recorded among the Land Records of Howard County, Maryland in Liber 394, At Folio 64 and being all of Parcel 1 as described in the aforesaid Liber 394 at Folio. 64; said parcel of land being more particularly described, as now surveyed, in the Maryland State Coordinate System NAD '83 Datum, as projected by Howard County Geodetic Control, as follows;

FISHER, COLLINS & CARTER, INC.

HOWARD COUNTY CIRCUIT COURT (Land Records) MDR 12722, p. 0271, MSA_CE53_12713. Date available 10/01/2010. Printed 06/04/2020.

CIVIL ERGIREEROND CONSULTANTS

10272 Ballicore Hallonal Pilo Liusofi City, Maryland 21942 410) 481-2855 *ax (410) 750-3784 BEGINNING FOR THE SAME at a point 33.0 feet, more or less, measured at right angles northerly from the centerline of paying as now exists, twenty-four (24) feet wide more or less, and being on the Northern margin of the sixty-six foot wide Right of Way of Frederick Road (Maryland Route 144), formerly known as the Baltimore to Frederick Turnpike Road;

. t. et. .

. 12

The second second second

, t. t. t.

一、 网络美国达 医手间炎 法经济

1.5

: (;;;

:

•

200

19 . ¹

4 1

LIBER | 2722 FOL10272

said point being opposite Manor Lane and at the beginning of the First or South 73°02'48" East 3140.09 foot Deed Line of Parcel 1, recorded in Liber 394 at Folio 64; thence running with Frederick Road and running with and binding on the said First Deed Line of Parcel 1, as now surveyed,

1)

South 73°02'02" East 3140.09 feet to a point at the beginning of the Second or South 25°58'45" East 2921.96 foot Deed Line of Parcel 1, recorded in Liber 394 at Folio 64; said Second Deed Line also being the following; the Third or North 18°43' West 323.01 foot line of the adjacent conveyance from Albert S. Singleton and Elsie B. Singleton, his wife, to Philip G. Wilkins and Corinne Elizabeth Wilkins, his wife, by deed dated June 29, 1963 and recorded among said Land Records in Liber 403 at Folio 669; and the North 26°02'23" West 1737.32 foot and the North 23°56'49" West 55.21 foot outlines, as shown on a plat entitled "Plat of Survey, Kiwanis Glub of Ellicott City, Inc." recorded among said Land Records as Plat No. 10119; and the North 19º061: West 180.00 foot outline, as shown on a subdivision plat entitled "Map of Pine Orchard " Meadow: Section Three" recorded among said Land Records in Plat Book 6 at Page 53; and also the North 18°39'00? West 532.45 foot outline, as shown on a subdivision plat entitled "Pine Orchard Meadows, Section Four" and recorded among said Land Records in Plat Book 17 at Page 75; thence intending to bind on and run with said Second Deed Line of Parcel 1, recorded in Liber 394 at Folio 64, as now surveyed, the following three (3) courses and distances:

n and an an early an early and an early and an early and an early and an early and an early and an early and an An early and the early and the early and the early and the early and the early and the early and the early and the Instance of the early and the early and the early and the early and the early and the early and the early and the Instance of the early and

HOWARD COUNTY CIRCUIT COURT (Land Records) MDR 12722, p. 0272, MSA_CE53_12713. Date available 10/01/2010. Printed 06/04/2020.

FISHER, COLLINS

CARTER.

CAVIL, EMBERGERING CONSULTANTS 2014 LAND BURYEYORS

10272 Bellimora Helkopel Pike Ellisen Cily, Morriend 21042 (410) 481-2855 Faz (410) 768-3784

INC.

LIBER 12722 FOLIO273



 South 24°00'53" East 55.21 feet to a 1 inch Iron Pipe found; thence still continuing with said Second Deed Line of Liber 394 at Folio 64.

South 25°53'43" East, 711.66 feet to a 0.4 foot by 0.4 foot wide Concrete
Monument found at the beginning of the Third or South 21°10'35" East 412.58
foot Deed Line of Parcel I, recorded in Liber 394 at Folio 64; said Third Deed
Line of Parcel I also being the North 14°00'00" West 136.25 foot outline, as
shown on the aforesaid plat of "Pine Orohard Meadows, Section Four" recorded
in.Plat Book: 17 at Folio 75, and also the South 14°00'00" East 276.25 foot
outline, as shown on a subdivision plat entitled "Pine Orohard Meadows,
Section 5" and recorded among said Land Records in Plat Book 22 at Folio 11;
thence binding along said Third Deed Line of Parcel 1; recorded in Liber 394 at
Folio 64, as now surveyed, and the second sin "FCC-106" Cap set at the second

beginning of the Fourth or South 87°17'37" West 523.99 foot Deed Line of

Parcel 1, recorded in: Liber 394 at Folio 64; said Fourth Deed Line of Parcel 1

also being the North 87º16'56" East 523.27 foot outline; as shown on Sheet 4 of

the subdivision-plats entitled "Centennial Manor, Section One, Area Three, Lots

٠.

85-108", recorded among said Land Records as Plat C.M.P. No. 7675, thence

.

. ji i

3

يعرجا فالمرجع فالتجار فالمرج

. . .

1

5)

4)

CIVIL ENGLICERUNG CONSULTANTS and LAND SURVEYORS 10272 Bollingson Halional Pito Hiltott City, Maryland 21042 4101 481-5555

FISHER, COLLINS

CARTER, INC.

35.0

410) 481-2856 *ax (410) 750-3784

LIBER 12722 FOL10274



HOWARD COUNTY CIRCUIT COURT (Land Records) MDR 12722, p. 0274, MSA_CE53_12713. Date available 10/01/2010. Printed 06/04/2020.

LIBER 12722 FOLIO 275

in Liber 394 at Folio 64; said Sixth Deed Line of Parcel 1 also being the North 62°06'38" East 641.44 foot outline, as shown on a subdivision plat entitled "Plat of Revision, Kingsbridge @ Burleigh Manor, Lots 763 thru 778" recorded among said Land Records as Plat No. 12607; thence binding along said Sixth Deed Line of Parcel I, recorded in Liber 394 at Folio 64, as now surveyed, South 62°05'05" West 641.25 feet to a Granite Stone found at the beginning of

the Seventh or North 73°14'48" West 4021.13 foot Deed Line of Parcel 1, recorded in Liber 394 at Folio 64; said Granite Stone also being at the end of the Seventh or South 73°14'48" East 4021.23 foot Deed Line of Parcel 2 described in the aforesaid Liber 394 at Folio 64; thence binding along the Seventh thru Thirteenth Deed Lines of Parcel 1, recorded in Liber 394 at Folio 64, and also binding reversely along the Seventh thru First Deed Lines of Parcel 2, recorded in Liber 394 at Folio 64, as now surveyed, for the following seven (7) courses and distances:

- 9) North 73°16'34" West 4018.45 feet to a Rebar with "FCC-106" Cap set at a point near the southeastern side of Manor Lane, being a twelve (12) foot wide gravel drive, as now exists; thence running with said Manor Lane,
- North 14°04'34" East 24.75 feet to a Rebar with "FCC-106" Cap set, thence crossing over said Manor Lane,
- North 65°25'26" West 2143.94 feet, to a 1.25 inch diameter Iron Pipe found; thence

HOWARD COUNTY CIRCUIT COURT (Land Records) MDR 12722, p. 0275, MSA_CE53_12713. Date available 10/01/2010. Printed 06/04/2020.

FISHER, COLLINS & CARTER, INC. CMTA EXECUTES DUM CONSIATIONTS MILLIO SURVEYORS

to272 Baltimore National Pike Ellicoit City, Maryland 20042 (410) 481-2855 Fax (410) 780-3784 8)

- North 70°10'26" West 1456.72 feet, to a 1.25 inch diameter Iron Pipe found;
 thence
- 13) North 78°34'25" West 791.22 feet to a point; thence
- 14) North 73°34'25" West 387.55 feet to a point; thence
- 15) South 69°10'35" West 31.37 feet to a point 23.8 feet, more or less, measured at right angles southeasterly from the centerline of paving, eighteen (18) feet wide, as now exists, of Folly Quarter Road, formerly known as the Vineyard Road; said point also being at the beginning of the Fourteenth or North 53°00'48" East 577.50 foot Deed Line of Parcel 1 recorded in Liber 394 at Folio 64; thence running with said Folly Quarter Road and running with and binding on the Fourteenth thru Sixteenth Deed Lines of Parcel 1, recorded in Liber 394 at Folio 64, as now surveyed, for the following three (3) courses and distances:
- 16) North 51°40'56" East 568.09 feet to a point; thence crossing into and running within said Folly Quarter Road, on or near the centerline of the existing paving thereof.
- 17) North 39°59'12" East 2458.50 feet to a point; thence
- 18) North 26°28'43" East 2376.00 feet to a point on or near the intersection of the centerlines of existing paving of said Folly Quarter Road with Frederick Road (Maryland Route 144), as now exists, twenty four (24) feet wide and formerly known as the Baltimore to Frederick Tumpike Road; said point also being the beginning of the Seventeenth or South 82°36'44" East 189.50 foot Deed Line of Parcel 1, recorded in Liber 394 at Folio 64; thence running within said Frederick

COLLINS & CARTER, INC. GVIL FRANKFORM

FISHER,

1027E Baitimore Notionel Pite Elifott Gity, Maryland 21642 (418) 481-2855 Fax (418) 750-3784

'LIBER'] 2722 FOL10277

Road, on or near the centerline of existing paving, and running with and binding on said Seventeenth Deed Line of Parcel 1 recorded in Liber 394 at Folio 64, as now surveyed,

- 19) South 82°54'28" East 189.50 feet to a point; thence leaving said centerline of
 Frederick Road and running with and binding the Eighteenth or North
 14°06'44" West 35.47 foot Deed Line of Parcel 1 recorded in Liber 394 at Folio
 64, as now surveyed,
- 20) North 17°12'33" West 32.08 feet to a point at the beginning of the Nineteenth or South 82°36'44" East 1394.98 foot Deed Line of Parcel 1 recorded in Liber 394 at Folio 64; thence running with the aforesaid Frederick Road and running with and binding on said Nineteenth Deed Line of Parcel 1 recorded in Liber 394 at Folio 64, as now surveyed,
- South 82°35'58" East 1394.98 feet to the point of beginning; containing 892.622
 acres of land, more or less.

SUBJECT TO an existing 20' Right-Of-Way for a Sewer Main described in a "Deed Of Easement" conveyed from Philip D. Carroll and Camilla Carroll to Howard County, Maryland by Deed dated December 24, 1994 and recorded among the Land Records of Howard County, Maryland in Liber 3422, Folio 357.

ALSO SUBJECT TO AND TOGETHER WITH all matters shown on a plat entitled "Amended Plat Of Easement, Sending Parcel, Property Of Philip Carroll And Camilla Carroll" and recorded among the Land Records of Howard County, Maryland as Plat No. 19928, and a

HOWARD COUNTY CIRCUIT COURT (Land Records) MDR 12722, p. 0277, MSA_CE53_12713. Date available 10/01/2010. Printed 06/04/2020. FISHER, COLLINS CARTER, INC. CIVIL ENGINEERING GOMEDIANTS d LAND SURVEYORS 10272 Beithmore Hellonai Pike Ellicoti City, Herrison 21042 [410] 491-2855 Fax (410) 750-3784

LIBER 12722 FOLIO278

"Deed Of Preservation Easement" from Philip D. Carroll and Camilla Carroll to The Howard County Conservancy, Inc. and Howard County, Maryland by Deed dated October 4, 2006 recorded among the aforesaid Land Records in Liber 10295, Folio 58.

ALSO SUBJECT TO AND TOGETHER WITH any and all terms and conditions that might exist to the rights, use and maintenance of Folly Quarter Road and Frederick Road, Maryland Route 144, to Howard County, Maryland and the Maryland State Highway Administration.

SUBJECT TO AND TOGETHER WITH all conditions, covenants, easemients, restrictions and rights-of-way of record, which may apply.

This description was prepared without the benefit of a Title Report.



Prepared By Checked By

WO #05022-3002

CARTER, CIVIL EXERCICERINA CONSULTANTS NOA LAND SURVEYORS

18272 Balilmona Kallonal Pite Ellicott City, Maryland 21842 (410) 461-2855 Fax (410) 750-3764

I:\2005\05022\dwg\05022-3001 Total Tract Exhibit.dwg, 3/18/2010 9:45:49 AM, dstreett, 1:1



LIBER | 2722 FOLD 279



HOWARD COUNTY CIRCUIT COURT (Land Records) MDR 12722, p. 0281, MSA_CE53_12713. Date available 10/01/2010. Printed 06/04/2020.



LINER | 2722 FOLID 28



ЕХНІВІТ З

March 18, 2010

DESCRIPTION OF A 221.064 ACRE PARCEL PART OF THE PROPERTY OF PHILIP D. CARROLL AND CAMILLA CARROLL SECOND AND THIRD ELECTION DISTRICTS HOWARD COUNTY, MARYLAND

BEING a parcel of land, situate and lying in the Second and Third Election Districts of Howard County, Maryland, said parcel of land being part of the land which by Will of Nina R. Carroll dated June 4, 1979, as amended by its First Codicil dated November 28, 1986 and its Second Codicil dated April 10, 1987; Nina R. Carroll having departed this life on February 11, 1989; said Will having been probated in the Surrogates Court of New York County, New York on April 11, 1989, and the Estate of Nina R. Carroll is filed in the Register of Wills of Howard County, Maryland as Estate No. 17-7868-276, said property also being described in a Mortgage dated October 31, 1962 and recorded among the Land Records of Howard County, Maryland in Liber 394 at Folio 64, and being part of Parcel 1 described in the aforesaid Liber 394 at Folio 64; said parcel being more particularly described, as now surveyed in the Maryland State Coordinate System NAD' 83 Datum, as projected by Howard County Geodetic Control, as follows:

FISHER, COLLINS & CARTER, INC. GVA. PRADELBARA CONSTRUCT

and LAND SURVEYORS 10272 Ballimore National Pita Eilloofi Gily, Maryland 21542 (410) 461-283 Fax (410) 750-3764 **BEGINNING FOR THE SAME** at a point on the Northern margin of the sixty-six foot wide Right of Way of Frederick Road (Maryland Route 144), formerly known as the Baltimore to Frederick Turnpike Road, said point being at the beginning of the Second or South 25°58'45'' East, 2921.96 foot Deed Line of the aforesaid Liber 394 at Folio 64 (Parcel 1); said Second Deed Line also being the following: the Third or North 18°43' West 323.01 foot line of the adjacent conveyance from Albert S. Singleton and Elsie B. Singleton, his wife, to Philip G. Wilkins and Corinne Elizabeth Wilkins, his wife, by deed dated June 29, 1963 and recorded among said Land Records in Liber 403 at Folio 669; and the North 26°02'23" West 1737.32 foot and the North 23°56'49" West 55.21 foot outlines, as shown on a plat entitled "Plat of Survey, Kiwanis Club of Ellicott City, Inc." recorded among said Land Records as Plat No. 10119; and the North 19°06' West 180.00 foot outline, as shown on a subdivision plat entitled "Map of Pine Orchard Meadow, Section Three" recorded among said Land Records in Plat Book 6 at Page 53; and also the North 18°39'00" West 532.45 foot outline, as shown on a subdivision plat entitled "Pine Orchard Meadows, Section Four" and recorded among said Land Records in Plat Book 17 at Page 75; thence intending to bind on and run with said Second Deed Line of Parcel 1, recorded in Liber 394 at Folio 64, as now surveyed, the following three (3) courses and distances:

- South 26°02'24" East 2,155.13 feet, passing over a Concrete Monument found 89.57 feet on line, to a Rebar with "FCC-106" Cap set; thence continuing with the aforesaid Second Deed Line of Liber 394 at Folio 64,
- South 24°00'53" East 55.21 feet to a 1 inch Iron Pipe found; thence still continuing with said Second Deed Line of Liber 394 at Folio 64,
- 3) South 25°53'43" East, 711.66 feet to a 0.4 foot by 0.4 foot wide Concrete Monument found at the beginning of the Third or South 21°10'35" East 412.58 foot Deed Line of Parcel 1, recorded in Liber 394 at Folio 64; said Third Deed Line of Parcel 1 also being the North 14°00'00" West 136.25 foot outline, as

INC. even exposed on source of the source o

FISHER,

COLLINS & CARTER,

2

shown on the aforesaid plat of "Pine Orohard Meadows, Section Four" recorded in Plat Book 17 at Folio 75, and also the South 14°00'00" East 276.25 foot outline, as shown on a subdivision plat entitled "Pine Orchard Meadows, Section 5" and recorded among said Land Records in Plat Book 22 at Folio 11; thence binding along said Third Deed Line of Parcel 1, recorded in Liber 394 at Folio 64, as now surveyed,

4) South 21°13'25" East 412.50 feet to a Rebar with "FCC-106" Cap set at the beginning of the Fourth or South 87°17'37" West 523.99 foot Deed Line of Parcel 1, recorded in Liber 394 at Folio 64; said Fourth Deed Line of Parcel 1 also being the North 87°16'56" East 523.27 foot outline, as shown on Sheet 4 of the subdivision plats entitled "Centennial Manor, Section One, Area Three, Lots 85-108", recorded among said Land Records as Plat C.M.P. No. 7675, thence binding along said Fourth Deed Line of Parcel 1, recorded in Liber 394 at Folio 64, as now surveyed,

5) South 87°11'19" West 523.27 feet to a Concrete Monument found at the beginning of the Fifth or South 07°19'03" East 2783.20 foot Deed Line of Parcel 1 recorded in Liber 394 at Folio 64; said Fifth Deed Line of Parcel 1 also being the following: the North 07°14'30" West 245.00 foot outline, as shown on the aforesaid plat of "Centennial Manor, Section One, Area Three, Lots 85-108" recorded as Plat C.M.P. No. 7675; and the North 07°14'30" West 930.98 foot outline, as shown on Sheets 3 and 2 of the subdivision plats entitled "Centennial Manor, Section One, Area One, Lots 1-22" recorded among said Land Records

HOWARD COUNTY CIRCUIT COURT (Land Records) MDR 12722, p. 0285, MSA_CE53_12713. Date available 10/01/2010. Printed 06/04/2020. FISHER, COLLINS CARTER, INC. TVIL ENGINEERING CONSULTANTS VID LAND SURVEYING 10272 Ballimora Hallenal Piko Slicoli Oly, Maryishd 21942 410) 461-2855 141 (410) 750-3764
'LIBER¹1 2722 FOLIO286

as Plat Nos. 6797 and 6796, respectively; and the North 06°45'30" West 600.00 foot outline, as shown on a subdivision plat entitled "Section Three, Chateau Ridge" recorded among said Land Records in Plat Book 21 at Folio 11; and the North 06°45'30" West 700.00 foot outline as shown on a subdivision plat entitled "Section Five, Chateau Ridge" recorded among said Land Records in Plat Book 22 at Folio 46; and also the North 06°45'30" West 300.00 foot outline, as shown on a subdivision plat entitled "Section Six, Chateau Ridge" recorded among said Land Records in Plat Book 22 at Folio 82; thence binding along said Fifth Deed Line of Parcel 1, recorded in Liber 394 at Folio 64, as now surveyed,

6) South 07°13'17" East 2779.90 feet to a Granite Stone found at the beginning of the Sixth or South 61°55'21" West 642.37 foot Deed Line of Parcel 1 recorded in Liber 394 at Folio 64; said Sixth Deed Line of Parcel 1 also being the North 62°06'38" East 641.44 foot outline, as shown on a subdivision plat entitled "Plat of Revision, Kingsbridge @ Burleigh Manor, Lots 763 thru 778" recorded among said Land Records as Plat No. 12607; thence binding along said Sixth Deed Line of Parcel 1, recorded in Liber 394 at Folio 64, as now surveyed,

South 62°05'05" West 641.25 feet to a Granite Stone found at the beginning of the Seventh or North 73°14'48" West 4021.13 foot Deed Line of Parcel 1, recorded in Liber 394 at Folio 64; said Granite Stone also being at the end of the Seventh or South 73°14'48" East 4021.23 foot Deed Line of Parcel 2 described in the aforesaid Liber 394 at Folio 64; thence running with and binding part of

HOWARD COUNTY CIRCUIT COURT (Land Records) MDR 12722, p. 0286, MSA_CE53_12713. Date available 10/01/2010. Printed 06/04/2020.

COLLINS & CARTER, INC. GVIL EMAILDOCEARS CONSOLVANTS And LAND SURVEYORS

FISHER,

7)

10272 Gallimus Reliansi Pike Elilorit GHy, Meryland 21042 410) 461-2255 Fox (410) 769-3784

LIBER 12722 FOLIO287

the aforesaid Seventh Deed Line of Liber 394, Folio 64, as now surveyed, the following course and distance:

- 8) North 73°16'34" West, 1,764.49 feet to a point; said point being on and a distance of 2,253.96 feet from the end of the Seventh or North 73°14'48" West, 4021.13 foot Deed Line of the aforesaid Liber 394 at Folio 64 (Parcel 1); thence leaving said Seventh line of Liber 394 at Folio 64 (Parcel 1) and running for new lines of division, the following twenty-five (25) courses and distances:
- 9) North 16°43'26" East, 858.48 feet to a point; thence
- 10) North 33°51'48" East, 651.81 feet to a point; thence
- 11) North 67°04'12" East, 462.42 feet to a point; thence
- 12) North 23°09'51" West, 507.53 feet to a point; thence
- 13) North 37°01'20" West, 903.13 feet to a point of curvature; thence
- 468.79 feet along the arc of a curve to the right, having a radius of 580.00 feet, a central angle of 46°18'34" and subtended by a chord bearing and distance of North 13°52'03" West, 456.13 feet to a point of tangency; thence
- 15) North 09°17'14" East, 113.42 feet to a point of curvature; thence
- 136.14 feet along the arc of a curve to the right, having a radius of 325.00 feet, a central angle of 24°00'04" and subtended by a chord bearing and distance of North 21°17'16" East, 135.15 feet to a point of reverse curvature; thence
- 131.70 feet along the arc of a curve to the left, having a radius of 170.00 feet, a central angle of 44°23'10" and subtended by a chord bearing and distance of North 11°05'43" East, 128.43 feet to a point of reverse curvature; thence

INC. GVAL EMAINEERING CONSIGNANTS END LAND SURVEYORS 18272 Builingte Hallenet Pite

FISHER,

COLLINS & CARTER,

filicott City, Margand 21042 410) 461-2855 43x (410) 760-3784

110ER | 2722 FOLIO 288

18) 89.85 feet along the arc of a curve to the right, having a radius of 285.00 feet, a central angle of 18°03'45" and subtended by a chord bearing and distance of North 02°03'59" West, 89.47 feet to a point of tangency; thence

19) North 06°57'53" East, 62.64 feet to a point of curvature; thence

- 20) 219.63 feet along the arc of a curve to the right, having a radius of 285.00 feet, a central angle of 44°09'14" and subtended by a chord bearing and distance of North 29°02'30" East, 214.24 feet to a point of reverse curvature; thence
- 283.12 feet along the arc of a curve to the left, having a radius of 275.00 feet, a central angle of 58°59'14" and subtended by a chord bearing and distance of North 21°37'30" East, 270.78 feet to a point of compound curvature; thence
- 22) 470.43 feet along the arc of a curve to the left, having a radius of 595.00 feet, a central angle of 45°18'00" and subtended by a chord bearing and distance of North 30°31'07" West, 458.27 feet to a point of tangency; thence
- 23) North 53°10'07" West, 259.20 feet to a point of curvature; thence
- 312.10 feet along the arc of a curve to the right, having a radius of 330.00 feet, a central angle of 54°11'16" and subtended by a chord bearing and distance of North 26°04'29" West, 300.60 feet to a point of tangency; thence
- 25) North 01°01'09" East, 232.32 feet to a point of curvature; thence
- 26) 121.72 feet along the arc of a curve to the left, having a radius of 150.00 feet, a central angle of 46°29'32" and subtended by a chord bearing and distance of North 22°13'37" West, 118.40 feet to a point of reverse curvature; thence

GVIL ENGLISHERDED CONSULTANTS and LAND SURVEYORS 10272 Ballimeta Halbanal Pila

FISHER, COLLINS

CARTER.

INC.

10272 Ballinoto Heliopal Pite Elicoli Oity, Nergiend 21042 (410) 481-2855 Fax (410) 750-3784

YEIDER 2722 FOLIO289

• •

	1	,
	27)	346.71 feet along the arc of a curve to the right, having a radius of 315.00 feet, a
		central angle of 63°03'47" and subtended by a chord bearing and distance of
		North 13°56'30" West, 329.47 feet to a point of tangency; thence
	28)	North 17°35'24" East, 225.85 feet to a point; thence
	29)	North 27°46'18" West, 35.58 feet to a point; thence
	3'0)	North 73°08'00" West, 180.91 feet to a point; thence
	31)	North 19°32'01" West, 18.41 feet to a point; thence
	32)	North 73°02'02" West, 328.95 feet to a point; thence
	33)	North 16°57'58" East, 65.50 feet to a point on the aforesaid Northern margin of
•		Frederick Road (Maryland Route 144); said point being on and a distance of
		1,432.93 feet from the beginning of the First or South 73°02'48" East, 3140.09
		feet Deed Line of the aforesaid Liber 394 at Folio 64 (Parcel 1); thence running
		with and binding on said First line of Liber 394 at Folio 64 (Parcel 1) and the
		aforesaid Northern margin of Frederick Road (Maryland Route 144), the
		following course and distance:
	34)	South 73°02'02" East, 1,707.16 feet to the point of beginning; containing
		221.064 acres of land, more or less.
		SUBJECT TO an existing 20' Right-Of-Way for a Sewer Main described in a
	"Deed Of Eas	ement" conveyed from Philip D. Carroll and Camilla Carroll to Howard County,
(174)TB		Deed dated December 24, 1994 and recorded among the Land Records of Howard land in Liber 3422, Folio 357.
ATANTS	County, tendly	ланы на Блобр ј444, го но јј /,
1 Pika 1242		

HOWARD COUNTY CIRCUIT COURT (Land Records) MDR 12722, p. 0289, MSA_CE53_12713. Date available 10/01/2010. Printed 06/04/2020. FISHER, COLLINS & CARTER, INC. CIVA. EXECUTERING CONSI and LAND SURVEYORS 10272 Daitioners Hollooni Eilicolt City, Naryland 211 (410) 401-2855 Fax (410) 760-3784

"LIBER | 2722 FOL 10290

ALSO SUBJECT TO AND TOGETHER WITH any and all terms and conditions that might exist to the rights, use and maintenance of Folly Quarter Road and Frederick Road, Maryland Route 144, to Howard County, Maryland and the Maryland State Highway Administration.

SUBJECT TO AND TOGETHER WITH all conditions, covenants, easements, restrictions and rights-of-way of record, or proposed, which may apply.

This description was prepared without the benefit of a Title Report.



Prepared By DBS Checked By

WO #05022-3002

FISHER, COLLINS & CARTER, INC. ovel endocervice constantions and land servicions

10272 Balümərə Həlbaə) Pitə Ellisofi City, Haryland 21042 410) 481-2055 Yaz (410) 780-3784 1:12005\05022\dwg\05022-3001 Parcel 3 Exhibit.dwg, 3/18/2010 10:01:05 AM, dstreett, 1:1



EXHIBIT 4

.

Attorney Title Opinion Certifying as to Legal and Equitable Owners

. LAW OFFICES OF TALKIN & OH, LLP COLUMBIA OFFICE 5100 DORSBY HALL: DRIVE ELLICOTT CITY, MARYLAND 21042-7870

(410) 964-0300 (301) 596-6500 Fam (410) 964-2008

June 22, 2010

Ms. Charlotte R. Dryden Real Estate Services Howard County Department of Public Works 3430 Courthouse Drive Ellicott City, Maryland 21043

Re: "Doughoregan Manor", 3500 Manor Lane, Ellicott City

Dear Ms. Dryden:

The following is a title report for the above-referenced property (the "Property").

Fee simple title in the Property is good and marketable as vested of record in Philip D. Carroll (also known as Philip Carroll) and Camilla Carroll, as tenants in common, as follows:

- 1. Deed from Charles Carroll, Junior to Philip A. Carroll, dated February 24, 1923 and recorded among the Land Records of Howard County, Maryland (the "Land Records") in Liber H.B.N. No. 117, folio 119.
- 2. Last Will and Testament of Philip A. Carroll (deceased July 8, 1957, survived by his wife, Nina R. Carroll) dated May 9, 1946, as amended by a Codicil thereto dated October 16, 1953, admitted to probate by Howard County Register of Wills (Estate No. 2046); Item Second devised all real property and improvements to his wife, Nina R. Carroll, for life, with a power of appointment (to devise through her Will upon her death) over all property located in Howard County, Maryland (including the Property).
- Last Will and Testament of Nina R. Carroll (deceased February 11, 1989) dated June 4, 1979, as amended by a Codicil thereto dated November 28, 1986 and a Second Codicil thereto dated April 10, 1987, admitted to probate by the Howard County Register of Wills (Estate No. 7868); Item Third, as

062210.04

ļ

Ms. Charlotte R. Dryden June 22, 2010 Page 2

> amended, exercised the power of appointment granted by the Will of Philip A. Carroll and devised the Property to "such of the lineal descendants of my son, Philip Carroll, as shall survive me, in equal shares, <u>per stirpes</u>, as tenants in common", and the then-surviving lineal descendants of Philip Carroll were Philip D. Carroll (also known as Philip Carroll) and Camilla Carroll.

The Property is known as 3500 Manor Lane, Ellicott City, Maryland 21042, and is located in the Third Election District of Howard County, Maryland, containing 876.562 acres of land, more or less (per SDAT records). The Property is shown on Tax Map 23, Grid 10, Parcel 71, property tax account number 03-281779.

The Property is not encumbered by any open financing.

The Property is subject to the following exceptions and agreements:

- Right of Way dated October 5, 1908, from Thomas P. O. Donnell and James O. Donnell to W. Raymond Cross, recorded among the Land Records in Liber W.W.L.C. 87, folio 662.
- 2. Deed of Easement dated May 23, 1977, from Nina R. Carroll to the Maryland Historical Trust, recorded among the Land Records in Liber 826, folio 542.
- Right of Way Easement dated May 20, 1980, from Nina R. Carroll to Baltimore Gas and Electric Company, recorded among the Land Records in Liber 1047, folio 333.
- 4. Right of Way Easement dated March 29, 1981, from Nina R. Carroll to Baltimore Gas and Electric Company, recorded among the Land Records in Liber 1051, folio 683.
- 5. Deed of Preservation Easement dated October 4, 2006, by and among Philip Carroll and Cantilla Carroll, The Howard County Conservancy, Inc., and Howard County, Maryland, recorded among the Land Records in Liber 10295, folio 58.
- Plat entitled "Density Sending Plat, Property of Philip Carroll and Camilla Carroll", which Plat is recorded among the Land Records as Plat Number 18572.

062210.04

082210.04

Ms. Charlotte R. Dryden June 22, 2010 Page 8

ι,

•

•

 Plat entitled "Amended Plat of Easement, Sending Parcel, Property of Philip Carroll and Camilla Carroll", which Plat is recorded among the Land Records as Plat Number 19928.

Title was run through May 28, 2010.

Copies of the title documents, encumbrances and exceptions noted above are enclosed. If you have any questions, please do not hesitate to get in touch with me.

Sincerely, Talkin & Oh, LLP

5

By: Jonathan E. Greenstein Of Counsel

enclosures

LIBER 12722 FOLM 296

113

MBN. 117/113

exemined per.

Received for Record 28" Peb 1923 at 9 o'alcok A. H. Seme day recorded and

Hast B. Malforerk.

OHARLES CARROLL, .478.DO Duly Junior, Cancelled DEED TO Revenue FHILIP A. CARROLL, Sramps.

THIS DEED Made this 24th day of February, in the year mineteen hundred and twenty-three, by and between Charles Carroll, Junior, otherwise known as Charles Bancroft Carroll, unmarried, of Howard County, in the State of

Story with states and the balance of the states of the sta

Maryland, party of the first part, and Philip A. Carroll, of the City and State of ' New York, party of the second part.

WITNESSETH: that for and in consideration of the sum of seventy-sight thousand (dollars (\$78,000.) this day paid, the reactly whereof is hereby soknowledged, the said party of the first part doth hereby grant and convey unto the said party of the second part, his heirs and assigns, in fee simple.

All those two tracts or parcels of land situate and lying in Howafd County, in the Beato of Maryland, which were conveyed to Anite Carroll by two decds; the first from Charles Carroll and Caroline Carroll, his wife, dated June 7th, 1866, and recorded 'on the eighth day of June 1866 among the Land Records of Howard County in Liber W. -W. W. No. 25, folio 316; etc., and the second from Matthias Hammohd and wife, dated : the 24th day of November 1867, and recorded among the said Land Records in Liber W. 'W. W. No. 27, folio 411, etc.

Being the same two tracts or parcels of land which by deed dated the second day of Hay 1912, and recorded among the Land Records of Howard County in Liber W. W. L. O. 1 No. 93, folio 232 etc., were granted and convoyed by Hoyal Pholps Carroll and Marion L. Carroll, his wife, and Philip A. Carroll to Charles Carroll, for life, and after his doubt to Charles Carroll, Jr., the granter herein, for life, with limitations

over, but with full power to the said Charles Carroll during his life and to the HOWARD COUNTY CIRCUIT (COURT (Land Records) (MGA CE 53-108) HEN 117, p. 0113. Printed 06/22/2010. Image available as of 11/06/2003.

LINER 12722 FOLM297

114

said Charles Carroll, Jr., the grantor herein, after the death of the said Cherles Curroll, to sell, give or in any other menner dispose of said two tracts or percels of land or any part thereof, or any interest therein, free from any of the conditions or provisions in said deed set forth, provided, however, that should said Charles Carroll or Charles Carroll, Jr. desire to sell, give or otherwise dispose of soid property, or any part thereof or any interest therein, pursuant to and in exercise of said power, the same shall first be offered to the said Royal Pholps Carroll, if living, and if not, to the said Philip A. Carroll, in fee and unencumbered with the buildings and improvements thereon, for the sum of sixty thousand (\$60,000) dollars, and in addition therets such sum of money as shall represent the fair value of all permanent improvements which shall have been placed upon said property by Charles Carroll or Charles Carroll, Jr.; the said Charles Carroll and the suid Royal Fhelps Carroll having departed this life, the said Charles Carroll, Jr. being desirous of solling and the said Philip A. Carroll being desirous of buying, and it having been egreed as to the fair value of all permanent improvements which have been placed on said property by said Charles Carroll and Charles Carroll; Jr.

TOGETHER WITH THE buildings and improvements thereon, and all the rights, reads, Allars, veys, waters, watercourses, privileges, escements, appurtenances and advantages to the same belonging or in anywise apportaining.

TO HAVE AND TO HOLD said described property unto and to the proper use of the said Philip A. Carroll, his heirs and assigns, in fee simple.

AND the said Charles Carroll, Junior, does hereby covenant that he will warrant specially the property hereby granted and conveyed and that he will execute such further assurances thereof as may be necessary.

WITNESS the hand and seal of the said grantor.

E. L. Drake.

. .

WITNESS:

Oharles Carroll, Jr. (SEAL)

STATE OF MARYLAND, CITY OF BALTIMORE, SBI

I HEREBY OBRIIFY that on this 24th day of February 1923, before we the subscriber 'a Notary Public of the State of Maryland, in and for the Oity of Baltimore aforesaid, , personally appeared Obarles Carroll, Junior, the granter bereinbefore named, and acknowledged the aforegoing deed to be his act.

AS WITNESS my hand and noterial seal. (SEAL'S PLACE)

Harry L. Drake Notary Public.

Olork.

Hart B. Mal

; Ressived for Record 1" May 1923 at 9 o'clock A. M. Same day recorded and examined per.

HOWARD COUNTY CIRCUIT COURT (Land Records) (MSA CE 53-100) HBN 117, p. 0114. Printed (8/22/2010, Imago available as of

No. 2441 Gola	te of PHILIP A.	OARROLL					
Late of NEW YORK COL	<u>мтх, м. х. </u>	july 800.		D_57.,		lestate	
PERSONS TO WHOM LOTTERS WERD GRANTED	0000 1	URREUS		- + - +	BAUTIC		
ina R. Carroll	\$10,000. New Amsterday	a Casualty Co.	hark	el.Venge	y & G	ามสะ พูล	5
and bilip Carroll Executors in Marylar	iđi l	•	1904 Balt:	finoro 2,	Md.	al Bank Blo	1g., ·
3 East 70th, Street			J J . 1	B. R _a ndo	l Carı	coll	!
low Y _o rk, N. Y.	ι · μ	· · · · · ·	Ι.	# ++\$ 1		,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,	
DATE	ROCEEDINGS	NAME OF VOLU	 Me	LIDER	No.	VELIO C	
	nanan nan nan salah. Tana s	Wills	····		/*% 1978		
oct. 8 Certified Copy of 1 Petition for Letter	n Testementery in Mary	· · · ·		R.L.F. R.L.F.	11. 版	303 · 215	
" Bond of Erecutors		Adm. Bonda			8	455	ļ
" Oath of Executors	nd & Amond at tath	th R		R.L.P.	8	455	·
Urder approving bo	nd & Grant of Letters so to Greditors	Proceedings		R.L.F. R.L.P.	44 44	216 217 ·	
" Order to appraise		11		R.L.P.	44	217	1
" Order to appreise :		1 B		R.L.P.	44	217	
" Appndatment of Rgs lov. 26 Report of Notice to			,	R.L.P.	144 144	217	
" Information Report	•	Filed		,			
D _e c. 10 Order passing Inve	ntory - Real Estate	Proceedings		R.L.P.	144	276	ŀ
" Inventory - Real E	state ntory - Personal Estate	Investories		R.L.P.		259	
1958 . " . Inventory - Person	-	Procescinge, Inventorèes		R.L.P.	1 44	.276	
Nar. 11 Patition & Order t			•	H.L.P.	<u>ц</u> ц	373	ļ
Oot, 21 Order passing Admi		Н	•	R.L.P.	hh	596	1
				1	111		1
" Administration Acc	ount - First & Final	Adm. Account	б. ,	R.L.F.	18	173	
	ount - First & Final	Adm, Account	e,	1	18		
	ount - First & Final	Adm. Account	<i>б.</i>	1	18		- - - - -
	ount - First & Final	Adm, Account	ef. '	1	18		
	ount - First & Final	Adm, Account	ef. '	1	18		
	ount - First & Final	Αάπ, Αοςουπτ	.	1	18		· · ·
	ount - First & Final	Adm, Aosount		1			· · ·
	ount - First & Final	Ι Αάπ, Αοςουπτ	.	1			,
	ount - First & Final	Adm, Aosount	e , .	1			
	ount - First & Final	Adm, Aocount		1			
	ount - First & Final	Ι Αάπ, Αοσουπτ		1			an darama ana ana ana ana a na ana ana ana ana ana
	ount - First & Final	Adm, Aocount	€.	1			
	ount - First & Final	Adm, Aocount	€.	1			
	ount - First & Final	Adm, Aosount	€.	1			
	ount - First & Final	Adm, Aocount	€.	1			
	ount - First & Final	Adm, Aocount	€.	1			
	ount - First & Final	Adm, Aocount	€.	1			
	ount - First & Final	Adm, Aocount	€.	1			
	ount - First & Final	Adm, Aocount	€.	1			
	ount - First & Final	Adm, Aocount	€.	1			

•

:

HOWARD COUNTY CIRCUIT COURT (Land Records) MDR 12722, p. 0298, MSA_CE53_12713. Date available 10/01/2010. Printed 06/04/2020.

•

•

. ,

•

 $(1)^{'}$

All those tracts of land which the said Philip A. Cerroll, late of New York City, owned situate and lying in the Second and Third Election Districts of Howard County binding on Carroll's Lane, Paul's Lane and the east side of Vineyard Road and known as Doughoregan Manor; 776 acres of which were devised to him under the Last Will and Testament of John Lee Carroll, late of Howard County, State of Maryland, of record in Liber RD of W No. 5, Folio 81, one of the Wills Records of Howard County aforesaid; 12042 acres of which was granted and conveyed unto Philip A. Carroll by deed dated the 24th day of February, 1923, and recorded among the Land Records of Howard County in Liber HBN No. 117, Folio 113 sto., from Charles Carroll, Jr.; 128 acres more or less of which was granted and conveyed unto Philip A. Carroll by deed dated the 15th. day of September, 1939, was recorded among the Land Records of Howard County in Liber BM Jr. No. 164, Folio 403, from Thomas P. O'Donnell and Ada D. O'Donnell, his wife, and five acres of which was granted and conveyed unto Philip A. Carroll by deed dated the 13th day of April, 1948, and recorded among the Land Records of Howard County in Liber MWB No. 203, Folio 483, from John Yarrow Eccles, executor. The said four tracts contained 2113 acres more or less, all of which said lots are adjoining. \$400,000.00

(.2.)

All' that farm in the Fifth Election District which the seid Philip A. Carroll owned, binding on Homewood Road and Clarksville Pike (old route 29) and known as the "Benedict Farm" containing 320 acres which were devised to him under the Last Will and Testament of John Lee Carroll, late of Howard County, State of Maryland, and recorded among the Wills Records of Howard County in Liber RD of W No. 5, Folio 81. (3)

All that tract of wood land in the Fifth Election District of Howard County which the said Philip A. Garroll owned, having no frontage on any road, but having an easement or right of way one-half mile long over the lands of R. G. Harper Carroll II, to Clarksville Pike near tract No. (2) above, containing 60 acres which was devised to Philip A. Carroll under the Last Will and Testament of John Lee Carroll, late of Howard County, State of Maryland, and recorded among the Wills Records of Howard County in Liber RD of W No. 5, Folio 81. (2) and (3) \$50,000.00 HOWARD COUNTY CIRCUIT COURT (Land Records) MDR 12722, p. 0301, MSA_CE53_12713. Date available 10/01/2010. Printed 06/04/2020.





HOWARD COUNTY CIRCUIT COURT (Land Records) MDR 12722, p. 0302, MSA_CE53_12713. Date available 10/01/2010. Printed 06/04/2020.



HOWARD COUNTY CIRCUIT COURT (Land Records) MDR 12722, p. 0304, MSA_CE53_12713. Date available 10/01/2010. Printed 06/04/2020.



LIBER | 2722 FOLIO 304



LIBER | 2722 FILLIO306



HOWARD COUNTY CIRCUIT COURT (Land Records) MDR 12722, p. 0306, MSA_CE53_12713. Date available 10/01/2010. Printed 06/04/2020.

HOWARD COUNTY CIRCUIT COURT (Land Records) MDR 12722, p. 0307, MSA_CE53_12713. Date available 10/01/2010. Printed 06/04/2020.





HOWARD COUNTY CIRCUIT COURT (Land Records) MDR 12722, p. 0308, MSA_CE53_12713. Date available 10/01/2010. Printed 06/04/2020.













2722 mm314

Linex

	PERSONAL REPRES	ENTATIVE	•	,		•	
,				·		• •	
e of <u>hey yoks, hey y</u>	ORX	died	<u>EEBBIIARY 1</u>	الرفق في الم	testate		
is to whom letters A Wine granted o	Luount 19 Hond	SURETTES		. 'sol	ICITORS	•	
, CARROLL A TER CARROLL ZJEGLER CARROLL ZJEGLER ERT BROTHERS AVENUE	e/a n/a	•		JOHN C. NUR 518 H. Char Salttwore, (301) 625-4	les Street Norvland 21	2 Q	
. HEU YORK 10136				· ·			
·	• •	••		•			
PROCES	sdings .		NAME OF VOLUM	n			
			HALLS OF YOLUM.	e liber	No. FOLID		
HILL - CERTIFIED COPY INO CODICILS (CERTIFIED COPY OF PI EXEMPLIFIED COPY OF PI) ED·CÓPIES)	ADH	PROCEEDING	5	·		
INTELAT ADDRULA	THENS	:	4 2 .	•			
RESIDENT ABERT FORNS A AND HARY CARTER			. .		•		
LIST OF RECIPIENTS OF NOTICE TO CAEDITORS	HARYLAND PROPER	нт г , •	• • •	· ·		•	
APPRAISAL OF HARYLAND	455E15	4		• •		•	
NARYLAND ESTATE TAX RE		•	•	1			
•	•	-	•	Marcin	· · · · · ·		
mended Meryland Estate	Tax Neturn ·			1	OFICHE	•	
				L.JAC	KETE	1	
application by Foreign Interitance Tax	Per. Rep to Fi	· ·			- I N .		
HEWALLOGICO TAX		ж ,		· · · · · · · · · · · · · · · · · · ·			
Notice of Inchoate Lien	1	,		· · · ·		· · ·	
	INS- UNITED STAT PERTY SUBJECT TO ES	,	N	· · · · · · · · · · · · · · · · · · ·		· ·	
Notice of Inchoate Lien ROM DEPT OF TRUBSURY-I ERTIFICATE DISCHARGING PROF	INS- UNITED STAT PERTY SUBJECT TO ES	,	N			· ·	
Notice of Inchoata Lien ROM DEPT OF TRUASURY-I ERTIFICATE DISCHARGING PROM Maria I. N	IRS~ UNITED STAT SERTY, SUBJECT TO ES 	es. State tax lite	•	· · · · · · · · · · · · · · · · · · ·		· ·	
Notice of Inchoata Lien ROM DEPT OF TRUASURY-I ERTIFICATE DISCHARGING PROM Maria I. N	IRS~ UNITED STAT SERTY, SUBJECT TO ES 	es. State tax lite	•			· ·	
Notice of Inchoata Lien ROM DEPT OF TRUASURY-I ERTIFICATE DISCHARGING PROM Maria I. N	IRS~ UNITED STAT SERTY, SUBJECT TO ES 	es. State tax lite	•	· · · · · · · · · · · · · · · · · · ·		· · · ·	
Notice of Inchoata Lien ROM DEPT OF TRUASURY-I ERTIFICATE DISCHARGING PROM Maria I. N	IRS~ UNITED STAT SERTY, SUBJECT TO ES 	es. State tax lite	•			· · · · ·	•
Notice of Inchoata Lien ROM DEPT OF TRUASURY-I ERTIFICATE DISCHARGING PROM Maria I. N	IRS~ UNITED STAT SERTY, SUBJECT TO ES 	es. State tax lite	•			· · · · ·	•
Notice of Inchoata Lien ROM DEPT OF TRUASURY-I ERTIFICATE DISCHARGING PROM Maria I. N	IRS~ UNITED STAT SERTY, SUBJECT TO ES 	es. State tax lite	•			·	•
Notice of Inchoata Lien ROM DEPT OF TRUASURY-I ERTIFICATE DISCHARGING PROM Maria I. N	IRS~ UNITED STAT SERTY, SUBJECT TO ES 	es. State tax lite	•			· ·	•
Notice of Inchoata Lien ROM DEPT OF TRUASURY-I ERTIFICATE DISCHARGING PROM Maria I. N	IRS~ UNITED STAT SERTY, SUBJECT TO ES 	es. State tax lite	•	· · · · · · · · · · · · · · · · · · ·			•
Notice of Inchoata Lien ROM DEPT OF TRUASURY-I ERTIFICATE DISCHARGING PROM Maria I. N	IRS~ UNITED STAT SERTY, SUBJECT TO ES 	es. State tax lite	•				•
Notice of Inchoata Lien ROM DEPT OF TRUASURY-I ERTIFICATE DISCHARGING PROM Maria I. N	IRS~ UNITED STAT SERTY, SUBJECT TO ES 	es. State tax lite	•				•
Notice of Inchoata Lien ROM DEPT OF TREASURY-I ERGIFICATE DISCHARGING PROM Marine L Him	IRS UNITED STAT EERTY, SUBJECT TO ES 4 1	es. State tax lie 	٧,	•		· . · ·	
Notice of Inchoata Lien ROM DEPT OF TREASURY-I ERGIFICATE DISCHARGING PROM Marine L Him	IRS UNITED STAT EERTY, SUBJECT TO ES 4 1	es. State tax lie 	٧,	•		· . · ·	
Notice of Inchoata Lien ROM DEPT OF TREASURY-I ERGIFICATE DISCHARGING PROM Marine L Him	IRS UNITED STAT EERTY, SUBJECT TO ES 4 1	es. State tax lie 	٧,	•		· . · ·	
Notice of Inchoata Lien ROM DEPT OF TREASURY-I ERGIFICATE DISCHARGING PROM Marine L Him	IRS UNITED STAT EERTY, SUBJECT TO ES 4 1	es. State tax lie 	٧,	•		· . · ·	
Notice of Inchoata Lien ROM DEPT OF TREASURY-I ERGIFICATE DISCHARGING PROM Marine L Him	IRS UNITED STAT EERTY, SUBJECT TO ES 4 1	es. State tax lie 	٧,	•		· . · ·	
Notice of Inchoata Lien ROM DEPT OF TRUASURY-I ERTIFICATE DISCHARGING PROM Maria I. N	IRS UNITED STAT EERTY, SUBJECT TO ES 4 1	es. State tax lie 	٧,	•		· . · ·	
Notice of Inchoata Lien ROM DEPT OF TREASURY-I ERGIFICATE DISCHARGING PROM Marine L Him	IRS UNITED STAT EERTY, SUBJECT TO ES 4 1	es. State tax lie 	٧,	•		· . · · · · · · · · · · · · · · · · · ·	

•

:

:

.

HOWARD COUNTY CIRCUIT COURT (Land Records) MDR 12722, p. 0315, MSA_CE53_12713. Date available 10/01/2010. Printed 06/04/2020.

.

•

10ER 1 2 7 2 2 FOL 10 3 1 6

I, NINA RYAN CARROLL, a resident of the City, County and the State of New York, do make, publish and declare this my LAST WILL AND TESTAMENT, hereby revoking all former Wills and Codicils made by we.

FIRST

By Article TENTH of an Indenture of Trust dated February 18, 1942, between DOROTHY H. RYAN as Grantor, and me, NINA M. RYAN, as Trustee, I am vested with the power to appoint a Successor Trustee to succeed me in the event of my death prior to the termination of the Trust, with all of the rights, duties, powers and immunities conferred upon the Trustees therein by the said Trust Indenture. Fursuant to such power, I hereby appoint my son, PHILIP CARROLL, as Successor Trustee to me with all of the powers, rights and duties possessed by me as Trustee.

SECOND

4642

A. Since my son PHILIP CARROLL now owns the tangible contents of Doughoregan Manor in Howard County, Maryland and is now farming the surrounding acres, I give and bequeath to him, but not by way of limitation, all farm and garden implements, equipment, livestock, and produce, together with all other tangible personal property there located used in connection with the farming operation. But if he is not

'LIBER 12722 FOLD317

1、11、1211、124 - 21 - Alia 用刘金的花时和如志的过去形式,在闲待到台级的12;

living then I give and bequeath the same to his son, PHILIP D. CARROLL: but if he is not living then to the Trustees of the Carroll Trust which is hereinafter provided for in subparagraph (c) of paragraph II of Article Third.

B. I give and bequeath the contents of my apartment at 33 Bast 70th Street, New York, N.Y. as follows:

> 1. To my son, PHILIP CARROLL, the two Shakespeare pictures, now in the living room, constituting part of a set at The Manor and all blue, white, gold and red china compotes, dishes, plates, urns, etc., constituting part of a set now at The Manor.

> To my son, JOHN LEE CARROLL, the painting of The Munor by Falix Kelly now in the library of my apartment.

 To my children, PHILIP CARROLL, JOHN LEE GARROLL and MARY CARTER CARROLL ZIEGLER, all my furs, jewelry, clothing and other personal effects, to be allocated among them as they should agree among themselves.

All the rest and remainder of the contents of my apartment, including, but not by way of limitation, the silverware, glass, china, furniture, pictures, rugs, books, ornaments and all other household goods and furnishings to my children, JOHN LEE CARROLL and MARY CARTER CARROLL ZIEGLER, to be allocated between them as they should agree.

In the event that any child of mine should predecease me, leaving children him or her surviving, then anything in this paragraph B to the contrary notwithstanding, such children shall receive and share equally the articles which my child would have received had he or she survived me.

~7.

4643

4.

有利用的结构是完全的非常常的的原则和自己的结构和自己的中心。 1111月

LIBER 12722 FOLD318

THIRD

In and by Article SECOND of the Last Will and Testament of my late husband, PHILIP A. CARROLL, dated May 9, 1946 together with Article FIRST of the Codicil thereto dated October 16, 1953, both of which were duly admitted to probate in the Surrogate's Court of New York County, New York, I was given a life estate in all of his real property situated in Howard County, Maryland, with all improvements thereon together with a general power to appoint the said property by Deed or Will.

Pursuant to the aforesaid power vested in me, I hereby appoint any and all of the aforesaid CARROLL lands and real property situated in Howard County, Maryland, together with any and all buildings and improvements thereon and appurtenances thereto, and any and all money or property representing the proceeds of any sale, mortgage or exchange of any such real property, and any investment or reinvestment thereof

as follows: **1.** I appoint Doughoregan Manor House together with about seven hundred and fifty (750) acres, more or less, of the surrounding lands (said house and surrounding lands being hereinafter referred to as "The Manor"), and all buildings and improvements thereon and all appurtenances, constituting that part of the real property lying north of the center line of Paul's Lane extending to the easterly

-3.

HOWARD COUNTY CIRCUIT COURT (Land Records) MDR 12722, p. 0318, MSA_CE53_12713. Date available 10/01/2010. Printed 06/04/2020

٠.,

LIBER | 2722 FOLIO3 | 9

boundary of said property, being that portion of said property which was conveyed to PHILIF A. CARROLL, by CHARLES CARROLL, JR., by Deed, dated February 24, 1923, and recorded in the land records of Howard Gounty on March 1, 1923, in Liber H. H. No. 117, Folio 113, to my son, PHILIP CARROLL, if living, or if he does not survive me, to his son, PHILIP D. CARROLL, if living, or, if PHILIP D. CARROLL does not survive me, to my son, JOHN LEE CARROLL, and my daughter, MARY CARTER CARROLL ZIEGLER, as trustees in trust however to hold and administer as provided in sub-paragraph (c) of paragraph II of this Article Third known as The Carrol1 Trust.

II. All of the remaining lands and real property in Howard County, Maryland over which I have the aforesaid power of appointment together with all buildings and improvements thereupon erected and all appurtenances and any and all money or property representing the proceeds of any sale, mortgage or exchange of any such property, or any investment or reinvestment thereof and I appoint as follows:

> (a) I appoint an undivided one-fourth interest therein to my daughter, MARY CARTER CARROLL ZIEGLER, for her own use absolutely, if living, but if she is not living, I appoint the said interest to the trustees of The Ziegler Trust provided for in the following sub-paragraph (b) as an addition to that trust.

HOWARD COUNTY CIRCUIT COURT (Land Records) MDR 12722, p. 0319, MSA_CE53_12713. Date available 10/01/2010. Printed 06/04/2020.

4645

LIDER 12722 FOLIO320

^국리 분야하 김,

《開創構業法語等的構成的情報的結果的"相比"。 I appoint an undivided one-fourth interest (b) to my son, JOHN LEE CARROLL, and my daughter, MARY CARTER CARROLL ZIEGLER, as trustees of a trust (The Ziegler Trust) in trust nevertheless to hold, manage, invest and reinvest the same and to collect the income therefrom and to pay or apply the net income at least quarterly to, or to the use of, such members of a class composed of the children of my daughter, MARY CARTER CARROLL ZIEGLER, in such amounts and proportions as my trustees, in their discretion, shall determine from time to time. In disbursing the income the trustees, in their uncontrolled discretion, may pay to, on apply the same to the use of, any one member of the class or apportion it for the . benefit of various members of the class to the exclusion of other members in such manner as they shall, from time to time, deem advisable without equality of treatment. Any net income not so distributed shall be added to the principal.

> Upon the death of the last surviving descendant of my husband, PHILIP A. CARROLL, living at the time of his death in 1957, or upon the twentieth anniversary of my death, whichever event first occurs, or upon the death of the last survivor of the children of MARY CARTER CARROLL ZIEGLER prior

4646

HOWARD COUNTY CIRCUIT COURT (Land Records) MDR 12722, p. 0320, MSA_CE53_12713. Date available 10/01/2010. Printed 06/04/2020.

LIBER 2722 FOLID 321

anishterasion territori anistation anitalishtes anit

thereto; this trust shall terminate and the trustees shall deliver and pay the principal then remaining in equal shares to each of the children of MARY CARTER CARROLL ZIEGLER then living, if any, and to the estate of each deceased child of MARY CARTER CARROLL ZIEGLER.

(c) In the event neither my son, PHILIP CARROLL, nor his son, PHILIP D. CARROLL, shall survive me I appoint an undivided one-half interest to my granddaughter, CAMILLA CARROLL, for her own use absolutely or to her estate if she should predecease me, but if either my aforesaid son or grandson should survive me, then in either such event the above appointment to my granddaughter, CAMILLA CARROLL shall be cancelled and shall be of no effect and in lieu thereof I appoint the aforesaid undivided one-half interest to my son, JOHN LEE CARROLL, and my daughter, MARY CARTER CARROLL ZIEGLER, as trustees of a trust (The Carroll Trust) in trust nevertheless to hold, manage, invest and reinvest the same to collect the income therefrom, and to pay or apply the net income therefrom at least quarterly to, or to the use of, such members of a class composed of the children of my son, JOHN LEE CARROLL; in such amounts and proportions as my trustees, in their discretion shall determine from time to time. In disbursing the income the Trustees,

6-

HOWARD COUNTY CIRCUIT COURT (Land Records) MDR 12722, p. 0321, MSA_CE53_12713. Date available 10/01/2010. Printed 06/04/2020.

47
LIBER 2722 FOL10322

13.114910

in their discretion may pay to, or apply the same to the use of, any one member of the class or apportion it for the benefit of various members of the class to the exclusion of other members in such manner as they shall, from time to time, deem advisable without equality of treatment. Any net income not so distributed shall be added to principal.

> Upon the death of the last surviving member of the descendants of my husband, PHILIP A. CARROLL, living at the time of his death in 1957 or upon the twentieth anniversary of my death, whichever event first occurs, or upon the death of the last survivor of the children of JOHN LEE CARROLL prior thereto, this trust shall terminate and the Trustees shall deliver and pay the principal then remaining in equal shares to each of the children of JOHN LEE CARROLL then living if any and to the estate of any deceased child of JOHN LEE CARROLL, provided however that if The Manor be included in said principal then and in that event The Manor shall be conveyed to such male child or children of JOHN LEE CARROLL as the trustees shall designate and to be charged against the distribution share or shares of said male or males so designated.

FOURTH

A large portion of the Maryland real property subject to

4648

LIBER 12722 FOLD 323

(期間将需要因為不均均均均均均均均均均均均均均均分子。)。

my aforesaid power of appointment is now subject to an easement granted by me to Maryland Historical Trust. Since all of the various parcels of real property subject to my power of appointment are contiguous or nearly so, and have been and now are being maintained and operated as a unit primarily as a farm, it is my wish that the beneficiaries under paragraph II of Article Third herein coordinate their efforts with those of the owner of Doughoregan Manor to the extent that the law allows so that the operation of the various parcels will be devoted to the same enterprise if possible and as much as possible.

, <u>FIFTH</u>

I appoint any and all property over which I may have a power of appointment by Will under Article Fifth of the last Will and Testament of my husband, the late PHILIP A. CARROLL, into three equal shares as follows:

I. I appoint one such share to my son, PHILIP CARROLL, absolutely, if he survives me, but if he does not, then to his son, PHILIP D. CARROLL and his daughter, CAMILLA CARROLL, in equal shares but if either predeceases me leaving descendants, such descendants shall take the share which my grandchild would have received had he or she survived me, but if there be none, to his or her estate.

LIBER 12722 FOL10324

腳

ł

II. I appoint one such share to my son, JOHN LEE CARROLL, absolutely, if he should survive me, or if he should predecease me, to his descendants me surviving, in equal shares per stirpes.

> III. I appoint one such share to my daughter, MARY CARTER CARROLL ZIEGLER, absolutely, if she should survive me, or if she should predecease me, to her descendants me surviving, in equal shares per stirpes.

IV. If the appointment of any share of the property over which I have a power of appointment by Will under Article Fifth of said Will of said PHILIF A. CARROLL should lapse or fail then I appoint such lapsed or failed share to the persons entitled to receive the balance of such property under the foregoing provisions of this Article in the proportions which their respective shares bear to each other.

4650

SIXTH

ALL THE REST, RESIDUE AND REMAINDER of my estate of whatsoever kind and wheresoever situated, I give, devise and bequeath as follows:

A. One third (1/3) thereof to my son, PHILIP CARROLL, if living, but if he is not living, then to his son, PHILIP D.

LIBER | 2722 FOLM325

CARROLL and his daughter, CAMILLA CARROLL, in equal shares, but if either of them be not living, then his or her share shall go to his or her descendants if any, me surviving and if none, then to his or her estate.

B. One-third (1/3) thereof to my son, JOHN LEE CARROLL if living, but if he is not living, to his descendants me surviving in equal shares per stirpes.

C. One-third (1/3) thereof to my daughter, MARY CARTER CARROLL ZIEGLER, if living, but if she is not living, then to her descendants me surviving in equal shares per stirpes.

D. Should any child of mine predecease me, leaving no descendants me surviving, then the share which the child would have received had he or she survived me, I give to my descendants who survive me in equal shares per stirpes.

SEVENTH.

I direct that all estate, inheritance and succession taxes including interest assessed against my estate by federal or state governments with respect to property required to be included in my gross tax estate passing by this Will or otherwise shall be charged to and paid out of the principal of my residuary estate without contribution thereto from the persons benefited, but if such taxes exceed my residuary estate after the payment of debts and administration expenses, then and in that event I

-10-

651

いせんないない

direct that the excess tax be charged to the beneficiaries under paragraph II of Article Third herein in proportion to the value of their respective benefits received, such value to be based upon that value assessed in the federal estate tax proceeding in my estate.

EIGHTH

I hereby appoint my sons, PHILIP CARROLL and JOHN LEE CARROLL, and my daughter, MARY CARTER CARROLL ZIEGLER, to be executors of this my Will and I direct that no bond or other security be required of any of them as executors or as trustees of any trust that may come into existence under Article Tenth of this Will.

In and by Article Third I have appointed my son, JOHN LEE CARROLL, and my daughter, MARY CARTER CARROLL ZIEGLER, as trustees of the trusts created therein. I direct that neither of them shall be required to file a bond or other security as trustee for the performance of his and her duties as such.

In the event of the death, resignation or failure to qualify of any of my aforesaid children as executor or trustee 553 I authorize and empower that child to appoint his or her own authoritude trustee or anecessor trustee. In the absence of such appointment i authorize and empower the other executors or trustees to appoint a substitute or successor executor or trustee as the case may be. Any appointments provided for

-11-

LIBER 1 2 7 2 2 FOL 10 3 2 7 .

herein shall be made in writing and shall be filed in the court where this Will is probated.

NINTH

and the second states of the second states and the second states of

In addition to the powers vested in them by law, I grant to my executors and to my trustees (as to the latter in addition to the powers set forth in Article FOURTH herein), with respect to any and all property, whether real or personal belonging to me at the time of my death or which may at any time be held by them hereunder, the following powers which may be exercised at any time during the administration of my estate, the continuance of any trust hereunder until actual distribution of all property in their hands:

(1) To retain such property as an investment without regard to the proportion scuh property or property of a similar character, so held, may bear to the entire amount of my estate, or of the trust in which such property is held, and whether or not the same be within the class of securities in which Trustees are authorized by law or any. rule of court to invest trust funds.

(2) To sell such property at either public or private sale for cash or on credit, to exchange such property, and to grant options for the purpose thereof.

-12-

4653

HOWARD COUNTY CIRCUIT COURT (Land Records) MDR 12722, p. 0327, MSA_CE53_12713. Date available 10/01/2010. Printed 06/04/2020.

nughanyanyanyanyanyanyanyanyanya

(3) To invest and reinvest in any property, in which a reasonably prudent person would invest, including but not by way of limitation, bonds, notes, debentures, mortgages, certificates of deposit, common and preferred stocks, and shares of interests in investment trusts, without being limited to the class of securities in which trustees are authorized by law or any rule of court to invest trust funds.

(4) To consent to and participate in any reorganization, consolidation, merger, combination, dissolution, change of capitalization or other changes of corporate structure, or any plan for any such change, and to consent to any contract, lease, mortgage, purchase, sale or other action by any corporation pursuant thereto.

4654

(5) To deposit any such property with any protective reorganization or similar committee, to delegate discretionary power thereto, and to pay part of its expenses and compensation and any assessments levied with respect to such property.

-13-

LIBER 2722 FOLM329

11111111111

(6) To exercise all conversion, subscription, voting and other rights of whatsoever nature pertaining to any such property, and to grant proxies, discretionary or otherwise, in respect thereto.

4655

HOWARD COUNTY CIRCUIT COURT (Land Records) MDR 12722, p. 0329, MSA_CE53_12713. Date available 10/01/2010. Printed 06/04/2020.

(7) To manage any such real property as if the absolute owner thereof, including but not by way of limitation the power to lease, or grant options to lease the same for any term or terms, although in excess of five years, and although any such terms may extend beyond the period of the trust under this my Will, without application to any court, and to enter into any covenants or agreements relating to the property so leased or any improvements which may then or thereafter be created thereon.

(8) To borrow money, and to pledge or mortgage any such property, for any purpose.

(9) To satisfy any general legacy or any part thereof, including any legacy to themselves as Trustees of the trusts under this Will, in kind; and in the division or distribution of my estate or the trusts under this Will, or any part thereof, including any distribution or payment

-14-

to themselves as Trustees of any trust under this Will, to make partition, division or distribution of property, or such payment, in kind, and for any such purpose to determine the value of any such property, which determination shall, to the extent permitted by law, be conclusive and not subject to review.

(10) To apportion between income and principal charges or expenses incurred by, or taxes assessed against my estate or the trusts hereunder, or to allocate the same wholly to principal or wholly to income, and any such apportionment or allocation shall, to the extent permitted by law, constitute full protection to my said Executors and Trustees with respect to any action taken or payments made in reliance therein.

(11) To appoint agents to set in his or their behalf, and to delegate discretionary powers to such agents.

(12) To extend the time of payment of any obligation at any time owing by or to my estate, or the trusts under this Will and to compromise, settle or submit to arbitration upon such terms as they may deem advisable, or to release with or without consideration, any claim in favor of

-15-

1001-1824897052

4656

LIBER | 2722 FOL1033 |

or against my estate or the trusts under this Will,

(13) To do all such acts, take all such proceedings and exercise all such rights and privileges, although not hereinbefore specifically mentioned, with relation to any such property as if the absolute owner thereof, and in connection therewith to make, execute and deliver any instrument and to enter into any covenants or agreements binding my estate or the trusts under this Will.

In distinguishing between income and principal hereunder, my Trustees shall not be required to amortize premiums paid for trust securities, and all stock and extraordinary dividends shall be treated as principal and added to the trust fund so far as permitted by law.

In the administation of my estate and of the trust hereunder, my Executors and my Trustees are hereby authorized to deposit any securities held by him or them in a custodian account in a bank or trust company, and to cause any such securities so deposited by them to be registered and held in the name of the nominee of such depositary without words indicating fiduciary capacity, and to pay the custodian fees and other charges of such bank or trust company for the custody of such securities, the collection of income and other

HOWARD COUNTY CIRCUIT COURT (Land Records) MDR 12722, p. 0331, MSA_CE53_12713. Date available 10/01/2010. Printed 06/04/2020.

LINER | 2722 FOLIO332

custodian services, and to charge the same against income of the estate, or of the trust as the case may be, and I direct that no part of such charge shall be deducted from or applied against the compensation of my individual Executors or my individual Trustees hereunder.

No one dealing with my Executors or my Trustees shall be obliged to look to the application of any moneys or other properties paid or delivered to my Executors or my Trustees.

<u>tenth</u>

If at any time any or all the principal of my rewiduary estate or any property, real or personal, appointed by any other Article of this my Will would, but for the provisions of this paragraph, vest in a minor, then anything hereinabove contained to the contrary notwithstanding, I give, devise, bequeath and appoint such property to my sons, PHILIP CARROLL and JOHN LEE CARROLL, and my daughter, MARY CARTER CARROLL ZIEGLER, as Trustees, IN TRUST, to hold, manage, invest and reinvest the same, to collect and receive the income therefrom and of the principal thereof as they in their absolute discretion shall deem advisable for the

-17-

3

LIDER 2722 FOLD 333

support, maintenance and education of such minor, and to accumulate any income not so applied until such minor shall attain the age of twenty-one years; and upon such minor attaining such age, my Trustees shall transfer and pay over to such minor the principal and accumulated income then in their hands hereunder; or in the event of the death of such minor prior to attaining the age of twenty-one years, my Trustees shall thereupon transfer and pay over the principal and accumulated income then in their hands hereunder to the personal representative of such minor.

In any case, where my Trustees are authorized to make payment to a minor, my Trustees may make application of income or principal to or for the use of such minor either by payment directly to such minor or by payment for the benefit of such minor; to the parent of such minor or to the guardian of such minor or to the person with whom such minor may reside, or by paying the expenses of such minor, to such minor and the receipt of such minor, parent or guardian or such other person for all payments so made or other evidence of such expenditures shall be sufficient to discharge my Trustees from further liability for the amount so paid or expended.

-18-

HOWARD COUNTY CIRCUIT COURT (Land Records) MDR 12722, p. 0333, MSA_CE53_12713. Date available 10/01/2010. Printed 06/04/2020.

5

ø

27.22 10.0334

ELEVENTH

Except as expressly provided herein by Article THIRD and FIFTH it is my intention not to exercise by this Will any power of appointment by Will which I may have at the time of my death.

IN WITNESS WHEREOF, I have hereunto set my hand and seal this 4th day of June 1979.

(L.S.) Adda 1 11 day

THE FOREGOING INSTRUMENT, was, on the day of the date thereof, signed, sealed, published and duclared by NINA RYAN CARROLL, the Testatrix therein named, as and for her LAST WILL AND TESTA-MENT, in the presence of us, who at her request, in her presence, and in the presence of each other, have hereunto subscribed our names as witnesses.

HOWARD COUNTY CIRCUIT COURT (Land Records) MDR 12722, p. 0334, MSA_CE53_12713. Date available 10/01/2010. Printed 06/04/2020.

esiding at) residing at residing at 06965 -19-INTERNAL ADMITTED IN PROMATE CA A TELE COPYL O colicily CLE?N September

1989

I, Nina Ryan Carroll, a resident of the City, County and State of New York do make, publish and declare this as a Codicil to my Last Will and Testament dated June 4, 1979.

LIBER' 1 2722 FOLD335

FIRST: I hereby revoke paragraph I of Article Third of my will and declare in its place and stead the following:

(a) I appoint Doughoregan Manor House together with about seven hundred and fifty (750) acres, more or less, of the surrounding lands (said house and surrounding lands being hereafter referred to as "The Manor"), and all buildings and improvements thereon and all appurtenances, constituting that part of the real property lying north of the center line of Paul's Lane extending to the easterly boundary of said property, being that portion of said property which was conveyed to PHILIP A. CARROLL, by CHARLES CARROLL, JR., by Deed, dated February 24, 1923, and recorded in the land records of Howard County on March 1, 1923, in Liber H. H. No. 117, Polio, to my grandson Philip D. Carroll and my granddaughter Camilla Carroll in equal shares as tenants in common. It is my wish that in the event that my granddaughter at anytime be vested with sole ownership of The Manor she will observe the Carroll family tradition that she exchange with some other male Carroll descendant her interest in The Manor for other Howard County realty of then-equal value owned by such other male

4661 (b) If neither grandchild survives me, then in that event I appoint The Manor as hereinabove describe to my son Philip . Carroll and to his estate.

LIDER | 2722 FOLIO336

SECOND As thus amended I hereby ratify and confirm my said Will of June 4, 1979 and declare that the Will and this Codicil together constitute my Last Will and Testament. In Witness Whereof I have hereunto set my hand and seal this 27; day of 100100 , 1986. R. Langel 5.) THE FOREGOING INSTRUMENT, was, THE FOREGOING INSTRUMENT, was, on the day of the date thereof, signed, sealed, published and declared by NINA RYAN CARROLL, the Testatrix therein named, as and for a codicil to her LAST WILL AND TESTAMENT, in the presence of us, who at her request, in her presence, and in the presence of each other. in the presence of each other, have hereunto subscribed our names as witnesses. residing at residing at/763.2 4662 Seite T Silvertune residing at y 06900 Coderil CLETH September 21, 1989 CILL .

HOWARD COUNTY CIRCUIT COURT (Land Records) MDR 12722, p. 0336, MSA_CE53_12713. Date available 10/01/2010. Printed 06/04/2020.

TO LAST WILL AND TESTAMENT

SECOND CODICIL

2722 FOLD337

LIBER

Oŕ

NINA RYAN CARROLL

I, NINA RYAN CARROLL, domiciled in the City, County and State of New York, do hereby make, publish and declare this to be the Second Codicil to my Last Will and Testament dated June 4, 1979, as amended by my First Codicil dated November 28, 1986.

r.

I delete the last sentence of Paragraph A of Article SECOND of my said Will and substitute therefor the following: "But if my son, PHILIP CARROLL, shall not survive me, then I give and bequeath the same to such of his lineal descendants as shall survive me, in equal shares, per stirpes."

I hareby amend Article THIRD of my said Will, as amended by my said First Codicil, in its entirety; Article THIRD shall now read as follows:

II.

4663

۲Ĩ, ć,

"In and by Article SECOND of the Last Will and Testament of my late husband, PHILIP A. CARROLL, dated May 9, 1946, together with Article FIRST of the Codicil thereto dated October 16, 1953, both of which were duly admitted to probate in the Surrogate's Court of New York County, New York, I was given a life estate in all of his real property rituated in Howard County, Maryland, with all improvements thereon together with a general power to appoint the said property by Deed or Will.

LIBER 12722 FOL10338

eth salaret factor

"Pursuant to the aforesaid power vested in me, I hereby appoint any and all of the aforesaid CARROLL lands and real property situated in Howard County, Maryland, together with any and all buildings and improvements thereon and appurtenances thereto, and any and all money or property representing the proceeds of any sale, mortgage or exchange of any such real property, and any investment or reinvestment thereof as follows:

"I. I appoint Doughoregan Manor House together with about seven hundred and fifty (750) acres, more or less, of the surrounding lands (said house and surrounding lands being hereinafter referred to as 'The Manor'), and all buildings and improvements thereon and all appurtenances, constituting that part of the real property lying north of the center line of Paul's Lane extending to the easterly boundary of said property, being that portion of said property which was conveyed to Philip A. Carroll, by Charles Carroll, Jr., by Deed, dated February 24, 1923, and recorded in the land records of Howard County on March 1, 1923, in Liber H. H.

"A. If I shall die before January 1, 1990, to such of the lineal descendants of my son, PHILIP CARROLL, as shall survive me, in equal shares, <u>per stirpes</u>; as tenants in common, but if there be none, to my son, PHILIP CARROLL, if he shall survive me.

"B. If I shall not die before January 1, 1990, to my son, PHILIP CARROLL, if he shall survive me, but if he does not, to such of his lineal descendants as shall survive me, in equal shares, per stirpes, as tenants in common.

"II. I appoint all of the remaining lands and real property in Howard County, Maryland, over which I have the aforesaid power of appointment together with all buildings and improvements thereupon erected and all appurtenances and any and all money or property representing the proceeds of any sale, mortgage or exchange of any such property, or any investment or reinvestment thereof, as follows:

"A. An undivided one-fourth interest therein to my daughter, MARY CARTER CARROLL ZIEGLER if she shall survive me, but if she does not, to such of her lineal descendants as shall survive me, in equal shares, <u>per</u> <u>stirpes</u>, as tenants in common.

"B. An undivided one-fourth interest therein (i) if I shall die before January 1, 1990, to such of the lineal

4664

m. R.C.

的時间的自己的行动的行动。

HOWARD COUNTY CIRCUIT COURT (Land Records) MDR 12722, p. 0338, MSA_CE53_12713. Date available 10/01/2010. Printed 06/04/2020.

descendants of my daughter, MARY CARTER CARROLL ZIEGLER, as shall survive me, in equal shares, per stirpes, as tenants in common; or (ii) if I shall not die before January 1, 1990, to my daughter, MARY CARTER CARROLL ZIEGLER, if she shall survive me, but if she does not, to such of her lineal descendants as shall survive me, in equal shares, per stirpes, as tenants in common.

"C. An undivided one-half interest therain (i) if I shall die before January 1, 1990, to such of the lineal descendants of my son, JOHN LEE CARROLL, as shall survive me, in equal shares, <u>per stirpes</u>, as tenants in common; or (ii) if I shall not die before January 1, 1990, to my son, JOHN LEE CARROLL, if he shall survive me, but if he does not, to such of his lineal descendants as shall survive me, in equal shares, <u>per stirpes</u>, as tenants in common.

"III. Should any child of mine predecease me, leaving no descendant surviving me, then the share which the child would have received had he or she survived me, I give to my descendants who survive me, in equal shares, per stirpes.

"IV. In the event that distribution of the property described in Paragraph I of this Article would be deemed a generation skipping transfer for federal tax purposes, I hereby direct my executors to allocate to said property the One Million Dollars (\$1,000,000) exemption applicable to generation skipping transfers pursuant to Section 2631 of the Internal Revenue Code, as amended. If said distribution would not be deemed a generation skipping transfer for federal tax purposes or to the extent the exemption is not needed to prevent imposition of the generation skipping tax. I hereby direct my executors to allocate said One Million Dollars (\$1,000,000) exemption, or the remaining amount of the exemption, among any property described in Paragraph II of this Article deemed to be generation skipping transfers for federal estate tax purposes in proportion to the value of said property."

III.

I amend Article EIGHTH by revoking in its entirety the second paragraph of said Article. I amend the third paragraph of said Article by deleting from the first sentence the phrase, "substitute trustee or successor trustee", and

4665

*,*73

LINER 1 2722 FOLIO 340

substituting therefor the phrase; "substitute or successor executor or trustee".

IV.

I amend the first paragraph of Article NINTH by deleting the phrase "(as to the latter in addition to the powers set forth in Article FOURTH herein)".

v.

I amend the first paragraph of Article TENTH of my said Will by deleting the phrase, "appointed by any other Article of this Will", and substituting therefor the phrase, "appointed by Article FIFTH or SIXTH of this Will",

'VI.

I hereby revoke my First Codicil dated November 28, 1986, to my said Last Will and Testament in its entirety being an amendment to Article THIRD which has been wholly restated in this Second Codicil.

4666

"14444"

M.R.C.

С

, **v**'

VII.

As thus amended, I hereby ratify and confirm my said Will of June 4, 1979 and declare that the Will and this Second

ડોટી દેશિય

へいゃじゃ

L.S. j

Codicil together constitute my Lash Will and Testament.

In Witness Whereof I have hereunto set my hand and

NINA

RYAN

CARROLL

20

O Ced

ଏଟ

S-(

seal this 10 day of April, 1987.

THE FOREGOING INSTRUMENT, was, on the day of the date thereof, signed, sealed, published and declared by NINA RYAN CARROLL, the Testatrix therein named, as and for a codicil to her LAST WILL AND TESTAMENT, in the presence of us, who at her request, in her presence, and in the presence of each other, have hereunto subscribed our names as witnesses.

siding esiding al esiding at 10

466'7

with webils merupa (7

CLERK CERTING

STATE OF NEW YORK

COUNTY OF NEW YORK

Each of the undersigned, individually and severally, being duly sworn, deposes and says:

85.:

The within Codicil was subscribed in our presence and sight at the end thereof by NINA RYAN CARROLL, the within named testatrix, on the formattion formattion formattion for the state of

Said testatrix at the time of making such subscription declared the instrument so subscribed to be her Codicil.

Each of the undersigned thereupon signed his and her name as a witness at the end of said Codicil, at the request of said testatrix and in their presence and sight and in the presence and sight of each other.

Said testatrix was, at the time of so executing said Codicil, over the age of eighteen years, and, in the respective opinions of the undersigned; of sound mind, memory and understanding and not under any restraint or in any respect incompetent to make a Codicil.

Said testatrix, in the respective opinions of the undersigned, could read, write and converse in the English language and was suffering from no defect of sight, hearing or speech, or from any other physical or mental impairment which would affect her capacity to make a valid Codicil. The Codicil was executed as a single, original instrument and was not executed in counterparts.

Each of the undersigned was acquainted with said testatrix at such time, and makes this affidavit at her request.

The within Codicil was shown to the undersigned at the time this affidavit was made, and was examined by each of them as to the signature of said testatrix and of the undersigned.

The foregoing instrument was executed by said testatrix and witnessed by each of the undersigned affiants

under the attorney-a utk day of Apr Notax 章 言 (20) Formly / 31, 1923 aw veik to before ríl, 1987 STAJ no 1 me this $\dot{\sim}$ Ś ,÷

, EBBIJ

2722 min343

an

HOWARD COUNTY CIRCUIT COURT (Land Records) MDR 12722, p. 0343, MSA_CE53_12713. Date available 10/01/2010. Printed 06/04/2020.

ESTATE OF NINA R. CARROLL

The Maryland assets consist of tangible and intengible personal property. The list of recipients under the will and codicils and their relationship is as follows:

Son: Philip Carroll f.R. 3500 Manor Lane Ellicott City, Md. 21043

Son: John Lee Carroll P.R 215 East 72nd Street New York, New York 10021

Daughter: Mary Carter Carroll Zeigler f. R. P.O. Box 146 Free Union, Va. 22940

Grandchildreni

HOWARD COUNTY CIRCUIT COURT (Land Records) MDR 12722, p. 0344, MSA_CE53_12713. Date available 10/01/2010. Printed 06/04/2020.

Philip D. Carroll & Apt. 252 1670 El Camino Real Menlo Park, CA 94025

Camilla Carroll 4 USARU APO New York 09053

John Lee Carroll, Jr. Box 199 Queenstown, Md. 21658

Thomas T. Carroll 215 E. 72nd Street New York, New York 10021

Genevieve Anne carroll 215 E. 72nd Street New York, New York 10021

Natalie Ziegler Zirchky 920 Kipling Drive Atlanta, GA 30318

Jessica Ziegler Cardew Apt. 3 922 E. 77th Street New York, New York 10021

Sophie Zeigler 1050 Sanchez St. San Francisco, CA 94114

22

(ill ro Taild he gives it wanted these by the pennell 1.11-11. prot.ourg asigu Asiling variali septedin perus in issuid fui i which ~ and the second second second second second second second second second second second second second second secon Address and a second second second second second second second second second second second second second second Address and a second second second second second second second second second second second second second second Dout !! . urbaan up Knowladged ਭੂ WULTHOO MI. UNV itens sin ny PUTAL DAVE PLAN ווה היה היהי היהוע היה והי והי והי היה היהות היהוו אוה when a thread a point of the and the fight of t - solution 14 ž \sim Auto Ana ang salata Si Oli kident of vin the storest , Chu W ARMER (A A. West of A. around the spinning of the second 614,0006.20 Will Spanik All the bar τ. ANR REAL OF in or HD 4 ingsun . or they 1 1 1 1 સં ·.

LIBER | 2722 FOL10346

10

826 LIBEROB2C FOLIO542

MARYLAND HISTORICAL TRUST DEED OF EASEMENT

THIS DEED OF EASEMENT, made this 23 d day of May 1977, by and between NINE R. CARROLL, Grantor, and the MARYLAND HISTORICAL TRUST, Grantee,

*** WITNESSETH ***

WHEREAS, Grantem is a body corporate and instrumentality of the State of Maryland oreated for the purpose generally of preserving and maintaining historical, aesthetic and cultural properties, all as is more particularly provided by law; and

WHEREAS, Grantor is the owner, pursuant to the Mill of Philip A. Carroll, dated May 9. 1946, and a Codicil therato dated October 16, 1953, of a life estate and a general power of appointment in improved real property known as Doughoregan Manor, located in the Second and Third Election Districts of Howard County, State, of Maryland, which property is hereinafter referred to as the Property and is more particularly described below; and

MERRAS, the Property, including all the land covered by this ensement, has substantial historic, aesthetic and cultural character and importance, and this easement will promote the preservation and maintenance of the Property and its historic aesthetic and cultural character; and

WHENEAS, Grantee has determined that this easement is exclusively for conservation purposes; and WHEREAS, Grantee is possessed with the power and duty to accept, hold and administer this easement;

NOW, THEREFORE, in consideration of the promises Grantor, as life tenant and in partial exercise of her power of appointment referred to above, gives and conveys to Grantee an easement (herwinafter the "Easement") in all of that certain lot or parcel of land known as Doughoregan Manor, together with all of the improvements thereon and appurtenances, rights, and interests thereunto belonging, the Property being more parcicularly described as follows:

LIBER 1 2 7 2 2 FOL 10 3 4 7

Begining for the same at a cut granite stone heretofore set on the Northern margin of the sixty-six foot wide right-of-way of the Baltimore to Frederick Turnpike Road, the said point being directly opposite Manor Lane, and at the end of the fifteenth or N 70 degrees W 181 perches line, of that land, which by deed dated June 7, 1866 and recorded among the Land Records of Howard County in Liber WWW 25, folio 316, etc. was granted and conveyed by Charles Carroll and Caroline Carroll, his wife, to Anita Carroll, wife of John Lee Carroli and running with the said road margin and reversely with the said fifteenth line plus a straight line prolongation thereof, with bearings corrected to "rue Meridian, as now surveyed; (1) S 73 degrees 0% minutes 48 seconds E 3140.09 feet to intersect the thirteenth outline at 2921.96 feet on said line, thence running reversely with said line and passing over a concrete monument now set on the Southern margin of the said road right-ofway. In .

and the second states and the second states

JBERD826 FOLIO543

1.0000

(2) S 25 degrees 58 minutes 45 seconds E 2921.96 feet to a concrete monument now set, thence running reversely with the twelfth and a part of the eleventh outlines respectively.

(3) S 21 degrees 10 minutes 35 seconds 2 412.58 feet to a granite stone heratofore set, thence

(4) S 87 degrees 17 minutes 37 seconds W 523.99 feet to a granite stone heretofore set at the end of the third line of that land, the said land being a part of the land herein described, which by deed dated November 26, 1867 and recorded among the said Land Records in Liber WWW 27, folio 411 etc., was granted and conveyed by Matthais Mammond and Clara Hammond, his wife, to Anita Carroll, wife of John Lee Carroll and running reversely with the said third line

(5) S 07 degrees 19 minutes 03 seconds E 2783.20 fast to A granite stone heretofors set at the end of the minth or N 62 degrees

À

LIBER | 2722 FOLO348

CHERRY CONTRACTOR STATES

E 34-3/4 perches line of the aforementioned conveyance from Charles Carroll and wife, thence running reversely with the said ninth line

(6) S 61 degrees 55 minutes 21 seconds W 642.37 feat to a granite stone heratorors set at the end of the eighth or N 62 degrees E 157.5 perches line of that tract designated as Lot J, Division 5 and recorded among the Partition Records of Howard County in Liber EPH No. 1, folio 35 etc., and running reversely with the eighth to second line inclusively

(7). S 51 degrees 52 minutes 51 seconds W 2549.50 feet
 to a large granite stone with the lattors CC cut therein, thence
 (8) S 27 degrees 49 minutes 20 seconds E 3024.41 feet
 to a granite stone heretofore set.

(9) S 61 degrees 59 minutes 09 seconds W 180.43 feet to a concrate monument now set,

(10) S 27 degrees 01 minute 31 seconds \$ 513.33 feet to
 a granite stone beretofore set,

(12) S 49 degrees 13 minutes 56 seconds W 956.26 feet to a tall granite stone heretofore set,

(12) N 85 degrees 56 minutes 08 seconds W 1573.23 feat to a concrete monument how set,

(13) S 76 degrees 03 minutes S2 seconds W 1021.30 feet to a cut sandstone bearing the letter A cut therein, thence running reversely with the firstly described outline of the said Lot J in combination reversely with the seventeenth or N 76 degrees W 147 perches line plus the sixteenth line of that tract designated as Lot K in the aforementioned Partition Records

(14) S 76 degrees 13 minutes 08 seconds W 3570.74 feet to a stone marked five which is now roset, thence running reversely with the fiftmenth and fourteenth outlines of Lot K.

3i J

then 1 2 7 2 2 Folio 3 4 9

Libero B 2 & FOLIO 5 4 5 🏻 🏶

(15) S 7B degrees 20 minutes W 561.00 feet passing over a stone heretofore set 35.61 feet from the end thereof, to a congrate monument now set on the west edge of an abandoned roadway, thence

(16) N 17 degrees 55 minutes 00 seconds W 2314.65 feet to a concrete monument now set at the point the persimmon tree once stood, thence running reversely with the combined thirteenth and twelfth outlines

(17) N 03 degrees 44 minutes 03 seconds N 354.05 feet to a point in the center of the stream between the abuttments of an abandoned bridge site, thence reversely with the eleventh and tenth outlines respectively

(10) N 15 degrees 17 minutes 33 seconds W 165.00 feet to a concrete monument now set,

(19) N 19 degrees 47 minutes 33 seconds W 592.00 feet to an iron pipe heretofore set in the bed of Carxoll's Mill Road, the said point also being at the end of the third or \$ 24-1/2 degrees W 23 parches line of the thirdly described or 20 acres 3 roods and 4 square perches parcel which by deed dated September 15, 1939 and recorded among the said Land Records in Liber BM Jr. No. 164, folio 403 etc. was granted and conveyed by Thomas P. O'Donnell and Ada D. O'Donnell, his wife, to Fhilip A. Carroll, and running three courses with the said Carroll's Mill Road and with the fourth, fifth and a part of the sixth lines of said third parcel

 (20) S 72 degrees 11 minutes 00 seconds W 57.75 feet to an iron pipe now set,

(21) N 87 degrees 49 minutes 00 seconds W 396.00 feet to an iron pipe now set,

(22) N 69 degrees 36 minutes 00 seconds W 206.70 feet to an iron pipe now set at the end of the third or S 26 degrees 50 minutes W 209.10 foot line of that land which by deed dated January 12, 1939 and recorded among the said Land Records in Liber BM Jr.

LIBER 12722 FOLIO350

CARDELE PROFESSION OF THE PROFESSION

No. 163 folio 258 was granted and conveyed by Thomas P. O'Donnall and Ada D. O'Donnell, his wife, to Isaac Johnson, the said parcel being an exception to the aforementioned thirdly described parcel and not the firstly described parcel as set forth in the said conveyance from O'Donnell to Carroll, thence running reversely with the third and second lines of the Johnson lot

LIBERD826.FULID546

(23) N 21 degraes 54 minutes 00 seconds E 209.10 feet to a concrete monument new set,

(24) N 69 degrees 36 minutes W 209.10 fast to a concrete monument now set at 161.23 fast on the eighth line of the aforementioned thirdly described.purce) thence with s.part of the eighth and [all.of the minth outlines

(25)N 21 degrees 54 minutes 00 seconds E 610.97 feet to a concrete monument now set on the West edge of a flintstone quarry, thence running with the combined tenth and eleventh outlines, and excluding the said quarry

(26) S 09 degrees 51 minutes 00 seconds E. 176.35 feet to an iron pipe now set, thence with the twelfth and thirteenth lines respectively

(27) E 80 degrees 96 minutes 00 seconds E. 13.20 feat to an iron pipe now set.

(28) 5 89 degrees 36 minutes E. 151.80 feet to a concrete monument now set, thence running with the fourteenth line of the said third parcel plus the seventh or N 2 degrees W. 9.8 perchas outline of the tract firstly described in the aforementioned conveyance from O'Donnell to Carroll

(29) N 00 degrees 50 minutes 33 seconds E. 628.36 feat to an iron pipe now set in a white oak stump, thence with the eighth, ninth, tenth and eleventh lines of the said firstly described parcel
(30) N 18 degrees 33 minutes 07 seconds W. 112.20 feet to an iron pipe now set in a black oak stump

LIBER | 2722 FOLIO35 |

(31) N 24 degrees 56 minutes.53 seconds E. 1468.58 feet a concrete monument new set

LIBERD826 FULINO

(32) S 65 degrees 03 minutes 07 seconds E. 394.22 feet to a stone heretofore set beside a guily,

(33) N 27 degrees 16 minutes 05 seconds E. 653.39 feat to a concrete monument now set at the end of the third or S 27-1/2 degrees E 25-3/4 perches lines of that land, which by deed dated March 31, 1927 and recorded among the said Land Records in Liber H.B.N. No. 129, folio 532, etc. and running reversely with the third, second, and a part of the first outlines of said land
(34) N 32 degrees 28 minutes 55 seconds W. 417.44 feet

. to a poplar tree

(35) N 42 degrees 41 minutes 06 seconds W. 364.81 feet to a stone heretofore set

(36) N 13 degrees 31 minutes 04 seconds E. 1056.53 feet to a concrete monument heretofore set at the end of the third or S 37-1/2 degrees E. 625 foot line of that land, which by deed dated April 13, 1948 and recorded among the said Land Records in Liber M.W.B. No. 203, folio 463 etc., was granted and conveyed by John Yarrow Eccles,

Executor of Will and Codicil of Belen Beatrice Cavendish Moyle Sherer, Late of Knightsbridge to Philip A. Carroll and running reversely with the said third line

(37) N 47 degrees 48 minutes 39 seconds N. 634.82 feet to the centerline of Vineyard Road as it is now located and passing over a concrate monument heretofore set on the Eastern boundary of said road, thence running four courses and distances with the said Vineyard Road and reversely with the second line of the said conveyance from Eccles to Carroll

(38) N 57 degrees 55 minutes 48 seconds E. 500.86 feet to the end of the twenty-first or 8 56 degrees W 35 perches line of

LIBER 1 2 7 2 2 FOLIO 3 5 2

the herein firstly mentioned conveyance to Anita Carroll, thence running reversely with the twenty-first to sixteenth lines of said land inclusively

LIBEROB26 FOLIO548

nga a

(39) N 53 degrees 00 minutes 48 seconds E. 577.50 feet,
(40) N 39 degrees 49 minutes 48 seconds E. 2458.50 feet,
(41) N 26 degrees 15 minutes 48 seconds E. 2376.00 feet
to the center-line intersection of Vineyard Road and the aforementioned Baltimore to Frederick Turnpike Road, thence one course
with the centerline of the said Turnpike.

(42) S 82 degrees 36 minutes 44 seconds E. 189.50 feet,

(43) N 14 degrees 06 minutes 44 seconds W. 35.47 feet to the Northern right-of-way margin,

(44) S 82 degrees 36 minutes 44 seconds B. 1394.98 feet to the point of the beginning, containing 2042.28 acres of land, more or less.

Exhibit A hereto consists of 40 pages, and includes as page 1 a schedule (which is recorded with this deed of easement) describing the documents, photographs and other things that are a part of the exhibit and that are filed at the offices of Grantee, that are not recorded herewith but are nonetheless as fully and completely incorporated into this deed of easement as though recorded herewith.

The Easement is subject to any and all presently existing valid oncumbrances, preaments and rights of way upon the Property. The Easement shall be of a duration of thirty (30) years from the data bereaf and shall then fully terminate. It is an easement in gross and as such is inheritable and assignable and runs with the land as a binding servitude and as an incorporeal property interest in the Property enforceable by Grantee, its

LIBER 1 2 7 2 2 FOLIO 3 5 3

successors and assigns with respect to the Property and against Grantor and Grantor's heirs, successors and assigns; and to that end Grantor covenants on behalf of herself, her heirs, successors and assigns, with Grantee, its successors and assigns, such covenants being deemed to run as a binding servitude until the date of termination, to do and to refrain from duing upon the Property each of the following stipulations, which contribute to the public purpose in that they aid significantly in the preservation and conservation of the Property:

BERO826 FOLIOS49

 No industrial or commercial activities, with the exception of farming (including the sale of farm produces to the public), shall be carried on on the Property, except such as can be carried on from a residential or farm building without alterations to the external appearance of the building.

2. Timberlands shall be managed in accordance with sound forestry practices, and treas may be selectively or clear out from time to time in such manner as will not permanently alter the character of such lands as forest lands. Notwithstanding the provisions of the praceding sentence, timberland may L: cleared for agricultural use or for any construction permitted by paragraph 4 hereof.

 No dump of ashes, sawlust, bark, trash, rubbish or any other unsightly or offensive material, except that which is produced by or maintained for agricultural use, may be permitted on the Property visible from the public roads and highways.
 No building or other structure shall be built or

maintained on the Property other than those buildings or structures which are on the date hereof located on the Property as Gasoribed and depicted in Exhibit A, except as follows: (a) fences and unpaved roads, when constructed or maintained for agricultural purposes, are not included within the meaning of "building" or "structure" as used herein;

LIBER 1 2 7 2 2 FOLIO 3 5 4

(b) any building or structure may be constructed and maintained for agricultural use, including residential houses for workers (including guards) on the Property and garages or other buildings or structures incident to such houses.

LIBERO825 FOLIOS50

(c) no more than eight (8) additional houses may be constructed and maintained on the Property, together with outbuildings and structures reasonably related therato, including (but not limited to) access words, wells, barns or stables, garages, swimming pools, tennis courts, garden structures and the like, provided that no such house shall be constructed except for the bonu fide personal residential use of one or more of Grantor's lineal descendants, including lineal descendant, by adoption. 5. *Without the express written permission of the

Director of the Maryland Historical Trust (hereinafter the "Officer"), no construction, alteration or remodeling or any other thing shall be undertaken or parmitted to be undertaken c. the existing structures numbered 1, 3, 4, 5, 7 and 22 on page 4 of Exhibit A which would affect their exterior, as described and depicted in Exhibit A; provided, however, that the maintenance, reconstruction, repair, repainting or refinishing of any of said exteriors, domago to which has resulted from casualty loss, deterioration or wear and tear, shall be permitted without such written permission of the Officer provided that such maintenance, reconstruction, repair, repainting or refinishing is performed in a manner that will not matarially altor the appearance thereof as they are as of this dato. The term exterior shall include general style and arrange- . ment of such exterior, including the kind and texture of building materials and the type and style of all exterior windows, doors, light fixtures, signs and other similar features, but not the color of paint or other finish.

.HOWARD COUNTY CIRCUIT COURT (Land Records) MDR 12722, p. 0354, MSA_CE53_12713. Date available 10/01/2010. Printed 06/04/2020.

LIBER | 2722 FOLIO355

6. Grantor agrees, to the extent reasonably financial feasible, to maintain the Manor House (Exhibit A, p. 4, No. 1) in good, olean and safe condition and shall maintain, repair and administer it to preserve its historic, aesthetic and cultural character and appearance as is described and depicted in Exhibit A; provided, that nothing herain shall require reconstruction if the Manor House is destroyed in whole or in part by casualty loss. This covenant is expressly limited to the Kanox House and does not apply to any other imporvement on the Property. The obligations of this affirmative covenant are expressly declared not to apply to the existing structures numbered 3, 4, 5, 7 and 22, which are subjected to a negative covenant only in paragraph 5 above.

LIBEROB26 FOLIO551

- 10 .-

C.

7. If at any time during the term of this Easement a public highway' is proposed to run across the Property between the axisting Masor Lane and the eastern boundary of the Property in A generally north-south direction, such public highway may be constructed and maintained if (a) the Officer finds that such highway would be less deleterious to the historic, aesthetic and cultural character of the Property than is the existing Manor Lane, and (b) the public right of way over the existing Manor Lane as it cronses all or substantially all of the Property is surrendered.

8. Grantes shall have the right to enter the Property for the purpose of inspecting the Property to determine whether there is compliance by Grantor with the terms of this Easement, provided that (a) unch right may be encoded to more frequently then once overy two years, (b) no more that one representative of Grantes shall be entitled to participate in each inspection, and (a) the interiors of all buildings and structures shall be exempt from inspection.

9. Upon any breach of the terms of this Easement by Grantor, Grantee shall have the following rights, which shall be oumulative and shall be in addition to any other rights and remedies available to grantee at law or in equity:

LINER | 2722 FOLIO356

(a) to require restoration of the Property to the con-, dition required by this Easement;

LIBERD826 FOLIOS52

ß

(b) to enjoin any breach or enforce any covenant hereof by <u>ex parts</u>, interlocutory, and final injunction, and

(c) to recover compensatory damages for any breach, which damages shall be applied to restoration of the Property to the condition required by this Essement.

No failurs on the part of the Grantee to enforce any covenant or provision herein nor the waiver of any right hereunder by Grantes shall discharge or invalidate such covenant or provision or any other govenant, condition, or provision hereof, or affect the right of Grantee to enforce the same in event of a subsequent brisch or default.

10. In any event where the terms of this Essement require the consent of the Officer, such consent shall be requested by notice to the Officer and consent shall be deemed to have been given within forty-five (45) days after receipt of notice by the Officer unless the Officer gives notice to Grantor of specific reason for disapproval. In any event where the Officer gives such notice of disapproval. Grantor may appeal the disapproval to the Board of Trustees of the Maryland Historical Trust for review by it or by such person or agency as may be designated by it to make such review. Appeal shall be made by notice to the Officer given within forty-five days of receipt of notice of disapproval from the Officer.

11. Any notico required to be given by this Basemant shall be in writing and may be given by certified or registered mail, with postago propaid and return receipt requested, addressed to the Officer, as follows:

st .

pirostor Maryland Historical Trust Shaw Houses 21. State Circle Annapolis, Md. 21401

: 1

LIDER 12722 FOLIO357

LIBERD826 FULIUS53 G - 12 ~ or to the Grantee or the Officer at such other address as the officer may from time to time designate by notice to Grantor. any notice given in the foregoing manner shall be deemed to have been given when deposited with the United States Post Office. 12. This Easement is for the purpose of promoting and 1.4 shall be construed so as to promote the proposes of the statutes oreating and governing Grantee and of Section 2-118 of the Real Property Article of the Annotated Code of Maryland and to preserve the bistoric, cultural, scenic and aesthetic character of the Prosecty. 13. Grantee agrees that it will hold the Basement exclusively for conservation purposes, i.s., that it will not transfer the Easement in exchange for money, other property, or services. TO HAVE AND TO HOLD unto the Maryland Historical Trust, Its successors and assigns, for thirty years. Witness the following signatures and seals, GRANTOR : witness : (SEAL) augus ORANTEE: WITNESS: Maryland Historical Trust (SEAL) DIRECTOR STATE OF HEAVENING COUNTY, to wit: I HEREBY CERTIFY that on this 23 day of MAY in the year $\frac{1}{1}$, before the subscriber, personally appeared Nina K, Carroll and acknowledged the foregoing deed to be her act. cary Publ 3-30-7 МУ mission expires: CHILETH P. LASI • • •

HOWARD COUNTY CIRCUIT COURT (Land Records) MDR 12722, p. 0357, MSA_CE53_12713. Date available 10/01/2010. Printed 06/04/2020.
LIDER 1 2 7 2 2 FOLTO 3 5 8

COEKOB26 FOLIOS54 STATE OF MARYLAND (Inulling COUNTY, to with I HEREBY CERTIFY that on this /_ day of in the year 22, before the subscriber, personally appeared Orwin C. Anlboft, Director of the Maryland Historical Trust, the foregoing daed to be the act of said Trust. agged cknowl \cap ATY PU 7/1/2: My commission expires; form and sufficiency t A______ 1977. ANTRAT <u>у</u>, 9 j . j •2

HOWARD-COUNTY CIRCUIT COURT (Land Records) MDR 12722, p. 0358, MSA_CE53_12713. Date available 10/01/2010. Printed 06/04/2020.



LIBER | 2722 FILLID 360

à

- -MONTO COMPANY EASEMENT EXHIBIT NO. SIGNED ORIGINAL ON ÷ 1 ~ FILE A DE ATTACHED CEE ATTACHED CRIPTION CRIPTION A DE A PAGES A OF A PAGES A, Page 2 3 H LIBERD820 FOLIOS56 ዪ ŧ MANOR <u>s</u>ys SCALE: 1" - 2000' PREPARED: 3/77 MARYLAND HISTORICAL 1 NATE BUILD . ö : . . ENLINT HOUSE E E 5 おとうに BURLEIGH MANOR THUST

HOWARD COUNTY CIRCUIT COURT (Land Records) MDR 12722, p. 0360, MSA_CE53_12713. Date available 10/01/2010. Printed 06/04/2020.

NORTH



2722 F01036

CIDER | 2722 FOL10362



HOWARD COUNTY CIRCUIT COURT (Lend Records) MDR 12722, p. 0362, MSA CE53_12713. Date available 10/01/2010. Printed 06/04/2020.

LUER | 2722 FOL10363 HO CIRCUIT COURT (Land Records) 1981, Liber CMP 1047, folio 0333 MSA CE53-1033

.

St		فريان ورويان والمرود والمرود والمرود والمرود والمرود والمرود والمرود والمرود والمرود والمرود والمرود والمرود و) ///	\$12148 * 11.11 TUNEILI	-T INTHING
	UEDI 0 6 7 HUD 8 3 3	(03.0/e3 5/7/80)	R/¥	31500	_
		•••		. ·	29.
		RIGHT OF WAY ADREEM	eht		
L	rurrectors, fictnoce, and unigra, fo and telephone lines, locitoring the a guys, conduits, tables, street light	tensury poles, crossing, closes, i a and equipment in, over, i	construct, operate and m wife, telephone and either rader and through the p	wites, anchors, reporty of the	A/R 31200 .
	Hill Road being known as	lay 3.3rd District, H	ioward. Cousty and	សហរិ វិនាស្ត្រជន	*
	PHILLIP A. CARAO	66			·
	64 64 6 22103 1947 9, 194 16 w 1119	Gounty in Liber K. I	nd reported among the B	and Records of otics 303 .	
	 yrnyandar, istay wires between an out down incen adjucant to the wir crested mader or over the line, The Hear are out to be be , side of Convoll Sill Road 1360 feat. 	e in provide ample clearon cauch beginning at Mar and certending in an e	pole is any building; an it. No building: or muc or Lans opposite th actorly direction :	d viz, top, or ium: are to be he porth spproxizately	880)
	"Unantor also grouts enouskronoss, for value r	and pointys to use us weived, the octating	polo line on grant	er ol all 11's property.	ARROLL, JUNA X.
	WITHESS MY hand WITNESS:	and stal this 2007	t day of HAY	19 3 12,	189
	у.х.	. C. J. LALO NDIA	R. CARROLZ.	(BBAL) VIA /IA	5.00
	STATE OF MARANE	with	·	τπατα (8BAL) ΓεέζΜΑ (25/2 Γςγ	
	1 HERBBY OERTIFY, ch before rat, characterister, a Noise	er on this 20 da Public of the State of Mary	r of MAY	US41	7/3/81 - 4 -
•		id, periodally appeared Mi		L	4
	sald aut and dued wild ac WiTNESS my brad and N	olarial stati.	Latastacion [].	h man "	
	(L-167).52.18 Res. 2/16		BENI P. LABAR	Notsingatur	
ଜ	*224 in ar prosence sign. B. J. & E. G. Im 401 Front St. Ro. Box 1475	d and couled the can	boon for a second secon	W ULDIN'S	•
5	Batto, ma 21203				
	ی و اورو هی دونه است. در مصور به د د	the shelled care prove and		: .,]
4. N. A. A.	• • • •	. ·			، بھر. ب
ι.	NC2 .				•
				·* }``	.'
	es all a contraction and		m when sites	an was a with	Sec.s
		i- ·			

LIDER 12722 FOL10364

HO CIRCUIT COURT (Land Records) 1981, Liber CMP 1051, folio 0633 MSA CE53-1037



um 10295 Aut058

Subdivision No. RE-06-02

DEED OF PRESERVATION EASEMENT

THIS DEED OF PRESERVATION EASEMENT ("Preservation Easement") made this day of <u>October</u>, 2006, by and between PHILIP CARROLL also known as Philip D. Carroll and CAMILLA CARROLL, individuals residing in Howard County, Maryland, having an address at 3500 Manor Lane, Ellicott City, Md, (the "Grantor") and THE HOWARD COUNTY CONSERVANCY, INC., a Maryland corporation and HOWARD COUNTY, MARYLAND, a body corporate and politic (hereinafter collectively referred to as the "Grantees").

WITNESSETH:

HOR HOR2 Oct 12: 2006 01:14 pg

00hā

WHEREAS, The Howard County Conservancy, Inc. is a not-for-profit tax exempt organization within the meaning of Section 501(c)(3) of the Internal Revenue Service Code and is established to promote the preservation and protection of natural resources within Howard County, Maryland; and

WHEREAS, Howard County, Maryland is a body corporate and politic which is authorized pursuant to the Howard County Zoning Regulations to be the granice of a preservation easement in the RC District (Rural Conservation); and

WHEREAS, the Grantees have been designated as two of the entities which may be parties to the Preservation Basement in accordance with the Howard County Zoning Regulations; and

WHEREAS, Grantor owns in fee simple 75 acres, more or less, of certain real property situate, lying and being in the Second Election District of Howard County, Maryland known as Parcel 71 on Tax Map 23 as shown on a Final Record Plat entitled, "Density Sending Plat Property of Philip Carroll and Camilla Carroll" and recorded as Plat No. 12272 among the Plat Book Records of Howard County, Maryland (the "Property"); and

WHEREAS, in consideration of the privilege of subdivision, the Grantor is willing to granta perpetual preservation easement on a portion of the Property pursuant to the Howard County Zoning Regulations and Subdivision and Land Development Regulations applicable to cluster subdivisions within the RC District (Rural Conservation), whereby the Grantor restricts and limits the use of that portion of the Property shown and described as Preservation Basement, consisting of 75 acres of land, more or less, as shown on the aforesaid Final Record Plat ("the Preservation Parcel"), pursuant to the terms and conditions, and for the purposes hereinafter set forth, and Grantees are willing to accept such preservation casement as to the Preservation Parcel; and

WHEREAS, Grantor and Grantees have a common purpose in conserving the dominant scenic, cultural, rural, agricultural, woodland and wetland character of the Preservation Parcel, and, except as hereinafter provided, preventing the use or development of the Preservation Parcel for any purpose or in any manner that would conflict with the maintenance of the Preservation Parcel in its open-space conditions; and

Office of Law

HOWARD COUNTER IN CLART COUNT (Land Records) [MSA CE 53-10285] Book MDR 10285, p. 0058. Printed 09/24/2008. Online 10/10/2005.

UKR 10295 F0L0059

WHEREAS, the Grantor and the Grantees acknowledge and agree that the primary purpose and use of the Preservation Parcel is Farming.

NOW, THEREFORE, in consideration of the mutual covenants, terms, conditions and restrictions hereinafter set forth, and other good and valuable consideration, the adequacy and receipt of which are hereby acknowledged, Grantor unconditionally and irrevocably hereby grants and conveys unto Grantees forever and in perpetuity a Preservation Basement of the nature and character and to the extent hereinafter set forth, with respect to the Preservation Parcel.

The purpose of this Preservation Easement is to preserve and protect the environment of the Preservation Parcel and to maintain permanently the open-space values of the Preservation Parcel and the dominant scenic, historic, cultural, rural, agricultural, woodland and wetland character of the Preservation Parcel.

To achieve these objectives, the following conditions and restrictions are set forth:

<u>ARTICLE I.</u>

1. As used in this Preservation Easement, the term "Grantor" means the Grantor, its personal representatives, heirs, successors and assigns and any other person(s) or entity(ies) now or hereafter having a legal interest in the fee simple title to the Preservation Parcel.

2. As used in this Preservation Easement, the term "Grantces" means the Grantees and their respective personal representatives, heirs, successors and assigns and any other person(s) or entity(ies) now or hereafter having a legal ownership interest in and benefit of the easement granted herein.

ARTICLE II. DURATION OF EASEMENT

This Preservation Easement shall be perpetual. It is an easement in gross and as such is inheritable and assignable in accordance with Article IX and runs with the land as an incorporeal interest in the Preservation Parcel, enforceable with respect to the Preservation Parcel by Grantees against Grantor.

ARTICLE III. PERMITTED USES AND ACTIVITIES

Those uses permitted on the Rural Conservation (RC) Preservation Parcels, principal uses permitted as a matter of right, accessory uses and conditional uses, pursuant to the Howard County Zoning Regulations, Section 104 as of July 12, 2001, are permitted, except to the extent such use is prohibited in Article IV hereof. Any other uses, which become permitted uses in RC Preservation Parcels pursuant to any amendments to the Howard County Zoning Regulations subsequent to July 12, 2001 shall require the prior written approval of both Grantees. Grantor and Grantees understand, stipulate, and agree that all principal uses permitted as a matter of right and all accessory uses are compatible uses and, further, that the compatibility of any conditional uses with the aforesaid uses will be determined on a case-by-case basis by the Howard County Board of Appeals upon petition for approval of a particular conditional use.

2

HOWARD COUNTY CIRCUIT COURT (Land Records) [MSA CE 53-10285] Book MDR 10295, p. 0059. Printed 09/24/2009. Online

10/18/2008.

4

September 22, 2005.idh.cs

TAR\LD/06/Philip Cerroll Property-125

USUR 1029.5 MINO60

In addition, the Grantor is permitted on the Preservation Parcel:

(i) To construct, improve, repair, restore, alter, remodel, and maintain all structures, including accessory structures, designed for the purpose of serving the permitted uses of the Preservation Parcel identified in the Howard County Zoning Regulations; and

(ii) To construct and maintain reasonable means of access to all permitted uses and structures both within and outside of the Preservation Parcel; and

(iii) If the Property is served by a shared sewage disposal facility, to install and use a shared subsurface wastewater disposal field, reserve fields and collector pipes in accordance with the terms of this Preservation Easement and the developer agreement and declaration of covenants executed by the Grantor.

ARTICLE IV. PROHIBITED AND RESTRICTED ACTIVITIES

The following uses and activities are prohibited and/or restricted on the Preservation Parcel:

1. Industrial uses, and commercial or residential uses other than those enumerated in Section 104 of the Zoning Regulations are prohibited on the Preservation Parcel unless any such use becomes permitted by amendments to the Zoning Regulations subsequent to July 12, 2001 and prior written approval of both Grantees is given.

2. Display of billboards, signs or advertisements is prohibited on or over the Preservation Parcel, except (a) to state solely the name and/or address of the Preservation Parcel and/or the owners; (b) to advertise the sale or lease of the Preservation Parcel; (c) to advertise the sale of goods or services produced by permitted uses on the Preservation Parcel; or (d) to commemorate the history of the Property, its recognition under state or federal laws; provided that no sign or billboard on the Preservation Parcel shall exceed four feet by four feet. Multiple signs shall be limited to a reasonable number, shall be placed at least 50 feet apart, shall not damage living trees, and shall be placed in accordance with applicable local regulations.

3. Dumping of soil, tresh, ashes, garbage, waste, abandoned vehicles, appliances or machinery, or other materials on the Preservation Parcel is prohibited, except that soil, rock, other earth materials, vegetative matter or compost may be placed (a) as reasonably necessary for agriculture and silviculture on the Preservation Parcel or (b) as may be reasonably necessary for the construction and/or maintenance of structures permitted under this Preservation Easement and means of access.

4. Excavating, dredging, mining, or removing loam, gravel, soil, rock, sand, coal, petroleum and other materials are prohibited, except (a) for the purpose of combating erosion or flooding, (b) for agriculture and silviculture on the Preservation Parcel, or (c) for the construction and/or maintenance of permitted structures, homesites, means of access and wildlife habitat.

Diking, draining, filling or removing wetlands is prohibited without the advance written approval of both Grantees.

HOWARD COUNTY CIRCUIT COURT (Land Records) [M9A CE 53-10285] Book MDR 10295, p. 0060. Printed 09/24/2088, OnEne 10/18/2005.

September 22, 2006 toh es

TAR\LD/06/Philip Carroll Property-125

LOUR 10295 FOLID 61

6. Managing and harvesting of all forested areas on the Preservation Parcel shall be in accordance with the Maryland Forest Practices Quidelines or comparable provisions of any guidelines or regulations which may replace the Forest Practices Guidelines in the future or as they may be amended from time to time.

7. No building, facility, or other structure shall be constructed on the Preservation Parcel after the date of execution of this Preservation Easement, except for that which is permitted in Article III hereof.

8. The total number of residential structures (including but not limited to principal dwelling units, farm tenant houses, mobile homes, caretakers' dwellings and accessory apartments) on the Preservation Parcel shall not exceed None.

9. If a shared subsurface wastewater disposal field, reserve fields and collector pipes are located on the Preservation Parcel, the following restrictions apply to that area of the Preservation Parcel where such fields and pipes are located:

(a) structures are prohibited, except that the Howard County Health Department may approve the installation of playground or athletic equipment of types which do not affect the operations or performance of the shared scwage disposal facility.

(b) earth moving, grading and other land disturbing activities are prohibited unless prior written approval by Howard County is given.

(c) coverings or toppings such as gravel, asphalt or concrete, which impede the growth of vegetation, are prohibited.

(d) the cultivation of crops (other than hay) or other farming practices which may disturb the soil are prohibited unless prior to written approval by the Maryland Department of the Environment ("MDE"), the Howard County Health Department and Howard County Soil-Conservation Service is given.

(e) planting of trees is prohibited. Existing trees need only be removed prior to the installation of the original and reserve subsurface wastewater disposal fields if required by the -Howard County Health Department.

10. The further subdivision of the Preservation Parcel is prohibited.

ARTICLE V. PRESERVATION PARCEL DESCRIPTION

The location and size of the Preservation Parcel is 75 acres of land, which is shown 1. and described as Preservation Basement on a Final Record Plat entitled "Density Sending Plat Property of Philip Carroll and Camilla Carroll" and recorded as Plat No. 18572 among the Plat Book Records of Howard County, Maryland.

HOWARD COUNTY CIRCUIT COURT (Land Records) [MSA CE 53-10285] Book MDR 10295, p. 0061. Printed 09/24/2008. Online 10/18/2006.

September 22, 2006.tdh.ci

TARU.D/06/Philip Carroll Property-125

um 10295 m0062

2. The existing improvements on the Preservation Parcel consist of None. In the event that either The Howard County Conservancy, inc. ("Conservancy") or the Audubon Society of Central Maryland, Inc. ("Audubon Society") is a Grantee of this Preservation Basement, color slides and aerial photographs of the Preservation Parcel obtained at the time of execution of this Preservation Basement shall be kept on file with the Conservancy and/or the Audubon Society, as appropriate, or their respective successors-in-interest in perpetuity. The color slides and aerial photographs shall be utilized to confirm the conditions existing at the time that this Preservation Basement is executed and shall be operative between the parties for such purposes.

ARTICLE VI. MAINTENANCE

1. Grantor shall be responsible for the maintenance of the Preservation Parcel consistent with the terms and conditions of this Preservation Easement.

2. In the event that the Audubon Society is a Grantee of this Preservation Easement, Grantor shall develop a maintenance plan for the Preservation Parcel in consultation with and agreed to by the Audubon Society in order to ensure that the Preservation Parcel is maintained to benefit wildlife and natural habitats.

3. Grantor shall establish and maintain a vegetative buffer strip along the N/A River (Creek, etc.). The minimum width of the buffer strip shall be seventy-five (75) feet along the N/A River (Creek, etc.), except as may be reasonably necessary for (a) forest or wildlife management; (b) recreational water uses and associated structures; (o) hunting, fishing, or trapping; or (d) access to water. Manure and compost shall not be stored within seventy-five (75) feet of streams. Pesticides, insecticides, herbicides or fertilizers shall not be used or deposited within seventy-five (75) feet of streams.

4. All rights reserved by Grantor or activities not prohibited by this Preservation Easement shall be exercised so as to prevent or to minimize damage to water quality, air quality, land/soil stability and productivity, wildlife, scenic and cultural values, and the natural topographic and open space character of the Preservation Parcel.

ARTICLE VII, ENFORCEMENT

1. Upon any breach of the terms of this Preservation Easement by Grantor, Grantees may, after reasonable notice to Grantor, require that the Preservation Parcel be restored promptly to the condition required by this Preservation Easement. In addition, each of the Grantees shall have the independent right to enforce, by any proceeding at law and/or in equity, all restrictions, covenants, conditions and provisions of this Preservation Easement. Grantees' aforesaid remedies shall be cumulative. There shall be and there is hereby created and declared to be a conclusive presumption that any violation or breach or attempted violation or breach of any of the aforesaid restrictions, covenants, conditions or and provisions cannot be adequately remedied by action at law or exclusively by recovery of damages.

2. No failure on the part of Grantees to enforce any covenant or provision hereof shall be deemed a waiver of the right to do so thereafter, nor shall a failure to enforce discharge or invalidate such covenant or any other covenant, condition, or provision hereof or affect the right of Grantees to HOWARD COMPARIAGED Samedia therewonth of Sambas quantize satisfunded 59/24/2008. Online 10/18/2008

September 22, 2006.1dh.cs	3	TARUD/06/Philip Carroli Property-125					
	•	•	· ·				

WB | 0295 MB063

'LIBER | 2722 FOLIO370

Grantees, their directors, officers, employees, contractors and agents, have the right, 3. after reasonable notice to the record owner of the Preservation Parcel, to enter the Preservation Parcel at reasonable times for the purposes of: (i), inspecting the Preservation Parcel to determine whether the Grantor is complying with the terms, conditions and restrictions of the Preservation Basement; and (ii) remediating any damage to the Preservation Parcel resulting from Grantor's breach of any of the aforesaid terms, conditions and restrictions. Further, if a shared subsurface wastewater disposal field is located on the Preservation Parcel, Howard County and representative of MDE have the right to enter the Preservation Parcel for the purpose of inspecting, maintaining, repairing and replacing the shared sewage disposal facility serving the Property; provided, that Howard County and/or MDE shall be required to give notice to Grantees prior to undertaking any construction. The Grantees and MDB shall coordinate inspection visits to the Preservation Parcel and shall coordinate written and oral communications to the Grantor in response to all of the foregoing inspections. This right of inspection does not include access to the interior of buildings and structures. The Grantees shall not be deemed guilty of any manner of trespass for any such entry, remediation, inspection, maintenance, repair or replacement.

4. . If Grantor is found to have breached any of Grantor's obligations under this Preservation Easement, Grantor shall reimburse Grantees for any costs and expenses incurred by Grantees in enforcing the terms of this Preservation Easement, including but not limited to court costs and reasonable attorney's fees, and in remediating any damage to the Preservation Parcel pursuant to Paragraph 3, above.

The Grantor agrees to indemnify, hold harmless and defend each of the Grantees, its 5. directors, officers, employees, contractors and agents from and against any and all claims, actions, demands, damages, liability and expenses in connection with loss of life, personal injury, bodily injury and/or damage to or loss of property that arises from the exercise by any Grantee of the rightof-entry granted herein and any activity undertaken on the Preservation Parcel in connection therewith; provided, that no Grantee shall be indemnified, held harmless or provided the cost of a defense by the Grantor for claims, actions, demands, damages, liability and expenses arising from that Grantee's own negligent act or omission, or that of its directors, officers, employees, contractors and agents.

Each Grantee has independent authority to enforce the provisions of this Preservation 6. Basement and may do so in its sole and absolute discretion; provided, that such authority shall not be deemed to create an obligation to enforce, above and beyond the enforcement obligations conferred on Howard County by law. In the event that any Grantes does not agree as to whether the Preservation Easement terms are being met, any Grantee may proceed, with reasonable advance notice to the other Grantee and the Grantor, with enforcement actions without the consent of the other Grantee.

ARTICLE VIII, PUBLIC ACCESS

The granting of this Preservation Easement does not convey to the public the right to enter the Preservation Parcel for any purpose other than using any trails or other public recreational facilities which now or hereafter exist on the Preservation Parcel as a permitted Public use pursuant to Article III of this Preservation Easement.

HOWARD COUNTY CIRCUIT COURT (Land Records) [MSA CE 53-10285] Book MDR 10295, p. 0063, Printed 09/24/2008. Online 10/18/2008. TARULD/06/Philip Carroli Property 125 6

September 22, 2006 (db.c.

UNA 10295 MM064

ARTICLE IX. MISCELLANBOUS

Each of the Grantees may assign, upon prior written notice to Grantor, its rights under 1. the Preservation Easement to Howard County, Maryland Environmental Trust ("MET"), Maryland Historical Trust ("MHT"), any land conservation organization that has been approved by resolution of the Howard County Council, or an incorporated homeowners' association and only with assurances that the purposes of this Preservation Easement will be maintained; provided, that there shall always be two Grantees or assignees. If any Grantee or assignee shall abandon this Preservation Easement or the rights and duties of enforcement herein set forth, or be dissolved and the terms of the dissolution fail to provide a successor, then such Grantee or assignee shall assign its rights to Howard County, MET, MHT, any land conservation organization that has been approved by resolution of the Howard County Council, or an incorporated homeowners' association; provided, that there shall always be two Grantees or assignees. In the event that such Grantee or assignee fails to make the aforesaid assignment, then, Grantor, its successor and/or assigns, shall institute in a court of competent jurisdiction a proceeding to appoint an appropriate successor as Grantee; provided, that there shall always be two Grantees or assignees. Any such successor shall be Howard County, MET, MHT, a land conservation organization that has been approved by resolution of the Howard County Council, or an incorporated homeowners' association. No assignment may be made by the Grantees or assignces of their rights under this Preservation Easement unless such Grantees or assignces, as a condition of such assignment, require the assignee to carry out the preservation purposes of this Preservation Basement,

2. Grantor shall notify Grantees in writing of the names and addresses of any party to whom the Preservation Parcel is to be transferred at or prior to the time said transfer is consummated. Grantor further agrees to make specified reference to this Preservation Easement in a separate paragraph of any subsequent deed or other legal instrument by which any interest in the Property or Preservation Parcel is conveyed.

 The provisions of this Preservation Easament do not replace, abrogate or otherwise set aside any local, state or federal laws, requirements or restrictions applicable to the Preservation Parcel.

4. This instrument sets forth the entire agreement of the parties with respect to the Preservation Easement and supersedes all prior discussions, negotiations, understandings or agreements relating to the Preservation Easement. Any amendments to this Preservation Easement shall be in writing, signed by each of the parties. If any provision is found to be invalid, the remainder of the provisions of this Preservation Easement, and the application of such provision to persons or circumstances other than those as to which it is found to be invalid, shall not be affected thereby.

5. Grantees shall record this instrument in timely fashion in the land records of Howard County, Maryland and may re-record it at any time as may be required to preserve their rights under this Preservation Easement.

HOWARD COUNTY CIRCUIT COURT (Land Records) [MSA CE 53-10285] Book MOR 10295, p. 0084. Printed 09/24/2008. Onine 10/10/2008.

September 22, 2006.1db.cs

TARUD/06/Phillip Corroll Property-125

THE REPORT OF BELLEVILLE AND LEASE AND LAND

1818 | 0295 100065

6. Any notices by Grantor to Grantees pursuant to any provision hereof shall be sent by registered or certified mail, return receipt requested, addressed to: The Howard County Conservancy, Inc., P.O. Box 175, Weedstock, Maryland 21163 and Howard County, Maryland, c/o Director, Department of Planning and Zoning, 3430 Court House Drive, Ellicott City, Maryland 21043 or to such other addresses as Grantees may establish in writing on notification to Grantor.

7. In any case where the terms of this Preservation Easement require the consent of Grantees, such consent shall be requested by notice to Grantees. Such consent shall be deemed to have been given unless within forty-five (45) days after receipt of notice Grantees mail notice to Grantor of disapproval and the reason therefore.

8. This Preservation Easement does not in any manner prohibit or otherwise restrict MDE from enforcing the requirements of COMAR 26.04.05, or any amendments thereto, and any other laws and regulations governing on site water and sewage disposal systems.

9. Neither the Conservancy nor the Audubon Society shall be subject to any claims for damages as a result of or in connection with this Preservation Easement or the exercise of or failure to exercise its rights hereunder. Each shall only be subject to claims for equitable relief.

TO HAVE AND TO HOLD unto Philip Carroll, Camilla Carroll and The Howard County Conservancy, Inc. their successors and assigns, forever. The covenants agreed to and the terms, conditions, restrictions and purposes imposed as aforesaid shall be binding upon Grantor and Grantees, their personal representatives, heirs, successors and assigns and all other successors to them in interest, and any person(s) or entity(ies) having a legal interest in the fee simple title to the Preservation Parcel or a legal ownership interest in and benefit of the Preservation Easement and shall be binding upon and continue as a servitude running in perpetuity with the Preservation Parcel.

All references herein to "Grantor" shall be deemed plural if more than one person has an Interest in the property herein conveyed to the Grantee. Any pronoun reference herein shall be deemed to apply to the appropriate gender or person, as the case may be. The term Grantor shall mean its respective successors or assigns.

IN WITNESS WHEREOF, Grantor and Grantees have hereunder set their hands and seals the day and year hereinabove written.

GRANTOR WITNESS: (SEAL) PHILIP CARROLL also known as Philip D. Carroll (SEAL)

HCWARD COUNTY CIRCUIT COURT (Lond Records) (MSA CE 53-10285) Book MDR 10285, p. 0065. Printed 09/24/2008. Online 10/18/2006. Schlember 22. 2006 lift.cs 8 TAR\LD/66/PhBp Carroll Property 125

LIBER 1 2 7 2 2 FOL 10 3 7 3

(SEAL)

ACCEPTED by the Grantees on this $\frac{44h}{1}$ day of October ATTEST: THE HOWARD COUNTY CONSERVANCY, INC. Carol M. Filiazak Caleb Could Carol Vilipozak By: Assistant Corporate Secretary ATTEST: 'By: Requel Sanudo Chief Administrative Officer APPROVED: James M. Iyvin, Director Department of Public Works APPROVED:

LINR | 0295 MR066

anh J. 'n

Marsha S. McLaughlin, Director Department of Planning and Zoning

APPROVED FOR SUFFICIENCY OF FUNDS:

whom Sharon F. Greisz, Director

Department of Finance ۰.

. :

APPROVED FOR FORM AND LEGAL SUFFICIENCY

Barbara M. Cook County Solicitor ۵V

HOWARD COUNTY CIRCUIT COURT (Land Records) [MSA CE 53-10286] Book MOR 10295, p. 0068. Printed 09/24/2008. Online 10/18/2008.

September 22, 2006.1dh.ce 1

a Maryland corporation Л

2006.

Ann Holnes Jones President

HOWARD COUNTY, MARYLAND

AL (SEAL) ames N. Robey County Executive

TARUD/06/Philip Carroll Property-125

UNA 10295 MIDO67

GRANTOR: STATE OF MARYLAND, <u>Howarb</u>, County to wit:

I HEREBY CERTIFY that on this $\underline{N7}^{th}$ day of <u>September</u>, 2006, before me, the subscriber, a Notary Public of the State of Maryland, in and for the County aforesaid, personally appeared Philip Carroll also known as Philip D. Carroll and Camilla Carroll the Grantor named in the within Deed of Preservation Basement, and each acknowledged the same to be his or her act.



Kender K. J Notary Public

My Commission Expires: 2-2-09

1.

GRANTEE: STATE OF MARYLAND, HOWARD, COUNTY TO WIT:

I HEREBY CERTIFY that on this $2^{1/2}$ day of <u>CCTOBER</u> 2006, before me, the subscriber, a Notary Public of the State of Maryland, in and for the County aforesaid, personally appeared Ann Homes Jones the President of The Howard County Conservancy, Inc., known to me (or satisfactorily proven) to be the person whose name is subscribed to the within instrument and (s)he acknowledged that (s)he executed the within and aforegoing instrument to be his/her act on behalf of said body corporate for the uses and purposes contained therein and in my presence signed and scaled the same.

AS WITNESS my Hand and Notarial Seal.

LINDA K. LEHMAN Notary Public State of Maryland Howard County My Comr

Notary Public

HOWARD COUNTY CIRCUIT COURT (Land Records) [MSA CE 53-10285] Book MDR 10285, p. 0067, Printed 09/24/2008, Online

•	H: 1192006. Beplember 22, 2006,1dh.cs	10	TARULD/06/Philip Curoll Property 125	
		•		

LIBER 1 2 7 2 2 FOLIO 3 7 5	-
UNIR 0295 MID068	ų
STATE OF MARYLAND, <u>Baltimal</u> COUNTY, TO WIT: I HEREBY CERTIFY that on this <u>400</u> day of <u>O'stalay</u> , 2006, before me, the subscriber, a Notary Public of the State of Maryland, in and for the County aforesaid, personally appeared lamate N Balting the County Formation of the State of Maryland, in and for the County aforesaid, personally	
appeared James N. Robey, the County Executive for Howard County, Maryland, a party to the within Deed of Preservation Basement, who acknowledged the same to be the act of the county and that he executed the foregoing Deed of Preservation Basement for the purposes therein contained by signing in my presence, the name of Howard County, Maryland, by himself as County Executive.	
AS WITNESS my Hand and Notarial Seal.	
My Commission Expires: 515 01	
AUBLION LADING	•
THIS IS TO CERTIFY that this instrument was prepared by Howard County, Maryland, a party named in the within Deed of Preservation Basement.	
Tina D. Hackett, Chief Real Estate Services Division	
,	
After Recording, Relurn To: Howard County, Maryland Real Estato Services Division 3430 Court House Drive Billcott City, Md 21043	
HOWARD COUNTY CIRCUIT COURT (Land Records) (MSA CE 53-10285) Book MDR 10295, p. 0068, Printed 08/24/2008, Online	
10/18/2005. September 22, 2005.idk.cs 11 TARV.D/06/Phillp Carroll Property-123	

ļ

,

.

.

HOWARD COUNTY CIRCUIT COURT (Land Records) MDR 12722, p. 0376, MSA_CE53_12713. Date available 10/01/2010. Printed 06/04/2020.

8266



HOWARD COUNTY CIRCUIT COURT (Land Records) MDR 12722, p. 0377, MSA_CE53_12713. Date available 10/01/2010. Printed 06/04/2020.

18.5.72





Ξ

HOWARD COUNTY-CIRCUIT COURT (Land Records) MDR 12722, p. 0378, MSA_CE53_12713. Date available 10/01/2010. Printed 06/04/2020.

110ER 1 2 7 2 2 FOL 10 3 7 9

											1			
·	11000	1. 17 M IT		נא זמי מו	NINTINI/	G & ZO	NINC	- *	BEFORE THE					
1						9 & ZO Philip E					G BOA)	rd of		
2 3	renu	oner: C	~amma	CALLU	11 AUU .	ւ ուսը բ	7. Vali	de la constante de la constante de la constante de la constante de la constante de la constante de la constante la constante de la constante de			COUN		ARYL	AND
5 4	Dove	lanmen	t Right	e and F	Zesnán	sibilitie	8					,		
5		ement	tugin	.,			~	*.						
6	*	*	ń	ň	*	*	*	*	*	*	*	*	*	*
~ 7		мот	ION:	To re	comm	end to th	ie Cou	nty Cou	ncil that	the De	velopm	ent Rig	hts and	
8					•			nt for the						1
9								by Cour						
10		VOT	E:	5-0				·	•					
11	*	*	*	*	*	ж.	*	*	*	¥	ŧŧ	*	*	* .
12						•								•
13	On Ju	une 17,	2010, ti	he Plan	ning B	oard of I	Howard	d Count	y, Mary	land, co	onsidere	d the pe	etition o	of i
14	1							oseđ De						
15	Agre	ement ("DRRA	A") for t	the Dou	ighoreg	an Mar	ıor.		•				
16		The 1	Petition	ers wer	e repre	sented b	y Sang	g Oh, Es	quire.	Also pro	esent an	d repres	senting.	the.
17	Petiti	ioner w	as Josej	oh Rutte	er.						•	••		· . ·
18										•	•			
19						<u>Public</u>	Testin	<u>10ny to</u>	<u>the Boa</u>	ard	•			
20	The DRRA with attached Exhibits 1-10 and Department of Planning and Zoning Technical													
21	Staff	Report	were p	resente	d to the	e Board	for its	conside	ration. I	The Dep	artmen	t of Pla	ming a	nđ
22	Zoning ("DPZ") recommended a finding of General Plan consistency between the DRRA and													
23	General Plan 2000 as amended by CB 9-2010.													
24		Mr.	Paul Jo	hnson c	of the C	ffice of	Law e	xplained	d to the	Board f	heir role	e, whiel	n is to	
25	determine whether or not the petition is consistent with the General Plan.													
26								as only						
27	DRRA is consistent with the General Plan and not to make recommendations on the details of the													
28	agreement not related to the consistency with the General Plan. He noted specific parts of the													
29	1 -											regulat	ions; er	hancing
30	the (County'	s park s	system;	agricu	ltural pr	eservat	tion; and	d histori	c presei	rvation.			
31				-				tion to t						
32								n was b						
33	sens	e of em	ergency	y which	i was n	ot allow	ing add	equate p	ublic se	rutiny a	ind ben	efiting	only the	;
	1							1						

.

:

• ،

LIDER 12722 FOLIO380

petitioner. Testimony noted several areas believed to be inconsistent with the General Plan including waste treatment, environmental degradation and fiscal impact.

Mr. Harry Carnes expressed concerns about sewage and waste water treatment and specific language included in the agreement pertaining to Burnside Drive. Comments were also critical of the technical staff report presented.

Mr. Victor Ilenda, speaking on behalf of the Chateau Ridge Lake Community Association 6 (CRLCA), reiterated that this is the first time Howard County will be entering into a DRRA and that 7 it should be done with the utmost caution. He referenced the section of the DRRA that states 8 Burnside Drive would remain closed, but noted that concept plans have been created for the proposal 9 and should also indicate where an alternate entrance would be made if needed to ensure a connection 10 to Burnside Drive wouldn't be opened. He stated that since the CRLCA was not an incorporated 11 entity, transfer of the proposed open space to block access to Burnside Drive should be to County 12 ownership not CRLCA's. He also noted concerns about and the potential impacts on the community 13 of on-site wastewater treatment. 14

Several persons spoke with concerns of Burnside Drive, which abuts a portion of the
Doughoregan Manor property to be developed.

Mr. Ken Aldrich provided the Board with several photos of current road conditions and
indicated how added traffic would only worsen those conditions.

Mr. Ted Baruch reiterated that the State Highway Administration's comments requested two
points of connection to the proposed development and suggested an another entrance besides
Burnside Drive be identified now to address this issue.

22 Mr. Larry Jeeter spoke in opposition of the proposed DRRA as currently written with 23 concerns including the opening of Burnside Drive, increasing traffic and adequate roads and 24 retention of existing environmental features of the area where development is proposed.

Ms. Christina Delmont-Small stated that the DRRA is not consistent with the General Plan based on a fallacy of composition and specifically noted concerns regarding sewage and the protection of environmental resources. Concerns were also raised regarding the availability of public records and a request was made to provide verification of the negotiation process leading to the proposed DRRA.

Mr. Kal Bhatti stated that proposal was inconsistent with the General Plan's smart growth policies noting that the addition of 325 dwelling units represented low density growth and that expanding public services perpetuates suburban sprawl and depletes needed resources.

1

2

3

4

5

"LIBER | 2722 FOLIO 38 |

Ms. Cathy Hudson stated that while the agreement has many good parts, there should be added emphasis on protecting and preserving the historic core of the property given its significance. Suggestions included stronger language that calls for no further development and further clarification on the intent of the preservation strategy rather than just saying there would be no subdivision.

Mr. Johnson noted that even with this agreement, the Carroll family is still subject to all other County laws and also that requirements set forth in the DRRA are above beyond and also run with the land and not the owners. Although the agreement proposes a ten year term, the County still retains all rights to change laws affecting the proposal based on the health, safety, and welfare of the County.

Mr. Oh's response to the public's testimony included further explanation of the intent to 11 12 ensure Burnside Drive stays closed. Exhibit 2 of the DRRA was highlighted to show that in order for the County to connect through Burnside Drive, they would have to condemn two homes as well as 13 14 the open space lot abutting Burmside Drive. Mr. Oh also stated that he didn't hear anything presented to the Board that specifically showed how the DRRA was inconsistent with the General 15 16 Plan. He also addressed questions of the Board regarding Ms. Hudson's testimony considering 17 landominiums, as well as ten years being sufficient to complete execution of the DRRA.

18 19

20

21

23

24

25

27

1 2

3

4

5

6

7 8

9 10

Board Discussion and Recommendation

Transportation

The Board agreed that language in the DRRA adequately addressed community concerns 22 regarding the closing of Burnside Drive although this issue is not related to the General Plan. The Board noted that the County's Adequate Public Facility Ordinance (APFO) would address concerns related to the adequacy of transportation infrastructure at the time new development occurs. Concerns raised about access to the proposed development would also be reviewed during the 26 County's subdivision review process.

Sewer

28 With regard to concerns raised by the additional sewage generated by the proposed development, the Board was in agreement that the General Plan Amendment (CB9-2010) to include 29 additional acreage into the Planned Service Area would not have been passed if the issue was not 30 adequately evaluated by the County. It was noted that the County's request to include an on-site 31 32 treatment facility was initially considered by the petitioners, yet after receiving much community 33 objection, has been removed as an option in favor of studying additional alternatives that would 3

· LIDER | 2722 FOL10382

handle wastewater nutrient reduction off-site as outlined in the agreement. While fiscal impacts associated with the proposal were seen as valid concerns, the Board agreed that it was a budget issue and outside the purview of evaluating the agreement on the basis of consistency with the General Plan and furthermore would not substitute its judgment for that of those with the knowledge and expertise to make a recommendation in the County's best interests.

Technical Report and Information

1

2

3

4

5

6

7

8

9

10

33

The Board agreed that the inclusion of information provided in previous Technical Staff Reports and Board Recommendations on the General Plan amendment and the Zoning amendment for Doughoregan Manor would enhance public understanding of the process as a whole and specific parts. The law enabling the County to enter into a DRRA has been in effect for a short period of time. This being the first DRRA the County has executed, the Board felt the public could benefit 11 from additional background, even if redundant. Staff also explained that citizens seeking to review 12 the DRRA files at the DPZ had to be redirected to the Office of Law, which has the files to respond 13 to a Freedom of Information Act request. The Board recommended that a better format for hearing a 14 DRRA, and all points included, be made if there were another request. 15

Historic Preservation 16

The Board had diverging opinions related to the preservation strategy for the historic core 17 included in the agreement. Although Doughoregan Manor is a National Historic Landmark, the 18 property is a place of private residence and deserves the benefit of allowing the petitioners to work 19 on their preservation efforts in good faith. The General Plan speaks broadly on the importance of 20 preserving historic resources, but does not specifically say how that should occur for individual 21 properties, nor does it say anything about the development of preservation plans for individual 22 resources other than through a comprehensive preservation plan developed by the County. To 23 suggest in this instance that a property plan needs to be part of the agreement, and the petitioner and 24 the County work together to preserve the site, could be construed as far reaching. The petitioner has 25 also invested considerable resources in the upkeep of the property. The County architectural 26 historian has been on the grounds numerous times to document the preservation activities now taking 27 place and noted the quality of work being employed. In addition, the petitioner has been negotiating 28 with the Howard County Conservancy on the final conservation easement that will prohibit any 29 future subdivision of the historic core. By these accounts, the intent to preserve the core is 30 memorialized in the various strategies outlined in the agreement. 31 One Board member, however, expressed disappointment that of all of the different parts of 32

the DRRA proposal, the one containing the least amount of detail is the preservation strategy for the

4

HOWARD COUNTY CIRCUIT COURT (Land Records) WDR 12722, p. 0382, MSA_CE53_12713. Date available 10/01/2010. Printed 06/04/2020.

118ER | 2722 FOL10383

historic core itself. In fact, the petitioners presented the proposal to expand the planned service area as a strategy to preserve the historic home. However, within the DRRA there is no mention of minimum preservation standards beyond limiting further subdivision of the core.

Consistency with the General Plan

1

2

3

4

5

6

28

29

30 31

32

33

The Board was in agreement that the DRRA is consistent with the following sections of the General Plan:

7 3.1 Ensure the critical mass of high quality, strategically located farmland is protected from
8 development: addressed in Article IV, Agricultural Preservation. Out of 900 total acres, placing 500
9 acres in agricultural preservation (in addition to 75 acres previously preserved) is a significant
10 critical mass of agricultural resources.

3.4 Protect Water Resources: addressed in Article III (3.3). The agreement is also consistent with
the General Plan Water Resources Element. Concentrating the area to be developed within the
Planned Water and Sewer Service Area prevents the proliferation of septic fields; there is also a
strategy to reduce the wastewater nutrient load from this development.

4.18 Enhance the County park system and recreational facilities: addressed in Article III (3.4.).
 Donation of land at no cost to be added to Kiwaniy-Wallis Park will allow expansion and enrichmet

Donation of land at no cost to be added to Kiwanis-Wallis Park will allow expansion and enrichment
 of current programs offered at this popular recreational facility.

5.14 Maintain or enhance the landscape character of roads: addressed as part of Exhibit 2 through
the orientation of housing along Route 144. Dedication of additional right-of way and the orientation
of homes along Route 144 are both appropriate and protect the character of Route 144.

21 5.18 Establish a comprehensive County-wide historic preservation program: addressed in Article

IV (4.2). The historic core easement will prevent future subdivision. Also the Concept Plan for
 development in Exhibit 2 locates all new development behind the topographical ridgeline, which will
 preserve existing vistas.

6.8 Secure better protection of environmental and landscape resources within new developments:
addressed as part of Article IV, Agricultural Preservation. The proposed R-ED zoning is the most
environmentally sensitive zone available within the PSA.

Additional Points Addressed by DRRA

During its meeting on the Doughoregan zoning petition ZB1087M, the Board noted several issues expected to be addressed in the DRRA. In reviewing the DRRA, the Planning Board found that the following points are adequately addressed:

• Places formally approximately 500 acres of the Property into the Agricultural Land Preservation: addressed as part of Article IV, Agricultural Preservation.

tHEER 1 2 7 2 2 FOLIO 3 8 4

۰.*.*

1	¢	Recordation of covenants to prevent further subdivision of the core 94 acres of property	1
2		around the	
3		Manor, with the Howard County Conservancy holding the easement: addressed in Article IV	ĺ
4		(4.2).	
5 6	ø	Compliance with fire safety requirement via sprinklers within the dwellings to eliminate any need for	
7		fire safety access through a secondary access at Burnside Road: addressed as part of plan to	
8		install sprinkler system throughout the proposed development.	
9	•	Prohibits any development access via Burnside Road. Addressed as part of Article III (3.1).	
10	•	Confirmation of the County's commitment to honor the intent of Resolution 43's termination	
11	_	of	Ì
12		Burnside Road: addressed as part of Article III (3.1)(B).	
13	ð	Confirmation of the donation of approximately of 34 to 36 acres of land to Howard County	
14		for the	
15		Department of Recreation and Parks: addressed in Article III (3.4.)	
16		Obligates the provision of a nitrogen pre-treatment plant and the expansion of sewer line	
17	•	capacity by entering into a Major Facilities Agreement with the Department of Public	
18		Works: addressed in Article III (3.2) (A) reassessment of nutrient concentration strategy	
19		confirmed by a wastewater flow study.	
20	•	Caps the number of dwelling units at 325 units as shown in the plan presented to the	
21		Planning Board: addressed as part of Article II (2.3) (A).	
22	•		
23		presented to the Planning Board: addressed as part of Article II (2.3) (B).	
24	•	Affirms that the development will: (1) Comply with all stormwater management	
25		requirements; (2) Will not proceed until all initial necessary public utility improvements are	
26		made; and (3) Comply with the Adequate Public Facilities Ordinance: addressed as part of	
27		Article II (2.4) (2.5) and Article III (3.1) (A).	1
28	9		
29		validity until	
30		the completion of the development, although it was expressed that perhaps a having a longe	r
31		10 year period might be prudent: addressed as part of Article IX (9.2) (A),	
32			

.

LINER | 2722 FOLIO385

Finally, the Board noted that from a legal perspective, the Option Agreement included in the DRRA is a strong tool that binds both parties to the terms of the DRRA.

Motion:

1

2 3 4

5

6

7

8 9

10

11

14

15 16

17 18

19 20

21

22

Mr. Grabowski made a motion that the Board make an advisory determination that the Developer's Rights and Responsibilities Agreement for the Doughoregan Manor is consistent with the General Plan as amended by CB 9-2010. Ms. CitaraManis seconded the motion. The motion passed by a vote of 5 to 0.

For the foregoing reasons, the Planning Board of Howard County, Maryland, on this 17th day of June, 2010, determines that the Development Rights and Responsibilities Agreement for the Doughoregan Manor proposed by Camilla Carroll and Philip D. Carroll, is consistent with the 12 Howard County General Plan, as amended by Council Bill 9-2010 as noted above. 13

HOWARD COUNTY PLANNING BOARD DI Linda A. Dombrowski. Chair

Grabowski, Vice-Chair David

 \mathcal{P} anno taraManis `amm∀

Paul Yelder

ATTEST: 27

28 29

30

whe S. M. Levallis Marsha S. McLaughlin, Executive Secretary





HOWARD COUNTY CIRCUIT COURT (Land Records) MDR 12722, p. 0387, MSA_CE53_12713. Date available 10/01/2010. Printed 06/04/2020.

HOWARD COUNTY OFFICE OF COUNTY EXECUTIVE 3430 Courthouse Drive B Ellicott City, Maryland 21043 # 410-313-2013

Ken Ulman Howard County Executive vww.howardcountymd.us FAX 410-313-3051 TDD 410-313-2323

May 18, 2010

VIA OVERNIGHT MAIL

Camilla Carroll and Philip Carroll o/o Camilla Carroll 3500 Manor Lane Ellicott City, Maryland 21042

Dear Ms. Carroll and Mr. Carroll:

Howard County, Maryland is pleased to offer to purchase the development rights as defined in Section 15.502 of the Howard County Code ("Development Rights") on the parcel of land consisting of approximately 500 acres, more or less, which you own in Howard County, Maryland and which is located at 3500 Manor Lane, Ellicott City, Howard County, Maryland (the "Land") pursuant to (i) an installment purchase agreement (the "Installment Purchase Agreement") between Howard County, Maryland, as purchaser (the "County") and Camilla Carroll and Philip Carroll, as sellers (the "Seller") and (ii) a Deed of Agricultural Land Preservation Easement between the County and the Seller (the "Deed of Easement"), upon and subject to the terms and conditions hereinafter set forth:

1. <u>Purchase Price</u>. The County offers to purchase the Development Rights from the Seller for a total purchase price of no more than \$19,100,000.00, (the "Purchase Price") which is based on a price of \$38,200.00 per acre, rounded to the next highest \$1,000.00. The Seller will receive 10% of the Purchase Price at settlement. The balance of the Purchase Price shall be paid to the Seller or Seller's assignee in 20 equal annual installments commencing on August 15, 2011 and on each August 15 thereafter. This Purchase Price is based on the following agreements and conditions with respect to the Land which shall be set forth in the Deed of Easement:

- (a) The Land is currently a portion of a parcel of record. The Land may be divided into no more than five (5) 50+ aore parcels;
- (b) Ten (10) one-acre lots may be subdivided from the Land upon the approval of the Agricultural Land Preservation Board and the Department of Planning and Zoning; and
- (c) A maximum of fifteen (15) tenant houses may be permitted on the Land upon the approval of the Agricultural Land Preservation Board and the Department of Planning and Zoning.

2. Interest on Unpaid Balance of Purchase Price. Interest on the unpaid balance of the Purchase Price shall accrue from the date of settlement ("Closing Date") and shall be paid to the Seller or Seller's assignce in 40 semiannual payments on February 15 and August 15 in each year after the Closing Date. Interest shall accrue and be payable at (a) the interest rate, as determined by the County's Director of Finance at the time of settlement of the first Batch 14 property to settle, which is equal to the greater of the average of the United States Treasury Yields or AAA Tax-Exempt General Obligation Yields, in each case for the closest available date to each principal payment installment date for the first Batch 14 property to settle, or (b) 4% per annum, whichever is lower. Interest shall be calculated on the basis of a 360-day year of twelve 30-day months.

3. <u>Documentation</u>. All instruments and documents required hereby or affecting the Land, or relating to the Seller's capacity and authority to sell the Development Rights and to execute the Documents and such other documents, instruments, opinions, assurances, consents and approvals as the County may request and all procedures connected herewith shall be subject to the approval, as to form and substance, of the County, the

Camilla Carroll and Philip Carroll property May 18, 2010 Page 2 of 4

County's counsel, and Miles & Stockbridge P.C. ("Bond Counsel"). All of the documents evidencing the installment purchase of the Development Rights (the "Documents"), including without limitation, the Installment Purchase Agreement and the Deed of Basement, shall be prepared by counsel for the County or by Bond Counsel. In addition to those items specifically set forth in this commitment, the Seller shall furnish to the County, prior to the Closing Date, such additional instruments, documents, opinions and materials as the County may require, all of which shall be satisfactory to the County in all respects.

4. <u>Conditions Precedent to Closing</u>. As a condition precedent to the County's obligation to close, not less than seven (7) days prior to the Closing Date:

(a) The Seller shall furnish to the County, a Subordination Agreement in recordable form from each mortgagee and other lienholder having a lien on all or any portion of the Land subordinating such person's interest in the Land to the rights of the County under the Deed of Easement, which Subordination Agreement shall be satisfactory in all respects to the County.

(b) If the Seller is not a natural person acting only in the Seller's individual capacity, Seller shall furnish to the County an opinion of Seller's Counsel, which is satisfactory in form and substance to the County, stating that Seller has the authority to convey the development rights to the County and has taken all actions necessary to validly exercise that authority.

(c) The County shall have received a title insurance binder with a commitment to issue a title insurance policy in the amount of the Purchase Price insuring the acquisition of the easement on the Land and ownership of the Development Rights by the County subject only to those exceptions to title as are approved by the County and its counsel, and with affirmative insurance on such matters as the County may require. The County will secure the title insurance.

(d) The County shall have verified with the Soil Conservation District that the Land is subject to an approved Soil Conservation and Water Quality Plan that reflects current conditions and activities on the Land.

5. <u>Expenses</u>. All costs relating to the recording of the Deed of Easement and any Subordination Agreement, all title examination charges, the premium for the title insurance policy, and the fees of Bond Counsel for a reasonable number of hours of time expended on consultation with legal or financial advisors of the Seller and the out-of-pocket expenses of Bond Counsel shall be paid by the County.

The Seller shall pay the fees and expenses of the Seller's own counsel and, if necessary, licensed engineer or surveyor, and all costs of preparation and recording of other documents, if any, required to perfect the title to the Land and provide a metes and bounds description for the Decd of Basement.

6. <u>Termination by County</u>. This commitment is being made in reliance upon the information supplied by the Seller to the County in connection with the sale of the Development Rights. If the County, acting in good faith, should determine that any such information or supporting representation of a material nature is false, inaccurate, incomplete or misleading, the County may rescind and cancel this commitment.

7. <u>Brokerage</u>. The County shall pay no fee or commission to any broker or agent in connection with the purchase of the Development Rights, and the Seller hereby agrees to indemnify and hold harmless the County against all claims for brokerage fees and commissions.

8. <u>Receipt of Opinion of Bond Counsel</u>. It is a condition precedent to the closing of the transaction contemplated hereby that the County and the Seller receive an opinion from Bond Counsel, dated the Closing Date, to the effect that under existing laws, regulations, rulings and decisions, interest paid under the Installment Purchase Agreement is not includable in the gross income of the Seller (or any holder of the Installment Purchase Agreement) for federal income tax purposes, which opinion may assume continuous compliance with certain

 \sim

~ C

Camilla Carroll and Philip Carroll property May 18, 2010 Page 3 of 4

covenants in the Tax Certificate and Compliance Agreement to be executed and delivered by the County on the date of delivery of the Certificate and may be otherwise limited in accordance with its terms.

9. <u>Acknowledgment of Seller with Regard to Tax Consequences of Transaction</u>. The Seller acknowledges that the Seller has made an independent investigation and has consulted with attorneys, accountants and others selected by the Seller with respect to all tax considerations related to the transaction contemplated hereby (other than the matter described in Section 8 hereof), and the Seller certifices that the Seller has not looked to or relied upon the County or any of its officials, agents or employees, or to Bond Counsel, with respect to any of such matters.

10. <u>Commitment Subject to Approval by Seller of Final Documents</u>. Final Documents for execution, consistent with this commitment, shall be satisfactory to the Seller in form and substance. In the event that the Documents are not acceptable to Seller for any reason (including the failure to execute a Development Rights and Responsibility Agreement) and cannot be made so, or this transaction is not acceptable to the Seller for any reason, Seller, at Seller's option, shall be released from this commitment.

11. Acceptance and Counterparts. To accept this commitment, a copy of this commitment must be signed (and the signature witnessed) by each person with an ownership interest in the property and the copy with original signature of the Seller, delivered to the County no later than May 20, 2010. This commitment may be executed in counterparts, each of which shall be considered an original and all of which shall, together, constitute a single instrument.

12. <u>Commitment Subject to Enactment of Approval Ordinance</u>. Because the Installment Purchase Agreement is a multi-year contract subject to the provisions of Section 612 of the County's charter, the Installment Purchase Agreement must be approved by ordinance ("Approval Ordinance") and the County is not authorized to execute and deliver the Installment Purchase Agreement until the Approval Ordinance is enacted. The Approval Ordinance will be prefiled for introduction to the County Council of Howard County (the "Council") on May 27, 2010, for the Approval Ordinance to be considered by the Council in June. If for any reason the Approval Ordinance is not enacted by October 1, 2010, this commitment shall be null and void and the County and Seller shall have no further obligation hereunder.

13. <u>Assignment Prohibited</u>. This commitment may not be assigned or in any way transferred by the Seller.

14. <u>Entire Agreement</u>. No statements, agreement or representations, oral or written, which may have been made to the Seller or to any employee or agent of the Seller, either by the County or by any employee, agent or broker acting on the Seller's behalf, with respect to the purchase of the Development Rights on the Land, including the Purchase Price, the Deed of Basement, or Installment Purchase Agreement, shall be of any force or effect, except to the extent stated in this commitment, and all prior agreements and representations with respect to the matters in this commitment are merged herein. This commitment may not be changed except by written agreement signed by the Seller and the County.

15. <u>Governing Law</u>. The Seller agrees that this commitment and the Documents shall be governed by and construed under the laws of the State of Maryland.

16. <u>Closing Date: Survival</u>. This transaction may be closed after all conditions precedent to closing have been met. Unless the Seller and the County enter into a Development Rights and Responsibilities Agreement concerning the Land and the Seller's property adjacent to the Land, this transaction must be fully closed by February 1, 2011 or this commitment shall be deemed null and void. Unless otherwise agreed to by the County, the closing shall take place in the main office of the County or at the office of Bond Counsel in Baltimore, .Maryland, as the County might designate. The terms and conditions of this commitment shall survive the closing; provided, however, that if any of the terms and conditions of this

LIDER 12722 FOLD390

Camilla Carroll and Philip Carroll property May 10, 2010 Page 4 of 4

commitment shall conflict with any of the terms and conditions of the Documents, the terms and conditions of the Documents shall prevail. The terms of this commitment shall supersede in full any prior commitment issued by the County in connection with the transaction contemplated hereby.

We are pleased to make this offer to you. Enclosed for your review is an updated score sheet and amortization schedule. Also attached is a sample deed of easement and a sample installment purchase agreement. Please indicate your acceptance of this commitment by signing and returning to us one of the executed originals of this letter no later than May 20, 2010. If not accepted prior to May 21, 2010, the offer set forth in this commitment letter terminates automatically and shall have no further force and effect.

ATTEST:

mie Robbins

Chief Administrative Officer

Approved for Form and Legal Sufficiency this 20 day of

2010: Mafgaret/Ann Nolan

County Solicitor

By:

HOWARD COUNTY, MARYLAND

Ken Ulman **County Executive**

THE FOREGOING TERMS AND CONDITIONS WITH RESPECT TO THE LAND ARE HEREBY AGREED TO AND ACCEPTED THIS DAY OF 2010.

Witness:

Witness:

Philip Carrol



HOWARD COUNTY CIRCUIT COURT (Land Records) MDR 12722, p. 0391, MSA_CE53_12713. Date available 10/01/2010. Printed 06/04/2020.

Exhibit 7

OPTION AGREEMENT

THIS OPTION AGREEMENT (this "Option"), is made this 23.4 day of $\underline{September}$, 2010 ("Effective Date") by and between by and between CAMILLA CARROLL and PHILIP D. CARROLL, individuals, (collectively referred to as the "Carrolls") and Howard County, MARYLAND, a body corporate and politic of the State of Maryland ("Howard County"). The Carrolls and Howard County are hereinafter referred to collectively as the "Parties".

RECITALS

1. The Carrolls own certain real property in Howard County, Maryland, described on Howard County Tax Map No. 23, Parcel 71 consisting of 892.6 AC \pm as shown and described on EXHIBIT 1 ("Doughoregan" or "Property"), which exhibit is attached and made a part of that Development Rights and Responsibilities Agreement that was contemporaneously executed between the Parties on <u>September</u> 23 2010 ("DRRA").

2. The Carrolls intend to develop a 221.1 AC \pm portion of Doughoregan as depicted on **EXHIBITS 2** and **3** (the "Site") of the DRRA with not more than 325 single family detached residential dwelling units substantially in the manner as set forth in EXHIBIT 2 (the "Project").

3. The Carrolls, furthermore, intend to sell and the County intends to purchase a perpetual "Agricultural Land Preservation Easement" on 500 acres of the Property more particularly described in the Exhibit A attached hereto ("Agricultural Preservation Parcel") subject to the terms and conditions contained in the Commitment Letter dated May 18, 2010 setting forth the terms and conditions for the County's purchase of the Development Rights on the Agricultural Preservation Parcel for \$19,100,000.00 (the "Purchase Price") between the County and the Carrolls, a copy of which is attached as Exhibit B (as amended in the DRRA to delete paragraph 10, the "Commitment Letter"), which is incorporated by reference.

4. Notwithstanding anything in the Commitment Letter to the contrary, either expressed or implied, the Parties intend that the obligations contained in the Commitment Letter shall be binding upon the Parties upon occurrence of the conditions precedent set forth in Section 1.2 below, whereupon the County shall exercise its rights under this Agreement and upon the failure of the conditions precedent set forth herein the Carrolls shall perform their obligations under this Agreement.

5. The Parties are entering into this Option to effectuate their mutual intent as specified in Recital No. 4.

NOW, THEREFORE, in consideration of the DRRA, the Commitment Letter, the recitals set forth herein, and for other good and valuable consideration, the receipt and adequacy of which they each acknowledge, Carrolls grant to Howard County and Howard

il Co

County accepts, the exclusive option to obtain the perpetual Agricultural Land Preservation Easement in, over, on, and through the Agricultural Preservation Parcel as set forth in the Commitment Letter as amended by the DRRA as set forth herein. The Carrolls and the County agree that the sale of the Agricultural Land Preservation Easement to the County is a material term of the DRRA and without the consideration of the sale of the Agricultural Land Preservation Easement to the County on the terms and conditions set forth in the Commitment Letter, the County would not have agreed to the DRRA.

Section 1. Option Term and Exercise of Option.

1.1 Term. The term of the Option ("Option Period") shall begin on the Effective Date of this Option, and shall expire (unless sooner exercised by written notice from the County) on January 1, 2013 ("Outside Date"). In the event any one of the conditions precedent to exercise set forth in Section 1.2.1-1.2.3 have not yet been determined as of the Outside Date, then the Option Period shall be extended automatically until (i) all conditions precedent have been determined or (ii) the termination of the DRRA. If any of the conditions precedent have not yet been determined as of the Outside Date, the Carrolls may take action to return the Property to a rural conservation zoning classification (now known as "RC"). Upon the final, unappealable approval of the rural classification zoning approval for the Site, the County shall terminate its right to acquire the Agricultural Land Preservation Easement under this Option Agreement and the Commitment Letter.

1.2 <u>Exercise</u>. Howard County shall exercise the Option upon the occurrence of all of the following conditions precedent by providing written notice to the Carrolls:

1.2.1 The final, unappealable approval of amendment(s) to the Howard County General Plan for extension of the Planned Service Area for water and sewerage for the Site including any amendment(s) to the Master Plan for Water and Sewerage and any application(s) for incorporation into the Howard County metropolitan district.

1.2.2 The final, unappealable approval of the DRRA; and

1.2.3 The final, unappealable approval of R-ED zoning approval for the Site.

1.3 Effect of Exercise. Upon the exercise of the Option, Howard County shall become entitled and obligated to purchase the Agricultural Land Preservation Easement from the Carrolls, and the Carrolls shall become obligated to sell the Agricultural Land Preservation Easement to Howard County on the terms and conditions set forth in the Commitment Letter. Furthermore, by exercising the Option, Howard County shall become obligated to the Carrolls, their heirs, successors and/or assigns to full performance under this Option.

CC
LIBER | 2722 FOL10394

1.4 <u>Effect of Conditions Precedent</u>. The Parties agree no Subdivision Plat for the Property creating any individual lots meeting the bulk requirements of R-ED zoning for single family detached housing shall be submitted by the Carrolls or reviewed by the County until all of the conditions precedent set forth in Sections 1.2.1 through 1.2.3 are satisfied.

1.5 <u>Failure of Conditions Precedent</u>. The Parties further agree that each shall take all actions required to promptly return the Property to a rural conservation zoning classification (now known as "RC") in the event any one of the conditions precedent in Sections 1.2.1 through 1.2.3 hereof is not satisfied. Upon the final, unappealable approval of the rural classification zoning approval for the Site, the County shall terminate its right to acquire the Agricultural Land Preservation Easement under the Commitment Letter.

Section 2. <u>Settlement</u>.

2.1 <u>Settlement Date</u>. Upon exercise of the Option, the Carrolls and Howard County shall set a mutually agreeable date for settlement to occur within ten (10) business days of the exercise of the Option. As set forth in the Commitment Letter, the County shall pay the Carrolls 10% of the Purchase Price of \$19,100,000.00 in cash and issue an installment purchase agreement for the balance of the Purchase Price and the Carrolls shall execute and deliver to County the Deed of Agricultural Land Preservation Easement as described in the Commitment Letter.

2.2 <u>Title</u>. The title to the Agricultural Land Preservation Easement shall be subject only to the effect of the matters recorded among the Land Records of Howard County and deemed acceptable to the County, as set forth in the Commitment Letter. While this Option is in effect, Carrolls shall not transfer all or any portion of their interest in the Agricultural Preservation Parcel or otherwise encumber the title to the Agricultural Land Preservation Easement by any lien, or other interest. The Carrolls agree to cause any lender or lien holder or other person having an interest in the Agricultural Preservation Parcel to subordinate their interest in the Agricultural Preservation Parcel to the County's Agricultural Land Preservation Easement and such subordination agreement shall be in a form acceptable to the County and recorded among the Land Records of Howard County.

Section 3. <u>Specific Performance</u>. If Howard County exercises its Option and the Carrolls fail to agree to a settlement date and consummate settlement, the Carrolls consent to Howard County obtaining a decree against them for specific performance as set forth herein. The Carrolls agree that the County entered into the Commitment Letter, this Option Agreement, and the DRRA because of the Carrolls' agreement to conserve and protect the $500 \pm$ acres pursuant to the Howard County Agricultural Land Preservation Act, and that there are no liquidated damages or other types of monetary damages that would be adequate to compensate the County for the Carrolls' breach of that promise. Therefore, Howard County may file, in lieu and as a substitute for monetary damages, a complaint against the Carrolls in the Circuit Court for Howard County for specific

Santo Do -seg MD 21042 5100 Do -seg MD 21042 Ell'mett C to , performance of their promise to execute and deliver to Howard County a Deed of Agricultural Land Preservation Easement, pursuant to the Commitment Letter, the DRRA, and this Option Agreement. A proposed form of complaint is attached hereto as Exhibit C. The Carrolls appoint and authorize [NAME, ADDRESS] as their agent to receive service of process and filings in the action. Except for such service, as an expedient compromise of the complaint and without admitting liability, the Carrolls knowingly, intelligently, and voluntarily waive all rights, defenses, and claims, from whatever source derived, both procedural and substantive, that they may have to Howard County's action to enforce its rights for specific performance of the acquisition of the Agricultural Land Preservation Easement, and the Carrolls consent to Howard County moving for entry of a consent decree against them for specific performance and to the entry of such a decree in the form attached hereto as Exhibit D.

LIBER 12722 FOLIO395

Section 4. General.

4.1. Notices. Any communication to be given to a party shall be in writing, shall be deemed to have been given on the 3rd business day after being sent as certified or registered mail in the United States mails, postage prepaid, return receipt requested, or on the next business day after being deposited with a reputable overnight courier service as follows:

4

MCC

In the case of the Carrolls notice shall be sent to:

Camilla and Phillip D. Carroll 3500 Manor Lane' Ellicott City, MD 21042

with a copy to:

Joseph Rutter 5300 Dorsey Hall Drive Ellicott City, MD 21042

Sang W. Oh, Esq. 5100 Dorsey Hall Drive Ellicott City, MD 21042

James L. Wright, Esq. 750 E. Pratt Street Suite 900 Baltimore, MD 21202

In the case of the County, notice shall be sent to: Director of Department of Planning and Zoning 3930 Courthouse Drive Ellicott City, Maryland 21043

With a copy to:

County Solicitor of Howard County 3930 Courthouse Drive Ellicott City, Maryland 21043

4.2. <u>Effect</u>; <u>Binding</u>; <u>Amendment</u>; <u>Counsel</u>. This Option shall become effective on its execution and delivery by each party. This Option may be amended only by a document signed by each party. The Parties agree that this Option shall run with the land and be binding upon and inure to the benefit of each party and their respective heirs, successors and assigns, and upon any and all successor owners of record of all or any portion of the Site. Each party has entered into this Option and the Commitment Letter after having the opportunity to receive advice of legal counsel and represents to the other that each understands their respective legal obligations under this Option.

4.3. <u>Governing Law; Jurisdiction</u>. This Option shall be governed by and construed and enforced in accordance with the laws of the State of Maryland. The parties hereby irrevocably accept and submit to the jurisdiction of the Circuit Court for Howard County, Maryland, in any action brought to enforce this Option, suit, action or proceeding and further waive any objection and any right of immunity on the ground of venue, the inconvenience of any forum or the jurisdiction of such courts or from the execution of judgments resulting therefrom.

4.4. <u>Rules of Construction</u>. The enumeration and headings of the sections of this Option are merely for convenience of reference and do not constitute representations or warranties, do not impose any obligations whatever and have no substantive significance. Unless the context otherwise requires, whenever used in this Option the singular will include the plural, the plural will include the singular, and the masculine gender will include the neuter or feminine gender and vice versa. As to the Carrolls, each of the obligations in this Option and the Commitment letter shall be joint and several. The Parties agree that neither shall be considered the primary drafter of this Option.

4.5 <u>Conflict with DRRA; Survival</u>. The Parties hereby agree that any conflict between the terms of the DRRA and this Option shall be interpreted to give this Option its full force and effect to allow the County to acquire the Agricultural Land Preservation Easement. The Parties further agree that this Option shall survive the invalidation of all or any portion of the DRRA.

WITNESS/ATTEST:

Dy A

CAMILLA/CARROLL (SEAL)

PHILIP D. CARROLL

LIBER | 2722 FOLIO 397

(SEAL)

AGREED and APPROVED:

HOWARD COUNTY, MARYLAND

ATTEST:

Jonnie R. Robbins Chief Administrative Officer

BY: (SEAL) Ken Ulmar

Howard County Executive

APPROVED AS TO FORM AND LEGAL SUFFICIENCY

eptember 2010. day of 🔬 this

Margaret Ann Nolan County Solicitor

[Notaries on Following Page]

LIBER 1 2722 FOLIO 398

. /
STATE OF MARYLAND, COUNTY, TO WIT:
I HEREBY CERTIFY that on this <u>944</u> day of <u>Septenbus</u> 2010, before me, the subscriber, a Notary Public of the State of Maryland, in and for the County aforesaid, personally appeared CAMILLA CARROLL, personally known to me or proven to be the individual named herein and executed this Option for the purposes stated therein.
AS WITNESS my Hand and Notatial Seal Find Delero A. 2 (e) BUIC Notary Public My Commission Expires:
STATE OF MARYLAND,
I HEREBY CERTIFY that on this <u>944</u> day of <u>State of Maryland</u> , in and for the County before me, the subscriber, a Notary Public of the State of Maryland, in and for the County aforesaid, personally appeared PHILIP D. CARROLL, personally known to me or proven to be the individual named herein and executed this Option for the purposes stated therein. AS WITNESS my Hand and Notarian Heat Here <u>555</u> Find Debio A. <u>2010</u> Notary Fublic
STATE OF MARYLAND, <u>Howard</u> COUNTY, TO WIT: I HEREBY CERTIFY that on this <u>33/d</u> day of <u>September</u> , 2010, before me, the subscriber, a Notary Public of the State of Maryland, in and for the County aforesaid, personally appeared KEN ULMAN, the County Executive for Howard County, Maryland, who acknowledged the same to be the act of the County and that he executed the
foregoing Option for the purposes therein contained by signing in my presence the name of Howard County, Maryland as County Executive.

AS WITNESS my Hand and Notarial Seal.

Hara Stello Min Motary Public

My Commission Expires: ____

LISA STELLO O'BRIEN NOTARY PUBLIC STATE OF MARYLAND My Commission Expires February 20, 2011

ATTORNEYS' CERTIFICATION

THIS IS TO CERTIFY that the undersigned are members, in good standing, of the Bar of the Court of Appeals of Maryland, and that the within instrument was prepared by the undersigned or under their supervision.

Upon Recordation Please Return To:

Sang W. Oh, Esq. Talkin & Oh, LLP 5100 Dorsey Hall Drive Ellicott City, MD 21042

· LIBER | 2722 FOLIOL+ 00

Exhibit A Description of Agricultural Land Preservation Easement (Ten Pages)

:

.

.

HOWARD COUNTY CIRCUIT COURT (Land Records) MDR 12722, p. 0400, MSA_CE53_12713. Date available 10/01/2010. Printed 06/04/2020.

.

;

Ś

.

EXHIBIT

March 18, 2010

DESCRIPTION OF A 500.000 ACRE AGRICULTURAL PRESERVATION EASEMENT CONSISTING OF 241.638 ACRE PART ONE AND 258.362 ACRE PART TWO PART OF THE PROPERTY OF PHILIP D. CARROLL AND CAMILLA CARROLL SECOND AND THIRD ELECTION DISTRICTS HOWARD COUNTY, MARYLAND

BEING a parcel of land, situate and lying in the Second and Third Election Districts of Howard County, Maryland, said parcel of land being part of the land which by Will of Nina R. Carroll dated June 4, 1979, as amended by its First Codicil dated November 28, 1986 and its Second Codicil dated April 10, 1987; Nina R. Carroll having departed this life on February 11, 1989; said Will having been probated in the Surrogates Court of New York County, New York on April 11, 1989, and the Estate of Nina R. Carroll is filed in the Register of Wills of Howard County, Maryland as Estate No. 17-7868-276, said property also being described in a Mortgage dated October 31, 1962 and recorded among the Land Records of Howard County, Maryland in Liber 394 at Folio 64, and being part of Parcel 1 described therein; said parcel being more particularly described, as now surveyed in the Maryland State Coordinate System NAD' 83 Datum, as projected by Howard County Geodetic Control, as follows:

FISHER COLLINS CARTER INC.

GIVA, HARMEHOKO CORSULTUNTU ING LAKO SURVEYING

10272 Baldmare Helber) / Pre Eliterit City, Mergland 2042 *410) 451-2855 *9x (410) 750-5784 (PART ONE) BEGINNING FOR THE SAME at a point on or near the intersection of the centerlines of existing paving of Folly Quarter Road with Frederick Road (Maryland Route 144); said point being at the beginning of the Seventeenth or South 82°36°44° East, 189.50 foot Deed Line of the aforesaid Liber 394 at Folio 64 (Parcel 1); thence running within said Frederick Road, on or near the centerline of paving, and running with and binding on all of the said Seventeenth line of said Liber 394 at Folio 64 (Parcel I), and with all of the Eighteenth, Nineteenth and a portion of the First line of said Liber 394 at Folio 64 (Parcel I), as now surveyed the following four (4) courses and distances:

1) South 82°54'28" Hest, 189,50 feet to a point; thence

2) North 17°12'33" West, 32.08 feet to a point; thence

3) South 82°35'58" East, 1,394.98 feet to a point; thence

4)

South 73°02'02" East, 2.22 feet to a point on and being a distance of 3,137.87 feet from the end of the aforesaid First or South 73°02'48" East, 3140.09 foot Deed Line of the aforesaid Liber 394 at Folio 64 (Parcel 1) and also being the Northern margin of the sixty-six foot wide Right of Way of Frederick Road (Maryland Route 144); thence leaving said First line of Liber 394 at Folio 64 (Parcel 1) and the Northern margin of Frederick Road (Maryland Route 144) to run over, across and through the aforesaid Liber 394, Folio 64 the following thirteen (13) courses and distances:

5) South 06°25'26" West, 152.29 feet to a point; thence

South 04°05'30" West, 282.20 feet to a point of curvature; thence

253.27 feet along the arc of a non-tangential curve to the left, having a radius of
 1,301.42 feet, a central angle of 11°09°01" and subtended by a chord bearing and
 distance of South 02°54'01" East, 252.87 feet to a point; thence

8) South 07°27'35" East, 191.10 feet to a point; thence

9) South 09°24'23" East, 392.33 feet to a point; thence

10) South 09°00'09" East, 293.76 feet to a point; thence

HOWARD COUNTY CIRCUIT COURT (Land Records) MDR 12722, p. 0402, MSA_CE53_12713. Date available 10/01/2010. Printed 06/04/2020.

FISHER, COLLINS

CARTER.

SAVE ENERGY AND CONSILIANTS

10272 Delucion Havonsi Pita Enicali City, Haryland Zioce (410) 455 - 2853 Fax (410) 750-3784

LAND SURVEYOR

INC.

2

LIBER 1 2722 FOLIDI, 03

11)	South 09°47'54" East, 393.83 feet to a point; thence
. 12)	South 09°03'17" East; 615.16 feet to a point; thence
13)	South 09°03'55" East, 291.12 feet to a point; thence
14)	North 79°49'03" West, 1,832.40 feet to a point; thence
15)	South 19°08'23" West, 1,237.51 feet to a point; thence
16)	North 77°07'13" West, 68.06 feet to a point; thence
17)	South 19°08'11" West, 1,248.08 feet to a point on and at a distance of 279.07 feet
•	from a 1.25 inch diameter Iron Pipe found at the beginning of the Tenth or North
	70°15'08" West, 1456.95 foot Deed Line of the aforesaid Liber 394 at Folio 64
•	(Parcel 1); thence running with and binding on the remaining portion of said
	Tenth line of Liber 394 at Folio 64 (Parcel 1), as now surveyed, the following
	course and distance:
18)	North 70°10'26" West, 1,177.65 feet to a 1.25 inch diameter from Pipe found at
	the end thereof; thence running with and binding on the Eleventh, Twelfth,
	Thirteenth, Fourteenth, Fifteenth and Sixteenth lines of said Liber 394 at Folio 64
	(Parcel 1), the following six (6) courses and distances:
19)	North 78°34'25" West, 791.22 feet to a point; thence
20)	North 73°34'25" West, 387.55 feet to a point; thence
21)	South 69°10'35" West, 31,37 feet to a point, thence
22)	North 51°40'56" East, 568.09 feet to a point; thence
23)	North 39°59'12" East, 2,458.50 feet to a point; thence
. 24)	North 26°28'43" East, 2,376.00 feet to the point of beginning, containing 241.638
	aures, more or less.

FISI COI &

HOWARD COUNTY CIRCUIT COURT (Land Records) MDR 12722, p. 0403, MSA_CE53_12713. Date available 10/01/2010. Printed 06/04/2020.

FISHER, COLLINS & CARTER, INC. 10272 Balümətə Kəllacəl Fiko Elitebi Diy, Marytanā 21042 (410) 401-2155 Fox (410) 750-8784

3

LIDER 12722 FOLIOLA 04

(PART TWO) BEGINNING FOR THE SAME at a point on the Northern margin of the sixty-six foot Right of Way of Frederick Road (Maryland Route 144), said point being on and a distance of 53.08 feet from the beginning of the First or South 73°02'48" East, 3140.09 foot Deed Line of the aforesaid Liber 394 at Folio 64 (Parcel 1); thence running with and binding on a portion of said First line of Liber 394 at Folio 64 (Parcel 1), and the Northern margin of Frederick Road (Maryland Route 144), the following course and distance:

- South 73°02'02° East, 1,379.85 feet to a point thereon; thence leaving the aforesaid First line of Liber 394 at Folio 64 and the Northern margin of Frederick Road (Maryland Route 144) and running for new lines of division, the following twenty-five (25) courses and distances:
- 2) South 16°57'58" West, 65.50 feet to a point; thence
- 3) South 73°02'02" East, 328.95 feet to a point; thence
- 4) South 19°32'01" East, 18.41 feet to a point; thence
- 5) South 73°08'00" East, 180.91 feet to a point; thence
- 6) South 27°46'18" East, 35.58 feet to a point; thence
- 10) South 17"35'24" West, 225.85 feet to a point of curvature; thence
- 346.71 feet along the aro of a curve to the left, having a radius of 315.00 feet, a central angle of 63°03'47" and subtended by a chord bearing and distance of South 13°36'30" East, 329.47 feet to a point of reverse curvature; thence
- 121.72 feet along the arc of a curve to the right, having a radius of 150.00 feet, a central angle of 46°29'32" and subtended by a chord bearing and distance of South 22°13'37" East, 118.40 feet to a point of tangency; thence

FISHER, COLLINS & CARTER, INC. GVA ENERGENCE PORSULATION

19272 Balilmore Halipael Pika Ellipi Elly, Marylind 21042 (410) 481-(2855 Fax (410) 750-(784

HOWARD COUNTY CIRCUIT COURT (Land Records) MDR 12722, p. 0404, MSA_CE53_12713. Date available 10/01/2010. Printed 06/04/2020.

LIBER | 2722 FOLK 05



HOWARD COUNTY CIRCUIT COURT (Land Records) MDR 12722, p. 0405, MSA_CE53_12713. Date available 10/01/2010. Printed 06/04/2020.

FISHER,

INC.

118ER | 2722 FOLIOL 06

-

. •		
· •		
	19)	136.14 feet along the arc of a curve to the left, having a radius of 325.00 feet, a
		central angle of 24°00'04" and subtended by a chord bearing and distance of
۰. · ·		South 21°17'16" West, 135.15 feet to a point of tangency; thence
	20)	South 09°17'14" West, 113.42 feet to a point of curvature; thence
	21)	468.79 feet along the aro of a curve to the left, having a radius of 580.00 feet, a
		central angle of 46°18'34" and subtended by a chord bearing and distance of
		South 13°52'03" East, 456.13 feet to a point of tangency; thence
	22)	South 37°01'20" East, 903.13 feet to a point; thence
	23)	South 23°09'51" Bast, 507.53 feet to a point; thence
	24)	South 67°04'12" West, 462.42 feet to a point; thence
	25)	South 33°51'48" West, 651.81 feet to a point; thence
	26)	South 16°43'26" West, 858,48 feet to a point on and a distance of 1,764.49 feet
		from a Granite Stone found and held at the beginning of the Seventh or North
-	-	73°14'48" West, 4021.13 foot Deed Line of the aforesaid Liber 394 at Folio 64
	- -	(Parcel 1); thence running with and binding on the remaining portion of said
		Seventh line of Liber 394 at Folio 64 (Parcel 1) and with all of the Eighth and a
		portion of the Ninth or North 65°30'08" West, 2143.94 foot Deed Line of said
· ·		Liber 394 at Folio 64 (Parcel 1), the following three (3) courses and distances:
	27)	North 73°16'34" West, 2,253.96 feet to a point; thence
FISHER, COLLINS	28)	North 14°04'34" East, 24.75 feet to a point; thence
& CARTER,	· 29)	North 65°25'29" West, 39.81 feet to a point; said point being on and a distance of
INC.		2,104,13 feet from a 1.25 inch Iron Pipe found at the end of the aforesaid Ninth
rad LAND CONFEVORS 1023 2: Baliloutes Hallogof Pita Lillouti DDy, Maryland 21642		line of Liber 394 at Folio 64 (Parcel 1); said point also being on the easterly
:410) 401-2835 Fox (410) 350-0784		
		6

HOWARD COUNTY CIRCUIT COURT (Land Records) MDR 12722, p. 0406, MSA_CE53_12713. Date available 10/01/2010. Printed 06/04/2020.

	30)	outline of a plat entitled "Amended Plat Of Easement, Sending Parcel, Property Of Philip Carroll and Camilla Carroll" and recorded as Plat No. 19928; thence leaving the aforesaid Ninth line of Liber 394 at Folio 64 (Parcel 1) and running with and binding on part of said easterly outline of Plat No. 19928, the following course and distance: North 14°13'15" East, 1,478.90 feet to a point thereon; thence leaving the aforesaid easterly outline of Plat No. 19928 and running over, across and through the aforesaid Liber 394, Folio 64 (Percel 1) the following twenty-two (22) courses
		and distances:
	31)	South 85°27'17" Bast, 421.22 feet to a point; thence
	32)	South 64°29'11" East, 192.86 feet to a point; thence
	33)	South 47°22'36" East, 333.45 feet to a point; thence
	34)	South 27°46'03" East, 102.62 feet to a point; thence
	-35)	South 73°56'56" East, 224.55 fact to a point; thence
	36)	South 82°25'01" Hast, 149.17 feet to a point; thence
	37)	North 12°14'25" East, 389.04 feet to a point; thence
	38)	North 23°29'52" West, 478.60 feet to a point; thence
	39).	North 16°15'45" West, 750.96 feet to a point; thence
	.40)	North 81°30'55" West, 730.33 feet to a point; thence
	41)	North 00°27'40" East, 207.41 feet to a point; thence
	42)	North 00°40'30" West, 76.57 feet to a point; thence
	43)	North 06°06'01" West, 117.35 feet to a point; thence
7 7 7	44)	North 09°05'56" West, 310.97 feet to a point; thence

FISHER, COLLINS & CARTER, INC. ovia theotering constant and theotering constant ovia theotering constant and theotering constant ovia theotering constant and theotering constant ovia theotering constant and theotering constant and theotering constant ovia theotering constant and theotering con

HOWARD COUNTY CIRCUIT COURT (Land Records) MDR 12722, p. 0407, MSA_CE53_12713. Date available 10/01/2010. Printed 06/04/2020.

10872 Sattimeta Kallavel Pixo Silicoli Bity, Maryiani 21042 (A10) 401-8855 *ax (410) 750-3784 7

.

.

tIDER | 2722 FOLIO 4 0 8



46) North 09°47'54" West, 393.78 feet to a point; thence

47) North 09°00'07" West, 293.32 feet to a point; thence

48) North 09°24'23" West, 391.82 feet to a point; thence

49) North 07°27'35" West, 190.70 feet to a point of curvature; thence

- 50) 243.37 feet along the aro of a curve to the right, having a radius of 1,251.42 feet, a central angle of 11°08'33" and subtended by a chord bearing and distance of North 02°55'27" West, 242.98 feet to a point of tangency; thence
- 51) North 04°05'30" East, 280,56 feet to a point; thence
- 52) North 06°25'26" East, 141.96 feet to the point of beginning; containing 258.362 acres, more or less.

SUBJECT TO AND TOGETHER WITH any and all terms and conditions that might exist to. the rights, use and maintenance of Folly Quarter Road and Frederick Road, Maryland Route 144, to Howard County, Maryland and the Maryland State Highway Administration.

SUBJECT TO AND TOGETHER WITH all conditions, covenants, easements, restrictions and rights-of-way of record, which may apply.

8

This description was prepared without the benefit of a Title Report,

Prepared By Checked By

WO #05022-3002

FISHER, COLLINS & CARTER, INC. CARTER, INC. CARTER, INC.

16272 Bettimoru Helkozai Pita Eliisoli City, Kasyked 21042 (410) 461-2833 Fez (410) 760-3784 HOWARD COUNTY CIRCUIT COURT (Land Records) MDR 12722, p. 0409, MSA_CE53_12713. Date available 10/01/2010. Printed 06/04/2020.

I:\2005\05022\dwg\05022-3001 Parcel 1 EXhibit.dwg, 3/18/2010 9:59:18 AM, dstreett, 1:1



UNR 12722 FILM 409

1:\2005\05022\dvg\05022-3001 Parcel 2 Exhibil.dvg, 3/18/2010 9:49:20 AM, dsbaett, 1:1



LINER | 2722 FOLIDIA | 0

Exhibit B

Commitment Letter to Acquire Agricultural Land Preservation Easement (Four Pages) .

.



HOWARD COUNTY OFFICE OF COUNTY EXECUTIVE 3430 Courthouse Drive Ellicott City, Maryland 21043 # 410-313-2013

Ken Ulman Howard County Executive www.howardcountymd.us PAX 410-313-3051 TDD 410-313-2323

May 18, 2010

<u>VIA OVERNIGHT MAIL</u> Camilla Carroll and Philip Carroll c/o Camilla Carroll 3500 Manor Lane Ellicott City, Maryland 21042

Dear Ms. Carroll and Mr. Carroll:

Howard County, Maryland is pleased to offer to purchase the development rights as defined in Section 15.502 of the Howard County Code ("Development Rights") on the parcel of land consisting of approximately 500 scres, more or less, which you own in Howard County, Maryland and which is located at 3500 Manor Lane, Ellicott City, Howard County, Maryland (the "Land") pursuant to (i) an installment purchase agreement (the "Installment Purchase Agreement") between Howard County, Maryland, as purchaser (the "County") and Camilla "Installment Purchase Agreement") between Howard County, Maryland, as purchaser (the "County") and Camilla Carroll and Philip Carroll, as sellers (the "Seller") and (ii) a Deed of Agricultural Land Preservation Easement between the County and the Seller (the "Deed of Basement"), upon and subject to the terms and conditions hereinafter set forth:

1. <u>Purchase Price</u>. The County offers to purchase the Development Rights from the Seller for a total purchase price of no more than \$19,100,000.00, (the "Purchase Price") which is based on a price of \$38,200.00 per agre, rounded to the next highest \$1,000.00. The Seller will receive 10% of the Purchase Price at settlement. The balance of the Purchase Price shall be paid to the Seller or Seller's assignee in 20 equal annual installments commencing on August 15, 2011 and on each August 15 thereafter. This Purchase Price is based on the following agreements and conditions with respect to the Land which shall be set forth in the Deed of Hasement:

- (a) The Land is currently a portion of a parcel of record. The Land may be divided into no more than five (5) 50+ acre parcels;
- (b) Ten (10) one-acre lots may be subdivided from the Land upon the approval of the Agricultural Land Preservation Board and the Department of Planning and Zoning; and
- Land Preservation Board and the Department of r failing and upon the approval of the
 (c) A maximum of fifteen (15) tenent houses may be permitted on the Land upon the approval of the
 Agricultural Land Preservation Board and the Department of Planning and Zoning.

2. Interest on Unpaid Balance of Purchase Price. Interest on the unpaid balance of the Furchase Price shall accrue from the date of settlement ("Closing Date") and shall be paid to the Seller or Seller's assignee in 40 semiannual payments on February 15 and August 15 in each year after the Closing Date. Interest shall accrue and be payable at (a) the interest rate, as determined by the County's Director of Finance at the time of settlement of the first Batch 14 property to settle, which is equal to the greater of the avorage of the United States Treasury Yields or AAA Tax-Exempt General Obligation Yields, in each case for the closest available date to each principal payment installment date for the first Batch 14 property to settle, or (b) 4% per annum, whichever is lower. Interest shall be calculated on the basis of a 360-day year of twelve 30-day months.

3. <u>Documentation</u>. All instruments and documents required hereby or affecting the Land, or relating to the Seller's capacity and authority to sell the Development Rights and to execute the Documents and such other documents, instruments, opinions, assurances, consents and approvals as the County may request and all procedures connected herewith shall be subject to the approval, as to form and substance, of the County, the

HOWARD COUNTY CIRCUIT COURT (Land Records) MDR 12722, p. 0412, MSA_CE53_12713. Date available 10/01/2010. Printed 06/04/2020.

Camilla Carroll and Philip Carroll property May 18, 2010 Page 2 of 4

County's counsel, and Miles & Stockbridge P.C. ("Bond Counsel"). All of the documents evidencing the installment purchase of the Development Rights (the "Documents"), including without limitation, the Installment Purchase Agreement and the Deed of Basement, shall be prepared by counsel for the County or by Bond Counsel. In addition to those items specifically set forth in this commitment, the Seller shall furnish to the County, prior to the Closing Date, such additional instruments, documents, opinions and materials as the County may require, all of which shall be satisfaotory to the County in all respects.

4. <u>Conditions Precedent to Closing</u>. As a condition precedent to the County's obligation to close, not less than seven (7) days prior to the Closing Date:

(a) The Seller shall furnish to the County, a Subordination Agreement in recordable form from each mortgagee and other lienholder having a lien on all or any portion of the Land subordinating such person's interest in the Land to the rights of the County under the Deed of Easement, which Subordination Agreement shall be satisfactory in all respects to the County.

(b) If the Seller is not a natural person acting only in the Seller's individual capacity, Seller shall furnish to the County an opinion of Seller's Counsel, which is satisfactory in form and substance to the County, stating that Seller has the authority to convey the development rights to the County and has taken all actions necessary to validly exercise that authority.

(c) The County shall have received a title insurance binder with a commitment to issue a title insurance policy in the amount of the Purchase Price insuring the acquisition of the assement on the Land and ownership of the Development Rights by the County subject only to those exceptions to title as are approved by the County and its counsel, and with affirmative insurance on such matters as the County may require. The County will secure the title insurance.

(d) The County shall have verified with the Soil Conservation District that the Land is subject to an approved Soil Conservation and Water Quality Plan that reflects current conditions and activities on the Land.

5. <u>Expenses</u>. All costs relating to the recording of the Deed of Easement and any Subordination Agreement, all title examination charges, the premium for the title insurance policy, and the fees of Bond Counsel for a reasonable number of hours of time expended on consultation with legal or financial advisors of the Seller and the out-of-pocket expenses of Bond Counsel shall be paid by the County.

The Seller shall pay the fees and expenses of the Seller's own counsel and, if necessary, licensed engineer or surveyor, and all costs of preparation and recording of other documents, if any, required to perfect the title to the Land and provide a metes and bounds description for the Deed of Basement.

6. <u>Termination by County</u>. This commitment is being made in reliance upon the information supplied by the Seiler to the County in connection with the sale of the Development Rights. If the County, acting in good faith, should determine that any such information or supporting representation of a material nature is false, inaccurate, incomplete or misleading, the County may resolud and cancel this commitment.

7. <u>Brokerage</u>. The County shall pay no fee or commission to any broker or agent in connection with the purchase of the Development Rights, and the Seller hereby agrees to indemnify and hold harmless the County against all claims for brokerage fees and commissions.

8. <u>Receipt of Opinion of Bond Counsel</u>. It is a condition precedent to the closing of the transaction contemplated hereby that the County and the Seller receive an opinion from Bond Counsel, dated the Closing Date, to the effect that under existing laws, regulations, rulings and decisions, interest paid under the Installment Purchase Agreement is not includable in the gross income of the Seller (or any holder of the Installment Purchase Agreement) for federal income tax purposes, which opinion may assume continuous compliance with certain

сC

Camilia Carroll and Philip Carroll property May 18, 2010 Page 3 of 4

covenants in the Tax Certificate and Compliance Agreement to be executed and delivered by the County on the date of delivery of the Certificate and may be otherwise limited in accordance with its terms.

9. <u>Acknowledgment of Seller with Regard to Tax Consequences of Transaction</u>. The Seller acknowledges that the Seller has made an independent investigation and has consulted with attorneys, accountants and others selected by the Seller with respect to all tax considerations related to the transaction contemplated hereby (other than the matter described in Section 8 hereof), and the Seller certifies that the Seller has not looked to or relied upon the County or any of its officials, agents or employees, or to Bond Counsel, with respect to any of such matters.

10. <u>Commitment Subject to Approval by Seller of Final Documents</u>. Final Documents for execution, consistent with this commitment, shall be satisfactory to the Seller in form and substance. In the event that the Documents are not acceptable to Seller for any reason (including the fullure to execute a Development Rights and Responsibility Agreement) and cannot be made so, or this transaction is not acceptable to the Seller for any reason, Seller, at Seller's option, shall be released from this commitment.

11. <u>Acceptance and Counterparts</u>. To accept this commitment, a copy of this commitment must be signed (and the signature witnessed) by each person with an ownership interest in the property and the copy with original signature of the Seller, delivered to the County po later than May 20, 2010. This commitment may be executed in counterparts, each of which shall be considered an original and all of which shall, together, constitute a single instrument.

12. <u>Commitment Subject to Enactment of Approval Ordinance</u>. Because the Installment Purchase Agreement is a multi-year contract subject to the provisions of Section 612 of the County's charter, the Installment Purchase Agreement must be approved by ordinance ("Approval Ordinance") and the County is not authorized to execute and deliver the Installment Purchase Agreement until the Approval Ordinance is enacted. The Approval Ordinance will be prefiled for introduction to the County Council of Howard County (the "Council") on May 27, 2010, for the Approval Ordinance to be considered by the Council in June. If for any reason the Approval Ordinance is not enacted by October 1, 2010, this commitment shall be null and void and the County and Seller shall have no further obligation hereunder.

13. <u>Assignment Prohibited</u>. This commitment may not be assigned or in any way transferred by the Seller.

14. <u>Entire Agreement</u>. No statements, agreement or representations, oral or written, which may have been made to the Seller or to any employee or agent of the Seller, either by the County or by any employee, agent or broker acting on the Seller's behalf, with respect to the purchase of the Development Rights on the Land, including the Purchase Price, the Deed of Easement, or Installment Purchase Agreement, shall be of any force or effect, except to the extent stated in this commitment, and all prior agreements and representations with respect to the matters in this commitment are merged herein. This commitment may not be changed except by written agreement signed by the Seller and the County.

15. <u>Governing Law</u>. The Seller agrees that this commitment and the Documents shall be governed by and construed under the laws of the State of Maryland.

16. <u>Closing Date: Survival</u>. This transaction may be closed after all conditions precedent to closing have been met. Unless the Seller and the County enter into a Development Rights and Responsibilities Agreement concerning the Land and the Seller's property adjacent to the Land, this transaction must be fully closed by February 1, 2011 or this commitment shall be deemed null and void. Unless otherwise agreed to by the County, the closing shall take place in the main office of the County or at the office of Bond Counsel in Baltimore, Maryland, as the County might designate. The terms and conditions of this conmitment shall survive the closing; provided, however, that if any of the terms and conditions of this Camilla Carroll and Philip Carroll property May 10, 2010 Page 4 of 4

commitment shall conflict with any of the terms and conditions of the Documents, the terms and conditions of the Documents shall prevail. The terms of this commitment shall supersede in full any prior commitment issued by the County in connection with the transaction contemplated hereby,

We are pleased to make this offer to you. Enclosed for your review is an updated score sheet and amortization schedule. Also attached is a sample deed of easement and a sample installment purchase agreement. Please indicate your acceptance of this commitment by signing and returning to us one of the executed originals of this letter no later than May 20, 2010. If not accepted prior to May 21, 2010, the offer set forth in this commitment letter terminates automatically and shall have no further force and effect.

ATTEST:

omie Robbins

Chief Administrative Officer

Approved for Form and Legal Sufficiency this 20 day of 2010:

Mafgaret/Ann Nolan

County Solicitor

HOWARD COUNTY, MARYLAND

By: Ken Ulman

County Executive

THE FOREGOING TERMS AND CONDITIONS WITH RESPECT TO THE LAND ARE HEREBY AGREED TO AND ACCEPTED THIS DAY OF 2010.

Witness:

Witness:

arrol

Philip Carroll

LIBER | 2722 FOLIOL | 6

				Compa	ann 101	speen		minov				
	ARD COUNTY, MARYLAN			ND	*	IN THI	E					
3430 Courthouse Drive Ellicott City, Maryland 21043						*	CIRCL	ЛТ СО	URT			
Plaintiff						*	FOR					
v.						*	HOWA	ARD C	OUNI	ſΥ		
CAMILLA CARROLL						*						
3500 Manor Lane Ellicott City, Maryland 21042					*	Case N	∛o.:					
SERV			Fra			*	·					
	Sang V 5100 I Ellicot	Dorsey	Hall D			*						
						*						
and					*							
PHILIP D. CARROLL 3500 Manor Lane												
Ellicott City, Maryland 21042						*						
SERVE ON:					*							
Sang W. Oh, Esq.						*					·	
5100 Dorsey Hall Drive Ellicott City, MD 21042					*							
Defendants						*						
	*	*	*	*	*	*	*	*	*	*	*	*

Exhibit C Complaint for Specific Performance

VERIFIED COMPLAINT FOR SPECIFIC PERFORMANCE

Howard County, Maryland (the "County"), by its attorneys, Margaret Ann Nolan, County Solicitor, and Gary W. Kuc, Senior Assistant County Solicitor, sues Camilla Carroll and Philip D. Carroll (the "Carrolls"), Defendants, for specific performance of

LIDER 12722 FOLIO 4 17

their promise to execute and deliver to the County a Deed of Agricultural Land Preservation Easement for 500 acres.

PARTIES

1. Plaintiff Howard County, Maryland is a body corporate and politic of the State of Maryland.

2. Defendant Camilla Carroll is a resident of Howard County, Maryland.

3. Defendant Philip D. Carroll is a resident of Howard County, Maryland.

4. The Carrolls own certain real property in Howard County, Maryland, described on Howard County Tax Map No. 23, Parcel 71 consisting of 892.6 acres \pm ("Doughoregan" or the "Property").

JURISDICTION

5. This Court has subject matter jurisdiction over this equitable action for specific performance pursuant to § 1-501 of the Courts and Judicial Proceedings Article of the Annotated Code of Maryland.

6. This Court has personal jurisdiction over the Carrolls pursuant to § 6-102 and § 6-103 of the Courts and Judicial Proceedings Article of the Annotated Code of Maryland.

7. Furthermore, the Carrolls have waived any defense based on lack of subject matter jurisdiction or lack of personal jurisdiction, pursuant to § 4 of an Option Agreement between the parties. A true and correct copy of the fully executed Option Agreement is attached hereto and incorporated as Exhibit A.

2

VENUE

8. Venue is proper in this Court pursuant to § 6-201 and § 6-202 of the Courts and Judicial Proceedings Article of the Annotated Code of Maryland.

9. Furthermore, the Carrolls have waived any defense based on venue pursuant to § 4 of the Option Agreement between the parties.

FACTS

10. The Howard County Agricultural Land Preservation Act is codified in § 15.500 through § 15.521 of the Howard County Code (the "Act"). A true and correct copy of the Act is attached hereto and incorporated as Exhibit B.

11. The purpose of the Act is to protect the health, safety, and well-being of present and future residents of Howard County by conserving and protecting agricultural land as an environmental resource of major importance and as the basis of a viable sector of the County's economy. § 15.500(c).

12. To effectuate the Act's purpose, the County acquires the development rights on eligible land by purchase of the land, by purchase of the development rights on the land, and by acceptance of dedicated remainders from cluster subdivisions as provided by law. § 15.500(c).

13. The County may acquire an agricultural land preservation easement on land in the County by, among other things, purchasing the development rights on eligible land. § 15.503(a).

3

LIBER | 2722 FOLIO 4 | 9

14. The parties have a valid and enforceable contract for the purchase by the County of the Carrolls' development rights in 500 acres of Doughoregan. The contract consists of a number of written agreements between the parties.

15. The first agreement is a Commitment Letter dated May 18, 2010. A true and correct copy of the fully executed Commitment Letter is attached hereto and incorporated as Exhibit C. The Commitment Letter contains the terms and conditions of the County's offer to purchase the Carrolls' development rights in their Agricultural Preservation Parcel, including the purchase price of \$19,100,000.00; an Installment Purchase Agreement concerning payment of the purchase price; and certain agreements and conditions for the land that is to be set forth in the Deed of Agricultural Land Preservation Easement (the "Deed of Easement"). A true and correct copy of the Deed of Easement is attached hereto and incorporated as Exhibit D.

16. On September ____, 2010, the parties entered into an Option Agreement. A true and correct copy of the fully executed Option Agreement is attached hereto and incorporated as Exhibit A. The Option Agreement grants the County the exclusive option to obtain the perpetual Deed of Easement in, over, and through the Agricultural Preservation Parcel described in the Commitment Letter, as amended by the parties' Development Rights and Responsibilities Agreement.

17. Contemporaneous with execution of the Option Agreement, the parties entered into a Development Rights and Responsibilities Agreement (the "DRRA"). A true and correct copy of the fully executed DRRA is attached hereto and incorporated as Exhibit E. The DRRA modifies the Commitment Letter by deleting paragraph 10. That

LIBER | 2722 FOLIO + 20

paragraph had granted the Carrolls to right to be released from the Commitment Letter if the final documents for execution were not satisfactory to them in form and substance.

18. By Council Bill 32-2010 adopted by the County Council of Howard County on July 6, 2010 and signed by the County Executive on July 8, 2010, the County approved the Installment Purchase Agreement. A true and correct copy of the approved Installment and Purchase Agreement is attached hereto and incorporated as Exhibit F.

19. All preconditions for the County's right to exercise its Option have occurred. Option Agreement § 1.2.

20. On [date to be inserted upon default], the County exercised its Option by giving written notice to the Carrolls. Option Agreement § 1.2. A true and correct copy of the notice is attached hereto and incorporated as Exhibit G.

21. The exercise of the Option entitled and obligated the County to purchase the Deed of Easement from the Carrolls, and obligated the Carrolls to sell the Deed of Easement to the County. Option Agreement § 1.3.

22. Upon exercise of the Option, the parties were required to set a mutually agreeable date for settlement to occur within ten (10) business days of the exercise of the Option. Option Agreement § 2.1.

23. The Carrolls failed to agree to a settlement date and consummate settlement, despite the County's demand to do so on date to be inserted upon default]. A true and correct copy of the demand is attached hereto and incorporated as Exhibit H.

24. At all relevant times, the County was ready, willing, and able to perform under the terms of the parties' contract and all preconditions for settlement were satisfied.

25. The Carrolls failed to schedule and attend settlement and deliver the Deed of Easement to the County and take the necessary action to satisfactorily perform under the parties' contract.

26. The Carrolls agree that there are no liquidated damages or other types of monetary damages that would be adequate to compensate the County for their failure to perpetually conserve and protect the 500 acres \pm pursuant to the Act as promised. Option Agreement § 4.

27. The Carrolls agree that the County entered into the Commitment Letter, the Option Agreement, and the DRRA because of the Carrolls' promise to conserve and protect the $500 \pm$ acres pursuant to the Act. Option Agreement § 4.

COUNT I -- SPECIFIC PERFORMANCE

28. The Plaintiff adopts by reference the allegations contained in paragraphs 1 through 27 of this Complaint with the same effect as if fully set forth.

29. The County and the Carrolls have a valid and enforceable contract for the purchase by the County of the Carrolls' development rights in $500 \pm acres$ of Doughoregan as set forth in the Deed of Easement.

30. The Carrolls breached the contract by failing to agree to settle and by not consummated settlement.

31. The County was ready, willing, and able to perform under the terms of the contract and all preconditions for settlement were satisfied.

32. The County has no adequate remedy at law.

WHEREFORE, Plaintiff Howard County, Maryland demands judgment that the contract between the parties be specifically enforced, and that the Defendants be ordered to execute and deliver the Deed of Easement to the County and such other documents necessary for the County to acquire the development rights on the 500 cares, more or less of Doughoregan, and for such other relief as the Court deems proper.

VERIFICATION

I solemnly affirm under the penalties of perjury that the contents of the foregoing Verified Complaint for Specific Performance are true to the best of my knowledge, information, and belief.

HOWARD COUNTY, MARYLAND

Date: _____

By:

County Executive

Respectfully submitted,

HOWARD COUNTY OFFICE OF LAW

Margaret Ann Nolan County Solicitor

Gary W. Kuc Senior Assistant County Solicitor George Howard Building 3430 Courthouse Drive Ellicott City, Maryland 21043 Counsel for Plaintiff 410-313-2103 410-212-3292

Date: _____

Exhibit D
Consent Decree for Specific Performance

HOWARD COUNTY, MARYLAND	*	IN THE				
Plaintiff	*	CIRCUIT COURT				
V.	*	FOR				
CAMILLA CARROLL, ET AL.	*	HOWARD COUNTY				
Defendants	*	Case No.:				
* * * * *	*	* * * * *				

CONSENT DECREE FOR SPECIFIC PERFORMANCE

Plaintiff, Howard County, Maryland (the "County") and Defendants, Camilla Carroll and Philip D. Carroll (the "Carrolls"), hereby represent and acknowledge that they agree to enter into this Consent Decree (collectively, the "Parties").

1. The Carrolls own certain real property in Howard County, Maryland, described on Howard County Tax Map. No. 23, Parcel 71 consisting of $892.6 \pm acres$ (the "Property"). The Carrolls intend to develop a 221.1 acre \pm portion of the Property with single family detached residential dwellings. The Howard County Agricultural Land Preservation Act is intended to protect the health, safety, and well-being of present and future residents of Howard County by conserving and protecting agricultural land as an environmental resource of major importance and as the basis of a viable sector of the County's economy. Howard County Code §§ 15.500 – 15.521 (the "Act").

2. The Parties entered into a contract for the purchase by the County of the Carrolls' development rights in $500 \pm acres$ of the Property pursuant to the Act. In exchange, the Carrolls agreed to execute and deliver to the County a Deed of Agricultural Land Preservation Easement regarding these rights (the "Deed of Easement").

12 CC

HOWARD COUNTY CIRCUIT COURT (Land Records) MDR 12722, p. 0424, MSA_CE53_12713. Date available 10/01/2010. Printed 06/04/2020.

HOWARD COUNTY CIRCUIT COURT (Land Records) MDR 12722, p. 0425, MSA_CE53_12713. Date available 10/01/2010. Printed 06/04/2020.

3. Howard County filed a Verified Complaint for Specific Performance against the Carrolls (the "Complaint"). The County alleged that all preconditions for settlement had occurred and the County was ready, willing, and able to perform the contract, and that the Carrolls breached the contract by failing to agree to a settlement date and consummate settlement regarding the Deed of Easement. The County demands judgment that the contract between the Parties be specifically enforced, and that the Carrolls be ordered to execute and deliver to the County the Deed of Easement. Contemporaneously with the Complaint, the County filed a Consent Motion for Entry of Consent Decree for Specific Performance, with a copy of this order attached (the "Consent Motion").

4. Prior to the commencement of this action, the Carrolls consented to the County suing them for specific performance in the event of their breach of the contract, and they knowingly, intelligently, and voluntarily waived all rights, defenses, and claims, from whatever source derived, both procedural and substantive, that they may have to the action, except for service of process and filings in the action. *See* Option Agreement § 4, attached as Exhibit A to the Complaint. The record in this action indicates that the County served the Complaint and Writs of Summonses, as well as the Consent Motion, on the appointed and authorized agent of the Carrolls.

5. The Parties have reached agreement concerning resolution of this action and consent to entry of this Consent Decree for Specific Performance.

NOW, THEREFORE, IT IS HEREBY ORDERED, ADJUDGED AND DECREED AS FOLLOWS:

2

NCC

6. That the Consent Motion be and the same is hereby GRANTED;

7. That within ten (10) business days of the entry of this Consent Decree the Carrolls execute and deliver to the County the Deed of Easement;

8. That contemporaneously with the Carrolls' delivery of the Deed of Easement, the County pay the Carrolls in accordance with the terms and conditions of the Parties' contract.

9. That this Consent Decree for Specific Performance shall automatically terminate and this action shall be closed upon the County's recordation of the Deed of Easement in this Land Records of this Court, subject to payment of any open court costs.

IT IS SO DECREED this _____ day of ______, 20__:

Judge, Circuit Court for Howard County

IT IS SO AGREED AND CONSENTED TO:

9,9,2010 Date

IT IS SO AGREED AND CONSENTED TO:

PHILIP D. CARROLL

IT IS SO AGREED AND CONSENTED TO:

9/ 10 Date

[NAME] 25 06 Attorneys for the Carrolls [NAME] Oh

IT IS SO AGREED AND CONSENTED TO:

Date

[NAME] Administrator, Howard County Agricultural Land Preservation Program

IT IS SO AGREED AND CONSENTED TO:

Date

Margaret Ann Nolan, County Solictor Gary W. Kuc, Senior Assistant County Solicitor



EXHIBIT 🖗

March 18, 2010

DESCRIPTION OF A 500.000 ACRE AGRICULTURAL PRESERVATION EASEMENT CONSISTING OF 241.638 ACRE PART ONE AND 258.362 ACRE PART TWO PART OF THE PROPERTY OF PHILIP D. CARROLL AND CAMILLA CARROLL SECOND AND THIRD ELECTION DISTRICTS HOWARD COUNTY, MARYLAND

BEING a parcel of land, situate and lying in the Second and Third Election Districts of Howard County, Maryland, said parcel of land being part of the land which by Will of Nina R. Carroll dated June 4, 1979, as amended by its First Codicil dated November 28, 1986 and its Second Codicil dated April 10, 1987; Nina R. Carroll having departed this life on February 11, 1989; said Will having been probated in the Surrogates Court of New York County, New York on April 11, 1989, and the Estate of Nina R. Carroll is filed in the Register of Wills of Howard County, Maryland as Estate No. 17-7868-276, said property also being described in a Mortgage dated October 31, 1962 and recorded among the Land Records of Howard County, Maryland in Liber 394 at Folio 64, and being part of Parcel 1 described therein; said parcel being more particularly described, as now surveyed in the Maryland State Coordinate System NAD' 83 Datum, as projected by Howard County Geodetic Control, as follows:

FISHER, COLLINS & CARTER, INC.

CIVIL ERODIEERING CONSIGTANTS And LAND SURVEYORS

10272 Halfinsola Hallondi Pike Elikoon Oliy, Maryland 21042 1410) 401-2815 Fax (410) 750-3784 (PART ONE) BEGINNING FOR THE SAME at a point on or near the intersection of the centerlines of existing paving of Folly Quarter Road with Frederick Road (Maryland Route 144); said point being at the beginning of the Seventeenth or South 82°36'44" East, 189.50 foot Deed Line of the aforesaid Liber 394 at Folio 64 (Parcel 1); thence running within said Frederick
-LIBER' | 2722 FOLIOL 30

Road, on or near the centerline of paving, and running with and binding on all of the said Seventeenth line of said Liber 394 at Folio 64 (Parcel 1), and with all of the Eighteenth, Nineteenth and a portion of the First line of said Liber 394 at Folio 64 (Parcel 1), as now surveyed the following four (4) courses and distances:

- 1) South 82°54'28" East, 189.50 feet to a point; thence
- 2) North 17°12'33" West, 32.08 feet to a point; thence
- 3) South 82°35'58" East, 1,394.98 feet to a point; thence
- 4) South 73°02'02" East, 2.22 feet to a point on and being a distance of 3,137.87 feet from the end of the aforesaid First or South 73°02'48" East, 3140.09 foot Deed Line of the aforesaid Liber 394 at Folio 64 (Parcel 1)and also being the Northern margin of the sixty-six foot wide Right of Way of Frederick Road (Maryland Route 144); thence leaving said First line of Liber 394 at Folio 64 (Parcel 1) and the Northern margin of Frederick Road (Maryland Route 144) to run over, across and through the aforesaid Liber 394, Folio 64 the following thirteen (13) courses and distances:
- 5) South 06°25'26" West, 152.29 feet to a point; thence
- 6) South 04°05'30" West, 282.20 feet to a point of curvature; thence
- 7) 253.27 feet along the arc of a non-tangential curve to the left, having a radius of
 1,301.42 feet, a central angle of 11°09°01" and subtended by a chord bearing and
 distance of South 02°54'01" East, 252.87 feet to a point; thence
- 8) South 07°27'35" East, 191.10 feet to a point; thence
- 9) South 09°24'23" East, 392.33 feet to a point; thence
- 10) South 09°00'09" East, 293.76 feet to a point; thence

EL CARTER, INC. AVIL INCULERANS CONSULTANTS IN LAND SERVEY/08

FISHER, COLLINS

19272 Ballinora Helional Pito Elicon City, Meryland 21042 (410) 461-2855 >ax (410) 760-3784

LIBER 12722 FOLIO 431

- 11) South 09°47'54" East, 393.83 feet to a point; thence
- 12) South 09°03'17" East; 615.16 feet to a point; thence
- 13) South 09°03'55" East, 291.12 feet to a point; thence
- 14) North 79°49'03" West, 1,832.40 feet to a point; thence
- 15) South 19°08'23" West, 1,237.51 feet to a point; thence
- 16) North 77°07'13" West, 68.06 feet to a point; thence
- South 19°08'11" West, 1,248.08 feet to a point on and at a distance of 279,07 feet from a 1.25 inch diameter Iron Pipe found at the beginning of the Tenth or North 70°15'08" West, 1456.95 foot Deed Line of the aforesaid Liber 394 at Folio 64 (Parcel 1); thence running with and binding on the remaining portion of said Tenth line of Liber 394 at Folio 64 (Parcel 1), as now surveyed, the following course and distance:
- 18) North 70°10'26" West, 1,177.65 feet to a 1.25 inch diameter Iron Pipe found at the end thereof; thence running with and binding on the Eleventh, Twelfth, Thirteenth, Fourteenth, Fifteenth and Sixteenth lines of said Liber 394 at Folio 64 (Parcel 1), the following six (6) courses and distances:
- 19) North 78°34'25" West, 791.22 feet to a point; thence
- 20) North 73°34'25" West, 387.55 feet to a point; thence
- 21) South 69°10'35" West, 31.37 feet to a point; thence
- 22) North 51°40'56" East, 568.09 feet to a point; thence
- 23) North 39°59'12" East, 2,458.50 feet to a point; thence
- North 26°28'43" East, 2,376.00 feet to the point of beginning, containing 241.638
 acres, more or less.

HOWARD COUNTY CIRCUIT COURT (Land Records) MDR 12722, p. 0431, MSA_CE53_12713. Date available 10/01/2010. Printed 06/04/2020.

COLLINS & CARTER, INC. GVIL ENCODERING CONSULTANTS and LAND SUBVEYINS 10272 Builtmure Mellocul Pilas EHILEDI CHY, Marriand 21D42 (410) 161-2485 Fox (410) 760-3784

FISHER.

'110ER | 2722 FOLIO 432

(FART TWO) BEGINNING FOR THE SAME at a point on the Northern margin of the sixty-six foot Right of Way of Frederick Road (Maryland Route 144), said point being on and a distance of 53.08 feet from the beginning of the First or South 73°02'48" East, 3140.09 foot Deed Line of the aforesaid Liber 394 at Folio 64 (Parcel 1); thence running with and binding on a portion of said First line of Liber 394 at Folio 64 (Parcel 1), and the Northern margin of Fréderick Road (Maryland Route 144), the following course and distance:

- South 73°02'02" East, 1,379.85 feet to a point thereon; thence leaving the aforesaid First line of Liber 394 at Follo 64 and the Northern margin of Frederick Road (Maryland Route 144) and running for new lines of division, the following twenty-five (25) courses and distances:
- 2) South 16°57'58" West, 65.50 feet to a point; thence
- 3) South 73°02'02" East, 328.95 feet to a point; thence
- 4) South 19°32'01" East, 18.41 feet to a point; thence
- 5) South 73°08'00" East, 180.91 feet to a point; thence
- 6) South 27°46'18" East, 35.58 feet to a point; thence
- 10) South 17°35'24" West, 225.85 feet to a point of curvature; thence
- 346.71 feet along the arc of a curve to the left, having a radius of 315.00 feet, a central angle of 63°03'47" and subtended by a chord bearing and distance of South 13°56'30" East, 329.47 feet to a point of reverse curvature; thence
- 121.72 feet along the arc of a curve to the right, having a radius of 150.00 feet, a central angle of 46°29'32" and subtended by a chord bearing and distance of South 22°13'37" East, 118.40 feet to a point of tangency; thence

HOWARD COUNTY CIRCUIT COURT (Land Records) MDR 12722, p. 0432, MSA_CE53_12713. Date available 10/01/2010. Printed 06/04/2020. FISHER COLLINS ß CARTER, INC. LEVIL ENROLLERING CONSULTAITS and LAND SURVEYORS 10272 Bollinsco National Piko Ellium City, Karyiand 21042 (410) 461-2834 Faz (410) 758-3784

LIBER | 2722 FOLIOL 33

10)	South 01°01'09" West, 232.32 feet to a	point of curvature; thence
-----	--	----------------------------

312.10 feet along the arc of a curve to the left, having a radius of 330.00 feet, a central angle of 54°11'16" and subtended by a chord bearing and distance of South 26°04'29" East, 300.60 feet to a point of tangency; thence

12) South 53°10'07" East, 259.20 feet to a point of curvature; thence

- 470.43 feet along the arc of a non-tangential curve to the right, having a radius of
 595.00 feet, a central angle of 45°18'00" and subtended by a chord bearing and
 distance of South 30°31'07" East, 458.27 feet to a point of compound curvature;
 thence
- 14) 283.12 feet along the arc of a curve to the right, having a radius of 275.00 feet, a central angle of 58°59'14" and subtended by a chord bearing and distance of South 21°37'30" West, 270.78 feet to a point of reverse curvature; thence
- 15) 219.63 feet along the arc of curve to the left, having a radius of 285.00 feet, a central angle of 44°09'14" and subtended by a chord bearing and distance of South 29°02'30" West, 214.24 feet to a point of tangency; thence
- 16) South 06°57'53" West, 62.64 feet to a point of curvature; thence
- 17) 89.85 feet along the arc of a curve to the left, having a radius of 285.00 feet, a central angle of 18°03'45" and subtended by a chord bearing and distance of South 02°03'59" East, 89.47 feet to a point of reverse curvature; thence
- 131.70 feet along the arc of a curve to the right, having a radius of 170.00 feet, a
 central angle of 44°23'10" and subtended by a chord bearing and distance of
 South 11°05'43" West, 128.43 feet to a point of reverse curvature; thence

HOWARD COUNTY CIRCUIT COURT (Land Records) MDR 12722, p. 0433, MSA_CE53_12713. Date available 10/01/2010. Printed 06/04/2020.

FISHER,

COLLINS

CARTER,

CIVIL ENRIGEERING CONSCIANTS ANG LAND SURVEYORS

1027£ Battimon Hational Pika Dilati Dily, Maryland 21042 (410) 461-2855 Fax (410) 750-5784

5

LIBER | 2722 FOLIO434

· .		
	19)	136.14 feet along the arc of a curve to the left, having a radius of 325.00 feet, a
		central angle of 24°00'04" and subtended by a chord bearing and distance of
		South 21°17'16" West, 135.15 feet to a point of tangency; thence
	20)	South 09°17'14" West, 113.42 feet to a point of curvature; thence
	21)	468.79 feet along the arc of a curve to the left, having a radius of 580.00 feet, a
		central angle of 46°18'34" and subtended by a chord bearing and distance of
		South 13°52'03" East, 456.13 feet to a point of tangency; thence
	22)	South 37°01'20" East, 903.13 feet to a point; thence
	23)	South 23°09'51" East, 507.53 feet to a point; thence
	24)	South 67°04'12" West, 462.42 feet to a point; thence
	25)	South 33°51'48" West, 651.81 feet to a point; thence
	26)	South 16°43'26" West, 858.48 feet to a point on and a distance of 1,764.49 feet
	,	from a Granite Stone found and held at the beginning of the Seventh or North
		73°14'48" West, 4021.13 foot Deed Line of the aforesaid Liber 394 at Folio 64
		(Parcel 1); thence running with and binding on the remaining portion of said
		Seventh line of Liber 394 at Folio 64 (Parcel 1) and with all of the Eighth and a
		portion of the Ninth or North 65°30'08" West, 2143.94 foot Deed Line of said
•		Liber 394 at Folio 64 (Parcel 1), the following three (3) courses and distances:
	27)	North 73°16'34" West, 2,253.96 feet to a point; thence
FISHER, COLLINS	28)	North 14°04'34" East, 24.75 feet to a point; thence
A CARTER,	29)	North 65°25'29" West, 39.81 feet to a point; said point being on and a distance of
		2,104,13 feet from a 1.25 inch Iron Pipe found at the end of the aforesaid Ninth
zad LAND SURVEYORS 10272, Dahlanara Hallonal Pike Silicott City, Katyland 21042 (410) 461-2065 Faz (410) 360-3764		line of Liber 394 at Folio 64 (Parcel 1); said point also being on the easterly
		6
Į.		· ·

HOWARD COUNTY CIRCUIT COURT (Land Records) MDR 12722, p. 0434, MSA_CE53_12713. Date available 10/01/2010. Printed 06/04/2020.

108812722 FOLIO435

outline of a plat entitled "Amended Plat Of Easement, Sending Parcel, Property Of Philip Carroll and Camilla Carroll" and recorded as Plat No. 19928; thence I leaving the aforesaid Ninth line of Liber 394 at Folio 64 (Parcel 1) and running with and binding on part of said easterly outline of Plat No. 19928, the following course and distance:

- 30) North 14°13'15" East, 1,478.90 feet to a point thereon; thence leaving the aforesaid easterly outline of Plat No. 19928 and running over, across and through the aforesaid Liber 394, Folio 64 (Parcel 1) the following twenty-two (22) courses and distances:
- 31) South 85°27'17" East, 421.22 feet to a point; thence
- 32) South 64°29'11" Bast, 192.86 feet to a point; thence
- 33) South 47°22'36" East, 333.45 feet to a point; thence
- 34) South 27°46'03" East, 102.62 feet to a point; thence
- 35) South 73°56'56" East, 224.55 feet to a point; thence
- 36) South 82°25'01" East, 149.17 feet to a point; thence
- 37) North 12°14'25" East, 389.04 feet to a point; thence
- 38) North 23°29'52" West, 478.60 feet to a point; thence
- 39) North 16°15'45" West, 750.96 feet to a point; thence
- 40) North 81°30'55" West, 730.33 feet to a point; thence
- 41) North 00°27'40" East, 207.41 feet to a point; thence
- 42) North 00°40'30" West, 76.57 feet to a point; thence
- 43) North 06°06'01" West, 117.35 feet to a point; thence
- 44) North 09°05'56" West, 310.97 feet to a point; thence



CTVIL ENGLISEERINIII CONSIA TAUTS and LAND SURVEYING

10272 Datimere Hallesei Pike Elikoti City, Marylead 21042 (410) 451-2635 Fax (410) 750-3784

LIBER | 2722 FOLIO 436

- 45) North 09°03'18" West, 615.62 feet to a point; thence
- 46) North 09°47'54" West, 393.78 feet to a point; thence
- 47) North 09°00'07" West, 293.32 feet to a point; thence
- 48) North 09°24'23" West, 391.82 feet to a point; thence

49) North 07°27'35" West, 190.70 feet to a point of curvature; thence

- 50) 243.37 feet along the arc of a curve to the right, having a radius of 1,251.42 feet, a central angle of 11°08'33" and subtended by a chord bearing and distance of North 02°55'27" West, 242.98 feet to a point of tangency; thence
- 51) North 04°05'30" East, 280.56 feet to a point; thence
- 52) North 06°25'26" East, 141.96 feet to the point of beginning; containing 258.362 acres, more or less.

SUBJECT TO AND TOGETHER WITH any and all terms and conditions that might exist to the rights, use and maintenance of Folly Quarter Road and Frederick Road, Maryland Route 144, to Howard County, Maryland and the Maryland State Highway Administration.

SUBJECT TO AND TOGETHER WITH all conditions, covenants, easements, restrictions and rights-of-way of record, which may apply.

8

This description was prepared without the benefit of a Title Report.

Prepared By Checked By

WO #05022-3002

HOWARD COUNTY CIRCUIT COURT (Land Records) MDR 12722, p. 0436, MSA_CE53_12713. Date available 10/01/2010. Printed 06/04/2020. **FISHER** COLLINS & CARTER. INC. CIVIL ENERGIESING CONSILITANTS and LAND SURVITORS 50272 Babimero Nailonal Pike Hikott City, Merylend 21042 (410) 461-2855 Fax (410) 750-3754

HOWARD COUNTY CIRCUIT COURT (Land Records) MDR 12722, p. 0437, MSA_CE53_12713. Date available 10/01/2010. Printed 06/04/2020.

1:12005\05022\dwg\05022-3001 Parcel 1 Exhibit.dwg, 3/18/2010 9:59:18 AM, dstreett, 1:1



UBER | 2722 FOLID 437







LINER | 2722 FOLING 39

EXHIBIT 20

March 18, 2010

DESCRIPTION OF A 94.558 ACRE PARCEL PART OF THE PROPERTY OF PHILIP D. CARROLL AND CAMILLA CARROLL SECOND AND THIRD ELECTION DISTRICTS HOWARD COUNTY, MARYLAND

BEING a parcel of land, situate and lying in the Second and Third Election Districts of Howard County, Maryland, said parcel of land being part of the land which by Will of Nina R. Carroll dated June 4, 1979, as amended by its First Codicil dated November 28, 1986 and its Second Codicil dated April 10, 1987; Nina R. Carroll having departed this life on February 11, 1989; said Will having been probated in the Surrogates Court of New York County, New York on April 11, 1989, and the Estate of Nina R. Carroll is filed in the Register of Wills of Howard County, Maryland as Estate No. 17-7868-276, said property also being described in a Mortgage dated October 31, 1962 and recorded among the Land Records of Howard County, Maryland in Liber 394 at Folio 64, and being part of Parcel 1 described in the aforesaid Liber 394 at Folio 64; said parcel being more particularly described, as now surveyed in the Maryland State Coordinate System NAD' 83 Datum, as projected by Howard County Geodetic Control, as follows:

BEGINNING FOR THE SAME at a point on the Northern margin of the sixty-six foot wide Right of Way of Frederick Road (Maryland Route 144), formerly known as the Baltimore to Frederick Turnpike Road, said point being on and a distance of 2.22 feet from the beginning of the First or South 73°02'48" East, 3140.09 foot Deed Line of the aforesaid Liber 394 at Folio 64 (Parcel 1); thence running with and binding on a portion of said First line of Liber 394 at

FISHER, COLLINS & . CARTER, INC. Hva Encorement constantions wellowed convertions

10278 Balkmore Hellenst Pite Siltent City, Maryland 21042 610) 461-2866 'az (410) 760-3784

LIBER 2722 FOLIO44

Folio 64 (Parcel 1) and the aforesaid Northern margin of Frederick Road (Maryland Route 144) the following course and distance:

- South 73°02'02" East, 50.86 feet to a point thereon, said point being a distance of 3,087.01 feet from the end thereof; thence leaving the aforesaid First line of Liber 394 at Folio 64 (Parcel 1) and the aforesaid Northern margin of Frederick Road (Maryland Route 144) and running over, across and through the aforesaid Liber 394, Folio 64 the following twenty-two (22) courses and distances:
- 2) South 06°25'26" West, 141.96 feet to a point; thence
- 3) South 04°05'30" West, 280.56 feet to a point of curvature; thence
- 243.37 feet along the arc of a curve to the left, having a radius of 1,251.42 feet, a central angle of 11°08'33" and subtended by a chord bearing and distance of South 02°55'27" East, 242.98 feet to a point of tangency; thence
- 5) South 07°27'35" East, 190.70 feet to a point; thence
- 6) South 09°24'23" East, 391.82 feet to a point; thence
- 7) South 09°00'07" East, 293.32 feet to a point; thence
- 8) South 09°47'54" East, 393.78 feet to a point; thence
- South 09°03'18" East, 615.62 feet to a point; thence
- 10) South 09°05'56" East, 310.97 feet to a point; thence
- 11) South 06°06'01" East, 117.35 feet to a point; thence
- 12) South 00°40'30" East, 76.57 feet to a point; thence
- 13) South 00°27'40" West, 207.41 feet to a point; thence
- 14) South 81°30'55" East, 730.33 feet to a point; thence
- 15) South 16°15'45" East, 750.96 feet to a point; thence

2

HOWARD COUNTY CIRCUIT COURT (Land Records) MDR 12722, p. 0441, MSA_CE53_12713. Date available 10/01/2010. Printed 06/04/2020.

FISHER, COLLINS & CARTER,

TVIL ENGLATERING CONSULTANTS nd LAND SURVEYORS

9272 Bahlmore Hallanal Pika Hooti City, Maryland 23042 -10) 481-2885 +1 (410) 750-3784

NC.

	16)	South 23°29'52" East, 478.60 feet to a point; thence
	17)	South 12°14'25" West, 389.04 feet to a point; thence
	18)	North 82°25'01" West, 149.17 feet to a point; thence
	19)	North 73°56'56" West, 224.55 feet to a point; thence
	20)	North 27°46'03" West, 102.62 feet to a point; thence
	21)	North 47°22'36" West, 333.45 feet to a point; thence
,	22)	North 64°29'11" West, 192.86 feet to a point; thence
	23)	North 85°27'17" West, 421.22 feet to a point on the easterly outline of a plat
		entitled "Amended Plat Of Easement, Sending Parcel, Property Of Philip Carroll
		and Camilla Carroll" and recorded among the Land Records of Howard County,
		Maryland as Plat No. 19928; thence running with and binding on the easterly and
		northerly outlines of said plat, the following three (3) courses and distances:
	24)	North 14°13'15" East, 205.25 feet to a point; thence
	25)	North 76°10'22" West, 1,044.43 feet to a point; thence
	26)	North 77°07'13" West, 1,128.53 feet to a point at the northwesterly corner of the
		aforesaid Plat No. 19928; thence leaving the outline of said Plat No. 19928 and
		continuing to run over, across and through the aforesaid Liber 394, Follo 64, the
		following eleven (11) courses and distances:
	27)	North 19°08'23" East, 1,237.51 feet to a point; thence
	28)	South 79°49'03" East, 1,832.40 feet to a point; thence
	29)	North 09°03'55" West, 291.12 feet to a point; thence
	30)	North 09°03'18" West, 615.16 feet to a point; thence
	31)	North 09°47'53" West, 393.83 feet to a point; thence
		3

FISHER, COLLINS & CARTER, INC.

TIVIL ENGLYEERING CONSIATANTS VIII LAND SURVEYORS

i0272 Oshimora Hallonal Pike Illicon City, Maryland 2 (042 410) 487-2158 'az (410) 760-3764

LIDER | 2722 FOLIO 4 4 3

- 32) North 09°00'09" West, 293.76 feet to a point; thence
- 33) North 09°24'23" West, 392.33 feet to a point; thence
- 34) North 07°27'35" West, 191.10 feet to a point of curvature; thence
- 35) 253.27 feet along the arc of a curve to the right, having a radius of 1,301.42 feet, a central angle of 11°09'01" and subtended by a chord bearing and distance of North 02°54'01" West, 252.87 feet to a point of tangency; thence
- 36) North 04°05'30" East, 282.20 feet to a point; thence
- 37) North 06°25'26" East, 152.29 feet to the point of beginning; containing 94.558 acres of land, more or less.

SUBJECT TO AND TOGETHER WITH any and all terms and conditions that might exist to the rights, use and maintenance of Folly Quarter Road and Frederick Road, Maryland Route 144, to Howard County, Maryland and the Maryland State Highway Administration.

SUBJECT TO AND TOGETHER WITH all conditions, covenants, easements, restrictions and rights-of-way of record, or proposed, which may apply.

This description was prepared without the benefit of a Title Report.



Prepared By Checked B

WO #05022-3002

HOWARD COUNTY CIRCUIT COURT (Land Records) MDR 12722, p. 0444, MSA_CE53_12713. Date available 10/01/2010. Printed 06/04/2020.

1:12005\05022\dwg\05022-3001 Parcel 4 Exhibit.dwg, 3/18/2010 9:54:00 AM, dstreett.



UBER | 2722 FOLID4 4 4

Attachment B - First Amendment to Doughoregan DRRA

LAW OFFICES OF TALKIN & OH, LLP COLUMBIA OFFICE 5100 DORSEY HALL DRIVE ELLICOTT CITY, MARYLAND 21042-7870

(410) 964-0300 (301) 596-6500 Fax: (410) 964-2008

March 23, 2020

<u>VIA FIRST CLASS MAIL</u> Calvin Ball, Howard County Executive 3430 Court House Drive Ellicott City, Maryland 21043

Re: Camilla Carroll and Philip D. Carroll, Petition for Amendment to Development Rights and Responsibilities Agreement

Dear County Executive Ball:

This firm represents Camilla Carroll and Philip D. Carroll (collectively, the "**Petitioners**"). The Petitioners and Howard County, Maryland are parties to that certain Development Rights and Responsibilities Agreement dated September 23, 2010 and recorded among the Land Records of Howard County, Maryland in Liber 12722, folio 248 (the "Current DRRA").

In accordance with Subtitle 17 of Title 16 of the Howard County Code, please accept this letter and the enclosed amendment (the "Amendment") as a petition to negotiate and execute a First Amendment to Development Rights and Responsibilities Agreement.

The purpose for the Amendment is to extend the term of the Current DRRA by a period of five (5) years. No other changes are proposed to be made to the Current DRRA.

Please be advised that a pre-submission community meeting regarding the Amendment was conducted on May 23, 2019 in accordance with the requirements of Section 16.1701(b)(1) of the Howard County Code.

Calvin Ball, Howard County Executive March 23, 2020 Page 2 of 2

After your review of the Amendment, we suggest initiating the negotiation process forthwith. Thank you.

Very truly yours,

TALKIN & OH, LLP

2 m sh

Sang W. Oh

SWO Enclosure

cc: VIA FIRST CLASS MAIL

Honorable Deb Jung, Chair, Howard County Council Honorable Liz Walsh, Vice-Chair, Howard County Council Honorable Opel Jones, Member, Howard County Council Honorable Christiana Mercer Rigby, Member, Howard County Council Honorable David Yungmann, Member, Howard County Council Amy Gowan, Director, Department of Planning and Zoning James Irvin, Director, Department of Public Works Gary W. Kuc, Howard County Solicitor

FIRST AMENDMENT TO DEVELOPMENT RIGHTS AND RESPONSIBILITIES AGREEMENT

THIS FIRST AMENDMENT TO DEVELOPMENT RIGHTS AND RESPONSIBILITIES AGREEMENT (this "Amendment"), is made as of the _____ day of ______, 2020 (the "Effective Date"), by and among CAMILLA CARROLL and PHILIP D. CARROLL, individuals (collectively, "Petitioner"), and HOWARD COUNTY, MARYLAND, a body corporate and politic of the State of Maryland ("Howard County"). Petitioner and Howard County are hereinafter referred to collectively as the "Parties".

RECITALS

1. Subtitle 3 of Title 7 of the Land Use Article of the Annotated Code of Maryland (the "DRRA Law") grants Howard County the authority to establish procedures and requirements for the consideration and execution of Development Rights and Responsibilities Agreements.

2. Subtitle 17 of Title 16 of the Howard County Code (the "County Ordinance") authorizes Howard County to amend previously executed Development Rights and Responsibilities Agreements.

3. The Parties hereto are parties to that certain Development Rights and Responsibilities Agreement dated September 23, 2010 and recorded among the Land Records of Howard County, Maryland (the "Land Records") in Liber 12722, folio 248 (the "Existing DRRA") regarding certain real property more particularly described and identified in the Existing DRRA.

4. Section 16.1701 of the County Ordinance provides the procedure for amending previously executed Development Rights and Responsibilities Agreements such as the Existing DRRA.

5. Additionally, Section 9.4 of the Existing DRRA provides that the Parties may amend the Existing DRRA "by mutual consent after Howard County holds a public hearing and complies with all applicable laws of the County Ordinance concerning amendment of a Development Rights and Responsibilities Agreement."

6. Section 9.2.A of the Existing DRRA provides that the Existing DRRA "shall terminate and be void on the tenth (10th) anniversary of the Effective Date of [the Existing DRRA], unless extended by an amendment to [the Existing DRRA] complying with all procedures required in [the Existing DRRA], the County Ordinance and Maryland Law"

7. The Parties desire to extend the term of the Existing DRRA for an additional

period of five (5) years.

8. This Amendment is intended to constitute an amendment to a Development Rights and Responsibilities Agreement as provided for in the DRRA Law and the County Ordinance.

9. On or about _____, 2020, Petitioner petitioned Howard County to enter into this Amendment.

10. On or about ______, 2020, Howard County reviewed this Amendment and determined to accept this Amendment and to initiate the process of considering an amendment to the Existing DRRA.

11. This Amendment was negotiated between Petitioner and the Howard County Executive.

12. A pre-submission community meeting regarding this Amendment was conducted in accordance with the requirements of the County Ordinance and Howard County law on May 23, 2019.

13. This Amendment was referred to the Howard County Planning Board (the "**Planning Board**") for an advisory determination of whether this Amendment is consistent with Howard County's general plan, PlanHoward 2030 (the "General Plan"). At a public meeting held on ______, 2020, the Planning Board determined that this Amendment was consistent with the General Plan. The recommendation of the Planning Board is attached hereto and incorporated herein by reference as <u>Exhibit A</u>.

14. On ______, 2020, the Howard County Council held a duly advertised public hearing on this Amendment in accordance with Howard County law, and approved this Amendment on ______, 2020 by Council Resolution ______.

NOW, THEREFORE, in consideration of the foregoing recitals, which are not merely prefatory but are hereby incorporated into and made a part of this Amendment, and the mutual covenants and agreements as set forth below, and for other good and valuable consideration, the receipt and sufficiency of which the Parties hereby acknowledge, Petitioner and Howard County hereby agree as follows:

ARTICLE I

AMENDMENT

1.1 <u>Term</u>. Section 9.2.A of the Existing DRRA is hereby deleted in its entirety

and is hereby replaced as follows:

"A. This Agreement shall constitute covenants running with the land and shall run with and bind the Property. This Agreement shall terminate and be void on September 23, 2025, unless extended by an amendment to this Agreement complying with all procedures required in this Agreement, the County Ordinance and Maryland Law or in accordance with Section 8.4 above or unless terminated by agreement of the Parties or as permitted by law."

1.2 <u>Nature, Survival, and Transfer of Obligations</u>. The Parties agree that this Amendment shall run with the land and be binding upon and inure to the benefit of Petitioner and their respective heirs, successors and assigns, and upon any and all successor owners of record of all or any portion of the Property (except owners of an individual lot improved as part of the Project and pursuant to a validly issued building permit). To assure that all such successors, assigns, and successor owners have notice of this Amendment and the obligations created by it, Petitioner agrees that they shall:

A. Have this Amendment recorded among the Land Records within twenty (20) days after the Effective Date of this Amendment; and

B. Incorporate, by reference, this Amendment into any and all real estate sales contracts entered into after the Effective Date of this Amendment for the sale of all or any portion of the Property; and

C. Prior to the transfer of all or any portion of the Property (except the transfer of an individual lot solely for use as a private residence), or any equitable interest therein, require the transferee to execute an enforceable written agreement, in a form reasonably satisfactory to Howard County, binding transferee to this Amendment.

1.3 <u>Binding Upon Successors and Assigns of Howard County</u>. Howard County agrees that, to the extent permitted by law, all obligations assumed by it under this Amendment shall be binding on it, its agencies, employees, governmental units, the Planning Board and its and their respective successors and assigns.

1.4 <u>Regulation and Master Plan Consistency</u>. Howard County has determined that this Amendment is consistent with the General Plan. The Planning Board has determined that this Amendment is consistent with the General Plan.

1.5 <u>Public Health, Safety and Welfare</u>. Howard County has determined that the terms and provisions of this Amendment will ensure that the public health, safety and welfare of the residents of Howard County are protected.

1.6 <u>Ratification</u>. The Parties hereby ratify and confirm all of the terms and

provisions of the Existing DRRA and acknowledge and agree that all of the terms and provisions of the Existing DRRA remain in full force and effect, except as otherwise expressly and specifically modified and amended by the terms and provisions of this Amendment. In the event of any conflict between the terms and provisions of the Existing DRRA, and the terms and provisions of this Amendment, the terms and provisions of this Amendment shall control.

ARTICLE II MISCELLANEOUS

2.1 <u>Time of Essence</u>. Time is of the essence in the performance of all terms and provisions of this Amendment.

2.2 <u>Authority to Execute</u>. Howard County and Petitioner hereby acknowledge and agree that all required notices, meetings, and hearings have been properly given and held by Howard County with respect to the approval of this Amendment and agree not to challenge this Amendment or any of the obligations created by it on the grounds of any procedural infirmity or any denial of any procedural right. Howard County hereby warrants and represents to Petitioner that the persons executing this Amendment on its behalf have been properly authorized to do so.

2.3 <u>Governing Law</u>. This Amendment shall be governed by and construed in accordance with the laws of the State of Maryland.

2.4 <u>Severability</u>. In case any one or more of the provisions contained in this Amendment shall for any reason be held invalid, illegal, or unenforceable in any respect, such invalidity, illegality, or unenforceability shall not affect any other provision of this Amendment, and this Amendment shall be construed as if such invalid, illegal, or unenforceable provision had never been contained in this Amendment.

2.5 <u>No Third Party Beneficiary Status</u>. The Parties specifically agree that this Amendment is not intended to create in the public or any member thereof, third party beneficiary status in connection with the performance of the obligations under this Amendment.

2.6 <u>Recitals</u>. The Parties acknowledge and agree that the recitals to this Amendment are true and correct, and such recitals are incorporated herein by reference.

2.7 <u>Defined Terms</u>. Capitalized terms used but not otherwise defined in this Amendment shall have the meanings given such terms in the Existing DRRA.

[Signatures on Following Pages]

IN WITNESS WHEREOF, the Parties have hereunto set their hands under seal on the date first above written.

WITNESS/ATTEST:

Camilla Carroll

(SEAL)

STATE OF _____, CITY/COUNTY OF _____, TO WIT:

I HEREBY CERTIFY, that on this _____ day of _____, 2020, before me, the subscriber, a Notary Public of the State aforesaid, personally appeared CAMILLA CARROLL, known to me (or satisfactorily proven) to be the person whose name is subscribed to the within instrument, and acknowledged that she executed the same for the purposes therein contained.

IN WITNESS WHEREOF, I hereunto set my hand and official seal.

Notary Public

[SEAL]

[Print Name of Notary]

My Commission expires:

[SIGNATURES CONTINUE ON FOLLOWING PAGES]

WITNESS/ATTEST:

_(SEAL)

Philip D. Carroll

STATE OF ______, CITY/COUNTY OF ______, TO WIT:

I HEREBY CERTIFY, that on this _____ day of _____, 2020, before me, the subscriber, a Notary Public of the State aforesaid, personally appeared PHILIP D. CARROLL, known to me (or satisfactorily proven) to be the person whose name is subscribed to the within instrument, and acknowledged that he executed the same for the purposes therein contained.

IN WITNESS WHEREOF, I hereunto set my hand and official seal.

Notary Public

[SEAL]

[Print Name of Notary]

My Commission expires: _____

[SIGNATURES CONTINUE ON FOLLOWING PAGE]

AGREED and APPROVED:

HOWARD COUNTY, MARYLAND

ATTEST:

BY: _____(SEAL)

Lonnie R. Robbins Chief Administrative Officer

Calvin Ball Howard County Executive

APPROVED AS TO FORM AND LEGAL SUFFICIENCY

this _____ day of _____ 2020.

Gary W. Kuc County Solicitor

STATE OF MARYLAND, _____ COUNTY, TO WIT:

I HEREBY CERTIFY that on this _____ day of _____, 2020, before me, the subscriber, a Notary Public of the State of Maryland, in and for the County aforesaid, personally appeared Calvin Ball, the County Executive for Howard County, Maryland, who acknowledged the within Amendment to be the act of the County and that he executed the foregoing Amendment for the purposes therein contained by signing in my presence the name of Howard County, Maryland as County Executive.

AS WITNESS my Hand and Notarial Seal.

Notary Public

My Commission Expires: _____

ATTORNEYS' CERTIFICATION

THIS IS TO CERTIFY that the undersigned are members, in good standing, of the Bar of the Court of Appeals of Maryland, and that the within instrument was prepared by the undersigned or under their supervision.

Upon Recordation Please Return To:

Sang W. Oh, Esq. Talkin & Oh, LLP 5100 Dorsey Hall Drive Ellicott City, Maryland 21042

EXHIBIT A

PLANNING BOARD RECOMMENDATION

.

See attached.



Attachment C Technical Staff Report General Plan Amendment

Marsha S. McLaughlin, Director

TECHNICAL STAFF REPORT

January 7, 2010

Planning Board Meeting of January 21, 2010 County Council Hearing to be scheduled

Case No. /Petitioner:	At the request of Camilla Carroll and Philip D. Carroll
Request:	Adjustment to the Planned Service Area boundary of the Master Plan for Water & Sewerage in conjunction with a related Petition to Amend the Zoning Map of Howard County for property located in Ellicott City, Howard County, Maryland.
Location:	Second and Third Election Districts Portion of Historic Doughoregan Manor at 3500 Manor Lane Ellicott City, Howard County, Maryland Tax Map 23, Grid 10, Part of Parcel 71
Area of Site:	221.1 +/- acres
Zoning:	RC-DEO: Rural Conservation – Density Exchange Option
Proposed Zoning:	R-ED: Residential – Environmental Development

Department of Planning and Zoning Recommendation:

<u>Approval</u>



I. BACKGROUND

Application Overview

- The owners of Doughoregan Manor are requesting that the General Plan be amended so that the easternmost, 221-acre portion of their property may be added to the Planned Service Area for Water and Sewerage. Their purpose is to create roughly 325 new building sites for single-family-detached residences, so that these sites may be sold in order to generate funds to restore and preserve the manor house which is a National Historic Landmark.
- A concurrent request to rezone a portion of this property (ZB 1087M) proposes that the parcel be rezoned to Residential – Environmental Design (R-ED) from the currently existing zoning designation of RC – Rural Conservation-Density Exchange Option (RC-DEO).
- The owners of Doughoregan Manor have applied to sell the County a permanent agricultural land preservation easement on 500 acres of farmland. The Agricultural Land Preservation Board recommended acquisition of an easement on the property. Upon signature of the commitment letter by the owners, the County Executive will request the County Council to approve the use of an installment purchase agreement to acquire the agricultural land preservation easement. Settlement on the easement acquisition can then be scheduled.
- The owners are also offering to donate about 34 acres of land above to be added to the Kiwanis-Wallas Park.
- The property owners are in the process of developing a strategy for preservation of the 90-acre historic core that includes the Manor and the immediately surrounding historic outbuildings.

General Information

- Doughoregan Manor was the country home of Charles Carroll III of Carollton, a signer of the Declaration of Independence for Maryland. Charles Carroll, planter, landowner, politician and US Senator, was the only Roman Catholic signer of the Declaration and its last surviving signer. Charles Carroll III is buried in the family chapel attached to the north end of the Manor. Doughoregan Manor is the only home of a signer of the Declaration of Independence that is still in family ownership.
- In May 2007, a 30-year historic easement placed on the property by the Maryland Historic Trust expired.
- Seventy-five acres of the property are already protected in a permanent preservation easement. This land is already preserved and is not included in any of the proposed requests mentioned above.
- Frederick Road (MD 144) is a part of the Historic National Road, a Maryland Scenic Byway, a Maryland Signed Bicycle Route and a County-designated Scenic Roadway. The portion of the roadway adjacent to the subject property is described in the Howard County scenic road survey document as: "predominantly residential, with some scattered commercial uses. Scenic features include the distant view of rolling farmland and forest and a row of mature trees at the road's edge at Doughoregan Manor, several picturesque farmsteads and homes, small areas of forest, a pond, and a small stream paralleling the road."

II. GENERAL PLAN AMENDMENT

General Plan 2000 allows for the inclusion of land in the Planned Service Area for water and sewer when a
new zoning designation is being sought for land located on property adjoining the Planned Service Area:

Although this General Plan does not propose an expansion of the Planned Service Area to accommodate future residential or commercial growth, it should be anticipated that in the future there may be isolated situations where minor adjustments may be appropriate. Any requests for a General Plan amendment for expansion of the Planned Service Area should be denied unless the following minimum criteria are met: the proposed expansion of the Planned Service Area is part of a proposed zoning and is consistent with the General Plan and Smart Growth policies, or the proposed expansion of the Planned Service Area is intended to provide for a public or institutional use such as a religious facility, charitable or philanthropic institution, or academic school. In each case sewer and water infrastructure capacity and costs shall be analyzed to confirm the feasibility and availability of scheduled capacity.



III. DOCUMENTS AFFECTED

General Plan 2000

If this proposed amendment were to be approved, then the 2000 General Plan Policies Map 2000/2020 would be adjusted to reflect the inclusion of the proposed site within the Planned Service Area boundary.

Master Plan for Water and Sewerage

If this amendment were approved, then the change on the 2000 General Plan Policies Map would need to be reflected in an amendment to the Master Plan for Water and Sewerage via a separate application to DPW.

Howard County Code

• If this amendment were approved as proposed, then a Development Rights and Responsibilities Agreement would be proposed by the owner to secure development rights and preclude any future rezoning request.

Zoning Map

 If the Petition to Amend the Zoning Map of Howard County were approved, then the portion of the property that is the subject of this proposed GPA would need to be rezoned from the current Rural Conservation-Density Exchange Option (RC-DEO) to a zoning designation intended for property located within the PSA.



IV. EVALUATION

General Plan 2000

The owner's request for a PSA expansion is consistent with the criteria in *General Plan 2000* for consideration as an amendment to the Planned Service Area (PSA). The owner's proposal is dependent on the approval of the Petition to Amend the Zoning Map of Howard County to prove either change or, in this case, mistake, as stated.

The following policies and objectives in General Plan 2000 are directly related to the proposed GPA 2010-1:

Policy 3.1: Ensure that a critical mass of high quality, strategically located farmland is protected from development.

- Purchase of Development Rights. Purchase agricultural preservation easements on at least 5,000 additional acres in the Rural West.
- Policy 4.18: Enhance the County park system and recreational facilities.
 - Acquisition Schedule. Accelerate acquisition of land to meet the County's long-term recreation needs since suitable sites are disappearing rapidly. Priority for park acquisition should be directed inside the Planned Service Area where population is greatest.
- Policy 5.14: Maintain or enhance the landscape character of roads.
 - Scenic Roads in the East. Strengthen requirements for view protection.
- Policy 5.18: Establish a comprehensive County-wide historic preservation program.
 - Broadening of Participation. Coordinate County historic preservation initiatives and programs with State and Federal programs and with historic preservation and interpretative programs of local nonprofit organizations.

Agency Comments

As requested by DPZ, local and State agencies reviewed the proposed GPA2010-1 and submitted comments.

The following agencies stated no objection to the application:

- Health
- Licenses, Inspections and Permits

The following agencies submitted specific comments (attached) that will be addressed during the Site Plan Development review by the Subdivision Review Committee:

- Fire & Rescue
- Howard County Public School System
- Police
- Public Works
- Recreation and Parks
- State Highway Administration

Water Resources Element

The County has proposed an Amendment to General Plan 2000, the "Water Resources Element" (WRE). The Planning Board has recommended approval of this General Plan Amendment (GPA) and the County Council is expected to consider this proposed Amendment in early 2010. Growth projections for the WRE address the potential request for a GPA to extend the Planned Service Area (PSA) to allow limited new residential development on a portion of Doughoregan Manor as part of a comprehensive strategy to protect this National Historic Landmark property.

The WRE offers the following analysis of any proposal to include Doughoregan Manor in the PSA:

"This increase is within the projected capacity of the water supply and sewage treatment system, however, the development of a portion of Doughoregan Manor will require an expansion of the current PSA. The development for Doughoregan Manor will not require any resizing of the water delivery system, but will require a significant expansion in pipe capacity at two locations in the sewage collection system.

Consideration of bringing a portion of Doughoregan Manor into the PSA is premised on permanent preservation of most of this National Historic Landmark property. It is not intended to signal the potential for inclusion of any other properties with existing or proposed multi-use septic systems adjacent to the PSA. Including additional properties would further increase flow to the Little Patuxent WRP and increase the need to achieve higher efficiency ENR treatment. To reduce flow and the nutrient concentration in flow sent to the Little Patuxent WRP from expansion of the PSA for Doughoregan Manor, wastewater from development at Doughoregan Manor should be treated on site before being discharged to the Little Patuxent WRP." (WRE, page 20)

The Doughoregan GPA request includes the following statements:

"As part of the new plan, we believe that the Department of Public Works will request that we increase the size of a portion of the line on the Property. We intend to accede to that request as well as adding nitrogen pre-treatment facilities, in effect treating not only the effluent from the new housing but also that from properties to the north and west of Doughoregan Manor. We also understand from discussions with the Department of Public Works that an existing undersized section on the Little Patuxent Interceptor will require replacement under a capital project prior to any units being constructed on our Property."

The proposal to increase the size and capacity of sewer lines and to add nitrogen pre-treatment facilities on the property is consistent with the recommendation in the Howard County Proposed Water Resources Element (WRE), a proposed amendment to General Plan 2000. The WRE recommends:

Policy 2: Ensure the adequacy of wastewater treatment capacity.

Action 2.2 Require that properties added to the current Planned Service Area, large redevelopment sites within the PSA and large sites with zoning intensification within the PSA minimize increases in flow and the nutrient concentration in flow sent to the wastewater treatment plants. (WRE, page 23)

The WRE also calls for such projects to address water conservation and reuse:

Policy 1: Ensure the safety and adequacy of the drinking water supply, and promote water conservation and reuse.

Action 1.2 Require that properties added to the current Planned Service Area, large redevelopment sites within the PSA and large sites with zoning intensification within the PSA implement water conservation and reuse practices and technology. (WRE. page 23)

V. CONCLUSION

The owner's request for a Planned Service Area boundary amendment is consistent with the criteria required for consideration by the County of an expansion to the Planned Service Area (PSA). It is associated with a proposed rezoning that is intended to achieve General Plan policies related to historic preservation, agricultural preservation, recreation and scenic roads.

VI. RECOMMENDATIONS

The Department of Planning and Zoning recommends approval of GPA2010-1.

Additionally, DPZ recommends including language in the proposed bill that if the parcel is developed and connected to the public water and/or sewerage system by a deadline specified in the bill, the planned service area expansion and the metropolitan district inclusion, if applicable, shall be null and void and the planned service area as it relates to the parcel shall revert to that in place prior to the council bill approving the expansion, without any additional action by the Council.

January 7, 2009

Manhe S. M. Laughlin

Marsha S. McLaughlin, Director of Planning & Zoning

Date

The file is available for public review in the Offices of the Department of Planning and Zoning, which are currently located in the Ascend One Building, located at 8930 Stanford Blvd in Columbia, Maryland, from Monday through Friday, 8:30 a.m. - 4:30 p.m.

APPLICATION

x

1

October 28, 2009

Honorable Ken Ulman, Howard County Executive Honorable Mary Kay Sigaty, Howard County Council, Chair Honorable Jennifer Terrasa, Howard County Council, Vice-Chair Honorable Calvin Ball, Howard County Council, Member Honorable Greg Fox, Howard County Council, Member Honorable Courtney Watson, Howard County Council, Member 3430 Courthouse Drive Ellicott City, MD 21043

> RE: Request for Amendment to General Plan Doughoregan Manor, Tax Map 23, Parcel 71, 892.6 AC ± (the "Property")

Dear Messrs. and Mesdames:

As you know, for approximately two years, our family worked with Erickson Retirement Communities to develop a continuing care retirement community ("CCRC") on a portion of our historic estate. We formulated a plan to preserve the remainder of one of the largest expanses of undeveloped land in Howard County.

As you also know, in February 2009, Erickson submitted a request for an amendment to the General Plan and Zoning Regulations of Howard County to facilitate their proposal and, thereby, begin the process of preserving Doughoregan Manor. In June 2009, Erickson formally withdrew their interest in the property and thus left the plan in limbo. The challenge of maintaining and preserving our property, however, persists. While the plan for a CCRC has not proven workable, viable strategies for long-term preservation remain.

We are writing to ask your help with a revised plan that we believe will solve the dilemma that our family faces as owners of Doughoregan. Your cooperation will allow us all to carry out our shared obligation as stewards of our past and present.

Following the withdrawal of Erickson, our family decided to retain complete control of the property and any future development proposals. Given our current financial circumstances, we cannot allow ourselves to be at the risk of another unanticipated decision by a third-party developer. Our decision requires more time and greater expense for our family; however, we believe that final and lasting solutions can be expeditiously achieved. Our family has invested heavily in the restoration of the main house, but there is much more that needs to be done soon in order to prevent irreversible deterioration. In addition, several historic outbuildings have reached a point where immediate action is needed or they will be lost. Doughoregan is more than the main house: the setting must be preserved. We intend to do that if the financial resources are there. October 28, 2009

Page 2 of 5

The revised plan has several components, all of which must be approved and implemented in order for our initiative to be a success. The components are as follows:

Development

Location

The eastern-most portion of the Property, which was previously identified and depicted on "Exhibit B" in a February 25, 2009 letter to you as the land to be acquired by Erickson, will continue to serve as the development area in the new plan. The proposed area is shaded and shown as 221.1 Acres \pm on the attached Exhibit to Accompany Petition to Amend the Zoning Map of Howard County, which is being provided to you as a courtesy. On this area will be sited approximately 325 single family detached units as shown on the attached Exhibit 2, a preliminary concept plan.

Purpose

The development plan is designed to be R-ED (Residential-Environmental Development), the lowest density and most environmentally conscientious development in the Howard County Zoning and Subdivision Regulations. With smaller lot sizes averaging 8,000 sq. ft. and larger requirements for open space, the proposed development will accommodate houses generally from 2,500 to 3,000 sq. ft. Exhibit 2 shows an innovative subdivision layout including an attractive neighborhood design, which will meet or go beyond the impending upgrades to Maryland's storm water regulations.

Benefits

Under the new plan, removal of large areas of existing forest that were part of the Erickson plan can be avoided. Connecting roads to the Ridge Lake and Chateau Ridge communities are also unnecessary. Development of the eastern boundary of Doughoregan best preserves and protects the setting of the historic buildings. The development of single family detached units will be harmonious with the existing residential development on adjacent properties and will avoid conflicts with ongoing farming operations on Doughoregan.

Traffic

Recently, we have discussed our plan with a number of interested citizens and community associations. Most concerns raised involved traffic on Rt. 144 and school capacity, which we understand and share as residents of this area. We believe, however, that we have sufficiently studied traffic impacts and can express with confidence that the new plan will not degrade our quality of life. We are committed to effecting a balance between preserving the existing scenic character of the National Road and the safety requirement for ingress and egress. We believe both can be achieved. The traffic studies that we have conducted show that peak hour trips do not add to the existing traffic on Rt. 144 to a substantial
October 28, 2009

Page 3 of 5

degree. In fact, the level of traffic generated by the proposed development is projected to be less than one third of that which would have been generated under the Erickson proposal. We also recognize we will be required to assure an adequate level of service at neighboring intersections.

Schools

We project the build-out of the development to occur approximately 12 to 15 years from now, based on the length of the permit approval process, actual construction and, of course, market factors. While we cannot predict exactly the number of school-aged children resulting from a particular development, we believe that absorption of the students from the 325 units over a 12 to 15 year period will be acceptable.

PSA

We will not require the zoning text amendment envisioned for the Erickson proposal. However, as before with the Erickson proposal, in order to accommodate this development, an extension of the planned service area for water and sewer ("PSA") along with a zoning map change will be required. Consequently, we ask that you consider the amendment to the PSA and Master Plan for Water and Se-werage Plan requested by Erickson in February 2009 as largely unchanged and remaining pending.

Public Works

As you may know, there are already existing and operating public sewer lines that traverse the proposed development area. As part of the new plan, we believe that the Department of Public Works will request that we increase the size of a portion of the line on the Property. We intend to accede to that request as well as adding nitrogen pre-treatment facilities, in effect treating not only the effluent from the new housing but also that from properties to the north and west of Doughoregan Manor. We also understand from discussions with the Department of Public Works that an existing undersized section on the Little Patuxent Interceptor will require replacement under a capital project prior to any units being constructed on our Property.

We have attached hereto a statement of justification from our legal counsel in support of the proposed amendment to the PSA.

Approval of the proposed amendment to the PSA will precede a Petition for an Amendment to the Zoning Map for Howard County to change the zoning of the development area of the Property from RC-DEO to R-ED. This Petition for rezoning has been filed with the Department of Planning and Zoning.

October 28, 2009

Page 4 of 5

Easement

The second component of the strategy to preserve the Manor is the placement of approximately 500 acres of the land along Folly Quarter Road and Rt. 144 under permanent preservation under the Howard County Agricultural Land Preservation Program. The 94.5 acres of land immediately surrounding the manor house and outbuildings will remain in our family's hands where we will continue to live for generations to come. While we do not plan for the family ever to leave the Manor, we are committed to the long-term preservation of the property and intend to investigate legal means of achieving that goal.

Dedication

The third component is our family's willingness to dedicate approximately 34 acres of land that is contiguous to Kiwanis Wallis Park. We believe the park provides an invaluable service to Howard County families. We welcome the opportunity to contribute to the welfare of our county's children.

Development Rights and Responsibilities Agreement

As a fourth component, we propose that the foregoing terms be set forth as stated in a Development Rights and Responsibilities Agreement, the authority and approval for which must be enacted by the governing body of Howard County. Attached hereto is a draft of the appropriate legislation.

The General Plan Amendment and draft of the Development Rights and Responsibilities Agreement are being provided to you with the request for their introduction as legislation. The application for rezoning is being submitted to the Department of Zoning in accordance with established procedures. We believe it is imperative for these elements to remain as a comprehensive package in order to produce a workable solution.

We and our consultants remain available to meet with each of you and members of the community to discuss your questions, comments or concerns. On behalf October 28, 2009

Page 5 of 5

of our family, we thank you for your consideration of this matter and look forward to continuing to work with you and the citizens of Howard County.

Sincere. Samilla Carroll

Philip D. Carroll

Marsha McLaughlin, Department of Planning and Zoning, Director James Irvin, Department of Public Works, Director cc: Margaret Ann Nolan, Office of Law, County Solicitor

LAW OFFICES OF TALKIN & OH, LLP COLUMBIA OFFICE 5100 DORSEY HALL DRIVE ELLICOTT CITY, MARYLAND 21042-7870

(410) 964-0300 (301) 596-6500 Fax: (410) 964-2008

October 28, 2009

Honorable Ken Ulman, Howard County Executive Honorable Mary Kay Sigaty, Howard County Council, Chair Honorable Jennifer Terrasa, Howard County Council, Vice-Chair Honorable Calvin Ball, Howard County Council, Member Honorable Greg Fox, Howard County Council, Member Honorable Courtney Watson, Howard County Council, Member 3430 Courthouse Drive Ellicott City, MD 21043

> RE: Justification In Support of General Plan Amendment Extension of Planned Service Area for Water and Sewer for Portion of Parcel 71 of Tax Map 25 (221.1 AC ±)

Dear Messers and Mesdames:

Please be advised that this firm represents Camilla and Phillip D. Carroll with respect to the above-referenced Property.

This letter is being provided to you as justification in support of amending the map of the Planned Service Area for water and sewer ("PSA") as contained in the 2000 General Plan and the Master Plan for Water and Sewerage. The proposed amendment requests the extension of water and sewer to a portion of Doughoregan Manor as specified on the attached "Exhibit to Accompany Petition to Amend the Zoning Map of Howard County – Doughoregan Manor." The proposed area for inclusion in the PSA is colored green and shown as 221.1 AC \pm (the "Site"). Currently, Doughoregan, including the Site, is zoned RC-DEO and is in the No Planned Service Area, despite the fact that actual sewer lines traverse portions of the Site.

In evaluating this request, we ask you to consider the following:

Page 98.1 of the 2000 General Plan specifies that "[a]ny requests for a General Plan amendment for expansion of the Planned Service Area should be denied unless the following minimum criteria are met: the proposed expansion of the Planned Service Area is part of a proposed zoning and is consistent with the General Plan and Smart Growth policies..."

Attached hereto is a Petition to Amend the Zoning Map of Howard County for the Site (the "Rezoning Application"). As you will note, the request is a change from RC-DEO to R-ED.

October 28, 2009 Page 2 of 3

This request is consistent with the tenets of the 2000 General Plan and Smart Growth policies for a number of reasons.

The 2000 General Plan ("GP") endorses preserving contiguous blocks of agricultural land and protecting the rural character in the Near West. GP at 44. The Near West is the area of the Rural West that is zoned Rural Conservation (RC) and is adjacent to the Planned Service Area. *Id.* Doughoregan is the largest tract of uncommitted residential land in the Near West and represents the greatest remaining opportunity for agricultural preservation in the Near West. If accomplished, the successful preservation of Doughoregan as proposed will result in the largest contiguous expanse of preserved farmland in the Near West and will be a key component in protecting its rural character.

The first tenet of Maryland's Smart Growth Goals is to save our most valuable resources before they are forever lost. The preservation of farmland and environmental resources in the Rural West has been the principal goal of the regulations for cluster subdivisions. As the 2000 General Plan acknowledges, however, the current cluster subdivision regulations have been very successful in protecting streams, wetlands, floodplains, steep slopes and forest, but they have been only partially successful in achieving the key goal of preserving farmland. GP at 47. Cluster subdivision design is frequently dictated by the location of the best soils for septic systems, resulting in the best soils being used for lots instead of farmland. *Id*.

Doughoregan is permitted under its existing zoning to develop over 400 single-family detached units on the Property. Locating these lots on the portions of Doughoregan with the soils most suitable for septic systems presents the undesirable option of sprawling the development across the Property and in locations that would adversely affect the agricultural character of this area. Residential development in accordance with the RC-DEO will also have an adverse impact of the character of two scenic roads: Folly Quarter Road and MD Route 144 (the "National Road). GP at 56. We believe the development that is permitted by right would not best serve the interests of our County.

The goals of the 2000 General Plan and Smart Growth policies would be better served by a proposal of concentrating the residential development to the eastern-most section of the Property in the area of the Site, where the number of proposed new housing with public water and sewer will be approximately 100 units less and more compatible with the homes on neighboring properties. The remainder of the Property will be left undisturbed, as-is and remain largely farmland. The General Plan discusses using a variety of tools to achieve its goal of protecting 30,000 acres in the Rural West. GP at 44. Each of these options may have limited applicability, but together they contribute to an effective land preservation strategy. *Id.* We believe that the proposed plan for Doughoregan is the type of combined strategy referenced in the General Plan.

Smart Growth principles also support development in areas where infrastructure is already in place or planned to support it. In addition to the preservation of farmland, one of the principal benefits of permanently protecting land in the Near West is to guard against the October 28, 2009 Page 3 of 3

potential for unwanted sprawl into the Rural West. With the requested General Plan Amendment, the boundary of the PSA will terminate on the west side of the Site. Abutting the new PSA boundary will be permanently preserved agricultural land, thereby eliminating the potential for any future PSA extension requests in this area.

The General Plan provides that "[i]n each case [of a proposed expansion of the PSA], sewer and water infrastructure capacity . . . and costs shall be analyzed to confirm the feasibility and availability of scheduled capacity." GP at 98.1. In this regard, please be advised that the extension of water line in this area (which is within the MD Route 144 Right-of-Way and adjoins the Property) and sewer line to serve the Site will not be an expense to the general public. To the contrary, all construction and fees will be private expense to be paid by the developer. We have had discussions with the Department of Pubic Works and are of the understanding that sewer and water capacity is available, particularly in light of the fact that we will be upgrading the size of the sewer line on-site and installing pre-treatment of new and existing effluent.

In terms of the adequacy of public facilities, we have already discussed the fact that a certain number of residential units are permitted by right on Doughoregan Manor. The traffic and school capacity to be generated by these units are already an existing condition. The proposed plan reduces the planned density for Doughoregan and envisions a long term projected build-out. In any event, the proposed units will be subject to all APFO tests pursuant to the General Plan, which must be met in order for the development to proceed.

And finally, the proposed development coupled with the 500 acres of land surrounding Doughoregan Manor being preserved with an Agricultural Preservation Easement under the County's program will enable the Carroll family to continue to be stewards of the Property as they have for the past three centuries. The proposed plan provide financial reservates to protect and restore the Manor and supporting historic structures, thereby furthering the County's historic preservation goals.

We look forward to continuing the community dialogue on these issues and welcome any questions or comments. Thank you.

Very truly yours,

TALKIN & OH, LLP

Ang V. ch By: Sang W. Oh

cc: Marsha McLaughlin, Department of Planning and Zoning, Director James Irvin, Department of Public Works, Director Margaret Ann Nolan, Office of Law, County Solicitor

Introduced
Public Hearing
Council Action
Executive Action
Bffective Date

:

County Council Of Howard County, Maryland

2009 Legislative Session

Legislative Day No.

Bill No. _____

Introduced by: _____

AN ACT to authorize Howard County to enter into Development Rights and Responsibilities Agreements pursuant to the authority granted by Article 66B, § 13.01, Md. Ann. Code, and to establish procedures and requirements for the consideration and execution of such agreements

Introduced and read first time, 2009. Ordered posted and hearing sche	åuled.
	By order
Having been posted and notice of time & place of hearing & title of Bill having been publi for a second time at a public hearing on, 2009.	shed according to Charter, the Bill was read
	By order
This Bill was read the third time on, 2009 and Passed, Passed with am	endments, Failed
· · · · ·	By order Stephen LeGendre, Administrator
Sealed with the County Seal and presented to the County Executive for approval thisd	y of, 2009 ata.m./p.m.
	By order
Approved/Vetoed by the County Executive, 2009	
	Ken Ulman, County Executive
NOTE: [[text in brackets]] indicates deletions from existing law; TEXT IN ALL CAPITA Strike-out indicates material deleted by amendment; Underlining indicates material adde	LS indicates additions to existing law; I by amendment

1	Section 1. Be it enacted by the County Council of Howard County, Maryland, that new
2	Subtitle 15 "Development Rights and Responsibilities Agreements" is added to Title 16
3	"Planning, Zoning and Subdivisions and Land Development Regulations" of the Howard
4	County Code to read as follows;
5	
6	Title 16. Planning, Zoning and Subdivisions and Land Development Regulations
7	SUBTITLE 15. DEVELOPMENT RIGHTS AND RESPONSIBILITIES AGREEMENTS
8	
9	SECTION 16.1500. PURPOSE.
10	THE PURPOSE OF THIS SUBTITLE IS TO PROTECT THE PUBLIC HEALTH, SAFETY, AND
11	WELFARE BY ENACTING THE AUTHORITY GRANTED BY SECTION 13.01 OF ARTICLE 66B
12	OF THE ANNOTATED CODE OF MARYLAND RELATING TO DEVELOPMENT RIGHTS AND
13	RESPONSIBILITIES AGREEMENTS AND ESTABLISHING PROCEDURES FOR SUCH
14	AGREEMENTS IN ACCORDANCE WITH THE PROVISIONS THE STATE CODE,
15	
16	SECTION 14.1501. DEFINITIONS.
17	IN THIS SECTION THE FOLLOWING WORDS HAVE THE MEANINGS INDICATED.
18	(A) "AGREEMENT" MEANS A DEVELOPMENT RIGHTS AND RESPONSIBILITIES
19	AGREEMENT.
20	(B) "COUNTY COUNCIL" MEANS THE HOWARD COUNTY COUNCIL.
21	(C) "COUNTY EXECUTIVE" MEANS THE HOWARD COUNTY EXECUTIVE.
22	(D) "PLANNING BOARD" MEANS THE HOWARD COUNTY PLANNING BOARD.
23	(E) "STATE CODE" MEANS THE MD. ANNOTATED CODE, AS AMENDED.
24	
25	SECTION 14.1502. APPLICABILITY.
26	ANY PERSON SATISFYING THE QUALIFICATIONS DESCRIBED IN ARTICLE 66B, §
27	13.01(C), OF THE STATE CODE, MAY PETITION THE COUNTY EXECUTIVE AND COUNTY
28	COUNCIL TO ENTER INTO A DEVELOPMENT RIGHTS AND RESPONSIBILITIES
29	AGREEMENT AS DESCRIBED IN ARTICLE 66B, § 13.01, OF THE STATE CODE.
	The bill of the bi

1	SECTION 14.1502. CONTENTS OF DEVELOPMENT RIGHTS AND RESPONSIBILITIES
2	AGREEMENTS.
3	AN AGREEMENT SHALL INCLUDE THE CONTENTS REQUIRED BY ARTICLE 66B, §
4	13.01(F)(1), OF THE STATE CODE, AND MAY INCLUDE THE CONTENTS ALLOWED BY
5	ARTICLE 66B, § 13.01(F)(2), OF THE STATE CODE.
б	
7	SECTION 14.1503. PROCEDURES.
8	(A) BEFORE ENTERING AN AGREEMENT, THE PERSON SATISFYING THE
9	QUALIFICATIONS DESCRIBED IN ARTICLE 66B, § 13.01(C), OF THE STATE CODE SHALL
10	PETITION THE COUNTY EXECUTIVE AND COUNTY COUNCIL TO ENTER INTO A
11	DEVELOPMENT RIGHTS AND RESPONSIBILITIES AGREEMENT AS DESCRIBED IN ARTICLE
12	66B, § 13.01, of the State Code.
13	(B) AN AGREEMENT MAY BE EXECUTED BY THE COUNTY EXECUTIVE ONLY AFTER A
14	PUBLIC MEETING BEFORE THE PLANNING BOARD AND A RECOMMENDATION BY THE
15	PLANNING BOARD THAT THE PROPOSED AGREEMENT IS CONSISTENT WITH THE
16	GENBRAL PLAN.
17	(C) AN AGREEMENT MAY BE EXECUTED BY THE COUNTY EXECUTIVE ONLY AFTER
18	THE COUNTY COUNCIL CONDUCTS A PUBLIC HEARING AND DETERMINES BY
19	RESOLUTION THAT THE COUNTY EXECUTIVE IS AUTHORIZED THE EXECUTE THE
20	AGREEMENT.
21	(D) WITH THE APPROVAL OF THE COUNTY COUNCIL, AN AGREEMENT MAY BE
22	CONSOLIDATED WITH A PUBLIC WORKS AGREEMENT OR WITH ANY OTHER PLAN OR
23	AGREEMENT REQUIRED FOR DEVELOPMENT OF A PROPERTY.
24	
25	SECTION 14.1504. AMENDMENTS OF AGREEMENTS.
26	(A) SUBJECT TO PARAGRAPH (B) OF THIS SUBSECTION AND AFTER A PUBLIC HEARING,
27	THE PARTIES TO AN AGREEMENT MAY AMEND THE AGREEMENT BY MUTUAL CONSENT.
28	(B) UNLESS THE PLANNING BOARD DETERMINES THAT THE PROPOSED AMENDMENT
29	TO THE AGREEMENT IS CONSISTENT WITH THE GENERAL PLAN, THE PARTIES MAY NOT
31	AMEND THE AGREEMENT.

.

ù,

3

1	SECTION 16.1505. TERMINATION OF AGREEMENTS; SUSPENSION.
2	(A) THE PARTIES TO AN AGREEMENT MAY TERMINATE THE AGREEMENT BY MUTUAL
3	CONSENT.
4	(B) IF THE COUNTY EXECUTIVE AND COUNTY COUNCIL DETERMINE THAT SUSPENSION
5	OR TERMINATION IS ESSENTIAL TO ENSURE THE PUBLIC HEALTH, SAFETY, OR WELFARE,
6	THE COUNTY EXECUTIVE AND COUNTY COUNCIL MAY SUSPEND OR TERMINATE AN
7	AGREEMENT AFTER A PUBLIC HEARING.
8 -	
9	SECTION 16.1506. APPLICABLE LAWS, REGULATIONS AND POLICIES.
10	(A) EXCEPT AS PROVIDED IN PARAGRAPH (B) OF THIS SUBSECTION, THE LAWS, RULES,
11	REGULATIONS, AND POLICIES GOVERNING THE USE, DENSITY, OR INTENSITY OF THE
12	REAL PROPERTY SUBJECT TO THE AGREEMENT SHALL BE THE LAWS, RULES,
13	REGULATION, AND POLICIES IN FORCE AT THE TIME THE PARTIES EXECUTE THE
14	AGREEMENT.
15	(B) IF THE COUNTY EXECUTIVE AND COUNTY COUNCIL DETERMINE THAT
16	COMPLIANCE WITH LAWS, RULES, REGULATIONS, AND POLICIES ENACTED OR ADOPTED
17	AFTER THE EFFECTIVE DATE OF THE AGREEMENT IS ESSENTIAL TO ENSURE THE HEALTH,
18	SAFETY, OR WELFARE OF RESIDENTS OF ALL OR PART OF THE JURISDICTION, AN
19	AGREEMENT MAY NOT PREVENT A LOCAL GOVERNMENT FROM REQUIRING A PERSON TO
20	COMPLY WITH THOSE LAWS, RULES, REGULATIONS OR POLICIES.
21	
22	SECTION 16.1507. RECORDING.
23	(A) AN AGREEMENT THAT IS NOT RECORDED IN THE LAND RECORDS OF HOWARD
24	COUNTY WITHIN 20 DAYS AFTE THE DAY ON WHICH THE PARTIES EXECUTE THE
25	AGREEMENT IS VOID.
26	(B) THE PARTIES TO AN AGREEMENT AND THEIR SUCCESSORS IN INTEREST ARE BOUND
27	TO THE AGREEMENT AFTER THE AGREEMENT IS RECORDED.
28	
29	Section 2. And Be it Further Enacted by the County Council of Howard County,
31	Marvland, that this Act shall become effective 61 days after its enactment.





- - -÷ f i și



AGENCY COMMENTS

.

HOWARD COUNTY DEPARTMENT OF FIRE AND RESCUE SERVICES

6751 Columbia Gateway Drive, Suite 400, Columbia, Maryland 21046 410-313-6000 • www.hcdfrs.org



William F. Goddard, III, Fire Chief

Ken Ulman, County Executive

December 9, 2009

Robert Lalush Planning and Zoning Ref: ZB-1087M Doughoregan Manor

To Planning and Zoning Board:

I have reviewed the plans for the proposed development of Doughoregan Manor. The development plans to build approximately 280 single family homes located off Frederick Road.

Currently, there is only one access point to enter the development. Per Title 17, Subtitle 1, Subsection 18.2.3.3.1 of the Howard County Fire Prevention Code, any development with over 100 residences may be required to have more than one access point available for emergency use. This is the main issue facing approval of the development from Fire and Rescue.

Listed below are recommendations for your consideration:

- A Fire Access Road could be created in the development that connects to Burnside Drive. This would be the easiest solution to the second access issue.
- If a second point of access cannot be agreed upon, then all homes would be required to have NFPA 13 D sprinkler systems installed.
- An access road would have to be dual lane the entire length except in those areas where stream crossings may be needed. A single bridge would be acceptable since no home would be located in/near the stream buffer and emergency apparatus could still access all homes in the development.

Battalion Chief Timothy Diehl Howard County Department of Fire and Rescue Services Office of the Fire Marshal





MEMORANDUM

DATE: January 5, 2010

TO: Mr. William Mackey, AICP Howard County DPZ

FROM:Joel Gallihue, AICPRE:GPA 2010-1 Doughoregan Manor / ZB 1087M

HCPSS staff has received the above referenced general plan amendment and petition to amend the zoning map and forwarded it to the Board of Education for their review. The proposal has been made for a property of approximately 221 acres located on the south side of MD Route 144 (Frederick Road) and west of the Kiwanis-Wallis Park. The proposal would facilitate a change in land use from what is presently farmland to residential. The residential units would not be age restricted so they will generate students. A previous proposal for this same part of Doughoregan Manor was made by Erickson Retirement Communities which would have been age restricted but that project was withdrawn.

The zoning petition states that 325 single family detached residential homes are proposed through the R-ED zone. The R-ED District is established to accommodate residential development at a density of two dwelling units per net acre in sensitive environmental or historic areas.

The current assigned schools for this area are Manor Woods ES, Burleigh Manor MS, and Marriotts Ridge HS. Five year average pupil generation ratios for the proposed unit type (single family detached) are presented for these schools and countywide in the following table:

· · · · · · · · · · · · · · · · · · ·	Assigned Schools	Countywide
Elementary	0.219	0.323
Middle	0.093	0.134
High	0.125	0.114

Board of Education policy envisions operation of schools between 90% and 110% utilization of capacity. Projections are developed annually. Projections that exceed 110% utilization inform planning for redistricting and capital projects. Manor Woods ES is currently projected to remain under 110% capacity until 2013. Burleigh Manor MS is currently projected to operate below 110% capacity until 2020. Marriott's Ridge HS is projected to operate below 110% capacity for the foreseeable future. Given the current five-year average pupil generation rates for the assigned schools, the proposed 325 SFD units would add approximately 71 elementary, 30 middle, and 41 high school students gradually over the 12-15 year development horizon.



WILLIAM J. MCMAHON Chief of Police

KEN ULMAN County Executive

HOWARD COUNTY DEPARTMENT OF POLICE 3410 Court House Drive, Ellicott City, Maryland 21043

December 23, 2009

Marsha S. McLaughlin, Director To: Department of Planning and Zoning

William J. McMahon, Chief From: Howard County Police Department

Doughoregan Manor Development- Police Response Subject:

This memorandum serves as the Police Department's response in reference to the development of the Doughoregan Manor. Overall, the Police Department has no concerns regarding an impact on services to the community due to the development of Doughoregan Manor. The police department has the following comments regarding the new community:

Police Manpower:

Any population growth and development creates a greater need for police resources to handle the increased calls for service in that area. In reviewing the development plans for Doughoregan Manor, the police department does not anticipate a significant impact on police resources.

Traffic Measures:

Due to the fact that the development's main access is Frederick Road, there may be an impact on vehicular traffic along that route. I would anticipate there would be increased traffic in this area; I do not believe it would cause a significant issues.

If you have any questions or concerns regarding this memorandum, please feel free to contact me at 410-313-2203.





Subject: Doughoregan Manor Request for Inclusion into the Planned Service Area

To: James Irvin, Director Department of Public Works

From: Don Lieu, Chief Utility Design Division

Date: January 7, 2010

We have reviewed the October 28, 2009 request submitted by Camilla and Philip Carroll for expansion of the Planned Service Area for the purpose of providing public water and sewer service to 325 homes on the eastern portion of the Doughoregan Manor property.

There is adequate capacity in the public water supply to provide service to the proposed development of Doughoregan Manor.

Regarding the public sewer system, there is a concern both on the treatment and conveyance facilities. The nutrient capacity at the Little Patuxent Water Reclamation Plant is a major concern. The limited excess capacity at the plant allows the County to absorb minor treatment excursions in the wastewater treatment process which would otherwise lead to discharge violations and monetary fines. By providing sewage treatment capacity (nutrient reduction) for a property previously not within the Planned Service Area, the capacity available for development to other properties currently within the Planned Service Area is reduced. Moreover, the available safety factor to handle treatment excursions is reduced by that amount.

In prior discussions with prospective developers of Doughoregan Manor, we expressed our preference for the wastewater to be treated and disposed of on the Doughoregan Manor site, i.e. disposal to a multi-use septic system. From the Bureau of Utilities point of view, the current development proposal, consisting of individual lots, would be consistent with the use of a shared septic system as described in the current County Code. This option provides for nutrient management, minimal conveyance impact and an existing stable financial model. With the reduction in flow produced with the most recent proposed land use, required drain field limitations becomes much more manageable.

In the event that this property is brought into the Planned Service Area and an on-site pretreatment system is constructed, we recommend that a rate structure be added to our utility system for this type of service. If on-site disposal of all or a portion of the wastewater from the Doughoregan Manor development is not possible, the wastewater should be treated to the limits of technology for nutrient removal before being discharged into the conveyance system.

In regard to the off-site sewer, there is a 300 foot section of 8-inch sewer from Doughoregan Manor to the 12-inch interceptor sewer which may need to be upsized. Although there is sufficient capacity in the 12-inch interceptor sewer between Doughoregan Manor and the Little Patuxent Interceptor to serve the additional 325 homes, 2,500 feet of the 27-inch Little Patuxent Interceptor from Old Annapolis Road to MD Route 108 will be required to be paralleled. The parallelization of this portion of the interceptor

sewer is currently in the capital budget as Capital Project S-6274, Upper Little Patuxent Parallel Sewer, to be funded in fiscal year 2015. The Upper Little Patuxent Parallel Sewer will need to be constructed before the additional homes can be built.

In regard to the existing on-site sewer, the Department of Public Works has not performed a capacity analysis on the on-site sewer system. A comprehensive utility plan noting the capacity and locations of the sewers within the subdivision is required from the developer during the subdivision process. The on-site sewers must be sized to provide sufficient capacity to support the upstream drainage area located east of Folly Quarter Road between US Route 40 and MD Route 144.

Please do not hesitate to contact me should you have any questions regarding the information provided.

cc: Ronald Lepson, Steve Gerwin, Jeff Welty

Comments from Gary J. Arthur, Director – Recreation and Parks 1/06/2010 ZB 1087M – Petition to Amend the Zoning Map of Howard County

Our Department supports this request to rezone 221.1 AC<u>+</u> of the subject property from RC-DEO to R-ED based on the need to maintain and preserve this nationally historic property. With this development, it will allow the Carroll's to maintain and preserve this historic property. This development would be consistent with existing developments across Frederick Road.

The build-out for this development is proposed over ten-twelve years. The thirty-four acres of dedicated land to the County will allow the county to improve the quality of life through the development of recreational facilities that would be jointly operated and maintained by Howard County Youth Programs, and the Department of Recreation and Parks. Presently, the thirty-four acres could potentially house fields, pathways along with environmental educational programs which would included natural resource protection and wildlife along with alleviating the already overcrowding conditions at Kiwanis Wallas Park by providing additional parking.



Martin O'Malley, Governor Anthony G. Brown, Lt. Governor Beverley K. Swaim-Staley, Secretary Neil J. Pedersen, Administrator

Maryland Department of Transportation

December 22, 2009

Ms. Marsha S. McLaughlin, Director Howard County Department of Planning and Zoning 3430 Courthouse Drive Ellicott City, Maryland 21043 RE: Howard County
 MD 144 A (Frederick Road)
 south side / west of US 40
 ZB 1087M
 Carroll Property (Doughoreagan)
 Mile Point 14.6 - 15.2

Dear Ms. McLaughlin:

Thank you for the opportunity to review and comment on the above referenced rezoning petition and amendment to the 2000 Howard County General Plan. The Maryland State Highway Administration, (SHA) has the following comments regarding this proposal.

As with all cases of land use, the SHA defers to Howard County to determine the appropriate use of the subject 221 acres. If approved by Howard County, SHA will require the necessary entrance and road improvements to support the additional site generated traffic. This includes traffic impacts to near-by intersections identified by the Howard County APFO.

The amended zoning petition seeks to allow the eastern-most portion of the property (approximately 221 acres) to be developed with 325 single family homes to be constructed under a long term build-out of 10 to 15 years. The proposed development is designed to be low density R-ED or Residential-Environmental Development.

The petition indicates that the remainder of the estate will be preserved under an agricultural preservation easement. SHA and HCDPW are jointly developing a project to construct a roundabout at the intersection of MD 144 and Folly Quarter Road. We suggest that the County modify the easement to set aside a portion of the property to the east of the intersection of MD 144 and Folly Quarter Road for possible construction or environmental needs for the project to lessen future financial impact to SHA and the County DPW.

The petition also briefly makes reference to the development of the aforementioned acreage as an assisted living development. ITE Trip generation rates comparing trips that can be expected to be generated by the former Senior Adult Housing and the trips that can be expected to be generated by the proposed Single Family detached housing is presented in the petition as prepared by the Traffic Group, Inc. The ITE trip generation data provided suggests that a considerable reduction in measured trips will be realized with the proposed Single-Family Detached Units. We concur with this assessment.

Ms. Marsha S. McLaughlin ZB 1087M December 22, 2009 Page Two

~ .

SHA is requesting that as part of the consideration of the rezoning request, the developer be required to dedicate right of way along the entire property frontage along MD 144. The dedication should be 40' from the existing centerline of MD 144 for future road widening and in keeping with the Howard County Master Plan.

SHA is recommending that one point of access be provided to MD 144. This is in keeping with the SHA desire to limit the number of access point to as few as possible to an urban arterial roadway. It would be preferable to have the western most proposed access serve as the sole access to MD 144. For this design to work we believe it is necessary to have the extension of Burnside Drive to serve as the second access to the property. The access location to MD 144 will require all necessary access improvements. These will include standard acceleration, deceleration, left turn lane and all necessary design elements to meet the term and conditions of an access permit issued by this office. These details will be determined at the time of the Sketch or Preliminary Plan submittal. If standards cannot be met, the entrance will be required to be relocated to a location that will yield the necessary improvements.

Normally SHA would request that development access be to a less traveled roadway such as the existing Folly Quarter Road or Manor Lane. In this case SHA recognizes the importance of maintaining a contiguous parcel which encompasses the historic and cultural resources of the Carroll Property. Requiring access to Folly Quarter Road or Manor Lane would dissect the property and detract from the significance of the property.

MD 144 is designated the Historic National Road National Scenic Byway. Because of the size of the Doughoregan Manor Residential Development, this proposal will have a substantial impact on the scenic byway's character-defining features. Therefore, it is most critical for the planning and design of this project include context sensitive architectural and site enhancement measures. In addition, the present layout does not represent historical development along the byway. Therefore, we strongly suggest that if the rezoning petition is approved and the project progresses into final design, that the layout of all residential lots adjacent to MD 144 replicate a traditional development that places all houses facing the scenic byway. The lots adjacent to MD 144 should have direct access from internal roads and alleys that spur from the development's entrance roadway. If the project is context sensitive designed, it will maintain the character and attractiveness of the byway, and support the byway travel experience.

Subject to the reduction of the access points on MD 144 to the one western access, and the connection of Burnside Drive to serve as the second access, along with intra-parcel connections, SHA has no objections to re-zoning approval.

Ms. Marsha S. McLaughlin ZB 1087M December 22, 2009 Page Three

SHA will require the opportunity to review and comment on all future Traffic Impact Studies, Sketch /Preliminary Plans, hydraulic computations, cultural resources studies and all necessary information to ultimately issue the required access permit from this office.

If you have any questions or comments, please contact Dan Doherty at 410-545-5584 or our toll free number in Maryland only 1-800-876-4742. You may also email him at (<u>ddoherty@sha.state.md.us</u>).

Sincerely, Steven D. Foster, Chief Engineering Access Permits Division

Cc: Mr. Dave Coyne Mr. John Concannon Mr. Mark Crampton Mr. Vaughn Lewis



Marsha S. McLaughlin, Director

TECHNICAL STAFF REPORT

June 3, 2010

Planning Board Meeting of June 17, 2010 County Council Hearing to be scheduled

Case No. /Petitioner;	At the request of Camilla Carroll and Philip D. Carroll
-----------------------	---

Request: Advisory determination by the Howard County Planning Board regarding General Plan consistency for the proposed Doughoregan Development Rights and Responsibilities Agreement (DRRA), pursuant to Section 16.1704 (2) of the Howard County Code and Article 66B, Section 13.01 of the Annotated Code of Maryland.

Location: Second and Third Election Districts Portion of Historic Doughoregan Manor at 3500 Manor Lane Ellicott City, Howard County, Maryland Tax Map 23, Grid 10, Part of Parcel 71

Area of Site: +/- 892 acres

Existing Zoning: RC-DEO: Rural Conservation – Density Exchange Option

Proposed Zoning: R-ED: Residential – Environmental Development for 221.1 acres, with the rest of the subject property remaining RC–DEO.

Department of Planning and Zoning Recommendation:

<u>Approval</u>

I. BACKGROUND

Application Overview

 The property owners of Doughoregan Manor submitted a proposed Development Rights and Responsibilities Agreement (DRRA) to formalize their multi-part proposal to restore and preserve most of this National Historic Landmark property (attached as Exhibit A).

- Pursuant to Section 16.1704 *Procedures* (B)(2) of the Howard County Code "An agreement may be executed by the County Executive only after: ... a public meeting before the Planning Board and a recommendation an advisory determination by the Planning Board that the proposed agreement is consistent with the Plan."
- On February 18, 2010, the Planning Board voted 4-0 to recommend approval of GPA 2010-1 that proposed adjusting the Planned Service Area boundary for water and sewer service to include 221 acres of the Doughoregan property.
- On February 18, 2010, the Planning Board voted 4-0 to recommend approval of ZB 1087M that proposed rezoning this 221-acre portion of the Doughoregan property to Residential – Environmental Design (R-ED) from the zoning designation of Rural Conservation – Density Exchange Option (RC-DEO).
- On March 1, 2010, the County Council via CB 4-2010 created a new Section 16.1700 *et. seq.* of the Howard County Code, pursuant to the authority granted by Article 66B, Section 13.01 of the Annotated Code of Maryland, to permit utilization of Development Rights and Responsibilities Agreements in Howard County.
- On April 5, 2010, the County Council via CB 9-2010 approved GPA 2010-1 adjusting the Planned Service Area boundary for water and sewer service to include approximately 221 acres of the Doughoregan property. The amended PSA boundary is attached as Exhibit B.
- The owners of Doughoregan applied to sell the County a permanent agricultural land preservation easement on 500 acres of farmland. The Agricultural Land Preservation Board recommended acquisition of an easement on the property. Upon signature of the commitment letter by the owners, the County Executive will request the County Council approve use of an installment purchase agreement to acquire the agricultural land preservation easement. Final settlement on the easement acquisition is part of this multi-part proposal.
- The owners are also offering to donate about 34 acres of land as an addition to the Kiwanis-Wallis Park.
- The property owners are also in the process of developing a strategy for preservation of the approximately 90-acre historic core that includes the Manor House and the numerous historic outbuildings.

General Information

- Doughoregan Manor was the country home of Charles Carroll III of Carrollton, a signer of the Declaration of Independence for Maryland. Charles Carroll, planter, landowner, politician and US Senator, was the only Roman Catholic signer of the Declaration and its last surviving signer. Charles Carroll III is buried in the family chapel attached to the north end of the Manor. Doughoregan Manor is the only home of a signer of the Declaration of Independence that is still in family ownership.
- Seventy-five acres of the property are already protected in a permanent preservation easement. This land is
 already preserved and is not included in any of the proposed requests mentioned above.

II, EVALUATION

The proposed Development Rights and Responsibilities Agreement commits the parties to the multi-part strategy for the subject property. GPA 2010-1 and ZB 1087M, which are the only two items subject to the Planning Board's review, were both recommended for approval by the Planning Board on February 18, 2010. At that time, the Planning Board found GPA 2010-1 to be consistent with *General Plan 2000*. The County Council concurred in adopting CB 9-2010 to amend the PSA boundary as shown in Exhibit B. Accordingly, the DRRA which includes this PSA boundary amendment is consistent with *General Plan 2000*.

Section 16.1704 (B)(2) of the Howard County Code sets forth both requirements and limitations related to the Planning Board review of a proposed DRRA for General Plan consistency: "(B) an agreement may be executed by the County Executive only after: ... (2) a public meeting before the Planning Board and a recommendation an advisory determination by the Planning Board that the proposed agreement is consistent with the Plan."

III. RECOMMENDATION

The Department of Planning and Zoning thus recommends a finding of General Plan consistency between the Development Rights and Responsibilities Agreement and General Plan 2000 as amended by CB 9-2101.

June 3, 2010 Date

Marsha S. McLaughlin, Director of Planning & Zoning

The file is available for public review in the Offices of the Department of Planning and Zoning, which are currently located in the Ascend One Building, located at 8930 Stanford Blvd in Columbia, Maryland, from Monday through Friday, 8:00 a.m. - 5:00 p.m.